

Geometry & Topology

Volume 28 (2024)

Homological invariants of codimension 2 contact submanifolds

Laurent Côté François-Simon Fauteux-Chapleau





Homological invariants of codimension 2 contact submanifolds

LAURENT CÔTÉ FRANÇOIS-SIMON FAUTEUX-CHAPLEAU

Codimension 2 contact submanifolds are the natural generalization of transverse knots to contact manifolds of arbitrary dimension. We construct new invariants of codimension 2 contact submanifolds. Our main invariant can be viewed as a deformation of the contact homology algebra of the ambient manifold. We describe various applications of these invariants to contact topology. In particular, we exhibit examples of codimension 2 contact embeddings into overtwisted and tight contact manifolds which are formally isotopic but fail to be isotopic through contact embeddings. We also give new obstructions to certain relative symplectic and Lagrangian cobordisms.

53D10, 53D42

1.	Introduction	2
2.	Geometric preliminaries	12
3.	Reeb dynamics near a codimension 2 contact submanifold	22
4.	Standard setups and tree categories.	29
5.	Intersection theory for punctured holomorphic curves	31
6.	Energy and twisting maps	41
7.	Enriched setups and twisted moduli counts	54
8.	Construction of the main invariants	66
9.	Augmentations and linearized invariants	75
10.	Invariants of Legendrian submanifolds	81
11.	Some computations	89
12.	Applications to contact topology	105
App	pendix. Connected sums of almost-contact manifolds	116
References		122

^{© 2024} MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

1 Introduction

1.1 Overview

The purpose of this paper is to introduce new invariants of codimension 2 contact submanifolds. Given a closed, co-oriented contact manifold (Y, ξ) and a codimension 2 contact submanifold V with trivial normal bundle, our main construction produces a unital, $\mathbb{Z}/2$ -graded $\mathbb{Q}[U]$ -algebra

(1-1)
$$CH_{\bullet}(Y,\xi,V;\mathfrak{r}).$$

Here $\mathfrak{r} = (\alpha_V, \tau, r)$ is a triple consisting of

- (i) a nondegenerate contact form α_V on *V* inducing $\xi_V := \xi|_V \cap TV$,
- (ii) a trivialization τ of $N_{Y/V}$, and
- (iii) a real number r > 0.

This triple is required to satisfy certain conditions (stated in Definition 3.3), and should be viewed as encoding a choice of Reeb dynamics in an infinitesimally small neighborhood of V. The (nonempty) set of all such triples is denoted by $\Re(Y, \xi, V)$.

The invariant (1-1) can be viewed as a deformation of the contact homology algebra $CH_{\bullet}(Y,\xi)$, as explained in Remark 1.1 below. In particular, U is a formal variable and there is a natural map

(1-2)
$$\operatorname{ev}_{U=1}: CH_{\bullet}(Y,\xi,V;\mathfrak{r}) \to CH_{\bullet}(Y,\xi)$$

obtained by setting U = 1.

The algebra $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$ is generated by (good) Reeb orbits for an auxiliary nondegenerate contact form λ on (Y, ξ) . See Pardon [61, Definition 2.59] for the notion of a good Reeb orbit; all nondegenerate Reeb orbits are good except for certain even-degree covers of simple Reeb orbits with odd Conley–Zehnder index. The form λ is required to be *adapted* to $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$, which means in particular that V is preserved by the Reeb flow of λ ; see Definition 3.4.

The differential is defined as in ordinary contact homology by counting pseudoholomorphic curves in the symplectization \hat{Y} , where the additional U variable keeps track of the intersection number of curves with the symplectization $\hat{V} \subset \hat{Y}$. More precisely, we fix an almost complex structure $J: \xi \to \xi$ which is compatible with the symplectic form $d\lambda$ and preserves $\xi|_V \cap TV \subset TY|_V$. We then consider \hat{J} -holomorphic curves in \hat{Y} , where $\hat{J} = -\partial_t \otimes \lambda + R_\lambda \otimes dt + J$. The differential is defined on generators by (roughly)¹ the formula

(1-3)
$$d(\gamma) = \sum_{\beta \in \pi_2(\widehat{Y}, \gamma \sqcup \gamma_1 \sqcup \cdots \sqcup \gamma_l)} \# \overline{\mathcal{M}}(\beta) U^{\widehat{V} * \beta + \Gamma^-(\beta, V)} \gamma_1 \ldots \gamma_l,$$

¹Strictly speaking, (1-3) should be refined as follows: (i) one should indicate that the virtual moduli counts depend on a choice of "virtual perturbation data"; (ii) one should indicate that the counts depend on the order of a certain group of automorphisms of the triple $(\gamma, \gamma_1 \sqcup \cdots \sqcup \gamma_l, \beta)$ which acts on γ by the identity; (iii) one must specify signs (or, more invariantly and following Pardon [61], orientations lines). See (6-7) for the precise formula.

where γ is a positive orbit (ie associated to the convex end of the symplectization) and $\gamma_1, \ldots, \gamma_l$ are negative orbits (ie associated to the concave end). We define $\Gamma^-(\beta, V) = \#\{\gamma_i \subset V\}$ to be the number of negative orbits of β which are contained in *V*.

We denote by $\overline{\mathcal{M}}(\beta)$ the compactification of the moduli space $\mathcal{M}(\beta)$ of \widehat{J} -holomorphic curves in the class β . Such moduli spaces are in general nontransversally cut out, so the moduli counts $\#\overline{\mathcal{M}}(\beta)$ appearing in (1-3) are defined as in Pardon's construction [60; 61] of contact homology, via his theory of virtual fundamental cycles. In particular, $\#\overline{\mathcal{M}}(\beta)$ is nonzero only for moduli spaces of virtual dimension zero. Finally, the pairing (-*-) is a count of intersections between \widehat{V} and β , which was introduced by Siefring [64; 54].

We also construct a closely related invariant

(1-4)
$$\widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r}),$$

which we call *reduced*. This is a unital, $\mathbb{Z}/2$ -graded \mathbb{Q} -algebra which is generated by Reeb orbits in the complement of *V*. The differential counts pseudoholomorphic curves which do not intersect \hat{V} . For appropriately chosen pairs $\mathfrak{r}, \mathfrak{r}' \in \mathfrak{R}(Y, \xi, V)$, we have a morphism of \mathbb{Q} -algebras

(1-5)
$$CH_{\bullet}(Y,\xi,V;\mathfrak{r}') \to \widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r})$$

The invariant $\widetilde{CH}_{\bullet}(-; -)$ is called *reduced* because it carries less information (in particular, it does not involve taking the kernel of an augmentation as in, for instance, reduced singular homology). However, it is easier to compute.

Remark 1.1 The $\mathbb{Q}[U]$ -algebra $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$ can be viewed as a deformation of the contact homology \mathbb{Q} -algebra $CH_{\bullet}(Y, \xi)$ in the following way. First, recall that for a ring R and a differential graded R-algebra (A, d), a (*formal*) deformation of (A, d) is the data of a differential $d_t := d + td_1 + t^2d_2 + \cdots$ on the R[[t]]-algebra A[[t]] satisfying the graded Leibnitz rule, where each d_i is an endomorphism of A; see Gerstenhaber and Wilkerson [30]. Now, let us set $U = e^t$ in (1-3) and expand in t. We then get

(1-6)
$$d(\gamma) = \sum_{\beta} \sum_{k=0}^{\infty} t^k \cdot (\#\overline{\mathcal{M}}(\beta)) \cdot \frac{(\widehat{V} * \beta + \Gamma^-(\beta, V))^k}{k!} \gamma_1 \cdots \gamma_l$$

Thus $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$ is indeed a deformation of ordinary contact homology, which can be recovered by sending $t \to 0$. In the case where $\Gamma^{-}(\beta, V) = 0$, the coefficient

$$\#\overline{\mathcal{M}}(\beta) \cdot \frac{(\widehat{V} * \beta + \Gamma^{-}(\beta, V))^{k}}{k!} = \#\overline{\mathcal{M}}(\beta) \cdot \frac{(\widehat{V} * \beta)^{k}}{k!}$$

could be interpreted as a count of rigid pseudoholomorphic curves which send k marked points in the source to the pseudoholomorphic divisor \hat{V} .

1.2 Energy and positivity of intersection

In order to ensure that (1-3) defines a differential over $\mathbb{Q}[U]$, we need to ensure that $\hat{V} * \beta + \Gamma^{-}(\beta, V) \ge 0$ whenever $\#\overline{\mathcal{M}}(\beta) \ne 0$. If $\mathcal{M}(\beta)$ is nonempty² and at least one of the asymptotic orbits of β is disjoint from V, then this is a consequence of the familiar phenomenon of positivity of intersection. Indeed, in this case, β admits a \hat{J} -holomorphic representative u which is not contained in \hat{V} . Positivity of intersection then implies that $\hat{V} * \beta = \hat{V} * u \ge 0$.

The situation is more complicated when all of the asymptotic orbits of β are contained in V. Indeed, in this case, the \hat{J} -holomorphic representatives of β may be contained in \hat{V} and positivity of intersection fails in general. However, one can show that there is a universal lower bound on the intersection number,

(1-7)
$$\widehat{V} * \beta \ge -\Gamma^{-}(\beta, V).$$

This explains the appearance of the correction term $\Gamma^{-}(\beta, V)$ in (1-3).

In order to construct $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$, it is not enough to define a differential: one also needs to define continuation maps, composition homotopies, etc. These maps are defined by counting curves in more complicated setups. For example, the continuation map is obtained by counting curves in a suitably marked exact relative symplectic cobordism $(\hat{X}, \hat{\lambda}, H)$.³ More precisely, one obtains an algebra map similar to (1-3) by counting \hat{J} -holomorphic curves in $(\hat{X}, \hat{\lambda})$ weighted by their intersection number with H, for a compatible almost complex structure \hat{J} which agrees with \hat{J}^{\pm} near the ends.

Unfortunately, for an arbitrary relative symplectic cobordism, a lower bound of the type (1-7) fails to hold. A key step in constructing the invariants (1-1) is to identify a sufficiently large class of relative symplectic cobordisms for which such a lower bound does hold. This leads us to introduce notions of energy for exact symplectic cobordisms and almost complex structures on exact relative symplectic cobordisms. These energy notions are developed in Section 6 and are of central importance in this paper.

We prove that a lower bound as in (1-7) holds under a certain condition which relates the behavior of λ^{\pm} near V^{\pm} to the energy of \hat{J} . We also prove analogous statements for other related setups. This allows us to prove that $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$ is well-defined, is it does not depend on the auxiliary contact form and almost complex structure. We also prove that an exact relative symplectic cobordism $(\hat{X}, \hat{\lambda}, H)$ induces a map

(1-8)
$$CH_{\bullet}(Y^+,\xi^+,V^+;\mathfrak{r}^+) \to CH_{\bullet}(Y^-,\xi^-,V^-;\mathfrak{r}^-)$$

provided that a certain inequality is satisfied, where the inequality involves \mathfrak{r}^{\pm} and the energy of the (sub)cobordism $H \subset (\hat{X}, \hat{\lambda})$.

²Since [61] uses virtual techniques to define contact homology without making any transversality assumptions, it is possible for the compactification $\overline{\mathcal{M}}(\beta)$ to be nonempty even if $\mathcal{M}(\beta)$ is empty. Positivity of intersection still holds when this happens, but the proof requires a bit more work. Details can be found in Sections 5.3 and 7.2.

³An exact relative symplectic cobordism is the data of an exact symplectic cobordism $(\hat{X}, \hat{\lambda})$ which looks like $(\hat{Y}^{\pm}, \hat{\lambda}^{\pm})$ near the ends, together with a codimension 2 symplectic submanifold $H \subset \hat{X}$ which looks like \hat{V}^{\pm} near the ends; see Definition 2.17 for the details.

Energy considerations play a similarly central role in our construction of the reduced invariant $\widetilde{CH}_{\bullet}(-;-)$. Although the continuation map for the reduced invariant does not count curves contained in H, one needs to ensure that sequences of curves disjoint from H does not degenerate into H. This requires hypotheses on the energy of the relevant cobordism. In general, the arguments involved in constructing $CH_{\bullet}(-;-)$ and $\widetilde{CH}_{\bullet}(-;-)$ turn out to be very similar.

Energy is not in general well-behaved under gluing symplectic cobordisms, unless one of them happens to be a symplectization. As a result, cobordism maps cannot be composed arbitrarily. This lack of functoriality of the invariants (1-1) and (1-4) can be remedied by considering variants of these invariants which are obtained by taking certain (co)limits over $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$; see Section 8.3. These variants are fully functorial but also seem harder to compute.

Remark 1.2 The apparent failure of positivity of intersection in the absence of energy bounds is not a deficiency of our method: if one could define maps (1-8) without additional hypotheses involving energy and the \mathfrak{r}^{\pm} , then this would imply that (1-1) and (1-4) are independent of \mathfrak{r} . To see that this cannot hold in general, consider an "irrational ellipsoid" $E(r_1, r_2) = \{z \in \mathbb{C}^2 \mid \pi \mid z_1 \mid^2 / r_1 + \pi \mid z_2 \mid^2 / r_2 \leq 1\} \subset \mathbb{R}^4$, where r_1/r_2 is irrational. Following Hutchings [41, Section 4.2], there are exactly two families of Reeb orbits $\{\gamma_1^k\}_{k \in \mathbb{N}_+}$ and $\{\gamma_2^k\}_{k \in \mathbb{N}_+}$, where $\gamma_i^1 \subset \{z_i = 0\} \cap \partial E(r_1, r_2)$ and γ_i^k denotes the *k*-fold cover of γ_i^1 . These orbits generate the (ordinary) contact homology of the 3-sphere, but the Conley–Zehnder indices of the γ_i^k are highly sensitive to r_1 and r_2 . If we could define the contact homology of the complement of (say) γ_1^1 , then it would be generated by the γ_2^k . One can verify that this is not an invariant, since changing the r_i does not change the contactomorphism type of $S^3 - \gamma_1^1$, but drastically changes the indices of the γ_2^k .

1.3 Legendrian invariants and the surgery formula

Contact homology is one of many invariants which can be constructed using the framework of Symplectic Field Theory (SFT). SFT was first introduced by Eliashberg, Givental and Hofer [24] and provides (among other things) a conjectural mechanism for constructing invariants in symplectic and contact topology by counting punctured pseudoholomorphic curves in symplectic manifolds with cylindrical ends.

In some of the later sections of this paper, we discuss how the invariants (1-1) and (1-4) are related to other SFT-type invariants. For computational purposes, it is particularly useful to explore the behavior of the invariants (1-1) and (1-4) under Weinstein handle attachment, following the work of Bourgeois, Ekholm and Eliashberg [7].

To this end, we introduce analogs of (1-1) and (1-4) for Legendrian submanifolds. With (Y, ξ, V) as above, suppose that $\Lambda \subset (Y - V, \xi)$ is a Legendrian submanifold. We then define (under mild topological assumptions) invariants

(1-9)
$$\mathcal{L}(Y,\xi,V,\Lambda;\mathfrak{r})$$
 and $\widetilde{\mathcal{L}}(Y,\xi,V,\Lambda;\mathfrak{r})$.

The first invariant can be thought of as a deformation of the Chekanov–Eliashberg dg algebra of $\Lambda \subset (Y, \xi)$, while the second invariant is a reduced version.

We describe a conjectural surgery exact sequence which relates (linearized versions of) the invariants (1-1), (1-4) and (1-9) under Weinstein handle attachments. This surgery exact sequence is an analog of the conjectural surgery exact sequence for linearized contact homology of Bourgeois, Ekholm and Eliashberg [7, Theorem 5.2].

Remark 1.3 The main invariants of this paper, (1-1) and (1-4), are constructed fully rigorously using Pardon's virtual perturbation framework [61]. However, our discussion of the surgery formula (and of the Legendrian invariants therein) requires certain transversality assumptions which are essentially the same as in [7]. We fully expect that if [7] can be made rigorous using Pardon's techniques [61], then extending this to our context should pose no substantial additional difficulties.

For the reader's convenience, *all statements in this paper which depend on unproved assumptions are labeled by a star.* The proofs of starred statements depend only on a limited set of assumptions, which are clearly identified in the text.

One could also attempt to define the invariants in this paper using other perturbation frameworks such as the polyfolds of Hofer, Wysocki and Zehnder [36; 37; 38] or the techniques of Ekholm [20], but we do not pursue this here.

1.4 Applications to contact and Legendrian embeddings

Transverse knots are important objects of study in three-dimensional contact topology. The notion of a codimension 2 contact embedding generalizes transverse knots to contact manifolds of arbitrary dimension. However, until recently, it was not understood whether the high-dimensional theory of codimension 2 contact embeddings is interesting from the perspective of contact topology, or whether it reduces entirely to differential topology.

Definition 1.4 Given a pair of contact manifolds (V^{2m-1}, ζ) and (Y^{2n-1}, ξ) , a *contact embedding* is a smooth embedding

such that $i^*(\xi_{i(V)} \cap di(TV)) = \zeta$. Such a map is also referred to as an isocontact embedding in the literature (see eg Casals and Etnyre [10], Eliashberg and Mishachev [25] and Pancholi and Pandit [59]), but we will not use this terminology. A contact submanifold $V \subset (Y, \xi)$ is a submanifold with the property that $\xi|_V \cap TV$ is a contact structure.

Observe that if 2n - 1 = 3 and 2m - 1 = 1 in the above definition, then we recover the familiar notion of a (parametrized) transverse knot. The following basic examples of codimension 2 contact embeddings will play an important role in this paper.

Example 1.5 (cf Definition 3.11 and [31]) Let $\pi: Y - B \to S^1$ be an open book decomposition which supports the contact structure ξ on Y. Then the binding $B \subset (Y, \xi)$ is a codimension 2 contact submanifold.

Example 1.6 (see Definition 10.1 and Definition 3.1 in [10]) Let (Y, ξ) be a contact manifold and let $\Lambda \hookrightarrow Y$ be a Legendrian embedding. Then the Weinstein neighborhood theorem furnishes an embedding

(1-11)
$$\tau(\Lambda): (\partial(D^*\Lambda), \xi_{\text{std}}) \hookrightarrow (Y, \xi),$$

which is canonical up to isotopy through codimension 2 contact embeddings. We refer to $\tau(\Lambda)$ as the *contact pushoff* of $\Lambda \hookrightarrow Y$. By abuse of notation, we will routinely identify $\tau(\Lambda)$ with its image.

As is customary in contact and symplectic topology, there is a notion of a *formal* contact embedding. This notion encodes certain necessary bundle-theoretic conditions which must be satisfied by any (genuine) contact embedding. It is then natural to seek to understand to what extent the space of genuine contact embeddings of (V, ξ) into (Y, ξ) differs from the space of formal contact embeddings.

In the case that V is a closed manifold of codimension at least 4 with respect to Y, or open and of codimension at least 2, then an h-principle due to Gromov (see Theorem 12.3.1 and Remark 12.3 of Eliashberg and Mishachev [25]) implies that the space of contact embeddings is essentially equivalent to the space of formal contact embeddings. Thus, in these settings, the theory of contact embeddings reduces to differential topology.

In contrast, a breakthrough result due to Casals and Etnyre [10] shows that this h-principle fails in general for codimension 2 contact embeddings of closed manifolds. More precisely, for $n \ge 3$, Casals and Etnyre [10, Theorem 1] exhibited a pair of contact embeddings of $(D^*S^{n-1}, \xi) = \partial_{\infty}(T^*S^{n-1}, \lambda_{can})$ into the standard contact sphere (S^{2n-1}, ξ_{std}) which are formally isotopic but are not isotopic through contact embeddings (here and throughout the paper, $\partial_{\infty}(-)$ denotes the ideal contact boundary). Building on these methods, Zhou [70] recently proved that there are in fact infinitely many formally isotopic contact embeddings of $\partial_{\infty}(T^*S^{n-1}, \lambda_{can})$ into the standard contact sphere which are not isotopic through contact embeddings, provided $n \ge 4$.

There has also been recent work to establish existence results for codimension 2 contact embeddings under certain conditions; see Lazarev [48] and Pancholi and Pandit [59]. This culminates in a full existence h-principle for codimension 2 contact embeddings due to Casals, Pancholi and Presas [13], which states that any formal codimension 2 contact embedding is formally isotopic to a genuine contact embedding.

The invariants constructed in this paper can be used to distinguish pairs of formally isotopic contact embeddings which are not isotopic through contact embeddings. We illustrate two types of applications, applying, respectively, to contact embeddings into overtwisted contact manifolds, and into the standard contact sphere.

Let us begin with the overtwisted case. In Construction 12.6, we describe a procedure for constructing pairs of formally isotopic contact embeddings into certain overtwisted contact manifolds which are not isotopic through contact embeddings. Here is a special case of this construction: let (Y, ξ) be an overtwisted contact manifold and fix an open book decomposition for (Y, B, π) which supports ξ ; see Section 3.3.

Let $i: B \to Y$ be the tautological inclusion of the binding. Using the relative *h*-principle for contact structures of Borman, Eliashberg and Murphy [5, Theorem 1.2], it can be shown that there exists a codimension 2 contact embedding $j: B \to Y$ which is formally isotopic to *i*, and such that $(Y - j(B), \xi)$ is overtwisted. (Construction 12.6 is more general, but this is the most important example.)

Theorem 1.7 (see Theorem 12.7) Let i, j and (Y, ξ) be constructed according to Construction 12.6, where (Y, ξ) is an overtwisted contact manifold and i and j are (formally isotopic) contact embeddings. Then i and j are not isotopic through contact embeddings. In fact, i is not isotopic to any reparametrization of j in the source, meaning that i(V) and j(V) are not isotopic as codimension 2 contact submanifolds of (Y, ξ) .

Theorem 1.7 can be proved using either of the invariants (1-1) or (1-4). To the best of our knowledge, it cannot be proved in general using invariants already in the literature. However, in the special case where i(B) is the binding of an open book decomposition (ie the example sketched above), then the conclusion of Theorem 1.7 follows from the fact that the complement of the binding of an open book decomposition is tight. This later fact is due to Etnyre and Vela-Vick [28, Theorem 1.2] in dimension 3; in higher dimensions, it follows from work of Klukas [45, Corollary 3], who proved (following an outline of Wendl [67, Remark 4.1]) the stronger statement that any local filling obstruction (such as an overtwisted disk) in a closed contact manifold must intersect the binding of any supporting open book.

In some cases (see Corollary 12.10), the embeddings i and j in fact coincide with the contact pushoffs of Legendrian embeddings. It is not hard to show that an isotopy of Legendrian embeddings induces an isotopy of their contact pushoffs. Thus the invariants (1-1) and (1-4) also distinguish certain Legendrian embeddings in overtwisted contact manifolds. To our knowledge, these embeddings cannot in general be distinguished using invariants already in the literature; see Remark 1.11.

Our second application concerns codimension 2 contact embeddings into the standard contact spheres $(S^{4n-1}, \xi_{\text{std}})$. More precisely, we use the reduced invariant (1-4) to distinguish formally isotopic contact embeddings of $(S^*S^{2n-1}, \xi) = \partial_{\infty}(T^*S^{2n-1}, \lambda_{\text{can}})$ into $(S^{4n-1}, \xi_{\text{std}})$, thus reproving the main result of Casals and Etnyre [10, Theorem 1] in dimensions 4n - 1 for n > 1.

Theorem* 1.8 (see Theorem* 12.18) Let (V, ξ) be the ideal boundary of $(T^*S^{2n-1}, \lambda_{can})$. Then for n > 1, there exists a pair of formally isotopic contact embeddings

(1-12)
$$i_0, i_1 : (V, \xi) \to (S^{4n-1}, \xi_{\text{std}})$$

which are not isotopic through contact embeddings.

The embeddings we exhibit turn out to coincide with those exhibited by Casals and Etnyre in their proof of [10, Theorem 1.1], although this fact is not entirely obvious; see Remark 12.21. However, the methods for distinguishing them are completely different. Casals and Etnyre consider double branched covers

9

along the contact submanifolds $i_0(V)$ and $i_1(V)$. Using symplectic homology, they prove that these branched covers do not admit the same fillings. This implies that $i_0(V)$ and $i_1(V)$ cannot be isotopic, since otherwise they would have contactomorphic branched covers.

In contrast, our proof of Theorem* 1.8 uses the invariant $\widetilde{CH}_{\bullet}(-; -)$ introduced in this paper. Roughly speaking, we prove Theorem* 1.8 by partially computing (linearizations of) $\widetilde{CH}_{\bullet}(-; -)$ associated to the two embeddings under consideration, and observing that they do not match. Our computations rely crucially on our version of the surgery formula discussed in Section 1.3 as well as the well-definedness of the invariants therein. This explains why this theorem statement is starred, following the convention stated in Remark 1.3. We also remark that although Theorem* 1.8 only applies to spheres of dimension 4n - 1, we expect that the same invariant also distinguishes embeddings into spheres of dimension 4n - 3. However, proving this would likely require more involved computations than those carried out in this paper.

1.5 Applications to symplectic and Lagrangian cobordisms

Consider a pair of contact manifolds (Y^{\pm}, ξ^{\pm}) and codimension 2 contact submanifolds $(V^{\pm}, \xi^{\pm}|_{V^{\pm}}) \subset (Y^{\pm}, \xi^{\pm})$. An exact relative symplectic cobordism from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) is a triple $(\hat{X}, \hat{\lambda}, H)$, where $(\hat{X}, \hat{\lambda})$ is an exact symplectic cobordism from (Y^+, ξ^+) to (Y^-, ξ^-) and $H \subset \hat{X}$ is a codimension 2 symplectic submanifold which coincides near the ends with the symplectization of V^{\pm} ; see Definition 2.17.⁴ In the special case where \hat{X} is the symplectization of Y^{\pm} and H is diffeomorphic to $\mathbb{R} \times V^{\pm}$, we speak of a *symplectic concordance* from V^+ to V^- . These notions were first considered by Bowden in his PhD thesis [9]. Using gauge theory, he exhibited certain restrictions on symplectic cobordisms between transverse links in contact 3-manifolds [9, Section 7].

The following theorem gives a constraint on exact symplectic cobordisms between certain pairs of codimension 2 contact submanifolds of an ambient overtwisted manifold. To the best of our knowledge, this is the first negative result in the literature on relative symplectic cobordisms in dimensions greater than three.

Theorem 1.9 Let V = i(B) and V' = j(B) be the codimension 2 contact submanifolds of the overtwisted contact manifold (Y,ξ) as described in Construction 12.6. Then there does not exist an exact relative symplectic cobordism $(\hat{X}, \hat{\lambda}, H)$ from (Y, ξ, V') to (Y, ξ, V) with $H^1(H, (-\infty, 0] \times V; \mathbb{Z}) =$ $H^2(H, (-\infty, 0] \times V; \mathbb{Z}) = 0$. In particular, there is no symplectic concordance from V' to V.

One can similarly consider Lagrangian cobordisms and concordances between Legendrian submanifolds. An exact Lagrangian cobordism from (Y^+, ξ^+, Λ^+) to (Y^-, ξ^-, V^-) is a triple $(\hat{X}, \hat{\lambda}, L)$, where $(\hat{X}, \hat{\lambda})$ is an exact symplectic cobordism from (Y^+, ξ^+) to (Y^-, ξ^-) and $L \subset \hat{X}$ is a Lagrangian submanifold which

⁴Note that our convention of regarding a cobordism as going *from* the convex end *to* the concave end is consistent with [61], but differs from most of the contact topology literature.

coincides near the ends with the Lagrangian cone of Λ^{\pm} ; see Definition 2.20. If \hat{X} is the symplectization of Y^- and $L = \mathbb{R} \times \Lambda^-$, one speaks of a *Lagrangian concordance* from Λ^+ to Λ^- .

The theory of Lagrangian cobordisms has been extensively developed in the literature from various perspectives; see eg Chantraine, Dimitroglou Rizell, Ghiggini and Golovko [15], Ekholm [19], Ekholm, Honda and Kálmán [23] and Sabloff and Traynor [63]. While much is known in ($\mathbb{R}^{2n+1}, \xi_{std}$) and certain other tight contact manifolds, we are not aware of any results constraining cobordisms and concordances in overtwisted contact manifolds; see Remark 1.11. The next theorem provides a first result in this direction.

Theorem 1.10 Let Λ and Λ' be the Legendrian submanifolds of the overtwisted contact manifold (Y, ξ) as constructed in Construction 12.9. Then Λ' is not concordant to Λ .

In contrast, a result of Eliashberg and Murphy [26, Theorem 2.2] implies that Λ is concordant to Λ' .

Remark 1.11 It is a basic fact that exact Lagrangian cobordisms induce morphisms on Legendrian contact homology which behave well under composition of cobordisms; see Etnyre and Ng [27, Section 5.1]. This leads to a myriad of interesting obstructions to the existence of Lagrangian cobordisms and concordances. One can also obtain many interesting obstructions using finite-dimensional invariants (which are closely related to Legendrian contact homology) coming from generating functions or sheaf theory; see eg Li [49] and Sabloff and Traynor [63].

One drawback of these approaches is that they are necessarily blind on overtwisted contact manifolds. Indeed, even if Legendrian contact homology could be rigorously defined in full generality following the framework of Eliashberg, Givental and Hofer [24, Section 2.8], it would provide no information for Legendrians in overtwisted contact manifolds: being a module over the contact homology algebra, it would vanish. In contrast, the invariants developed in this paper do give information about Legendrians even in the overtwisted case.

Our final application states that certain Lagrangian concordances cannot be displaced from a codimension 2 symplectic submanifold. More precisely, let $(Y, \xi) = \text{obd}(T^*S^{n-1}, \text{id})$ and let $V \subset Y$ be the binding of the open book. Let $\Lambda \subset Y$ be the zero section of a page and let Λ' be obtained by stabilizing Λ in the complement of V. It can be shown (see Casals and Murphy [11, Proposition 2.9]) that $\Lambda \subset (Y, \xi)$ is a loose Legendrian; hence Λ and Λ' are Legendrian isotopic in (Y, ξ) and, in particular, concordant.

Theorem* 1.12 Any Lagrangian concordance from Λ' to Λ must intersect the symplectization of V.

In contrast, work of Eliashberg and Murphy [26, Theorem 2.2] implies that there exists a Lagrangian concordance from Λ to Λ' which is disjoint from the symplectization of *V*. Our proof of Theorem* 1.12 uses the deformed versions of the Chekanov–Eliashberg dg algebra in (1-9). Hence the statement is starred according to the convention stated in Section 1.3.

1.6 Context and related invariants

The invariants constructed in this paper, when specialized to contact 3-manifolds, are related to other invariants in the literature. The most closely related invariant is due to Momin [53]. Given a contact 3-manifold (Y^3, ξ) , Momin considers the set of pairs (λ, L) where λ is a contact form and $L \subset Y$ is a link of Reeb orbits of λ . Two such pairs (λ, L) and (λ', L') are said to be equivalent if L = L' and each component orbit (and all its multiple covers) has the same Conley–Zehnder index. Under certain assumptions on (Y, λ, L) , Momin defines an invariant which we denote by $CH^{\text{mo}}_{\bullet}(Y, [(\lambda, L)])$. This is a \mathbb{Z} -graded \mathbb{Q} -vector space which depends only on Y and the equivalence class of (λ, L) .

The invariant constructed by Momin is in general distinct from the invariants described in this paper. In particular, he considers cylindrical contact homology, whereas we work with ordinary contact homology. However, in the special case where (Y^3, ξ) is the standard contact sphere — or more generally a subcritical Stein manifold with $c_1(\xi) = 0$ — and $L \subset (Y, \xi)$ is a collection of Reeb orbits which bound a symplectic submanifold $H \subset B^4$, then we expect that

(1-13)
$$CH^{\mathrm{mo}}_{\bullet}(Y,[\lambda,L]) = \widetilde{CH}^{\widetilde{\epsilon}}_{\bullet}(Y,\xi,L;\mathfrak{r})$$

for suitable \mathfrak{r} which depends on the equivalence class of (λ, L) . Here the right-hand side of (1-13) denotes the linearization of $\widetilde{CH}_{\bullet}(Y, \xi, L; \mathfrak{r})$ with respect to an augmentation $\widetilde{\epsilon}$ induced by the relative filling $(B^4, \lambda_{\text{std}}, H)$; recall that an augmentation of a dg algebra is a morphism to the ground ring, viewed as a dg algebra concentrated in degree zero. See Section 9.3 for details.

Momin's work has led to beautiful applications to Reeb dynamics on contact 3–manifolds; see eg Alves and Pirnapasov [3] and Hryniewicz, Momin and Salomão [40]. It would be interesting to explore whether the invariants developed in this paper can be used in studying Reeb dynamics in higher dimensions.

Another related invariant is Hutchings' "knot-filtered" embedded contact homology [41]. The setting for this invariant is a contact 3-manifold (Y, ξ) with $H_1(Y; \mathbb{Z}) = 0$. Given a transverse knot $L \subset (Y, \xi)$ and an irrational parameter $\theta \in \mathbb{R} - \mathbb{Q}$, Hutchings defines a filtration on embedded contact homology with values in $\mathbb{Z} + \mathbb{Z}\theta$ which is an invariant of (L, θ) . The basic idea is to choose a contact form $\xi = \ker \lambda$ so that L is a Reeb orbit, and to filter the generators of embedded contact homology by their linking number with L. Positivity of intersection considerations imply that the differential decreases the linking number for orbits which are disjoint from L. However, the situation is more complex when the differential involves L, which explains why the filtration is only valued in $\mathbb{Z} + \mathbb{Z}\theta$.

One could presumably carry over Hutchings' construction to the context of (cylindrical) contact homology in dimension 3. We expect that the resulting invariant would carry related information to the one defined by Momin or to the invariants constructed in this paper. However, we do not have a precise formulation of what this relationship should be.

We remark that the invariants introduced by Momin and Hutchings are built using techniques from 4-dimensional symplectic topology which cannot be generalized to higher dimensions. In contrast, the

invariants introduced in this paper are constructed by a different approach, which ultimately relies on Pardon's robust virtual fundamental cycles package [60].

In a slightly different direction, we also wish to highlight work of Ekholm, Etnyre, Ng and Sullivan [21], which is similar in spirit to the present work. Recall that the knot contact homology of a framed link $K \subset \mathbb{R}^3$ is an invariant which can be defined as the Legendrian dg algebra of the conormal lift of K; see Ekholm, Etnyre, Ng and Sullivan [22] and Ng [56]. If K is a transverse knot with respect to the standard contact structure, the authors define in [21] a two-parameter deformation of knot contact homology by weighting holomorphic curve counts by their intersection number with a pair of canonically defined complex submanifolds in the symplectization. The resulting deformed dg algebra is an invariant of the transverse knot type of K. Unfortunately, we do not know a precise relationship between this invariant and the ones introduced in this paper.

Notation and conventions All manifolds in this paper are assumed to be smooth. If M is a manifold, a *ball* $B \subset M$ is an open subset diffeomorphic to the open unit disk and whose closure is embedded and diffeomorphic to the closed unit disk. If (M, ω) is symplectic, a *Darboux ball* $B \subset M$ is a ball which is symplectomorphic to the open unit disk equipped with (some rescaling of) the standard symplectic form.

Unless otherwise specified, all contact manifolds considered in this paper are compact without boundary and co-oriented. Given such a contact manifold $(Y, \xi = \ker \lambda)$, the Reeb vector field associated to the contact form λ will be denoted by R_{λ} . We will let $\xi_V := \xi|_V \cap TV$ denote the contact structure induced by ξ on a contact submanifold $V \subset (Y, \xi)$.

Acknowledgements We thank Yasha Eliashberg for suggesting this project, and for many helpful discussions. We also benefited from discussions and correspondence with Cédric De Groote, Georgios Dimitroglou Rizell, Sheel Ganatra, Oleg Lazarev, Josh Sabloff and Kyler Siegel. We are grateful to John Etnyre for pointing us to multiple useful references. Finally, we wish to thank the referee for many helpful comments and suggestions.

Côté was partially supported by a Stanford University Benchmark Graduate Fellowship and by the National Science Foundation under grant DMS-1926686.

2 Geometric preliminaries

2.1 Symplectic cobordisms

Let (Y, ξ) be a closed co-oriented contact manifold. The *symplectization* of (Y, ξ) is the exact symplectic manifold (SY, λ_Y) where $SY \subset T^*Y$ is the total space of the bundle of positive contact forms on Y (ie a point $(p, \alpha) \in T^*Y$ is in SY if and only if $\alpha : T_pY \to \mathbb{R}$ vanishes on ξ_p and the induced map $T_pY/\xi_p \to \mathbb{R}$ is an orientation-preserving isomorphism) and λ_Y is the restriction of the tautological Liouville form on T^*Y . Given a choice of positive contact form α for (Y, ξ) , there is a canonical identification

(2-1)
$$\sigma_{\alpha} : (\mathbb{R} \times Y, e^{s_{\alpha}}) \to (SY, \lambda_{Y})$$

given by $\sigma_{\alpha}(s, p) = (p, e^{s}\alpha_{p})$. We will refer to $(\hat{Y}, \hat{\alpha}) := (\mathbb{R} \times Y, e^{s}\alpha)$ as the symplectization of (Y, α) .

A subset $U \subset SY$ will be called a *neighborhood of* $+\infty$ (resp. of $-\infty$) if it contains $\sigma_{\alpha}([N, \infty) \times Y)$ (resp. $\sigma_{\alpha}((-\infty, -N] \times \mathbb{R})$ for N > 0 sufficiently large — note that this notion doesn't depend on the choice of α .

Definition 2.1 Given a contactomorphism $f: (Y, \xi) \to (Y, \xi)$, we define its *symplectic lift*

(2-2)
$$\widetilde{f}: (SY, \lambda_Y) \to (SY, \lambda_Y), \quad (p, \alpha) \mapsto (f(p), \alpha \circ (df_p)^{-1}).$$

One can verify that $\tilde{f}^* \lambda_Y = \lambda_Y$, so \tilde{f} is in particular a symplectomorphism. There is a canonical bijection between

- (i) contact vector fields on (Y, ξ) ,
- (ii) sections of TY/ξ ,
- (iii) linear Hamiltonians on the symplectization (recall that H is *linear* if ZH = H, where Z denotes the Liouville vector field).

The correspondence between (i) and (ii) is clear; the correspondence between (ii) and (iii) takes a section σ to the Hamiltonian $H(p, \alpha) = \alpha(\sigma(p))$. In particular, the symplectic lift of a (timedependent) family of contactomorphisms is induced by a (time-dependent) family of linear Hamiltonians; cf [14, Proposition 2.2].

Definition 2.2 Let (Y^+, ξ^+) and (Y^-, ξ^-) be closed co-oriented contact manifolds. An *exact symplectic cobordism* from (Y^+, ξ^+) to (Y^-, ξ^-) is an exact symplectic manifold $(\hat{X}, \hat{\lambda})$ equipped with embeddings

 $(2-3) e^+: SY^+ \to \hat{X},$

$$(2-4) e^-: SY^- \to \widehat{X},$$

satisfying the properties

- $(e^{\pm})^* \hat{\lambda} = \lambda_{Y^{\pm}}$, and
- there exists a neighborhood U⁺ ⊂ SY⁺ of +∞ and a neighborhood U⁻ ⊂ SY⁻ of -∞ such that the restriction of e[±] to U[±] is proper, the images e⁺(U⁺) and e⁻(U⁻) are disjoint and the complement X \ (e⁺(U⁺) ∪ e⁻(U⁻)) is compact.

Definition 2.3 (cf [61, Section 1.3]) Let (Y^+, λ^+) and (Y^-, λ^-) be closed manifolds equipped with contact forms. A (*strict*) *exact symplectic cobordism* from (Y^+, λ^+) to (Y^-, λ^-) is an exact symplectic manifold $(\hat{X}, \hat{\lambda})$ equipped with embeddings

(2-5)
$$e^+ \colon \mathbb{R} \times Y^+ \to \hat{X},$$

$$(2-6) e^- : \mathbb{R} \times Y^- \to \hat{X}$$

satisfying the properties

- $(e^{\pm})^* \hat{\lambda} = \hat{\lambda}^{\pm}$, and
- there exists an N ∈ R such that the restrictions of e⁺ to [N,∞)×Y⁺ and of e⁻ to (-∞, -N]×Y⁻ are proper and that the images e⁺([N,∞)×Y⁺) and e⁻((-∞, -N]×Y⁻) are disjoint and together cover a neighborhood of infinity (ie the complement of their union is compact).

Notation 2.4 Let $(\hat{X}, \hat{\lambda})$ be an exact symplectic cobordism from (Y^+, ξ^+) to (Y^-, ξ^-) in the sense of Definition 2.2. Given any choice of contact forms λ^{\pm} on (Y^{\pm}, ξ^{\pm}) , one can obtain from \hat{X} a cobordism from (Y^+, λ^+) to (Y^-, λ^-) in the sense of Definition 2.3 by precomposing the embeddings (2-3)–(2-4) with the canonical identifications $\mathbb{R} \times Y^{\pm} \to SY^{\pm}$ induced by λ^{\pm} . We will denote this cobordism by $(\hat{X}, \hat{\lambda})_{\lambda^-}^{\lambda^+}$, or simply by $\hat{X}_{\lambda^-}^{\lambda^+}$ when this creates no ambiguity.

Similarly, any cobordism $(\hat{X}, \hat{\lambda})$ in the sense of Definition 2.3 can be viewed as a cobordism in the sense of Definition 2.2 as well.

Remark 2.5 In light of the above discussion, Definitions 2.2 and 2.3 are essentially equivalent. However, it will be convenient for us to be able to discuss symplectic cobordisms without fixing a particular choice of contact forms on the ends, so we adopt Definition 2.2 as our main definition moving forward.

Example 2.6 (symplectizations) The symplectization (SY, λ_Y) of a contact manifold (Y, ξ) is canonically endowed with the structure of an exact symplectic cobordism in the sense of Definition 2.2 by letting $e^+ = e^- = \text{id}$. The additional data of a pair of contact forms λ^+ and λ^- for (Y, ξ) endows (SY, λ_Y) with the structure of a strict exact symplectic cobordism in the sense of Definition 2.3, and we write $(SY, \lambda_Y)_{\lambda^-}^{\lambda^+}$.

Definition 2.7 Let $(\hat{X}^{01}, \hat{\lambda}^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12})$ be exact symplectic cobordisms from (Y^0, ξ^0) to (Y^1, ξ^1) and from (Y^1, ξ^1) to (Y^2, ξ^2) , respectively. Fix a real number $t \ge 0$ and let $\mu_t : SY^1 \to SY^1$ denote multiplication by e^t . The *t*-gluing of \hat{X}^{01} and \hat{X}^{12} , denoted by $\hat{X}^{01} \#_t \hat{X}^{12}$, is the smooth manifold obtained by gluing \hat{X}^{01} and \hat{X}^{12} along the maps

$$\begin{array}{c} SY^1 \xrightarrow{(2-4)} \widehat{X}^{01} \\ \downarrow \mu_t \\ SY^1 \xrightarrow{(2-3)} \widehat{X}^{12} \end{array}$$

Since $\mu_t^* \lambda_{Y^1} = e^t \lambda_{Y^1}$, there is, for any $s \in \mathbb{R}$, a Liouville form on $\hat{X}^{01} \#_t \hat{X}^{12}$ which agrees with $e^{t+s} \hat{\lambda}^{01}$ on \hat{X}^{01} and with $e^s \hat{\lambda}^{12}$ on \hat{X}^{12} . We will denote it by $\hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12}$. Note that $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12})$ is canonically equipped with the structure of an exact symplectic cobordism from (Y^0, ξ^0) to (Y^2, ξ^2) via the embeddings

$$\begin{split} SY^0 & \xrightarrow{\mu_{-t} - s} SY^0 \xrightarrow{(2-3)} \widehat{X}^{01} \to \widehat{X}^{01} \, \#_t \, \widehat{X}^{12}, \\ SY^2 & \xrightarrow{\mu_{-s}} SY^2 \xrightarrow{(2-4)} \widehat{X}^{02} \to \widehat{X}^{01} \, \#_t \, \widehat{X}^{12}. \end{split}$$

The precise choice of *s* doesn't really matter since the forms $\hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12}$ for $s \in \mathbb{R}$ are all constant multiples of each other. When t = 0, it is natural to choose s = 0, and we will denote the resulting cobordism simply by $(\hat{X}^{01} \# \hat{X}^{12}, \hat{\lambda}^{01} \# \hat{\lambda}^{12})$. There is no obvious choice for t > 0, but for the sake of definiteness we set $\hat{\lambda}^{01} \#_t \hat{\lambda}^{12} := \hat{\lambda}^{01} \#_{t,-t/2} \hat{\lambda}^{12}$ and will refer to $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{12})$ as "the" t-gluing of $(\hat{X}^{01}, \hat{\lambda}^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12})$.

Remark 2.8 When t = s = 0, it follows directly from the definition that the gluing operation is associative: $((\hat{X}^{01} \# \hat{X}^{12}) \# \hat{X}^{23}, (\hat{\lambda}^{01} \# \hat{\lambda}^{12}) \# \hat{\lambda}^{23})$ and $(\hat{X}^{01} \# (\hat{X}^{12} \# \hat{X}^{23}), \hat{\lambda}^{01} \# (\hat{\lambda}^{12} \# \hat{\lambda}^{23}))$ are canonically isomorphic.

Remark 2.9 Multiplication by e^t on SY corresponds to translation by t in the \mathbb{R} coordinate under the identification $SY \cong \mathbb{R} \times Y$ induced by a choice of contact form on Y. Definition 2.7 is therefore consistent with the notion of "t-gluing" in [61, Section 1.5].

Definition 2.10 Let $(\hat{X}^1, \hat{\lambda}^1)$ and $(\hat{X}^2, \hat{\lambda}^2)$ be cobordisms from (Y^+, ξ^+) to (Y^-, ξ^-) . An *isomorphism* of exact symplectic cobordisms $\phi: (\hat{X}^1, \hat{\lambda}^1) \to (\hat{X}^2, \hat{\lambda}^2)$ consists of a diffeomorphism $\phi: \hat{X}^1 \to \hat{X}^2$ such that $\phi^* \hat{\lambda}^2 = \hat{\lambda}^1$ and which is compatible with the ends in the sense that the diagram



commutes.

Example 2.11 Let $(\hat{X}, \hat{\lambda})$ be an exact symplectic cobordism from (Y^+, ξ^+) to (Y^-, ξ^-) . Then for any $t \ge 0$ and $s \in \mathbb{R}$, the glued cobordisms $(SY^+ \#_t \hat{X}, \lambda_{Y^+} \#_{t,s} \hat{\lambda})$ and $(\hat{X} \#_t SY^-, \hat{\lambda} \#_{t,s} \lambda_{Y^-})$ are canonically isomorphic to $(\hat{X}, \hat{\lambda})$.

Definition 2.12 A one-parameter family of exact symplectic cobordisms from (Y^+, ξ^+) to (Y^-, ξ^-) is a manifold \hat{X} equipped with a family of Liouville forms $\{\hat{\lambda}^t\}_{t \in I}$ (where $I \subset \mathbb{R}$ is an interval), together

with embeddings

$$(2-7) e_t^+: SY^+ \to \hat{X}$$

$$(2-8) e_t^-: SY^- \to \hat{X}$$

as in Definition 2.2. We will always assume that the family is fixed at infinity, meaning that for every compact subinterval $[a, b] \subset I$,

- $\{\hat{\lambda}^t\}_{t \in [a,b]}$ is constant outside of a compact subset of \hat{X} , and
- $\{e_t^+\}_{t \in [a,b]}$ (resp. $\{e_t^-\}_{t \in [a,b]}$) is independent of t on some neighborhood of $+\infty$ in SY^+ (resp. of $-\infty$ in SY^-).

Two cobordisms $(\hat{X}^0, \hat{\lambda}^0)$ and $(\hat{X}^1, \hat{\lambda}^1)$ are said to be *deformation equivalent* if there exists a oneparameter family $(\hat{W}, \hat{\mu}^t)_{t \in [0,1]}$ such that $(\hat{X}^0, \hat{\lambda}^0)$ is isomorphic to $(\hat{W}, \hat{\mu}^0)$ and $(\hat{X}^1, \hat{\lambda}^1)$ is isomorphic to $(\hat{W}, \hat{\mu}^1)$. The deformation class of a cobordism $(\hat{X}, \hat{\lambda})$ will be denoted by $[\hat{X}, \hat{\lambda}]$.

Lemma 2.13 Given a pair of cobordisms $(\hat{X}^{01}, \hat{\lambda}^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12})$ as in Definition 2.7, the glued cobordisms $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{12})_{t \in [0,\infty)}$ form a one-parameter family.⁵ Similarly, there is a one-parameter family $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12})_{s \in \mathbb{R}}$ for any fixed $t \ge 0$. In particular, we have that the deformation class $[\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12}]$ is independent of both t and s.

Proof We will construct a two-parameter family $\phi_{t,s}: \hat{X}^{01} \# \hat{X}^{12} \to \hat{X}^{01} \#_t \hat{X}^{12}$ of diffeomorphisms, with $\phi_{0,0} = id$, such that the forms $\phi_{t,s}^*(\hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12})$ agree with $\hat{\lambda}^{01} \# \hat{\lambda}^{12}$ outside of a compact set (depending on t, s) and form a smooth family.

In order to simplify the notation, we fix contact forms λ^i on (Y^i, ξ^i) for i = 0, 1, 2, so that we can view the symplectization of Y^i as a product $\mathbb{R} \times Y^i$. For C > 0 sufficiently large, we can decompose the cobordisms \hat{X}^{01} and \hat{X}^{12} as

(2-9)
$$\widehat{X}^{01} = (-\infty, 1] \times Y^1 \cup \overline{X}^{01} \cup [C, \infty) \times Y^0,$$

(2-10)
$$\hat{X}^{12} = (-\infty, -C] \times Y^2 \cup \bar{X}^{12} \cup [-1, \infty) \times Y^1,$$

where $\overline{X}^{01} \subset \widehat{X}^{01}$ is a compact submanifold with boundary $\{1\} \times Y^1 \sqcup \{C\} \times Y^0$, and similarly for \overline{X}^{12} . This induces a decomposition of $\widehat{X}^{01} \#_t \widehat{X}^{12}$ of the form

(2-11)
$$\hat{X}^{01} \#_t \hat{X}^{12} = (-\infty, -C] \times Y^2 \cup \overline{X}^{12} \cup [-1, t+1] \times Y^1 \cup \overline{X}^{01} \cup [C, \infty) \times Y^0$$

for any $t \ge 0$. Hence, in order to define $\phi_{t,s}$, it suffices to make a choice of

• a smooth family of diffeomorphisms $f_t: [-1, 1] \rightarrow [-1, t + 1]$ which coincide with the identity near -1 and with translation by t near 1,

⁵Strictly speaking, the underlying manifold of $\hat{X}^{01} \#_t \hat{X}^{12}$ depends on t, so in order to obtain a family in the sense of Definition 2.12 one needs to choose suitable diffeomorphisms $\hat{X}^{01} \#_t \hat{X}^{12} \cong \hat{X}^{01} \# \hat{X}^{12}$.

- a smooth family of diffeomorphisms g_{t,s}: [C, ∞) → [C, ∞) which coincide with the identity near C and with translation by -t s at infinity, and
- a smooth family of diffeomorphisms $h_{t,s}: (-\infty, -C] \to (-\infty, -C]$ which coincide with the identity near -C and with translation by -s at infinity.

We of course also require that f_0 , $g_{0,0}$ and $h_{0,0}$ be the identity on their respective domains.

Proposition 2.14 The deformation class of $(\hat{X}^{01} \# \hat{X}^{12}, \hat{\lambda}^{01} \# \hat{\lambda}^{12})$ only depends on the deformation classes of $(\hat{X}^{01}, \hat{\lambda}^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12})$.

Proof Let $(\hat{W}^{01}, \hat{\mu}^{01,s})_{s \in [0,1]}$ and $(\hat{W}^{12}, \hat{\mu}^{12,s})_{s \in [0,1]}$ be one-parameter families of exact symplectic cobordisms from (Y^0, ξ^0) to (Y^1, ξ^1) and from (Y^1, ξ^1) to (Y^2, ξ^2) , respectively. The negative end (2-8) of $(\hat{W}^{01}, \hat{\mu}^{01,s})$ will be denoted by $e_s^-: SY^1 \to \hat{W}^{01}$ and the positive end (2-7) of $(\hat{W}^{12}, \hat{\mu}^{12,s})$ will be denoted by $e_s^+: SY^1 \to \hat{W}^{12}$. By definition, we can find a neighborhood $U^+ \subset SY^1$ of $+\infty$ and a neighborhood $U^- \subset SY^1$ such that the restriction of e_s^{\pm} to U^{\pm} is independent of *s*. This common restriction will be denoted by e^{\pm} .

Fix a large t > 0 so that the intersection $V := \mu_t^{-1}(U^+) \cap U^-$ is nonempty. Let $\hat{W}^{01} \#_V \hat{W}^{12}$ be the space obtained by gluing $\hat{W}^{01} \setminus e^-(U^- \setminus V)$ and $\hat{W}^{12} \setminus e^+(U^+ \setminus \mu^t(V))$ along the maps

$$V \xrightarrow{e^{-}} \widehat{W}^{01} \setminus e^{-}(U^{-} \setminus V)$$

$$\downarrow^{\mu_{t}}$$

$$U^{+} \xrightarrow{e^{+}} \widehat{W}^{12} \setminus e^{+}(U^{+} \setminus \mu^{t}(V))$$

As a smooth manifold, $\hat{W}^{01} \#_V \hat{W}^{12}$ is canonically identified with $\hat{W}^{01} \#_t \hat{W}^{12}$. Thus we can view $\hat{\mu}^{01,s} \#_t \hat{\mu}^{12,s}$ as a Liouville form on $\hat{W}^{01} \#_V \hat{W}^{12}$ for each *s*, making $(\hat{W}^{01} \#_V \hat{W}^{12}, \hat{\mu}^{01,s} \#_t \hat{\mu}^{12,s})_{s \in [0,1]}$ into a one-parameter family of cobordisms. In particular, it follows that $(\hat{W}^{01} \#_t \hat{W}^{12}, \hat{\mu}^{01,0} \#_t \hat{\mu}^{12,0})$ and $(\hat{W}^{01} \#_t \hat{W}^{12}, \hat{\mu}^{01,1} \#_t \hat{\mu}^{12,1})$ are deformation equivalent.

Corollary 2.15 There is a well-defined gluing operation on deformation classes of exact symplectic cobordisms given by

(2-12) $[\hat{X}^{01}, \hat{\lambda}^{01}] \# [\hat{X}^{12}, \hat{\lambda}^{12}] = [\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12}]$

for any $t \ge 0$ and $s \in \mathbb{R}$.

Proposition 2.16 The gluing operation (2-12) is associative.

Proof This follows from Remark 2.8.

Geometry & Topology, Volume 28 (2024)

We now discuss submanifolds in exact symplectic cobordisms. Let $V \subset (Y, \xi)$ be a contact submanifold. There is a canonical embedding

$$(2-13) (SV, \lambda_V) \to (SY, \lambda_Y)$$

which takes a pair $(p, \alpha_p) \in SV$ to the unique pair $(p, \tilde{\alpha}_p) \in SY$ such that $\tilde{\alpha}_p(w) = \alpha_p(w)$ for some (and hence any) $w \in T_pV \setminus (\xi_p \cap T_pV)$.

Definition 2.17 Let $V^+ \subset (Y^+, \xi^+)$ and $V^- \subset (Y^-, \xi^-)$ be contact submanifolds of the same codimension, and let $(\hat{X}, \hat{\lambda})$ be an exact symplectic cobordism from (Y^+, ξ^+) to (Y^-, ξ^-) . We say that a smooth submanifold $H \subset \hat{X}$ is *cylindrical* with ends V^{\pm} if it is closed (as a subset) and there exist neighborhoods $U^{\pm} \subset SY^{\pm}$ of $\pm \infty$ such that

(2-14)
$$(e^{\pm})^{-1}(H) \cap U^{\pm} = SV^{\pm} \cap U^{\pm},$$

where $e^{\pm}: SY^{\pm} \to \hat{X}$ are the ends (2-3)–(2-4) of $(\hat{X}, \hat{\lambda})$.

If *H* is a *symplectic* cylindrical submanifold of $(\hat{X}, \hat{\lambda})$, then we say that $(\hat{X}, \hat{\lambda}, H)$ is an *exact relative symplectic cobordism* from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) . Note that in this case, the restrictions of e^{\pm} to $SV^{\pm} \cap U^{\pm}$ endow $(H, \hat{\lambda}|_H)$ with the structure of an exact symplectic cobordism from $(V^+, \xi^+_{V^+})$ to $(V^-, \xi^-_{V^-})$.

Example 2.18 If *V* is a contact submanifold of (Y, ξ) , then, as noted above, *SV* can be viewed as a symplectic submanifold of (SY, λ_Y) , and (SY, λ_Y, SV) is canonically endowed with the structure of an exact relative symplectic cobordism in the sense of Definition 2.17 by letting $e^+ = e^- = id$.

Notation 2.19 Let \hat{X} , $\hat{\lambda}$ and H be as in Definition 2.17. As explained in Notation 2.4, a choice of contact forms ker $\lambda^{\pm} = \xi^{\pm}$ endows $(\hat{X}, \hat{\lambda})$ with the structure of a *strict* relative symplectic cobordism. We analogously speak of a *strict* relative exact symplectic cobordism and write $(\hat{X}, \hat{\lambda}, H)_{\lambda^{-}}^{\lambda^{+}}$ when we wish to emphasize that we are fixing contact forms λ^{\pm} on the ends.

Let $\Lambda \subset (Y, \xi)$ be a Legendrian submanifold. The Lagrangian cone of Λ is the Lagrangian submanifold

(2-15)
$$L = \{(p, \alpha) \in SY \subset T^*Y \mid p \in \Lambda\} \subset (SY, \lambda_Y).$$

Definition 2.20 Let $\Lambda^+ \subset (Y^+, \xi^+)$ and $\Lambda^- \subset (Y^-, \xi^-)$ be Legendrian submanifolds and let $(\hat{X}, \hat{\lambda})$ be an exact symplectic cobordism from (Y^+, ξ^+) to (Y^-, ξ^-) . We say that a Lagrangian submanifold $L \subset (\hat{X}, \hat{\lambda})$ is *cylindrical* with ends Λ^{\pm} if it is closed (as a subset) and there exist neighborhoods $U^{\pm} \subset SY^{\pm}$ of $\pm \infty$ such that

(2-16)
$$(e^{\pm})^{-1}(L) \cap U^{\pm} = L^{\pm} \cap U^{\pm},$$

where $e^{\pm}: SY^{\pm} \to \hat{X}$ are the ends (2-3)–(2-4) of $(\hat{X}, \hat{\lambda})$ and L^{\pm} are the Lagrangian cones of Λ^{\pm} . A triple $(\hat{X}, \hat{\lambda}, L)$ is called an *(exact) Lagrangian cobordism* from (Y^+, ξ^+, Λ^+) to (Y^-, ξ^-, Λ^-) .

Definition 2.21 The set of equivalence classes of cylindrical codimension 2 submanifolds of \hat{X} with ends V^{\pm} , where two submanifolds are equivalent if they are isotopic via a compactly supported (smooth) isotopy, will be denoted by $\Omega_{2n-2}(\hat{X}, V^+ \sqcup V^-)$.

Definition 2.22 A contact submanifold $V \subset (Y, \lambda)$ is said to be a *strong contact submanifold* if it is (setwise) invariant under the Reeb flow of λ on Y. We will also say that $(\hat{X}, \hat{\lambda}, H)_{\lambda^{-}}^{\lambda^{+}}$ is a *strong* relative exact symplectic cobordism if both $V^{+} \subset (Y^{+}, \lambda^{+})$ and $V^{-} \subset (Y^{-}, \lambda^{-})$ are strong contact submanifolds.

Definition 2.23 Let $(\hat{X}^{01}, \hat{\lambda}^{01}, H^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12}, H^{12})$ be exact relative symplectic cobordisms from (Y^0, ξ^0, V^0) to (Y^1, ξ^1, V^1) and from (Y^1, ξ^1, V^1) to (Y^2, ξ^2, V^2) , respectively. For any sufficiently large real number $t \ge 0$, $H^{01} \#_t H^{12}$ sits naturally inside $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{12})$ as a symplectic submanifold, and $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{12}, H^{01} \#_t H^{12})$ is a relative cobordism from (Y^0, ξ^0, V^0) to (Y^2, ξ^2, V^2) . We will refer to it as the *t*-gluing of $(\hat{X}^{01}, \hat{\lambda}^{01}, H^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12}, H^{12})$.

Definition 2.24 Let $(\hat{X}^1, \hat{\lambda}^1, H^1)$ and $(\hat{X}^2, \hat{\lambda}^2, H^2)$ be relative cobordisms from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) . An *isomorphism of exact relative symplectic cobordisms* $\phi: (\hat{X}^1, \hat{\lambda}^1, H^1) \rightarrow (\hat{X}^2, \hat{\lambda}^2, H^2)$ is an isomorphism $\phi: (\hat{X}^1, \hat{\lambda}^1) \rightarrow (\hat{X}^2, \hat{\lambda}^2)$ in the sense of Definition 2.10 which maps H^1 diffeomorphically onto H^2 .

Example 2.25 Suppose that $(\hat{X}, \hat{\lambda}, H)$ is an exact relative symplectic cobordism from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) . Then for any $t \ge 0$, the glued cobordisms $(SY^+, \lambda_{Y^+}, SV^+) \#_t (\hat{X}, \hat{\lambda}, H)$ and $(\hat{X}, \hat{\lambda}, H) \#_t (SY^-, \lambda_{Y^-}, SV^-)$ are defined and canonically isomorphic to $(\hat{X}, \hat{\lambda}, H)$.

Definition 2.26 A one-parameter family of exact relative symplectic cobordisms from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) is a manifold \hat{X} equipped with a family of Liouville forms $\{\hat{\lambda}^t\}_{t \in I}$, a family of symplectic submanifolds $H^t \subset (\hat{X}, \hat{\lambda}^t)$, and embeddings

 $(2-17) e_t^+: SY^+ \to \hat{X},$

$$(2-18) e_t^-: SY^- \to \hat{X}$$

as in Definition 2.17. We will always assume that the family is fixed at infinity, meaning that for every compact subinterval $[a, b] \subset I$,

- $\{\hat{\lambda}^t\}_{t\in[a,b]}$ and $\{H^t\}_{t\in[a,b]}$ are constant outside of a compact subset of \hat{X} , and
- {e_t⁺}_{t∈[a,b]} (resp. {e_t⁻}_{t∈[a,b]}) is independent of t on some neighborhood of +∞ in SY⁺ (resp. of -∞ in SY⁻).

Two relative cobordisms $(\hat{X}^0, \hat{\lambda}^0, H^0)$ and $(\hat{X}^1, \hat{\lambda}^1, H^1)$ are said to be *deformation equivalent* if there exists a one-parameter family $(\hat{W}, \hat{\mu}^t, K^t)_{t \in [0,1]}$ such that $(\hat{X}^0, \hat{\lambda}^0, H^0)$ is isomorphic to $(\hat{W}, \hat{\mu}^0, K^0)$ and $(\hat{X}^1, \hat{\lambda}^1, H^1)$ is isomorphic to $(\hat{W}, \hat{\mu}^1, K^1)$. The deformation class of a cobordism $(\hat{X}, \hat{\lambda}, H)$ will be denoted by $[\hat{X}, \hat{\lambda}, H]$.

Example 2.27 Given $(\hat{X}^{01}, \hat{\lambda}^{01}, H^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12}, H^{12})$ as in Definition 2.23, the glued cobordisms $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{12}, H^{01} \#_t H^{12})_{t \in [N,\infty)}$ form a one-parameter family for N > 0 sufficiently large. Similarly, for any fixed $t \gg 0$, $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12}, H^{01} \#_t H^{12})_{s \in \mathbb{R}}$ is a one-parameter family. As in Lemma 2.13, it follows that the deformation class

(2-19)
$$[\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12}, H^{01} \#_t H^{12}]$$

is independent of $t \gg 0$ and $s \in \mathbb{R}$.

Proposition 2.28 The deformation class of $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{12}, H^{01} \#_t H^{12})$ only depends on the deformation classes of $(\hat{X}^{01}, \hat{\lambda}^{01}, H^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12}, H^{12})$.

Proof The proof of Proposition 2.14 also works in the relative case as long as t > 0 is chosen large enough.

Corollary 2.29 There is a well-defined gluing operation on deformation classes of exact relative symplectic cobordisms given by

(2-20)
$$[\hat{X}^{01}, \hat{\lambda}^{01}, H^{01}] \# [\hat{X}^{12}, \hat{\lambda}^{12}, H^{12}] = [\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_{t,s} \hat{\lambda}^{12}, H^{01} \#_t H^{12}]$$

for any $t \gg 0$ and $s \in \mathbb{R}$.

Proposition 2.30 The gluing operation (2-20) is associative.

Proof Let $(\hat{X}^{i,i+1}, \hat{\lambda}^{i,i+1}, H^{i,i+1})$ be a relative cobordism from (Y^i, ξ^i, V^i) to $(Y^{i+1}, \xi^{i+1}, V^{i+1})$ for $i \in \{0, 1, 2\}$, and fix $t_1, t_2 \gg 0$. Note that $((\hat{X}^{01} \#_{t_1} \hat{X}^{12}) \#_{t_2} \hat{X}^{23}, (H^{01} \#_{t_1} H^{12}) \#_{t_2} H^{23})$ and $(\hat{X}^{01} \#_{t_1} (\hat{X}^{12} \#_{t_2} \hat{X}^{23}), H^{01} \#_{t_1} (H^{12} \#_{t_2} H^{23}))$ can be canonically identified as pairs of smooth manifolds. Hence, it suffices to show that there exist $s_1, s_2 \in \mathbb{R}$ such that

(2-21)
$$(\hat{\lambda}^{01} \#_{t_1,s_1} \hat{\lambda}^{12}) \#_{t_2,s_2} \hat{\lambda}^{23} = \hat{\lambda}^{01} \#_{t_1,s_1} (\hat{\lambda}^{12} \#_{t_2,s_2} \hat{\lambda}^{23}).$$

One can easily see from Definition 2.7 that taking $s_1 = 0$ and $s_2 = -t_2$ works.

2.2 Homotopy classes of asymptotically cylindrical maps

Definition 2.31 [68, Section 6.1] Suppose that $(\hat{X}, \hat{\lambda})$ is an exact symplectic cobordism from (Y^+, λ^+) to (Y^-, λ^-) . Given a closed surface Σ and finite subsets $p^+, p^- \subset \Sigma$ (corresponding respectively to positive and negative punctures), a smooth map $u : \Sigma - (p^+ \sqcup p^-) \to \hat{X}$ is said to be *asymptotically cylindrical* if it converges exponentially near each puncture $z \in p^+ \sqcup p^-$ to a trivial cylinder over a Reeb orbit.

More precisely, given any choice of translation invariant metric on $\mathbb{R} \times Y^{\pm}$, we require that there exist a choice of holomorphic cylindrical coordinates near each $z \in p^{\pm}$ such that *u* takes the form

(2-22)
$$u(s,t) = \exp_{(Ps,\gamma_z(t))} h(s,t)$$

for |s| large, where γ_z is a Reeb orbit of period P and h(s, t) is a vector field which decays to zero with all its derivatives as $|s| \to \infty$ (ie these properties hold for $s \gg 0$ if $z \in p^+$ and for $s \ll 0$ if $z \in p^-$).

Geometry & Topology, Volume 28 (2024)

Remark 2.32 There is also a notion of an asymptotically cylindrical submanifold which will not be needed in this paper.

Definition 2.33 (cf [61, Section 1.2(I)]) Let $(\hat{X}, \hat{\lambda})$ be an exact symplectic cobordism from (Y^+, λ^+) to (Y^-, λ^-) and let Γ^{\pm} be a finite set of Reeb orbits in (Y^{\pm}, λ^{\pm}) .

By truncating the ends of \hat{X} , we obtain a compact submanifold $X_0 \subset \hat{X}$ with boundary $\partial X_0 = Y^+ \sqcup Y^-$. We define the set of homotopy classes $\pi_2(\hat{X}, \Gamma^+ \sqcup \Gamma^-)$ by

(2-23)
$$\pi_2(\hat{X}, \Gamma^+ \sqcup \Gamma^-) := [(S, \partial S), (X_0, \Gamma^+ \sqcup \Gamma^-)] / \operatorname{Diff}(S, \partial S),$$

where S is a compact connected oriented surface of genus 0 equipped with a homeomorphism $\partial S \rightarrow \Gamma^+ \sqcup \Gamma^-$, and Diff $(S, \partial S)$ is the group of diffeomorphisms of S which fix ∂S pointwise. (The notation [-, -] here stands for *homotopy classes* of maps of pairs.)

Remark 2.34 The right-hand side of (2-23) is independent of the choice of truncation X_0 up to canonical bijection. In the case where $(\hat{X}, \hat{\lambda}) = (\mathbb{R} \times Y, e^s \lambda)$ is the symplectization of a contact manifold (Y, λ) , we can take $X_0 = \{0\} \times Y$ and (2-23) becomes identical to [61, equation (1.2)].

For any choice of truncation $X_0 \subset \hat{X}$, there is a canonical retraction $\pi : \hat{X} \to X_0$ induced by quotienting by the Liouville flow (more precisely, one should quotient by the backwards Liouville flow at the positive end and by the forwards Liouville flow at the negative end). If $u: \Sigma - (p^+ \sqcup p^-) \to \hat{X}$ is an asymptotically cylindrical map, then the composition $\pi \circ u$ can be extended to a map

(2-24)
$$\overline{u}: (\overline{\Sigma}, \partial \overline{\Sigma}) \to (X_0, \Gamma^+ \sqcup \Gamma^-),$$

where $\overline{\Sigma}$ is a compactification of $\Sigma - (p^+ \sqcup p^-)$ obtained by adding one boundary circle for each puncture. The homotopy class $[u] \in \pi_2(\widehat{X}, \Gamma^+ \sqcup \Gamma^-)$ of u is defined to be the equivalence class of (2-24).

Definition 2.35 Let $(\hat{X}, \hat{\lambda}, L)$ be an exact Lagrangian cobordism from $(Y^+, \lambda^+, \Lambda^+)$ to $(Y^-, \lambda^-, \Lambda^-)$; see Definition 2.20.

Given a surface with boundary Σ and finite subsets p^+ , $p^- \subset int(\Sigma)$ and c^+ , $c^- \in \partial \Sigma$, a smooth map $u: \Sigma - (p^{\pm} \cup c^{\pm})$ is said to be cylindrical if it converges asymptotically near each interior puncture to a trivial cylinder over a Reeb orbit, and it converges exponentially near each boundary puncture to a trivial strip over a Reeb chord.

Let Γ^{\pm} be a finite set of Reeb orbits in (Y^{\pm}, λ^{\pm}) and let $\Gamma_{\Lambda^{\pm}}$ be a finite *ordered* set of Reeb chords of $\Lambda^{\pm} \subset (Y^{\pm}, \lambda^{\pm})$. We let p^{\pm} be a finite set equipped with bijections $\gamma^{\pm} : p^{\pm} \to \Gamma^{\pm}$ and we let $c = c^{+} \sqcup c^{-}$ be a finite ordered set equipped with order-preserving bijections $a^{\pm} : c^{\pm} \to \Gamma_{\Lambda^{\pm}}$. We let

(2-25)
$$\pi_2(\hat{X};\Gamma_{\Lambda^+},\Gamma_{\Lambda^-},\Gamma^+,\Gamma^-)$$

be the set of equivalence classes of maps from $\Sigma - (p^{\pm} \cup c^{\pm})$ to \hat{X} which are asymptotic to γ_p^{\pm} at $p \in p^{\pm}$ (resp. a_c^{\pm} at $c \in c^{\pm}$), where two such maps u, v are equivalent if there exists a compactly supported diffeomorphism ϕ of $\Sigma - (p^{\pm} \cup c^{\pm})$ such that u and $v \circ \phi$ are homotopic (through cylindrical maps).

3 Reeb dynamics near a codimension 2 contact submanifold

3.1 The Conley–Zehnder index

A hermitian vector bundle (E, J, ω) is a vector bundle *E* together with an almost complex structure *J* and a symplectic structure ω such that *J* is compatible with ω . An *asymptotic operator* on a hermitian vector bundle (E, J, ω) over S^1 is a real-linear differential operator $A : \Gamma(E) \to \Gamma(E)$ which, in some (and hence any) unitary trivialization, takes the form

(3-1)
$$A: C^{\infty}(S^1, \mathbb{R}^{2n}) \to C^{\infty}(S^1, \mathbb{R}^{2n}), \quad \eta \mapsto -J \partial_t \eta - S(t)\eta,$$

where $t \in S^1$ and $S: S^1 \to \text{End}(\mathbb{R}^{2n})$ is a loop of symmetric matrices. The asymptotic operator A is said to be nondegenerate if 0 is not an eigenvalue.

Fix a hermitian vector bundle (E, J, ω) over S^1 and a unitary trivialization τ . Given a nondegenerate asymptotic operator A, we can obtain a nondegenerate path of symplectic matrices by solving the ordinary differential equation

(3-2)
$$(-J\partial_t - S(t))\Psi(t) = 0, \quad \Psi(0) = \mathrm{id}.$$

Conversely, given a nondegenerate path of symplectic matrices, we can recover a nondegenerate asymptotic operator by solving (3-2) for S(t).

The *Conley–Zehnder index* $CZ(A) = CZ(\Psi) \in \mathbb{Z}$ is an integer-valued invariant which can be associated equivalently to a nondegenerate asymptotic operator equipped with a unitary trivialization τ or to a nondegenerate path of symplectic matrices. It only depends on τ up to homotopy through unitary trivializations. We refer the reader to [68, Section 3.4] or [33] for a detailed overview of the Conley–Zehnder index.

Definition 3.1 Let $(Y, \xi = \ker \lambda)$ be a contact manifold and let γ be a Reeb orbit of period P > 0, parametrized so that $\lambda(\gamma') = P$. Given a choice of $d\lambda$ -compatible almost complex structure J on ξ , we can define the asymptotic operator $A_{\gamma} \colon \Gamma(\gamma^*\xi) \to \Gamma(\gamma^*\xi)$ by $A_{\gamma} = -J(\nabla_t - P \nabla R_{\lambda})$, where ∇ is some symmetric connection on Y.

The Conley–Zehnder index of a Reeb orbit γ relative to a trivialization τ of $\gamma^* \xi$ will be denoted by $CZ^{\tau}(\gamma) := CZ^{\tau}(A_{\gamma}).$

Let us now consider a contact manifold $(Y^{2n-1}, \xi = \ker \lambda)$ and a strong contact submanifold $(V^{2n-3}, \lambda|_V)$. Observe that the contact distribution splits naturally along V as

(3-3)
$$\xi|_V = \xi|_V^\top \oplus \xi|_V^\perp,$$

where $\xi|_V^{\top} = \xi|_V \cap TV$ and $\xi|_V^{\perp}$ is the symplectic orthogonal complement of $\xi|_V^{\top} \subset \xi|_V$ with respect to $d\lambda$. Suppose that *J* is a $d\lambda$ -compatible almost complex structure on ξ which respects the splitting (3-3). Let $\gamma: S^1 \to V$ be a Reeb orbit. Since *J* respects the above splitting, then so does the associated asymptotic operator, which we can therefore write as $A_{\gamma} = A_{\gamma}^{\top} \oplus A_{\gamma}^{\perp}$. If we choose a unitary trivialization τ of $\xi|_{\gamma}$ which is also compatible with the splitting, we can define $CZ_T^{\tau}(\gamma) := CZ^{\tau}(A_{\gamma}^{\top})$ and $CZ_N^{\tau}(\gamma) := CZ^{\tau}(A_{\gamma}^{\perp})$. We call these respectively the tangential and normal Conley–Zehnder indices of γ with respect to τ .

We define the integers

(3-4)
$$\alpha_N^{\tau;-}(\gamma) := \lfloor \operatorname{CZ}_N^{\tau}(\gamma)/2 \rfloor,$$

(3-5)
$$\alpha_N^{\tau;+}(\gamma) := \lceil \operatorname{CZ}_N^{\tau}(\gamma)/2 \rceil.$$

Let $p_N(\gamma) = \alpha_N^{\tau;+}(\gamma) - \alpha_N^{\tau;-}(\gamma) \in \{0, 1\}$ be the *(normal) parity* of γ , and observe that it is independent of the choice of trivialization. We have

(3-6)
$$CZ_N^{\tau}(\gamma) = 2\alpha_N^{\tau;-}(\gamma) + p_N(\gamma) = 2\alpha_N^{\tau;+}(\gamma) - p_N(\gamma),$$

from which it also follows that $p_N(\gamma) \equiv CZ_N^{\tau}(\gamma) \mod 2$.

There is a canonical isomorphism

$$(3-7) \qquad \qquad \xi|_V^{\perp} \xrightarrow{\sim} N_{Y/V}$$

where $N_{Y/V}$ denotes the normal bundle of $V \subset Y$. If $(V, \xi_V) \subset (Y, \xi)$ are co-oriented and hence oriented, then $N_{Y/V}$ is also oriented and (3-7) is orientation-preserving. If we assume that $N_{Y/V}$ is trivial, then it follows that (3-7) induces a bijection between homotopy classes of trivializations of $N_{Y/V}$ compatible with the orientation and homotopy classes of unitary trivializations on $\xi|_V^{\perp}$. Since the Conley–Zehnder index only depends on the homotopy class of unitary trivializations, we may define $CZ_N^{\tau}(\gamma)$ with respect to any homotopy class of trivializations τ of $N_{Y/V}$ compatible with the orientation.

3.2 Normal dynamics and adapted contact forms

We now state some important definitions, which will be used throughout this paper.

Definition 3.2 A *trivial-normal contact pair* (or just *TN contact pair*) is a datum (Y, ξ, V) consisting of a closed co-oriented contact manifold (Y, ξ) and a co-oriented codimension 2 contact submanifold $V \subset (Y, \xi)$ with trivial normal bundle $N_{Y/V}$.

An important example of a TN contact pair is the binding of a contact open book decomposition; see Definition 3.10. A choice of (homotopy class of) trivialization τ on $N_{Y/V}$ is called a *framing* and we say that $(V, \tau) \subset Y$ is a *framed* codimension 2 submanifold. Note that we do not assume in Definition 3.2 that V and Y are nonempty. For future reference, we let ξ_{\emptyset} denote the unique contact structure on the empty set.

Definition 3.3 Given a TN contact pair (Y, ξ, V) with *V* nonempty, let $\Re(Y, \xi, V)$ be the set of triples $\mathfrak{r} = (\alpha_V, \tau, r)$ where

• $\alpha_V \in \Omega^1(V)$ is a nondegenerate contact form for (V, ξ_V) ,

- τ is a homotopy class of trivializations of $N_{Y/V}$ (ie a framing) which is compatible with the orientation, and
- *r* > 0 is a strictly positive real number, and we have (1/*r*)ℤ ∩ S(α_V) = Ø, where S(α_V) is the action spectrum of α_V.

If $V = \emptyset$ (with Y possibly also empty), we define $\Re(Y, \xi, \emptyset) = \{(\alpha_{\emptyset}, \tau_{\emptyset}, 0)\}$, where α_{\emptyset} and τ_{\emptyset} are understood as a contact form and normal trivialization on the empty set.

Definition 3.4 Given a TN contact pair (Y, ξ, V) and $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V)$, we say that a contact form ker $\lambda = \xi$ is *adapted* to \mathfrak{r} if

- λ is nondegenerate,
- $\lambda|_V = \alpha_V$,
- *V* is a strong contact submanifold of (Y, λ) (see Definition 2.22), and
- we have $CZ_N^{\tau}(\gamma) = 1 + 2\lfloor rP_{\gamma} \rfloor$ for all Reeb orbits $\gamma \subset V$, where P_{γ} is the period of γ .

In the case that $V = \emptyset$, any contact form λ is considered to be adapted to the unique element $(\alpha_{\emptyset}, \tau_{\emptyset}, 0) \in \Re(Y, \xi, \emptyset)$. Given a contactomorphism $f: (Y, \xi, V) \to (Y', \xi', V')$, we write $f_*\mathfrak{r} = (f_*\alpha_V, f_*\tau, r) \in \Re(Y', \xi', V')$. If $\phi_t: V \to Y$ is an isotopy of contact embeddings where ϕ_0 is the tautological embedding $V \xrightarrow{\text{id}} V \subset Y$ and $\phi_1(V) = V'$, then ϕ_t extends to a family of contactomorphisms f_t . We then write $(\phi_1)_*\mathfrak{r} := (f_1)_*\mathfrak{r}$; this is independent of the choice of extension.

We say that λ is *positive elliptic* near $V \neq \emptyset$ if it is adapted to some $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V)$; we refer to r > 0 as the *rotation parameter*.

Remark 3.5 Our insistence on allowing the case where $Y = \emptyset$ in the above definitions is explained by the need to treat Liouville manifolds as special cases of Liouville cobordisms in the arguments of Section 7.

We will prove in Proposition 3.9 that adapted contact forms always exist, ie for any TN contact pair (Y, ξ, V) and $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V)$ there exists a contact form adapted to \mathfrak{r} . The first step is to construct a suitable local model.

Construction 3.6 For $0 < \epsilon \le 1$, let $D^2 \subset \mathbb{R}^2$ be the standard disk of radius ϵ (in the sequel, we will often denote this disk by D_{ϵ}^2). Let (V, α_V) be a contact manifold and let $\phi : D^2 \to \mathbb{R}_{>0}$ be a smooth positive function which has a nondegenerate critical point at 0 and satisfies $\phi(0) = 1$. We define

(3-8)
$$\alpha_V^{\phi} = \frac{1}{\phi} (\alpha_V + \lambda_{D^2}),$$

where $\lambda_{D^2} = \frac{1}{2}(x \, dy - y \, dx)$ is the usual Liouville form on D^2 . This is a contact form on $V \times D^2$ whose restriction to $V = V \times \{0\}$ coincides with α_V . Its Reeb vector field is given by

(3-9)
$$R_{\phi} = (\phi - Z_{D^2}\phi)R_V + X_{\phi},$$

where $Z_{D^2} = \frac{1}{2}(x\partial_x + y\partial_y)$ is the Liouville vector field of λ_{D^2} and $X_{\phi} = -(\partial_y \phi)\partial_x + (\partial_x \phi)\partial_y$ is the Hamiltonian vector field of ϕ with respect to the symplectic form $\omega_{D^2} = d\lambda_{D^2}$. Our assumptions on ϕ imply that $R_{\phi} = R_V$ on $V \times \{0\}$, so that (V, α_V) is a strong contact submanifold of $(V \times D^2, \alpha_V^{\phi})$. We will let

(3-10)
$$S_{\phi} = \begin{pmatrix} \partial_{xx}\phi(0) & \partial_{yx}\phi(0) \\ \partial_{xy}\phi(0) & \partial_{yy}\phi(0) \end{pmatrix} \in \mathbb{R}^{2 \times 2}$$

denote the Hessian of ϕ at the origin. Since S_{ϕ} is symmetric and nondegenerate, its eigenvalues are real and nonzero, so its signature Sign (S_{ϕ}) is one of 0, ± 2 . We will say that ϕ is *hyperbolic* if Sign $(S_{\phi}) = 0$, *positive elliptic* if Sign $(S_{\phi}) = 2$ and *negative elliptic* if Sign $(S_{\phi}) = -2$. In the elliptic case, we define $c_{\phi} = \sqrt{\det(S_{\phi})}/(2\pi)$; this is a positive real number since $\det(S_{\phi}) > 0$. Finally, we note that the splitting $(\xi_{\phi})|_{V} = (\xi_{\phi})|_{V}^{\top} \oplus (\xi_{\phi})|_{V}^{\perp}$ —see equation (3-3)—is given by $(\xi_{\phi})|_{V}^{\top} = \xi_{V}$ and $(\xi_{\phi})|_{V}^{\perp} = T_{0}D^{2}$. We will let τ_{ϕ} denote the trivialization of $(\xi_{\phi})|_{V}^{\perp}$ by $\{\partial_{x}, \partial_{y}\}$.

We say that α_V^{ϕ} is *nondegenerate on* V if every Reeb orbit of α_V is nondegenerate when viewed as a Reeb orbit of α_V^{ϕ} .

Proposition 3.7 Carrying over the notation of Construction 3.6, suppose that α_V is nondegenerate. If ϕ is elliptic, then α_V^{ϕ} is nondegenerate on *V* if and only if $(1/c_{\phi})\mathbb{Z} \cap S(\alpha_V) = \emptyset$, where $S(\alpha_V)$ denotes the action spectrum of α_V .

Proof Let γ be a Reeb orbit of period P contained in V. Recall that γ is nondegenerate if and only if its asymptotic operator is nondegenerate. Choose a trivialization τ and an almost complex structure J on $(\xi_{\phi})|_{\gamma}$ which preserve the splitting $(\xi_{\phi})|_{\gamma} = (\xi_V)|_{\gamma} \oplus T_0 D^2$ and coincide with τ_{ϕ} and J_0 respectively on $T_0 D^2$, where J_0 denotes the standard almost complex structure on $\mathbb{R}^2 = T_0 D^2$. The asymptotic operator A_{γ} is compatible with this splitting and can therefore be written as $A_{\gamma} = A_{\gamma}^{\top} \oplus A_{\gamma}^{\perp}$. The tangential part A_{γ}^{\top} is nondegenerate since it coincides with the asymptotic operator of γ as a Reeb orbit in V. The normal part A_{γ}^{\perp} is given explicitly by

(this follows from a short computation using the formula for the Reeb vector field R_{ϕ} in Construction 3.6). Define a path Ψ of symplectic matrices by $\Psi(t) = \exp(tP \cdot J_0 S_{\phi})$. Then A_{γ}^{\perp} is nondegenerate if and only if $\Psi(1)$ doesn't have 1 as an eigenvalue. If ϕ is elliptic, then the eigenvalues of $\Psi(1)$ are $\exp(\pm iP \sqrt{\det(S_{\phi})})$. Hence, γ is nondegenerate if and only if $P \sqrt{\det(S_{\phi})}$ is not an integer multiple of 2π , ie $P \notin (1/c_{\phi})\mathbb{Z}$. It follows that λ_{ϕ} is nondegenerate if and only if $(1/c_{\phi})\mathbb{Z} \cap S(\alpha_V) = \emptyset$, as claimed.

The important feature of Construction 3.6 is that the normal Conley–Zehnder indices of the Reeb orbits in V can be computed explicitly.

Proposition 3.8 Assume α_V^{ϕ} is nondegenerate on V. If ϕ is elliptic, then

(3-12)
$$CZ_N^{\tau_\phi}(\gamma) = \pm (1 + 2\lfloor c_\phi P_\gamma \rfloor)$$

for every Reeb orbit γ contained in V, where $P_{\gamma} > 0$ denotes the period of γ and the sign is + or – depending on whether ϕ is positive elliptic or negative elliptic.

Proof We have $CZ_N^{\tau_{\phi}}(\gamma) = CZ(\Psi)$, where $\Psi(t) = \exp(tP \cdot J_0 S_{\phi})$ is the path of symplectic matrices defined in the proof of Proposition 3.7; see [68, Section 3.4]. Proposition 41 of [33] implies that

(3-13) $CZ(\Psi) = \pm (1 + 2\lfloor c_{\phi} P \rfloor)$

if $\operatorname{Sign}(S_{\phi}) = \pm 2$.

Proposition 3.9 Fix a TN contact pair (Y, ξ, V) and an element $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V)$. Then there exists a contact form λ on (Y, ξ) which is adapted to \mathfrak{r} .

Proof Let ϕ be as in Construction 3.6. The standard neighborhood theorem for contact submanifolds (see [29, Theorem 2.5.15]) implies that the inclusion map $V \to Y$ extends to a contact embedding $\iota: (V \times D_{\epsilon}^2, \ker(\alpha_V^{\phi})) \to (Y, \xi)$ such that $\iota^* \tau$ is homotopic to τ_{ϕ} , for some $\epsilon > 0$ sufficiently small. Hence there exists a contact form λ for ξ such that $\iota^* \lambda = \alpha_V^{\phi}$ near V. In addition to choosing ϕ so that λ is adapted to τ , we also need to make sure that λ can be modified away from V so that it becomes nondegenerate. By [2, Theorem 13], this can be achieved by choosing a ϕ such that the following two conditions are satisfied:

- α_V^{ϕ} is nondegenerate on V.
- All the Reeb orbits of α_V^{ϕ} in $V \times D_{\epsilon}^2$ are contained in V.

Let us set $\phi = 1 + \pi r(x^2 + y^2)$. Proposition 3.7 implies that α_V^{ϕ} is nondegenerate on V since

$$c_{\phi} = \frac{\sqrt{(2\pi r)^2}}{2\pi} = r.$$

Since $R_{\phi} = R_V + X_{\phi}$, every Reeb orbit γ of α_V^{ϕ} is of the form $\gamma = (\gamma_V, \gamma_{\phi})$, where γ_V is an orbit of R_V and γ_{ϕ} is an orbit of X_{ϕ} with the same period P > 0. From the formula $X_{\phi} = -2\pi r \gamma \partial_x + 2\pi r \chi \partial_y$, we see that if γ_{ϕ} were not constant, we would have $P \in (1/r)\mathbb{Z}$, contradicting our assumption on r; see Definition 3.3. Thus γ is contained in V.

3.3 Open book decompositions

In this section, we consider normal Reeb dynamics for bindings of open book decompositions. We begin by recalling the definition of an open book decomposition; we refer to [62, Section A.1; 29, Section 4.4.2] for a historically informed survey of this theory.

Geometry & Topology, Volume 28 (2024)

Definition 3.10 An *open book decomposition* (Y, B, π) of a closed, oriented *n*-manifold *Y* consists of the following data:

- (i) An oriented, closed, codimension 2 submanifold $B \subset Y$ with trivial normal bundle.
- (ii) A fibration $\pi: Y B \to S^1$ which coincides with the angular coordinate in some neighborhood $B \times \{0\} \subset B \times D^2 = B \times \{(x, y) \mid x^2 + y^2 < 1\}.$

The submanifold $B \subset Y$ is called the *binding* and the fibers of π are called *pages*.

Observe that the data of an open book decomposition induces a natural trivialization of the normal bundle to the binding. We also recall what it means for an open book decomposition to support a contact structure.

Definition 3.11 [31] Given an odd-dimensional manifold Y^{2n-1} , an open book decomposition (Y, B, π) is said to *support* a contact structure ξ if there exists a contact form $\xi = \ker \alpha$ such that the following properties hold:

- (i) The restriction of α to *B* is a contact form.
- (ii) The restriction of $d\alpha$ to any page $\pi^{-1}(\theta)$ is a symplectic form.
- (iii) The orientation of *B* induced by α coincides with the orientation of *B* as the boundary of the symplectic manifold $(P_{\theta}, d\alpha)$, where $P_{\theta} = \pi^{-1}(\theta)$ is any page.

Such a contact form is called a *Giroux form* (and is also said in the literature to be *adapted* to the open book decomposition).

Remark 3.12 Condition (ii) in the above definition is equivalent to the Reeb vector field of α being transverse to the pages.

For future convenience, we state the following definition.

Definition 3.13 Let \mathcal{G} be the set of TN contact pairs (Y, ξ, V) having the property that ξ is supported by an open book decomposition $\pi: Y - V \to S^1$ with binding V.

Lemma 3.14 Let (Y, B, π) be an open book decomposition supporting the contact structure ξ . Let α_B be a contact form for (B, ξ_B) and let $f : [0, 1) \to \mathbb{R}$ be a positive smooth function such that f(0) = 1 and f'(r) < 0 for r > 0. Then there exists a Giroux form α and an embedding $\phi : B \times D_{\epsilon}^2 \to Y$ (for some small $\epsilon > 0$) with the following properties:

- (1) $\alpha|_B = \alpha_B$.
- (2) The projection $\pi \circ \phi$ is given by $(r, \theta) \mapsto \theta$ on $B \times D_{\epsilon}^2 B \times \{0\}$.
- (3) $\phi^* \alpha = f(r)(\alpha|_B + \lambda_{D^2}).$

Proof The proof is in two steps. First, we will show that there exists a Giroux form $\tilde{\alpha}$ such that $\tilde{\alpha}|_B = \alpha_B$. Second, we will construct α by modifying $\tilde{\alpha}$ so that (1)–(3) are satisfied.

Step 1 Let α' be an arbitrary Giroux form for (Y, B, π) . Since $\alpha'|_B$ and α_B define the same contact structure, we can write $\alpha_B = (1+h)(\alpha'|_B)$ for some smooth function $h: B \to \mathbb{R}$. We may assume without loss of generality that $h \ge 0$ everywhere, since any positive constant multiple of α' is also a Giroux form.

It is shown in the proof of [18, Proposition 2] that there exists a tubular neighborhood $B \times D_{\epsilon}^2$ of the binding on which $\pi = \theta$ and $\alpha' = g(\alpha'|_B + \lambda_{D^2})$, where $g: B \times D_{\epsilon}^2 \to \mathbb{R}$ is a positive smooth function satisfying $g \equiv 1$ on $B \times \{0\}$, $\lambda_{D^2} = \frac{1}{2}(x \, dy - y \, dx)$ and $\epsilon > 0$ is a suitably small constant. Note that $g(\alpha'|_B + \lambda_{D^2})$ is a Giroux form on $(B \times D_{\epsilon}^2, B \times \{0\}, \pi = \theta)$ if and only if $\partial g/\partial r < 0$ for r > 0.

Let $\sigma: [0, \epsilon] \to \mathbb{R}$ be a nonincreasing smooth function such that $\sigma(r) = 1$ for r near 0 and $\sigma(r) = 0$ for r near ϵ . Set $\tilde{g} := (1 + \sigma(r)h)g$. Then $\partial_r \tilde{g} = \partial_r \sigma \cdot hg + (1 + \sigma h) \cdot \partial_r g < 0$. Now we define $\tilde{\alpha}$ by replacing g with \tilde{g} .

Step 2 By the previous step, we may fix a Giroux form $\tilde{\alpha}$ so that $\tilde{\alpha}|_B = \alpha_B$. Appealing again to the proof of [18, Proposition 2], there exists a tubular neighborhood $B \times D_{\epsilon'}^2$ of the binding on which $\pi = \theta$ and $\tilde{\alpha} = \gamma(\tilde{\alpha}|_B + \lambda_{D^2})$, where $\gamma : B \times D_{\epsilon'}^2 \to \mathbb{R}$ is a positive smooth function satisfying $\gamma \equiv 1$ on $B \times \{0\}$, $\lambda_{D^2} = \frac{1}{2}(x \, dy - y \, dx)$ and $\epsilon' > 0$ is a suitably small constant. Again, we have that $\gamma(\tilde{\alpha}|_B + \lambda_{D^2})$ is a Giroux form on $(B \times D_{\epsilon'}^2, B \times \{0\}, \pi = \theta)$ if and only if $\partial \gamma / \partial r < 0$ for r > 0.

Let $\delta: B \times D^2_{\epsilon'} \to \mathbb{R}$ be a positive smooth function such that $\delta = f$ near $B \times \{0\}, \delta = \gamma$ near $B \times \partial D^2_{\epsilon'}$, and $\partial_r \delta < 0$ for r > 0. Let α be the unique contact form on Y which coincides with $\tilde{\alpha}$ outside the image of ϕ and satisfies $\phi^* \alpha = \delta(\tilde{\alpha}|_B + \lambda_{D^2})$. Then α is a Giroux form and satisfies conditions (1)–(3). \Box

Corollary 3.15 Consider an open book decomposition (Y, B, π) which supports a contact structure ξ and let τ denote the induced trivialization of the normal bundle of $B \subset Y$. Choose an element $\mathfrak{r} = (\alpha_B, \tau, r) \in \mathfrak{R}(Y, \xi, B)$. Then there exists a Giroux form α which is adapted to \mathfrak{r} ; see Definition 3.4.

Proof Let $\kappa = \pi r$ and define $f(s) = (1 + \kappa s^2)^{-1}$ for $s \in [0, 1)$. Since f(0) = 1 and f'(s) < 0, it follows that there exists a Giroux form $\tilde{\alpha}$ satisfying the conditions stated in Lemma 3.14. As we observed in the proof of Proposition 3.9, there exists a neighborhood \mathcal{U} of B with the properties that

- $\tilde{\alpha}$ is nondegenerate on \mathcal{U} , and
- all the Reeb orbits in \mathcal{U} are contained in B.

According to [2, Theorem 13], we can obtain a nondegenerate contact form by multiplying $\tilde{\alpha}$ by a smooth function $g: Y \to \mathbb{R}_+$ with $g \equiv 1$ near *B*. Moreover, we can assume that g - 1 is arbitrarily C^1 -small and hence that $g\tilde{\alpha}$ is still a Giroux form.

Since $g\tilde{\alpha} = \alpha$ near *B*, it follows that $(g\tilde{\alpha})|_B = \alpha_B$ and that *B* is a strong contact submanifold with respect to $g\tilde{\alpha}$. Finally, the last point in Definition 3.4 can be verified just as in the proof of Proposition 3.8. \Box

4 Standard setups and tree categories.

Contact homology is defined in [61] by counting pseudoholomorphic curves (and, more generally, pseudoholomorphic buildings) in four setups. To keep track of the combinatorics of these curves, Pardon introduces certain categories of decorated trees. We briefly review this formalism here, referring the reader to [61, Section 2.1] for details.

4.1 Standard setups

Setup I A datum \mathcal{D} for Setup I consists of a triple (Y, λ, J) , where Y is a closed manifold, λ is a nondegenerate contact form on Y and J is a $d\lambda$ -compatible almost complex structure on $\xi = \ker \lambda$.

Setup II A datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, \hat{\lambda}, \hat{J})$ for Setup II consists of

- data $\mathcal{D}^{\pm} = (Y^{\pm}, \lambda^{\pm}, J^{\pm})$ as in Setup I,
- an exact symplectic cobordism $(\hat{X}, \hat{\lambda})$ with positive end (Y^+, λ^+) and negative end (Y^-, λ^-) , and
- a $d\hat{\lambda}$ -tame almost complex structure \hat{J} on \hat{X} which agrees with \hat{J}^{\pm} at infinity.

Setup III A datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, (\hat{X}, \hat{\lambda}^t, \hat{J}^t)_{t \in [0,1]})$ for this setting consists of

- data $\mathcal{D}^{\pm} = (Y^{\pm}, \lambda^{\pm}, J^{\pm})$ as in Setup I,
- a family of exact symplectic cobordisms (X̂, λ̂^t)_{t∈[0,1]} with positive end (Y⁺, λ⁺) and negative end (Y⁻, λ⁻), and
- a $d\hat{\lambda}^t$ -tame almost complex structure \hat{J}^t on \hat{X} which agrees with \hat{J}^{\pm} at infinity.

Note that for every $t_0 \in [0, 1]$, there is a datum $\mathcal{D}^{t=t_0} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, \hat{\lambda}^{t_0}, \hat{J}^{t_0})$ as in Setup II.

Setup IV A datum $\mathcal{D} = (\mathcal{D}^{01}, \mathcal{D}^{12}, (\hat{X}^{02,t}, \hat{\lambda}^{02,t}, \hat{J}^{02,t})_{t \in [0,\infty)})$ for this setting consists of

• data

$$\mathcal{D}^{01} = (\mathcal{D}^0, \mathcal{D}^1, \hat{X}^{01}, \hat{\lambda}^{01}, \hat{J}^{01})$$
 and $\mathcal{D}^{12} = (\mathcal{D}^1, \mathcal{D}^2, \hat{X}^{12}, \hat{\lambda}^{12}, \hat{J}^{12})$

as in Setup II, where $\mathcal{D}^i = (Y^i, \lambda^i, J^i)$ for i = 0, 1, 2,

- a family of exact symplectic cobordisms $(\hat{X}^{02,t}, \hat{\lambda}^{02,t})_{t \in [0,\infty)}$ with positive end (Y^0, λ^0) and negative end (Y^2, λ^2) , which for t large coincides with the t-gluing of $(\hat{X}^{01}, \hat{\lambda}^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12})$, and
- a $d\hat{\lambda}^{02,t}$ -tame almost complex structure $\hat{J}^{02,t}$ on $\hat{X}^{02,t}$ which agrees with \hat{J}^{0} and \hat{J}^{2} at infinity, and is induced by \hat{J}^{01} and \hat{J}^{02} for t large.

4.2 Trees

For each setup, Pardon [61, Section 2.1] defines a category S_* for * = I, II, III, IV, which depends on some datum \mathcal{D} . Each object $T \in S$ is a decorated tree (or forest) representing a certain class of pseudoholomorphic curves (or more generally of buildings). Geometrically, the vertices correspond to curves, edges correspond to asymptotic orbits, and the decorations keep track of additional information (such as the homology classes of the components, and their "level" in the SFT compactification).

A morphism of trees in S_* consists of two pieces of data. First, a contraction of some edges, with the important caveat that only certain contractions are allowed, which depend on the decorations on the tree. Second, one specifies some additional data on external edges, which depends on the decorations of the external edges. Geometrically, a morphism of trees correspond to gluing holomorphic curves, and the data which one specifies on the external edges encodes different ways of moving asymptotic markers. For $T \in S_*$, we let Aut(T) denote the group of automorphisms of T. Given a morphism $T' \to T$, we let Aut(T'/T) \subset Aut(T') be the subgroup of automorphisms of T' which are compatible with $T' \to T$.

In each category S_* , there is an operation called *concatenation* whose input is a collection of trees (satisfying certain conditions, and with additional matching data), and whose output is a single tree. Geometrically, concatenations of trees correspond to "stacking" holomorphic buildings. The precise rules for concatenations are rather involved and depend on the individual setups.

Remark 4.1 A datum \mathcal{D} for Setups II, III, IV determines multiple categories of trees: this is because such a datum itself contains (by definition) data for multiple setups. We always follow the notation of [61, Section 2.1] to denote the resulting tree categories. So, for example, if $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, \hat{\lambda}, \hat{J})$ is a datum for Setup II, we write $\mathcal{S}_{\text{II}} := \mathcal{S}_{\text{II}}(\mathcal{D}), \mathcal{S}_{\text{I}}^+ := \mathcal{S}_{\text{I}}(\mathcal{D}^+)$ and $\mathcal{S}_{\text{I}}^- := \mathcal{S}_{\text{I}}(\mathcal{D}^-)$. Similarly, a datum for Setup III determines categories $\mathcal{S}_{\text{III}}, \mathcal{S}_{\text{II}}^{t=0}, \mathcal{S}_{\text{II}}^{t=1}, \mathcal{S}_{\text{I}}^{\pm}$, and a datum for Setup IV determines categories $\mathcal{S}_{\text{IV}}, \mathcal{S}_{\text{II}}^{01}, \mathcal{S}_{\text{I}}^{12}, \mathcal{S}_{\text{I}}^0, \mathcal{S}_{\text{I}}^1, \mathcal{S}_{\text{I}}^2$.

4.3 Virtual moduli counts

To an object $T \in S_*$, we can associate a moduli space $\mathcal{M}(T)$ [61, Section 2.3] which carries an action of Aut(T) — this action corresponds geometrically to changing asymptotic markers. Note that $T \in S_*$ has a well-defined notion of index and virtual dimension [61, Definition 2.42]. The compactified moduli space $\overline{\mathcal{M}}(T)$ is defined by (see [61, Definition 2.13])

(4-1)
$$\overline{\mathcal{M}}(T) := \bigsqcup_{T' \to T} \mathcal{M}(T') / \operatorname{Aut}(T'/T).$$

Theorem 1.1 in [61] provides a perturbation datum $\theta \in \Theta_*(\mathcal{D})$ and associated virtual moduli counts $\#\overline{\mathcal{M}}(T)_{\theta}^{\text{vir}} \in \mathbb{Q}$ (which are zero for $\operatorname{vdim}(T) \neq 0$) satisfying the *master equations*

(4-2)
$$0 = \sum_{\operatorname{codim}(T'/T)=1} \frac{1}{|\operatorname{Aut}(T'/T)|} \# \overline{\mathcal{M}}(T')_{\theta}^{\operatorname{vir}}$$

Homological invariants of codimension 2 contact submanifolds

and

(4-3)
$$\#\overline{\mathcal{M}}(\#_i T_i)_{\theta}^{\text{vir}} = \frac{1}{|\operatorname{Aut}(\{T_i\}_i / \#_i T_i)|} \prod_i \#\overline{\mathcal{M}}(T_i)_{\theta}^{\text{vir}}$$

By standard arguments, this can be used to define the various maps involved in the definition of contact homology (such as the differential d) and show that they satisfy the expected relations (such as $d^2 = 0$); see [61, Section 1.7].

5 Intersection theory for punctured holomorphic curves

5.1 Definition of the Siefring intersection number

We will make use in this paper of an intersection theory for asymptotically cylindrical maps and submanifolds. The four-dimensional theory was constructed by Siefring [64] and assigns an integer to a pair of asymptotically cylindrical maps in a 4–dimensional symplectic cobordism; see also the book by Wendl [69]. The higher-dimensional theory, also due to Siefring, assigns an integer to the pairing of a codimension 2 (asymptotically) cylindrical hypersurface with a (asymptotically) cylindrical map. A detailed overview can be found in [54].

Consider a strong exact symplectic cobordism $(\hat{X}, \hat{\lambda})$ from (Y^+, λ^+) to (Y^-, λ^-) . Let $(V^{\pm}, \lambda^{\pm}|_V) \subset (Y^{\pm}, \lambda^{\pm})$ be strong contact submanifolds and let $H \subset \hat{X}$ be a codimension 2 submanifold with cylindrical ends $V^+ \sqcup V^-$.

We let τ denote a choice of trivialization of $\xi^{\pm}|_{V^{\pm}}^{\perp}$ along every Reeb orbit in V^{\pm} . We require that the trivialization along a multiply covered orbit be pulled back from the chosen trivialization along the underlying simple orbit. Let $u: \Sigma - (p_u^+ \sqcup p_u^-) \to \hat{X}$ be a map which is positively/negatively asymptotic at $z \in p_u^{\pm}$ to the Reeb orbit γ_z . Now set

$$(5-1) u \bullet_{\tau} H := u^{\tau} \cdot H,$$

where u^{τ} is a perturbation of u which is transverse to H and constant with respect to τ at infinity, and $(-\cdot -)$ is the usual algebraic intersection number for transversely intersecting smooth maps. While (5-1) depends on the choice of trivialization τ , Siefring showed that this count can be corrected so as to become independent of τ . This leads to the following definition.

Definition 5.1 [54, Section 2] The generalized (or Siefring) intersection number $u * H \in \mathbb{Z}$ of u and H is defined by

(5-2)
$$u * H = u \bullet_{\tau} H + \sum_{z \in \boldsymbol{p}_u^+} \alpha_N^{\tau;-}(\gamma_z) - \sum_{z \in \boldsymbol{p}_u^-} \alpha_N^{\tau;+}(\gamma_z).$$

Proposition 5.2 The intersection number u * H only depends on the equivalence classes of u in $\pi_2(\hat{X}, \Gamma^+ \sqcup \Gamma^-)$, and H in $\Omega_{2n-2}(\hat{X}, V^+ \sqcup V^-)$; see Definition 2.21.

Proof The intersection number $u^{\tau} \cdot H$ is clearly invariant under compactly supported isotopies of H. Given a truncation $X_0 \subset \hat{X}$, we can proceed as in Section 2.2 to associate to u^{τ} a map $\overline{u}^{\tau} : \overline{\Sigma} \to X_0$. Let $H_0 = H \cap X_0$. If we choose X_0 sufficiently large (so that H is cylindrical in its complement), then H_0 will be a submanifold with boundary $\partial H_0 = H_0 \cap \partial X_0 = V^+ \sqcup V^-$. Note that $\overline{u}^{\tau} \cdot H_0$ only depends on $[u] \in \pi_2(\hat{X}, \Gamma^+ \sqcup \Gamma^-)$. Moreover, we have $\overline{u}^{\tau} \cdot H_0 = u^{\tau} \cdot H$; indeed, if X_0 is sufficiently large, then the intersections of \overline{u}^{τ} with H_0 are exactly the same as those of u^{τ} with H.

5.2 Positivity of intersection

We now discuss positivity of intersection for the Siefring intersection number. Given a contact manifold $(Y, \xi = \ker \lambda)$ and an almost complex structure J on ξ , we adopt the usual convention of letting \hat{J} denote the induced almost complex structure on the symplectization. An almost complex structure on a cobordism $(\hat{X}, \hat{\lambda})$ between two contact manifolds (Y^{\pm}, λ^{\pm}) is called *cylindrical* if it agrees at infinity with \hat{J}^{\pm} for some choice of $d\lambda^{\pm}$ -compatible almost complex structures J^{\pm} on ker (λ^{\pm}) .

Proposition 5.3 [54, Corollary 2.3 and Theorem 2.5] Let $(\hat{X}, \hat{\lambda})$ be an exact symplectic cobordism from (Y^+, λ^+) to (Y^-, λ^-) . Let *u* and *H* denote an asymptotically cylindrical map and a cylindrical submanifold of codimension 2 in \hat{X} , respectively.

Suppose that u and H are \hat{J} -holomorphic for some cylindrical almost complex structure \hat{J} on \hat{X} which is compatible with $d\hat{\lambda}$. If the image of u is not contained in H, then $\text{Im}(u) \cap H$ is a finite set and

$$(5-3) u * H \ge u \cdot H.$$

(Note that, by ordinary positivity of intersection for two pseudoholomorphic submanifolds, this implies that $u * H \ge 0$, and that Im(u) and H are disjoint if u * H = 0.)

When the image of u is contained in H, positivity of intersection does not hold. The following computation, which will be useful to us later, is one example of this. The notation $\hat{\gamma}$ refers to the trivial cylinder $\mathbb{R} \times S^1 \to \hat{Y}$ over the Reeb orbit γ ; similarly, $\hat{V} = \mathbb{R} \times V \subset \hat{Y}$ is the cylinder over the strong contact submanifold V.

Corollary 5.4 Let γ be a Reeb orbit in Y. If γ is contained in V, then

(5-4)
$$\widehat{\gamma} * \widehat{V} = -p_N(\gamma).$$

Proof By definition,

(5-5)
$$\hat{\gamma} * \hat{V} = \hat{\gamma}^{\tau} \cdot \hat{V} + \alpha_N^{\tau;-}(\gamma) - \alpha_N^{\tau;+}(\gamma).$$

We can choose the perturbation $\hat{\gamma}^{\tau}$ so that its image is disjoint from \hat{V} . The result follows since $\alpha_N^{\tau;+}(\gamma) - \alpha_N^{\tau;-}(\gamma) = p_N(\gamma)$ by definition.

Remark 5.5 If γ is disjoint from V, then $\hat{\gamma} * \hat{V} = 0$.

Corollary 5.4 shows that positivity of intersection fails for curves contained in \hat{V} . However, we still have a lower bound on the intersection number $u * \hat{V}$ when $u = \hat{\gamma}$ is a trivial cylinder, namely, $\hat{\gamma} * \hat{V} \ge -1$ since $p_N(\gamma) \in \{0, 1\}$. In the remainder of this section, we show that if (Y, ξ, V) is a TN contact pair and λ is positive elliptic near V, then the intersection number $u * \hat{V}$ is bounded below for *all* asymptotically cylindrical curves u contained in \hat{V} . We also give an analogous result for cylindrical submanifolds of symplectic cobordisms $H \subset (\hat{X}, \omega)$ with trivial normal bundle.

Proposition 5.6 Fix a TN contact pair (Y, ξ, V) and a datum $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V)$. Consider a contact form λ on (Y, ξ) which is adapted to \mathfrak{r} , and an almost complex structure J on ξ which is compatible with $d\lambda$ and which preserves ξ_V . Suppose that u is a \hat{J} -holomorphic curve whose image is entirely contained in \hat{V} . If λ is positive elliptic near V, then $u * \hat{V} \ge 1 - p_u$, where p_u denotes the number of punctures (positive and negative) of u.

Proof We have by definition that

(5-6)
$$\alpha_N^{\tau;-}(\gamma_z) = \lfloor CZ_N^{\tau}(\gamma_z)/2 \rfloor = \lfloor rP_z \rfloor \qquad \text{for } z \in p_u^+$$

(5-7)
$$\alpha_N^{\tau;+}(\gamma_z) = \lceil \operatorname{CZ}_N^{\tau}(\gamma_z)/2 \rceil = 1 + \lfloor rP_z \rfloor \quad \text{for } z \in p_u^-,$$

where P_z denotes the period of the Reeb orbit γ_z . Using the trivial bounds $x - 1 < \lfloor x \rfloor \le x$ and the fact that $u^{\tau} \cdot \hat{V} = 0$, we obtain

(5-8)
$$u * \hat{V} > \sum_{z \in p_u^+} (rP_z - 1) - \sum_{z \in p_u^-} (1 + rP_z) \ge -p_u + r \left(\sum_{z \in p_u^+} P_z - \sum_{z \in p_u^-} P_z \right).$$

The fact that u is \hat{J} -holomorphic implies that $\sum_{z \in p_u^+} P_z - \sum_{z \in p_u^-} P_z$ is nonnegative; see [68, page 60]. Thus $u * \hat{V} \ge 1 - p_u$, as desired.

We will need an analog of Proposition 5.6 for cobordisms. Note that if $V \subset Y$ is a codimension 2 contact submanifold, then the normal bundle of $\hat{V} = \mathbb{R} \times V \subset \mathbb{R} \times Y = \hat{Y}$ can be identified with the pullback of $\xi|_{V}^{\perp}$ under the projection $\hat{V} \to V$. Hence, any trivialization τ of $\xi|_{V}^{\perp}$ induces a trivialization of the normal bundle of \hat{V} , which we will denote by $\hat{\tau}$.

Proposition 5.7 Fix TN contact pairs $(Y^{\pm}, \xi^{\pm}, V^{\pm})$ and elements $\mathfrak{r}^{\pm} = (\alpha_V^{\pm}, \tau^{\pm}, r^{\pm}) \in \mathfrak{R}(Y^{\pm}, \xi^{\pm}, V^{\pm})$. Let λ^{\pm} be contact forms on (Y^{\pm}, ξ^{\pm}) which are adapted to \mathfrak{r}^{\pm} , and let $(\hat{X}, \hat{\lambda}, H)_{\lambda^{-}}^{\lambda^{+}}$ be a strong relative symplectic cobordism from (Y^{+}, ξ^{+}, V^{+}) to (Y^{-}, ξ^{-}, V^{-}) . We assume that there exists a global trivialization τ of the normal bundle of H which coincides with $\hat{\tau}^{\pm}$ near $\pm \infty$.

Let \hat{J} be an almost complex structure on \hat{X} which is cylindrical and compatible with $d\lambda^{\pm}$ outside a compact set, and such that H is \hat{J} -holomorphic. Let u be an asymptotically cylindrical map in \hat{X} which is \hat{J} -holomorphic and whose image is entirely contained in H. Following the notation of Proposition 5.6, if λ^+ is positive elliptic near V^+ , then $u * H > -p_u + r^+ \sum_{z \in p_u^+} P_z - r^- \sum_{z \in p_u^-} P_z$. In particular, $u * H \ge 0$ if u has no negative puncture.

Proof We have

(5-9)
$$u * H = u^{\tau} \cdot H + \sum_{z \in \boldsymbol{p}_u^+} \alpha_N^{\tau;-}(\gamma_z) - \sum_{z \in \boldsymbol{p}_u^-} \alpha_N^{\tau;+}(\gamma_z).$$

We can choose the perturbation u^{τ} so that it is disjoint from H, so $u^{\tau} \cdot H = 0$. We now argue as in Proposition 5.6 to find that

(5-10)
$$u * H > \sum_{z \in p_u^+} (r^+ P_z - 1) - \sum_{z \in p_u^-} (1 + r^- P_z) \ge -p_u + r^+ \sum_{z \in p_u^+} P_z - r^- \sum_{z \in p_u^-} P_z.$$

5.3 The intersection number for buildings

In this section, we use Siefring's intersection theory to define an intersection number for buildings of asymptotically cylindrical maps and buildings of asymptotically cylindrical codimension 2 submanifolds. Since the differences between S_{I} , S_{II} , S_{III} and S_{IV} don't matter for this purpose, we start by defining a category \hat{S} of labeled trees which only keeps track of the information needed for intersection theory; in particular, there are obvious "forgetful" functors $S_* \to \hat{S}$.

The category $\hat{S} = \hat{S}(\{\hat{X}^{ij}\}_{ij})$ depends on the following data:

- (i) An integer $m \ge 0$ and a collection of m + 1 co-oriented contact manifolds (Y^i, ξ^i) , each equipped with a choice of contact form λ^i for $0 \le i \le m$.
- (ii) For each pair of integers $0 \le i \le j \le m$, an exact symplectic cobordism $(\hat{X}^{ij}, \hat{\lambda}^{ij})$ with positive end (Y^i, ξ^i) and negative end (Y^j, ξ^j) . We require that $\hat{X}^{ii} = SY^i$ be the symplectization of Y^i and that $\hat{X}^{ik} = \hat{X}^{ij} # \hat{X}^{jk}$ for $i \le j \le k$; this makes sense in light of Remark 2.8.

An object $T \in \hat{S}$ is a finite directed forest, ie a finite collection of finite directed trees. We require that every vertex has a unique incoming edge. Edges which are adjacent to only one vertex are allowed; we will refer to them as input or output edges depending on whether they are missing a source or a sink. The other edges will be called interior edges. We also have the following decorations:

- For each edge e ∈ E(T), a symbol *(e) ∈ {0,...,m} such that *(e) = 0 for input edges and *(e) = m for output edges, together with a Reeb orbit γ_e in (Y^{*(e)}, λ^{*(e)}).
- For each vertex $v \in V(T)$, a pair $*(v) = (*^+(v), *^-(v)) \in \{0, \dots, m\}^2$ such that $*^+(v) \le *^-(v)$, and a homotopy class $\beta_v \in \pi_2(\hat{X}^{*(v)}, \gamma_{e^+(v)} \sqcup \{\gamma_{e^-}\}_{e^- \in E^-(v)})$, where $e^+(v)$ denotes the unique incoming edge of v and $E^-(v)$ denotes its set of outgoing edges. We require that $*(e^+(v)) = *^+(v)$ and $*(e^-) = *^-(v)$ for every $e^- \in E^-(v)$.

Remark 5.8 Geometrically, these decorations specify how different curves and orbits fit together to form a holomorphic building. For example, suppose m = 1 and T is a tree with one vertex v and one input edge e. If *(e) = 0 and *(v) = 00, then T describes a curve in SY^0 with one positive puncture and no negative punctures. If *(e) = 0 and *(v) = 01, then T describes a curve in \hat{X}^{01} with one positive puncture and no negative punctures. This labeling scheme of course follows [61, Section 2.1].
We will let Γ_T^+ and Γ_T^- denote the collections of Reeb orbits associated to the input and output edges of an object $T \in \hat{S}$. In the case where *T* is a tree, the unique element of Γ_T^+ will be denoted by γ_T^+ . A morphism $\pi: T \to T'$ consists of a contraction of the underlying forests (meaning that *T'* is identified with the forest obtained by contracting a certain subset of the interior edges of *T*) subject to the following conditions:

- For every noncontracted edge $e \in E(T)$, we require that $*(\pi(e)) = *(e)$ and $\gamma_{\pi(e)} = \gamma_e$.
- For every vertex $v \in V(T)$, we have $*^+(\pi(v)) \le *^+(v)$ and $*^-(\pi(v)) \ge *^-(v)$.
- For every vertex $v' \in V(T')$, we require that $\beta_{v'} = \#_{\pi(v)=v'}\beta_v$.

Note that for any morphism $T \to T'$, we have $\Gamma_T^+ = \Gamma_{T'}^+$ and $\Gamma_T^- = \Gamma_{T'}^-$.

Remark 5.9 For every $T \in \hat{S}$, we get a morphism $T \to T_{\max}$ by contacting all of the interior edges of T. Each component of T_{\max} is a tree with a unique vertex. In the case where T is connected, we will write $\beta_T := \#_v \beta_v \in \pi_2(\hat{X}^{0m}, \gamma_T^+ \sqcup \Gamma_T^-)$ for the homotopy class labeling the unique vertex of T_{\max} . Note that for every morphism $T \to T'$, we have $T_{\max} = T'_{\max}$. In particular, if T and T' are trees, then $\beta_T = \beta_{T'}$.

Definition 5.10 Let $T \in \hat{S}$ and let $\{T_i\}_i$ denote its connected components. The *intersection number* T * H of T with a codimension 2 cylindrical submanifold $H \subset \hat{X}^{0m}$ is defined to be

(5-11)
$$T * H = \sum_{i} \beta_{T_i} * H.$$

By Proposition 5.2, this intersection number only depends on the class of H in $\Omega_{2n-2}(\hat{X}^{0m}, V^0 \sqcup V^m)$. By Remark 5.9, it is "invariant under gluing":

Proposition 5.11 Let $T, T' \in \hat{S}$. If there exists a morphism $T \to T'$, then T * H = T' * H.

Suppose now that m = 0, so that objects $T \in \hat{S}$ represent buildings of curves in the symplectization \hat{Y} of a single contact manifold $(Y, \lambda) := (Y^0, \lambda^0)$, and that $H = \hat{V} := \mathbb{R} \times V$ is the trivial cylinder over some strong contact submanifold $V \subset Y$ of codimension 2. In that case, the intersection number T * H can be expressed more explicitly as follows.

Proposition 5.12 For any $T \in \hat{S}$, we have

(5-12)
$$T * \hat{V} = \sum_{v \in V(T)} \beta_v * \hat{V} - \sum_{e \in E^{\text{int}}(T)} \hat{\gamma}_e * \hat{V}.$$

Proof The proof will be by induction on the number of interior edges. If this number is zero, then (5-12) is true by definition. Otherwise, pick an edge $e \in E^{int}(T)$ and contract it to obtain a morphism $\pi: T \to T'$ where T' has one less interior edge than T. We can assume inductively that T' satisfies (5-12). Since $T * \hat{V} = T' * \hat{V}$, it suffices to show that

(5-13)
$$\beta_{v^+} * \hat{V} + \beta_{v^-} * \hat{V} - \hat{\gamma}_e * \hat{V} = \beta_{v'} * \hat{V},$$

where v^+ and v^- are the source and sink of *e*, respectively, and $v' = \pi(v^+) = \pi(v^-)$.

To do this, start by picking curves $u_{\pm}: \Sigma^{\pm} \to \hat{Y}$ representing the classes $\beta_{v^{\pm}}$. Fix a choice of cylindrical coordinates near the positive puncture of u^- and near the negative puncture of u^+ corresponding to e. We can assume that u_{\pm} is cylindrical at infinity, so that there exists a constant C > 0 such that

(5-14)
$$u_{\pm}(s,t) = (P_e \cdot s, \gamma_e(t)) \quad \text{for } \mp s \ge C,$$

where P_e is the period of the orbit corresponding to e. Now let

(5-15)
$$\overline{\Sigma}^+ = \Sigma^+ \setminus ((-\infty, -3C) \times S^1),$$

(5-16)
$$\overline{\Sigma}^{-} = \Sigma^{-} \setminus ((3C, \infty) \times S^{1}),$$

and let $\Sigma = \Sigma^+ \# \Sigma^-$ be obtained by identifying $[-3C, -C] \times S^1 \subset \overline{\Sigma}^+$ with $[C, 3C] \times S^1 \subset \Sigma^-$ via translation by 4*C*. The curve $u_+ \# u_-$: $\Sigma \to \hat{Y}$ which is given by $\tau_{2CP_e} \circ u_+$ on $\overline{\Sigma}^+$ and $\tau_{-2CP_e} \circ u_-$ on $\overline{\Sigma}^-$ (where $\tau_s : \hat{Y} \to \hat{Y}$ denotes translation by *s*) then represents the homotopy class $\beta_{v^+} \# \beta_{v^-} = \beta_{v'}$. Choose a trivialization τ of $\xi|_V^{\perp}$ along the relevant Reeb orbits and use it to produce perturbations u_{\pm}^{τ} and $(u_+ \# u_-)^{\tau}$ as in Section 5.1. We can do this in such a way that $(u_+ \# u_-)^{\tau}$ is obtained by gluing u_+^{τ} and u_-^{τ} . Then

(5-17)
$$(u_+ \# u_-)^{\tau} \cdot \hat{V} = u_+^{\tau} \cdot \hat{V} + u_-^{\tau} \cdot \hat{V},$$

so

$$\begin{aligned} (u_{+} \# u_{-}) * \hat{V} &= u_{+}^{\tau} \cdot \hat{V} + u_{-}^{\tau} \cdot \hat{V} + \alpha_{N}^{\tau;-}(\gamma^{+}) - \sum_{z \in \mathbf{p}_{u_{+}}^{-} \# u_{-}} \alpha_{N}^{\tau;+}(\gamma_{z}) \\ &= u_{+}^{\tau} \cdot \hat{V} + u_{-}^{\tau} \cdot \hat{V} + \alpha_{N}^{\tau;-}(\gamma^{+}) + \alpha_{N}^{\tau;+}(\gamma_{e}) - \sum_{z \in \mathbf{p}_{u_{+}}^{-}} \alpha_{N}^{\tau;+}(\gamma_{z}) - \sum_{z \in \mathbf{p}_{u_{-}}^{-}} \alpha_{N}^{\tau;+}(\gamma_{z}) \\ &= u_{+} * \hat{V} + u_{-} * \hat{V} + \alpha_{N}^{\tau;+}(\gamma_{e}) - \alpha_{N}^{\tau;-}(\gamma_{e}) \\ &= u_{+} * \hat{V} + u_{-} * \hat{V} + p_{N}(\gamma_{e}). \end{aligned}$$

We have $p(\gamma_e) = -\hat{\gamma}_e * \hat{V}$ by Corollary 5.4, so this implies that

(5-18)
$$\beta_{v'} * \hat{V} = (u_+ \# u_-) * \hat{V} = u_+ * \hat{V} + u_- * \hat{V} - \hat{\gamma}_e * \hat{V} = \beta_{v^+} * \hat{V} + \beta_{v^-} * \hat{V} - \hat{\gamma}_e * \hat{V},$$

as desired.

Definition 5.13 Given $T \in \hat{S}$, we say that *T* is *representable by a holomorphic building* if there exists a $d\lambda$ -compatible almost complex structure *J* on ξ such that, for every vertex $v \in V(T)$, the homotopy class $\beta_v \in \pi_2(\hat{Y}, \gamma_{e^+(v)} \sqcup \{\gamma_{e^-}\}_{e^- \in E^-(v)})$ admits a \hat{J} -holomorphic representative. We say that *T* is representable by a \hat{J} -holomorphic building if we wish to specify \hat{J} .

Corollary 5.14 Let $T \in \hat{S}$. Suppose that there exists a morphism $T' \to T$ and a $d\lambda$ -compatible almost complex structure J on ξ such that T' is representable by a \hat{J} -holomorphic building. Suppose also that \hat{V} is \hat{J} -holomorphic. If λ is positive elliptic near V, then $T * \hat{V} \ge -\Gamma^-(T, V)$, where $\Gamma^-(T, V)$ denotes the number of output edges e of T such that γ_e is contained in V.

Proof By Proposition 5.11, $T * \hat{V} = T' * \hat{V}$. By Proposition 5.12,

(5-19)
$$T' * \widehat{V} = \sum_{v \in V(T')} \beta_v * \widehat{V} - \sum_{e \in E^{\text{int}}(T')} \widehat{\gamma}_e * \widehat{V}.$$

According to Proposition 5.3, we have $\beta_v * \hat{V} \ge 0$, unless the holomorphic representative of β_v is entirely contained in \hat{V} , in which case Proposition 5.6 tells us that $\beta_v * \hat{V} \ge -\#E^-(v)$. By Corollary 5.4, we have

(5-20)
$$-\#E^{-}(v) = \sum_{e \in E^{-}(v)} \hat{\gamma}_{e} * \hat{V}$$

Given $v \in V(T')$, let us denote by $\Gamma^-(v, V)$ the number of output edges $e \in E^-(v)$ such that $\gamma_e \subset V$. Appealing again to Proposition 5.12, we have

(5-21)
$$T'^* \hat{V} = \sum_{v \in V(T')} \beta_v * \hat{V} - \sum_{e \in E^{\text{int}}(T')} \hat{\gamma}_e * \hat{V}$$
$$= \sum_{v \in V(T')} \left(\beta_v * \hat{V} - \sum_{e \in E^-(v)} \hat{\gamma}_e * \hat{V} \right) + \sum_{e \in E^-(T)} \hat{\gamma}_e * \hat{V}$$
$$= \sum_{v \in V(T')} (\beta_v * \hat{V} + \Gamma^-(v, V)) - \Gamma^-(T', V)$$
$$\geq -\Gamma^-(T, V),$$

where we have used the fact that T and T' have the same exterior edges in the last line. This completes the proof.

More generally, suppose we are given the following data, where m is now allowed to be any nonnegative integer:

- For each $0 \le i \le m$, a strong contact submanifold $V^i \subset Y^i$ of codimension 2.
- For each $0 \le i \le j \le m$, a homotopy class $\eta_{ij} \in \Omega_{2n-2}(\hat{X}^{ij}, V^i \sqcup V^j)$. We require that $\eta_{ii} := [\hat{V}_i]$ be the homotopy class of $\hat{V}^i = \mathbb{R} \times V^i$, and that $\eta_{ik} = \eta_{ij} \# \eta_{jk}$ for any $i \le j \le k$.

Let $\eta := \eta_{0m} \in \Omega_{2n-2}(\hat{X}^{0m}, V^0 \sqcup V^m).$

Proposition 5.15 Let $T \in \hat{S}$. Then

(5-22)
$$T * \eta = \sum_{v \in V(T)} \beta_v * \eta_{*(v)} - \sum_{e \in E^{\operatorname{int}}(T)} \widehat{\gamma}_e * \widehat{V}_{*(e)}$$

Proof We will say a vertex $v \in V(T)$ is a *symplectization vertex* if *(v) = ii for some *i*, and we will call it a *cobordism vertex* otherwise. This induces a partition $E^{int}(T) = E^{ss}(T) \sqcup E^{sc}(T) \sqcup E^{cc}(T)$ of the set of interior edges according to the types of the vertices they are adjacent to — here the superscripts s and c stand for "symplectization" and "cobordism", respectively. Similarly, the set of exterior edges admits a partition $E^{ext}(T) = E^{s}(T) \sqcup E^{c}(T)$.

We can (and will) assume without loss of generality that $E^{ss}(T)$ is empty. Indeed, let $T \to T'$ be the morphism obtained by contracting all the edges in $E^{ss}(T)$. Replacing T with T' doesn't change the left-hand side of (5-22) by Proposition 5.11 and doesn't change the right-hand side by Proposition 5.12.

Let $I = V(T) \sqcup E^{c}(T) \sqcup E^{cc}(T)$ and choose a family of curves $\{u_i\}_{i \in I}$ with the following properties:

- For each $v \in V(T)$, u_v is a curve in the homotopy class β_v which is cylindrical at infinity.
- For each $e \in E^{c}(T) \sqcup E^{cc}(T)$, $u_{e} = \hat{\gamma}_{e}$ is the trivial cylinder over the Reeb orbit γ_{e} .

For t > 0 sufficiently large, we can glue the u_i to obtain a curve u in $X^{00} \#_t X^{01} \#_t X^{11} \#_t \cdots \#_t X^{mm} \cong X^{0m}$ representing T. We can also choose representatives H_{ij} of η_{ij} so that $H_{ii} = \hat{V}_i$ and $H := H_{0m}$ coincides with $H_{00} \#_t H_{01} \#_t \cdots \#_t H_{mm}$.

As in the proof of Proposition 5.12, we can choose perturbations u^{τ} and $\{u_i^{\tau}\}$ so that

(5-23)
$$u^{\tau} \cdot H = \sum_{i \in I} u_i^{\tau} \cdot H_i$$

where $H_i := H_{*(v)}$ for $i = v \in V(T)$ and $H_i := \hat{V}_{*(e)}$ for $i = e \in E^c(T) \sqcup E^{cc}(T)$. The difference $\sum_i u_i * H_i - u * H$ is therefore equal to

(5-24)
$$\sum_{e \in E^{sc}(T) \sqcup E^{c}(T)} \alpha_{N}^{\tau;-}(\gamma_{e}) - \alpha_{N}^{\tau;+}(\gamma_{e}) + 2 \sum_{e \in E^{cc}(T)} \alpha_{N}^{\tau;-}(\gamma_{e}) - \alpha_{N}^{\tau;+}(\gamma_{e}) = \sum_{e \in E^{sc}(T) \sqcup E^{c}(T)} \hat{\gamma}_{e} * \hat{V}_{*(e)} + 2 \sum_{e \in E^{cc}(T)} \hat{\gamma}_{e} * \hat{V}_{*(e)} = \sum_{e \in E^{int}(T)} \hat{\gamma}_{e} * \hat{V}_{*(e)} + \sum_{e \in E^{c}(T) \sqcup E^{cc}(T)} \hat{\gamma}_{e} * \hat{V}_{*(e)}.$$

Since $u_i * H_i = \hat{\gamma}_e * \hat{V}_{*(e)}$ for $i = e \in E^c(T) \sqcup E^{cc}(T)$, we conclude that

(5-25)
$$u * H = \sum_{v \in V(T)} u_v * H_{*(v)} - \sum_{e \in E^{int}(T)} \widehat{\gamma}_e * \widehat{V}_{*(e)},$$

which implies (5-22).

Definition 5.16 Given $T \in \hat{S}$, we say that *T* is *representable by a holomorphic building* if for every vertex $v \in V(T)$, there exists an adapted almost complex structure \hat{J}^v on $\hat{X}^{*(v)}$ such that

 $\beta_v \in \pi_2(\hat{X}^{*(v)}, \gamma_{e^+(v)} \sqcup \{\gamma_{e^-}\}_{e^- \in E^-(v)})$ and $\eta_{*(v)} \in \Omega_{2n-2}(\hat{X}^{*(v)}, V^{*^+(v)} \sqcup V^{*^-(v)})$

admit \hat{J}^v -holomorphic representatives.

Proposition 5.17 Let $T \in \hat{S}$. Suppose that there exists a morphism $T' \to T$ where T' is representable by a holomorphic building. Suppose that λ^i is positive elliptic near V^i for all $0 \le i \le m$, and let $r_i > 0$ be the rotation parameter; see Definition 3.4. If $\beta_v * \eta_{*(v)} \ge -\#\{E^-(v)\}$, then $T * \eta \ge -\Gamma^-(T, V^m)$. (Recall that $\eta := \eta_{0m}$.)

Proof By Proposition 5.11, we have $T * \eta = T' * \eta$. Given $v \in V(T')$, let us denote by $\Gamma^{-}(v, V^{*-(v)})$ the number of output edges $e \in E^{-}(v)$ such that $\gamma_{e} \subset V^{*-(v)}$.

Arguing as in the proof of Corollary 5.14, we obtain from Proposition 5.15 that

(5-26)
$$T' * \hat{V} = \sum_{v \in V(T')} \beta_v * \eta_{*(v)} - \sum_{e \in E^{int}(T')} \hat{\gamma}_e * \hat{V}_{*(e)}$$
$$= \sum_{v \in V(T')} \left(\beta_v * \eta_{*(v)} - \sum_{e \in E^{-}(v)} \hat{\gamma}_e * \hat{V}_{*(e)} \right) + \sum_{e \in E^{-}(T')} \hat{\gamma}_e * \hat{V}_{*(e)}$$
$$= \sum_{v \in V(T')} (\beta_v * \hat{V} + \#\Gamma^{-}(v, V^{*^{-}(v)})) - \Gamma^{-}(T', V)$$
$$\geq -\Gamma^{-}(T, V),$$

where we have used the fact that T and T' have the same exterior edges in the last line. This completes the proof.

5.4 The intersection number for cycles

For future reference, we collect some basic facts about intersection numbers for cycles in oriented manifolds. This subsection takes places entirely in the smooth category and does not involve any contact topology.

Definition 5.18 Let M be an oriented, compact manifold of dimension n, possibly with boundary. Let $S_1, S_2 \subset M$ be disjoint closed embedded submanifolds of M. (We allow the S_i to intersect ∂M , in which case the S_i are required to be embedded submanifolds after enlarging M by a collar). Then we can define a pairing

$$(5-27) \qquad -\cdots : H_k(M, S_1; \mathbb{Z}) \times H_{n-k}(M, S_2; \mathbb{Z}) \to \mathbb{Z}, \quad (A, B) \mapsto A \cdot B,$$

where $A \cdot B$ is a signed count of intersections between cycles representing A and B. More precisely, we represent cycles by C^{∞} chains; by general position, these chains may be assumed to intersect transversally after an arbitrarily small perturbation which does not affect their homology class. It is a folklore result, which is beyond the scope of this paper, that the resulting count is well-defined and graded-symmetric; see eg [32, Section 2.3].

We note that the intersection number in Definition 5.18 could be defined under much milder hypotheses, but this is not necessary for our purposes. If A and B are (the pushforward of the fundamental class of) oriented manifolds, then $A \cdot B$ coincides with the usual intersection number for submanifolds. By abuse of notation, we will view the intersection pairing as being defined on both homology classes and oriented submanifolds.

Definition 5.19 Fix a closed manifold *Y* of dimension $m \ge 3$ and a closed codimension 2 submanifold $V \subset Y$. Suppose that $H_1(Y; \mathbb{Z}) = H_2(Y; \mathbb{Z}) = 0$.

Let $\gamma: S^1 \to Y - V$ be a loop with embedded image. The *linking number* of γ with respect to V is denoted by $link_V(\gamma)$ and defined by

(5-28)
$$\operatorname{link}_{V}(\gamma) := V \cdot C_{\gamma},$$

where C_{γ} is a cycle bounding γ (which exists since $H_1(Y; \mathbb{Z}) = 0$). This is well-defined due to our assumption that $H_2(Y; \mathbb{Z}) = 0$.

Suppose now that $\Lambda \subset Y - V$ is a submanifold with $\pi_0(\Lambda) = \pi_1(\Lambda) = 0$. Let $c : [0, 1] \to Y - V$ be a path with embedded image having the property that $c(0), c(1) \in \Lambda$. Let $\overline{c} : S^1 \to Y - V$ be a loop with embedded image obtained by connecting c(1) to c(0) by a path in Λ . The (*path*) linking number of c with respect to V is denoted by link_V($c; \Lambda$) and is defined by setting

(5-29)
$$\operatorname{link}_{V}(c;\Lambda) := V \cdot C_{\overline{c}},$$

where $C_{\overline{c}}$ is a cycle bounding \overline{c} . This is independent of \overline{c} since $\pi_1(\Lambda) = 0$, and independent of $C_{\overline{c}}$ since $H_2(Y;\mathbb{Z}) = 0$.

Remark 5.20 Fix an open book decomposition (Y, B, π) and let $\gamma: S^1 \to Y - B$ be a loop. Then it is not hard to show that we have $link_B(\gamma) = deg(\pi \circ \gamma)$.

Similarly, suppose $\Lambda \subset Y$ is a submanifold which is contained in a page of (Y, B, π) . Let $c: [0, 1] \to Y - B$ be a path with the property that $c(0), c(1) \in \Lambda$. Then the composition $\pi \circ c: [0, 1] \to S^1$ induces a map $\overline{c}: [0, 1]/\{0, 1\} \to S^1$. We then have $\operatorname{link}_B(c; \Lambda) = \deg \overline{c}$.

Lemma 5.21 Let Y^{\pm} be oriented manifolds with $Y^{+} \neq \emptyset$ and let $B^{\pm} \subset Y^{\pm}$ be oriented submanifolds. Let *W* be an oriented, smooth cobordism from Y^{+} to Y^{-} and let $H \subset W$ be an oriented subcobordism from B^{+} to B^{-} , ie *H* is an embedded submanifold which admits a collar neighborhood near the boundary of *W*. Suppose that $H_{1}(Y^{\pm}; \mathbb{Z}) = H_{2}(Y^{\pm}; \mathbb{Z}) = H_{2}(W, Y^{+}; \mathbb{Z}) = 0$.

Let Σ be a Riemann surface with k + 1 boundary components labeled $\gamma^+, \gamma_1^-, \ldots, \gamma_k^-$. Suppose that $u: (\Sigma, \partial \Sigma) \to (W, \partial W)$ is a smooth map sending γ^+ into $Y^+ - B^+$ and γ_i^- into $Y^- - B^-$.

Then

(5-30)
$$\operatorname{link}_{B^+}(\gamma^+) - \sum_{i=1}^{k} \operatorname{link}_{B^-}(\gamma_i^-) = H \cdot u(\Sigma),$$

where we have identified the boundary components of Σ with the restriction of u to these components.

,

Proof Choose a 2-chain $B^- \in C_2(Y^-; \mathbb{Z})$ with $\partial B^- = \gamma_1^- \cup \cdots \cup \gamma_k^-$. Glue B^- to $u(\Sigma)$ along the γ_i^- and call the resulting chain $C \in C_2(W; \mathbb{Z})$. We now have $H \cdot C = H \cdot u(\Sigma) + \sum_{i=1}^k \operatorname{link}_{B^-}(\gamma_i^-)$.

By the long exact sequence of the triple (W, Y^+, γ^+) and our assumption that $H_2(W, Y^+; \mathbb{Z}) = 0$, the natural map $H_2(Y, \gamma^+; \mathbb{Z}) \to H_2(W, \gamma^+; \mathbb{Z})$ is surjective. Let $\tilde{C} \in H_2(Y^+, \gamma^+; \mathbb{Z})$ be a lift of $C \in H_2(W, \gamma^+; \mathbb{Z})$. Then $H \cdot C = H \cdot \tilde{C} = B^+ \cdot \tilde{C} = \text{link}_{B^+}(\gamma^+)$.

Lemma 5.22 We carry over the setup and notation from Lemma 5.21. In addition to the data considered there, let $\Lambda^{\pm} \subset Y^{\pm} - B^{\pm}$ be an oriented, smooth submanifold and let $\Lambda \subset W$ be an oriented subcobordism from Λ^{+} to Λ^{-} which is disjoint from *H*. We suppose in addition that $\pi_{0}(\Lambda^{\pm}) = \pi_{1}(\Lambda^{\pm}) = 0$.

Let Σ be a closed, oriented surface of genus zero with s + 1 boundary components labeled $\gamma^*, \gamma_1, \ldots, \gamma_n$. For $\sigma \in \mathbb{N}_+$, we place 2σ disjoint marked points on γ^* , thus partitioning γ^* into 2σ subintervals. Let us label these subintervals by the symbols $c^+, b_{01}, c_1^-, b_{12}, c_2^-, \ldots, b_{(\sigma-1)\sigma}, c_{\sigma}^-, b_{\sigma 0}$, in the order induced by the orientation.

Suppose now that $u: (\Sigma, \partial \Sigma) \to (W, \partial W \cup \Lambda)$ is a smooth map sending $(c^+, \partial c^+)$ into $(Y^+ - B^+, \Lambda^+)$, sending $(c_i^-, \partial c_i^-)$ into $(Y^- - B^-, \Lambda^-)$, sending $b_{i(i+1)}$ into Λ , and sending the γ_i^- into $Y^- - B^-$.

Then

(5-31)
$$\operatorname{link}_{B^+}(c^+;\Lambda^+) - \sum_{i=1}^{\sigma} \operatorname{link}_{B^-}(c_i^-;\Lambda^-) - \sum_{i=1}^{s} \operatorname{link}_{B^-}(\gamma_i^-) = H \cdot u_*[\Sigma]$$

where we have again identified the boundary components of Σ with the restriction of u to these components.

6 Energy and twisting maps

6.1 Twisting maps

In order to define invariants of codimension 2 contact submanifolds, we will proceed as follows. First, we will use Siefring's intersection theory to define maps $\psi : S \to \mathcal{R}$. Here \mathcal{R} could be any \mathbb{Q} -algebra, though we will only use $\mathcal{R} = \mathbb{Q}[U]$ and $\mathcal{R} = \mathbb{Q}$. We will then use these maps to define "twisted" moduli counts

(6-1)
$$\#_{\psi}\overline{\mathcal{M}}(T)^{\operatorname{vir}} := \#\overline{\mathcal{M}}(T)^{\operatorname{vir}} \cdot \psi(T) \in \mathcal{R}.$$

The maps ψ will have the property that

(6-2)
$$\psi(T') = \psi(T)$$
 for every morphism $T' \to T$, and

(6-3)
$$\psi(\#_i T_i) = \prod_i \psi(T_i),$$

which implies that the *master equations* (4-2) and (4-3) still hold if $\#\overline{\mathcal{M}}^{\text{vir}}$ is replaced by $\#_{\psi}\overline{\mathcal{M}}^{\text{vir}}$.

The properties which must be satisfied by the maps ψ in order to obtain twisted counts which are suitable for defining our invariants can be conveniently axiomatized in the notion of a *twisting map*. We now define precisely this notion in each of the four setups.

Setup I Fix a datum \mathcal{D} for Setup I. Let $\mathcal{S}_{I}^{\neq \emptyset}$ denote the full subcategory of \mathcal{S}_{I} spanned by objects T for which the moduli space $\overline{\mathcal{M}}(T)$ is nonempty.

Definition 6.1 Let \mathcal{R} be a \mathbb{Q} -algebra. The set $\Psi_{I}(\mathcal{D}; \mathcal{R})$ of \mathcal{R} -valued *twisting maps* consists of all maps $\psi: S_{I}^{\neq \varnothing}(\mathcal{D}) \rightarrow \mathcal{R}$ satisfying the following two properties:

- For any morphism $T' \to T$, $\psi(T') = \psi(T)$.
- For any concatenation $\{T_i\}_i$, $\psi(\#_i T_i) = \prod_i \psi(T_i)$.

Fix a twisting map $\psi \in \Psi_{I}(\mathcal{D}; \mathcal{R})$. Let

(6-4)
$$CC_{\bullet}(Y,\xi,\psi)_{\lambda} := \bigoplus_{n\geq 0} \operatorname{Sym}_{\mathcal{R}}^{n} \left(\bigoplus_{\gamma\in\mathcal{P}_{good}} \mathfrak{o}_{\gamma} \right)$$

be the free supercommutative $\mathbb{Z}/2$ -graded unital \mathcal{R} -algebra generated by the good Reeb orbits. The grading of a Reeb orbit is given by its *parity*, which is defined as

(6-5)
$$|\gamma| = \operatorname{sign} \det(I - A_{\gamma}) \in \{\pm 1\} = \mathbb{Z}/2,$$

where A_{γ} is the linearized Poincaré return map of ξ along γ ; see [61, Section 2.13]. Recall that a Reeb orbit is good if and only if it is not bad; a Reeb orbit γ is bad if it is an even multiple of some simple Reeb orbit γ_s such that γ and γ_s have different parity [61, Definition 2.49].

Theorem 1.1 of [61] provides a set of perturbation data $\Theta_{I}(\mathcal{D})$ and associated virtual moduli counts $\#\overline{\mathcal{M}}_{I}(T)_{\theta}^{\text{vir}} \in \mathbb{Q}$ satisfying (4-2) and (4-3). We define the twisted moduli counts

(6-6)
$$\#_{\psi}\overline{\mathcal{M}}_{\mathrm{I}}(T)_{\theta}^{\mathrm{vir}} := \#\overline{\mathcal{M}}_{\mathrm{I}}(T)_{\theta}^{\mathrm{vir}} \cdot \psi(T) \in \mathcal{R}.$$

It follows easily from Definition 6.1 that the twisted moduli counts also satisfy (4-2) and (4-3). We may therefore endow $CC_{\bullet}(Y, \xi, \psi)_{\lambda}$ with a differential $d_{\psi,J,\theta}$ which is given by

(6-7)
$$d_{\psi,J,\theta}(\mathfrak{o}_{\gamma^+}) = \sum_{\mu(T)=1} \frac{1}{|\operatorname{Aut}(T)|} \cdot \#_{\psi} \overline{\mathcal{M}}_{\mathrm{I}}(T)_{\theta}^{\operatorname{vir}} \mathfrak{o}_{\Gamma^-},$$

where the sum is over all trees $T \in S_I(\mathcal{D})$ representing curves with positive orbit γ^+ and negative orbits $\Gamma^- \to \mathcal{P}_{good}$.

The homology of $(CC_{\bullet}(Y, \xi, \psi)_{\lambda}, d_{\psi,J,\theta})$ is a supercommutative $\mathbb{Z}/2$ -graded unital \mathcal{R} -algebra, which is denoted by

(6-8)
$$CH_{\bullet}(Y,\xi,\psi)_{\lambda,J,\theta}$$

43

Setup II Fix a datum \mathcal{D} for Setup II. Suppose now we are given a map of \mathbb{Q} -algebras $m: \mathcal{R}^+ \to \mathcal{R}^$ and twisting maps $\psi^{\pm} \in \Psi_{\mathrm{I}}(\mathcal{D}^{\pm}; \mathcal{R}^{\pm})$.

Definition 6.2 The set $\Psi_{II}(\mathcal{D}; \psi^+, \psi^-)$ consists of all maps $\psi : S_{II}^{\neq \varnothing}(\mathcal{D}) \to \mathcal{R}^-$ satisfying the following two properties:

- For any morphism $T' \to T$, $\psi(T') = \psi(T)$.
- For any concatenation $\{T_i\}_i$,

(6-9)
$$\psi(\#_i T_i) = \left(\prod_{T_i \in \mathcal{S}_{\mathrm{I}}^+} m(\psi^+(T_i))\right) \left(\prod_{T_i \in \mathcal{S}_{\mathrm{II}}} \psi(T_i)\right) \left(\prod_{T_i \in \mathcal{S}_{\mathrm{I}}^-} \psi^-(T_i)\right).$$

Fix a twisting map $\psi \in \Psi_{II}(\mathcal{D}; \psi^+, \psi^-)$. Theorem 1.1 of [61] provides a set of perturbation data $\Theta_{II}(\mathcal{D})$ together with a forgetful map

(6-10)
$$\Theta_{\mathrm{II}}(\mathcal{D}) \to \Theta_{\mathrm{I}}(\mathcal{D}^+) \times \Theta_{\mathrm{I}}(\mathcal{D}^-)$$

and associated virtual moduli counts $\#\overline{\mathcal{M}}_{\mathrm{II}}(T)_{\theta}^{\mathrm{vir}} \in \mathbb{Q}$. We define the twisted moduli counts

(6-11)
$$\#_{\psi}\overline{\mathcal{M}}_{\mathrm{II}}(T)_{\theta}^{\mathrm{vir}} := \#\overline{\mathcal{M}}_{\mathrm{II}}(T)_{\theta}^{\mathrm{vir}} \cdot \psi(T) \in \mathcal{R}^{-}.$$

For any $\theta \in \Theta_{II}(\mathcal{D})$ mapping to $(\theta^+, \theta^-) \in \Theta_I(\mathcal{D}^+) \times \Theta_I(\mathcal{D}^-)$, we obtain a unital \mathcal{R}^+ -algebra map

(6-12)
$$\Phi(\hat{X},\hat{\lambda},\psi)_{\hat{J},\theta}: CC_{\bullet}(Y^+,\xi^+,\psi^+)_{\lambda^+,J^+,\theta^+} \to CC_{\bullet}(Y^-,\xi^-,\psi^-)_{\lambda^-,J^-,\theta^-},$$

which maps \mathfrak{o}_{γ^+} to

(6-13)
$$\sum_{\mu(T)=0} \frac{1}{|\operatorname{Aut}(T)|} \cdot \#_{\psi} \overline{\mathcal{M}}_{\mathrm{II}}(T)_{\theta}^{\mathrm{vir}} \mathfrak{o}_{\Gamma^{-}},$$

with the sum over all trees $T \in S_{II}(\mathcal{D})$ representing curves with positive orbit γ^+ and negative orbits $\Gamma^- \to \mathcal{P}_{good}(Y^-)$. This is a chain map, since it follows from Definition 6.2 that the twisted moduli counts satisfy (4-2) and (4-3).

Setup III Fix a datum \mathcal{D} for Setup III. There are three types of concatenations $\{T_i\}_i$ in $\mathcal{S}_{III} = \mathcal{S}_{III}(\mathcal{D})$:

- (1) $\{T_i\} \subset S_{\mathrm{I}}^+ \sqcup S_{\mathrm{II}}^{t=0} \sqcup S_{\mathrm{I}}^-$, in which case $\mathfrak{s}(\#_i T_i) = \{0\}$.
- (2) $\{T_i\} \subset S_{\mathrm{I}}^+ \sqcup S_{\mathrm{II}}^{t=1} \sqcup S_{\mathrm{II}}^-$, in which case $\mathfrak{s}(\#_i T_i) = \{1\}$.
- (3) $\{T_i\} \subset S_{\mathrm{I}}^+ \sqcup S_{\mathrm{III}} \sqcup S_{\mathrm{I}}^-$ and $T_i \in S_{\mathrm{III}}$ for a unique $i = i_0$, in which case $\mathfrak{s}(\#_i T_i) = \mathfrak{s}(T_{i_0})$.

(Here $S_{I}^{\pm}, S_{II}^{t \in \{0,1\}}, S_{III}$ are tree categories determined by \mathcal{D} , following the notation of [61, Section 2.1].) Suppose now we are given a map of \mathbb{Q} -algebras $m: \mathcal{R}^{+} \to \mathcal{R}^{-}$ and twisting maps $\psi^{\pm} \in \Psi_{I}(\mathcal{D}^{\pm}; \mathcal{R}^{\pm}), \psi^{0} \in \Psi_{II}(\mathcal{D}^{t=0}; \psi^{+}, \psi^{-})$ and $\psi^{1} \in \Psi_{II}(\mathcal{D}^{t=1}; \psi^{+}, \psi^{-})$.

Definition 6.3 The set $\Psi_{\text{III}}(\mathcal{D}; \psi^0, \psi^1)$ consists of all maps $\psi : S_{\text{III}}^{\neq \varnothing}(\mathcal{D}) \to \mathcal{R}^-$ satisfying the following properties:

- For any morphism $T' \to T$, $\psi(T') = \psi(T)$.
- For any concatenation $\{T_i\}_i$ of the first type,

(6-14)
$$\psi(\#_i T_i) = \left(\prod_{T_i \in \mathcal{S}_{\mathrm{I}}^+} m(\psi^+(T_i))\right) \left(\prod_{T_i \in \mathcal{S}_{\mathrm{II}}^{t=0}} \psi^0(T_i)\right) \left(\prod_{T_i \in \mathcal{S}_{\mathrm{I}}^-} \psi^-(T_i)\right).$$

• For any concatenation $\{T_i\}_i$ of the second type,

(6-15)
$$\psi(\#_i T_i) = \left(\prod_{T_i \in \mathcal{S}_{\mathrm{I}}^+} m(\psi^+(T_i))\right) \left(\prod_{T_i \in \mathcal{S}_{\mathrm{II}}^{t=1}} \psi^1(T_i)\right) \left(\prod_{T_i \in \mathcal{S}_{\mathrm{I}}^-} \psi^-(T_i)\right).$$

• For any concatenation $\{T_i\}_i$ of the third type,

(6-16)
$$\psi(\#_i T_i) = \left(\prod_{T_i \in \mathcal{S}_{\mathbf{I}}^+} m(\psi^+(T_i))\right) \psi(T_{i_0}) \left(\prod_{T_i \in \mathcal{S}_{\mathbf{I}}^-} \psi^-(T_i)\right).$$

Fix a twisting map $\psi \in \Psi_{III}(\mathcal{D}; \psi^0, \psi^1)$. Theorem 1.1 of [61] provides a set of perturbation data $\Theta_{III}(\mathcal{D})$ together with a forgetful map $\Theta_{III}(\mathcal{D}) \to \Theta_{II}(\mathcal{D}^0) \times_{\Theta_I(\mathcal{D}^+) \times \Theta_I(\mathcal{D}^-)} \Theta_{II}(\mathcal{D}^1)$ and associated virtual moduli counts $\#\overline{\mathcal{M}}_{III}(T)^{\text{vir}}_{\theta} \in \mathbb{Q}$; note that the fiber product is defined with respect to (6-10). We define the twisted moduli counts

(6-17)
$$\#_{\psi}\overline{\mathcal{M}}_{\mathrm{III}}(T)_{\theta}^{\mathrm{vir}} := \#\overline{\mathcal{M}}_{\mathrm{III}}(T)_{\theta}^{\mathrm{vir}} \cdot \psi(T) \in \mathcal{R}^{-}.$$

If $(\hat{X}, \hat{\lambda}^t)$ is a family of exact cobordisms, then for any $\theta \in \Theta_{\text{III}}(\mathcal{D})$, we obtain an \mathcal{R}^+ -linear map

(6-18)
$$K(\hat{X}, \{\lambda^t\}_t, \psi)_{\hat{J}^t, \theta} \colon CC_{\bullet}(Y^+, \xi^+, \psi^+)_{\lambda^+, J^+, \theta^+} \to CC_{\bullet+1}(Y^-, \xi^-, \psi^-)_{\lambda^-, J^-, \theta^-}$$

which sends the monomial $\prod_{i \in I} \mathfrak{o}_{\gamma_i^+}$ to

(6-19)
$$\sum_{\operatorname{vdim}(\{T_i\}_{i\in I})=0} \frac{1}{|\operatorname{Aut}(\{T_i\}_{i\in I})|} \cdot \#_{\psi} \overline{\mathcal{M}}_{\operatorname{III}}(\{T_i\}_{i\in I})_{\theta}^{\operatorname{vir}} \prod_{i\in I} \mathfrak{o}_{\Gamma_i^-}$$

with the sum over trees $T_i \in S_{\text{III}}(\mathcal{D})$ with positive orbit γ_i^+ and negative orbits $\Gamma_i^- \to \mathcal{P}_{\text{good}}(Y^-)$.

Equations (4-2) and (4-3) applied to the twisted moduli counts imply that this is a chain homotopy between $\Phi(\hat{X}, \hat{\lambda}^0, \psi^0)_{\hat{J}^0, \theta^0}$ and $\Phi(\hat{X}, \hat{\lambda}^1, \psi^1)_{\hat{J}^1, \theta^1}$ and hence that the induced maps on homology

(6-20)
$$CH_{\bullet}(Y^{+},\xi^{+},\psi^{+})_{\lambda^{+},J^{+},\theta^{+}} \xrightarrow{\Phi(\hat{X},\hat{\lambda}^{0},\psi^{0})_{\hat{J}^{0},\theta^{0}}} \Phi(Y^{-},\xi^{-},\psi^{-})_{\lambda^{-},J^{-},\theta^{-}} \xrightarrow{\Phi(\hat{X},\hat{\lambda}^{1},\psi^{1})_{\hat{J}^{1},\theta^{1}}} CH_{\bullet}(Y^{-},\xi^{-},\psi^{-})_{\lambda^{-},J^{-},\theta^{-}}$$

are equal.

Setup IV Fix a datum \mathcal{D} for Setup IV. There are three types of concatenations $\{T_i\}_i$ in $\mathcal{S}_{IV} = \mathcal{S}_{IV}(\mathcal{D})$:

- (1) $\{T_i\} \subset S^0_{\mathrm{I}} \sqcup S^{02}_{\mathrm{II}} \sqcup S^2_{\mathrm{I}}$, in which case $\mathfrak{s}(\#_i T_i) = \{0\}$.
- (2) $\{T_i\} \subset S^0_{\mathrm{I}} \sqcup S^{01}_{\mathrm{II}} \sqcup S^{12}_{\mathrm{II}} \sqcup S^2_{\mathrm{II}}$. in which case $\mathfrak{s}(\#_i T_i) = \{\infty\}$;
- (3) $\{T_i\} \subset S_{\mathrm{I}}^0 \sqcup S_{\mathrm{IV}} \sqcup S_{\mathrm{I}}^2$ and $T_i \in S_{\mathrm{IV}}$ for a unique $i = i_0$, in which case $\mathfrak{s}(\#_i T_i) = \mathfrak{s}(T_{i_0})$.

(We again follow the notation of [61, Section 2.1] for tree categories determined by \mathcal{D} .) Suppose now we are given maps of \mathbb{Q} -algebras $m^{01}: \mathcal{R}^0 \to \mathcal{R}^1$ and $m^{12}: \mathcal{R}^1 \to \mathcal{R}^2$, and twisting maps

(6-21)
$$\begin{aligned} \psi^i \in \Psi_{\mathrm{I}}(\mathcal{D}^i; \mathcal{R}^i) & \text{for } i = 0, 1, 2, \\ \psi^{ij} \in \Psi_{\mathrm{II}}(\mathcal{D}^{ij}; \psi^i, \psi^j) & \text{for } ij = 01, 12, 02. \end{aligned}$$

Set $m^{02} = m^{12} \circ m^{01} \colon \mathcal{R}^0 \to \mathcal{R}^2$.

Definition 6.4 The set $\Psi_{\text{IV}}(\mathcal{D}; \{\psi^{ij}\})$ consists of all maps $\psi: \mathcal{S}_{\text{IV}}^{\neq \varnothing}(\mathcal{D}) \to \mathcal{R}^2$ satisfying the following properties:

- For any morphism $T' \to T$, $\psi(T') = \psi(T)$.
- For any concatenation $\{T_i\}_i$ of the first type,

(6-22)
$$\psi(\#_{i}T_{i}) = \left(\prod_{T_{i} \in \mathcal{S}_{I}^{0}} m^{02}(\psi^{0}(T_{i}))\right) \left(\prod_{T_{i} \in \mathcal{S}_{II}^{02}} \psi^{02}(T_{i})\right) \left(\prod_{T_{i} \in \mathcal{S}_{I}^{2}} \psi^{2}(T_{i})\right).$$

• For any concatenation $\{T_i\}_i$ of the second type,

$$(6-23) \quad \psi(\#_{i}T_{i}) = \left(\prod_{T_{i} \in \mathcal{S}_{I}^{0}} m^{02}(\psi^{0}(T_{i}))\right) \left(\prod_{T_{i} \in \mathcal{S}_{I}^{01}} m^{12}(\psi^{01}(T_{i}))\right) \left(\prod_{T_{i} \in \mathcal{S}_{I}^{1}} m^{12}(\psi^{1}(T_{i}))\right) \left(\prod_{T_{i} \in \mathcal{S}_{I}^{2}} \psi^{12}(T_{i})\right) \left(\prod_{T_{i} \in \mathcal{S}_{I}^{2}} \psi^{2}(T_{i})\right)$$

• For any concatenation $\{T_i\}_i$ of the third type,

(6-24)
$$\psi(\#_i T_i) = \left(\prod_{T_i \in \mathcal{S}_{\Gamma}^0} m^{02}(\psi^0(T_i))\right) \psi(T_{i_0}) \left(\prod_{T_i \in \mathcal{S}_{\Gamma}^2} \psi^2(T_i)\right).$$

Fix a twisting map $\psi \in \Psi_{\text{IV}}(\mathcal{D}; \{\psi^{ij}\})$. Theorem 1.1 of [61] provides a set of perturbation data $\Theta_{\text{IV}}(\mathcal{D})$ together with a forgetful map $\Theta_{\text{IV}}(\mathcal{D}) \to \Theta_{\text{II}}(\mathcal{D}^{02}) \times_{\Theta_{\text{I}}(\mathcal{D}^0) \times \Theta_{\text{I}}(\mathcal{D}^2)} (\Theta_{\text{II}}(\mathcal{D}^{01}) \times_{\Theta_{\text{I}}(\mathcal{D}^1)} \Theta_{\text{II}}(\mathcal{D}^{12}))$ and associated virtual moduli counts $\#\overline{\mathcal{M}}_{\text{IV}}(T)^{\text{vir}}_{\theta} \in \mathbb{Q}$ —here again, the fiber product is defined using (6-10). We define the twisted moduli counts

(6-25)
$$\#_{\psi}\overline{\mathcal{M}}_{\mathrm{IV}}(T)_{\theta}^{\mathrm{vir}} := \#\overline{\mathcal{M}}_{\mathrm{IV}}(T)_{\theta}^{\mathrm{vir}} \cdot \psi(T) \in \mathcal{R}^{2}.$$

As in the previous section, we obtain an \mathcal{R}^0 -linear map

(6-26)
$$CC_{\bullet}(Y^0,\xi^0,\psi^0)_{\lambda^0,J^0,\theta^0} \to CC_{\bullet+1}(Y^2,\xi^2,\psi^2)_{\lambda^2,J^2,\theta^2},$$

which is a chain homotopy between the maps

$$\Phi(\hat{X}^{02}, \hat{\lambda}^{02}, \psi^{02})_{\hat{J}^{02}, \theta^{02}} \quad \text{and} \quad \Phi(\hat{X}^{12}, \hat{\lambda}^{12}, \psi^{12})_{\hat{J}^{12}, \theta^{12}} \circ \Phi(\hat{X}^{01}, \hat{\lambda}^{01}, \psi^{01})_{\hat{J}^{01}, \theta^{01}},$$

so that we get the commuting diagram

(6-27)
$$\begin{array}{c} \Phi(\hat{X}^{01}, \hat{\lambda}^{01}, \psi^{01})_{\hat{j}^{01}, \theta^{01}} & CH_{\bullet}(Y^{1}, \xi^{1}, \psi^{1})_{\lambda^{1}, J^{1}, \theta^{1}} & \Phi(\hat{X}^{12}, \hat{\lambda}^{12}, \psi^{12})_{\hat{j}^{12}, \theta^{12}} \\ CH_{\bullet}(Y^{0}, \xi^{0}, \psi^{0})_{\lambda^{0}, J^{0}, \theta^{0}} & \xrightarrow{\Phi(\hat{X}^{02}, \hat{\lambda}^{02}, \psi^{02})_{\hat{j}^{02}, \theta^{02}}} & CH_{\bullet}(Y^{2}, \xi^{2}, \psi^{2})_{\lambda^{2}, J^{2}, \theta^{2}} \end{array}$$

6.2 The energy of a (strict) symplectic cobordism

In this section, we introduce a notion of energy for (families of strict) exact symplectic cobordisms, and for certain classes of almost complex structures.

Notation 6.5 Recall that a strict exact symplectic cobordism from (Y^+, λ^+) to (Y^-, λ^-) is the data of an exact symplectic cobordism $(\hat{X}, \hat{\lambda})$ and embeddings

(6-28)
$$e^{\pm}: (\mathbb{R} \times Y^{\pm}, \hat{\lambda}^{\pm}) \to (\hat{X}, \hat{\lambda}),$$

which preserve the Liouville forms and satisfy certain additional properties stated in Definition 2.3. When we consider strict exact symplectic cobordisms in this section, we will routinely abuse notation by identifying subsets of $\mathbb{R} \times Y^{\pm}$ with their image under e^{\pm} . We hope that this abuse will make this section easier to read without introducing any substantial ambiguities.

We begin with the following definition.

Definition 6.6 Let $(\hat{X}, \hat{\lambda})$ be a strict exact symplectic cobordism from (Y^+, λ^+) to (Y^-, λ^-) . A *Type A cobordism decomposition* is the data of a pair of hypersurfaces

(6-29)
$$\mathcal{H}_{-} = \{-C_{-}\} \times Y^{-} \text{ and } \mathcal{H}_{+} = \{C_{+}\} \times Y^{+}$$

for $C_{\pm} \in \mathbb{R}$, such that

(6-30)
$$((-\infty, -C_{-}) \times Y^{-}) \cap ((C_{+}, \infty) \times Y^{+}) = \emptyset.$$

The intersection in (6-30) takes place inside \hat{X} ; if $Y^- = \emptyset$, we set $\mathcal{H}_- = \emptyset$, $C_- = 0$ and we consider that (6-30) is tautologically satisfied. We let $\Sigma(\hat{X}, \hat{\lambda}) = \Sigma(\hat{X}, \hat{\lambda}; \lambda^+, \lambda^-)$ be the set of all such Type A cobordism decompositions.

Remark 6.7 Since we are working with *strict* cobordisms, the real numbers $C_{\pm} \in \mathbb{R}$ are uniquely determined by the hypersurfaces \mathcal{H}_{\pm} . The data of the pair $(\mathcal{H}_{-}, \mathcal{H}_{+})$ is therefore equivalent to the data of the pair (C_{-}, C_{+}) .

Definition 6.8 Let $(\hat{X}, \hat{\lambda})$ be as in Definition 6.6 and let $\sigma \in \Sigma(\hat{X}, \hat{\lambda})$ be a Type A cobordism decomposition. We let $\mathcal{E}(\sigma) := C_{-} + C_{+}$ be the *energy* of the decomposition σ . We define

(6-31)
$$\mathcal{E}(\hat{X},\hat{\lambda}) = \mathcal{E}(\hat{X},\hat{\lambda};\lambda^+,\lambda^-) := \inf_{\sigma \in \Sigma(\hat{X},\hat{\lambda})} \mathcal{E}(\sigma) \in \mathbb{R} \cup \{-\infty\}$$

to be the energy of $(\hat{X}, \hat{\lambda})$; this is well-defined since a cobordism decomposition clearly always exists. We note that the energy may in general be negative.

Given $C \in \mathbb{R}$, let $\Sigma(\hat{X}, \hat{\lambda})_{\leq C} \subset \Sigma(\hat{X}, \hat{\lambda})$ (resp. $\leq C$) denote the subset of cobordism decompositions of energy strictly less than *C* (resp. at most *C*).

Lemma 6.9 (energy of a symplectization) Suppose that $(\hat{X}, \hat{\lambda}) = (SY, \lambda_Y)$ is a symplectization which is endowed with the canonical structure of a (strict) exact symplectic cobordism from (Y, λ^+) to (Y, λ^-) ; see Example 2.6. Fix $f: Y \to \mathbb{R}$ such that $\lambda^+ = e^f \lambda^-$. Then $\mathcal{E}(SY, \lambda_Y) \leq -\min f$.

Proof Let $e^{\pm}: (\mathbb{R} \times Y, \hat{\lambda}^{\pm}) \to (SY, \lambda_Y)$ be the canonical identifications induced by λ^{\pm} . Let $\mathcal{H}_+ = \{0\} \times Y$ in the coordinates induced by e^+ . This means that $\mathcal{H}_+ = \{(f(y), y) \mid y \in Y\} \subset \mathbb{R} \times Y$ in the coordinates induced by e^- . Now given any $C_- > -\min f$, we can let $\mathcal{H}_- = \{-C_-\} \times Y$ in the coordinates induced by e^- . It follows that $\mathcal{E}(SY, \hat{\lambda}_Y) \leq C_-$. Since $C_- > -\min f$ was arbitrary, the claim follows.

Remark 6.10 If we assume in addition that $\lambda^+ = \lambda^-$, then it is easy to verify that in fact $\mathcal{E}(SY, \lambda_Y) = 0$.

Lemma 6.11 We have $\mathcal{E}(\hat{X}, \hat{\lambda}) = -\infty$ if and only if $Y^- = \emptyset$.

Proof Suppose that Y^- is nonempty and choose a cobordism decomposition σ for $(\hat{X}, \hat{\lambda})$ given by a pair of hypersurfaces $\mathcal{H}_-, \mathcal{H}_+ \subset \hat{X}$. Let (X, λ) be the truncated Liouville cobordism with negative boundary \mathcal{H}_- and positive boundary \mathcal{H}_+ . Observe that the image of the negative boundary under the Liouville flow must touch the positive boundary in some finite time $T < \infty$ —indeed, this follows from the fact that (X, λ) has finite volume. Given any other cobordism decomposition σ' , we now have $\mathcal{E}(\sigma') \geq \mathcal{E}(\sigma) - T$.

Suppose now that Y^- is empty. Then (6-30) is a vacuous condition. Since the backwards Liouville flow of any slice $\{C_+\} \times Y^+$ is defined for all time, it follows that we can find a cobordism decomposition of arbitrarily negative energy.

Lemma 6.12 Fix a strict exact symplectic cobordism $(\hat{X}, \hat{\lambda})$ from (Y^+, λ^+) to (Y^-, λ^-) . Then $\Sigma(\hat{X}, \hat{\lambda})_{\leq C} \subset \Sigma(\hat{X}, \hat{\lambda})$ is

- (a) nonempty for $C > \mathcal{E}(\hat{X}, \hat{\lambda})$,
- (b) *path-connected for all* $C \in \mathbb{R}$. (Note that the empty set is path-connected.)

If moreover $(\hat{X}, \hat{\lambda}) = (SY, \lambda_Y)$ is a symplectization and $\lambda^+ = \lambda^-$, then $\Sigma(\hat{X}, \hat{\lambda})_{\leq 0}$ is nonempty and path-connected.

Proof Note first that (a) is tautologically true. Next, note that (b) is obvious when $Y^- = \emptyset$. It therefore remains to prove (b) under the assumption that $Y^- \neq \emptyset$.

Let us consider a pair of cobordism decompositions $\sigma, \sigma' \in \Sigma(\hat{X}, \hat{\lambda})_{\leq C}$. By definition, σ and σ' are entirely determined by the constants $-C_{-}, C_{+} \in \mathbb{R}$ (resp. $-C'_{-}, C'_{+}$), where we are following the notation of Definition 6.6.

Suppose first that $-C_{-} = -C'_{-}$. Up to relabeling σ and σ' , we can assume that $C_{+} \leq C'_{+}$. Now just translate C'_{+} in the negative direction until $C'_{+} = C_{+}$. This translation defines a one-parameter family of cobordism decompositions taking σ' to σ , whose energy is clearly bounded by $\max(\mathcal{E}(\sigma), \mathcal{E}(\sigma')) = \mathcal{E}(\sigma') < C$. An analogous argument works if we now suppose $C_{+} = C'_{+}$ and $-C_{-} \neq -C'_{-}$.

Suppose finally that $-C_{-} \neq -C'_{-}$ and $C_{+} \neq C'_{+}$. Up to relabeling σ and σ' , we can assume that $C_{+} < C'_{+}$. If $-C'_{-} < -C_{-}$, then we translate $-C'_{-}$ in the positive direction until $-C'_{-} = -C_{-}$. If instead $-C_{-} < -C'_{-}$, then we simultaneously translate $-C_{-}$ and C_{+} in the positive direction until either $-C_{-} = -C'_{-}$ or $C_{+} = C'_{+}$. This takes us back to the case treated in the previous paragraph.

Finally, if $(\hat{X}, \hat{\lambda}) = (SY, \lambda_Y)$ is a symplectization with $\lambda^+ = \lambda^-$, then any Type A cobordism decomposition $\sigma \in \Sigma(\hat{X}, \hat{\lambda})_{\leq 0}$ has vanishing energy (Remark 6.10) and is equivalent to a choice of hypersurface $\mathcal{H} = \mathcal{H}_- = \mathcal{H}_+ = \{\tilde{C} \times Y\}$. The space of such choices is in natural bijection with \mathbb{R} , so it is in particular nonempty and connected.

Definition 6.13 Let $(\hat{X}, \hat{\lambda}_t)_{t \in [0,1]}$ be a one-parameter family of (strict) exact symplectic cobordisms; cf Definition 2.12. A one-parameter family of Type A cobordism decompositions is just the data of a family of hypersurfaces

(6-32)
$$\mathcal{H}_{-}(t) = \{-C_{-}(t)\} \times Y^{-} \text{ and } \mathcal{H}_{+}(t) = \{C_{+}(t)\} \times Y^{+}$$

such that

(6-33)
$$((-\infty, -C_{-}(t)) \times Y^{-}) \cap ((C_{+}(t), \infty) \times Y^{+}) = \emptyset.$$

(If $Y^- = \emptyset$, we again set $\mathcal{H}_-(t) = \emptyset$, $C_-(t) = 0$ and we consider that (6-33) is tautologically satisfied.) We let $\Sigma(\hat{X}, \hat{\lambda}_t)_{t \in [0,1]}$ be the set of all such families of cobordism decompositions. (Note that $\Sigma(\hat{X}, \hat{\lambda}_{t_0})$ is a Type A cobordism decomposition for each fixed choice of t_0 .)

Definition 6.14 With the notation as above, with define the energy of a family of Type A cobordism decompositions $\sigma \in \Sigma(\hat{X}, \hat{\lambda}_t)_{t \in [0,1]}$ to be $\mathcal{E}(\sigma) := \sup_t (C_-(t) + C_+(t))$.

Let $(\hat{X}^{01}, \hat{\lambda}^{01})$ (resp. $(\hat{X}^{12}, \hat{\lambda}^{12})$) be a strict exact symplectic cobordism from (Y^0, λ^0) to (Y^1, λ^1) (resp. from (Y^1, λ^1) to (Y^2, λ^2)). Let $(\hat{X}, \hat{\lambda}_t)_{t \in [0,\infty)}$ be a one-parameter family of strict exact symplectic

cobordisms which agrees for $t \ge a$ large enough with the *t*-gluing $(\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{02})_{t \in [a,\infty)}$; see Definition 2.7. For $t \ge a$, note that there are canonical Liouville embeddings

$$\begin{split} \iota_{0,t} &: (\hat{X}^{01}, \hat{\lambda}^{01}) \xrightarrow{\mu_{t/2}} (\hat{X}^{01}, e^{t/2} \hat{\lambda}^{01}) &\to (\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{02}), \\ \iota_{2,t} &: (\hat{X}^{12}, \hat{\lambda}^{12}) \xrightarrow{\mu_{-t/2}} (\hat{X}^{12}, e^{-t/2} \hat{\lambda}^{12}) \to (\hat{X}^{01} \#_t \hat{X}^{12}, \hat{\lambda}^{01} \#_t \hat{\lambda}^{02}). \end{split}$$

Definition 6.15 A *Type B cobordism decomposition* of $(\hat{X}, \hat{\lambda}_t)_{t \in [0,\infty)}$ is the data of a family of hypersurfaces

(6-34)
$$\mathcal{H}_2(t) = \{-C_2(t)\} \times Y^2 \text{ and } \mathcal{H}_0(t) = \{C_0(t)\} \times Y^0,$$

and a Liouville embedding $([-C_1(t), \tilde{C}_1(t)] \times Y^1, e^s \lambda_1) \hookrightarrow (\hat{X}, \hat{\lambda}_t)$ such that

(6-35)
$$(-\infty, -C_2(t)) \times Y^2, \quad (-C_1(t), \tilde{C}_1(t)) \times Y^1 \text{ and } (C_0(t), \infty) \times Y^0$$

are pairwise disjoint. (In the case that $Y^- = \emptyset$, we set $\mathcal{H}_2(t) = \emptyset$ and $C_2(t) = 0$, and replace (6-35) by the condition that $(-C_1(t), \tilde{C}_1(t)) \times Y^1$ and $(C_0(t), \infty) \times Y^0$ are pairwise disjoint.)

We let

(6-36)
$$\mathcal{H}_1(t) = \{-C_1(t)\} \times Y^1 \quad \text{and} \quad \widetilde{\mathcal{H}}_1(t) = \{\widetilde{C}_1(t)\} \times Y^1.$$

This data is required to satisfy the following hypotheses:

- (1) $\tilde{C}_1(0) = -C_1(0).$
- (2) For t large enough, \mathcal{H}_0 and $\tilde{\mathcal{H}}_1(t)$ (resp. $\mathcal{H}_1(t)$ and $\mathcal{H}_2(t)$) are in the image of the canonical embedding $\iota_{0,t}$ (resp. $\iota_{2,t}$). Moreover, their preimages define a Type A decomposition on $(\hat{X}^{01}, \hat{\lambda}^{01})$ (resp. on $(\hat{X}^{12}, \hat{\lambda}^{12})$) which is independent of t.

We let $\Sigma_B((\hat{X}, \hat{\lambda}_t)_{t \in [0,\infty)})$ denote the set of all such cobordism decompositions. We will write $\Sigma(-)$ instead of $\Sigma_B(-)$ when the subscript is understood from the context. As in Remark 6.7, note that the data of the hypersurfaces $\mathcal{H}_2(t)$, $\mathcal{H}_1(t)$, $\tilde{\mathcal{H}}'_1(t)$ and $\mathcal{H}_0(t)$ is equivalent to the data of the constants $C_2(t)$, $C_1(t)$, $\tilde{C}_1(t)$ and $C_0(t)$.

Definition 6.16 It follows from property (1) of Definition 6.15 that a Type B cobordism decomposition $\sigma^{02} \in \Sigma_B((\hat{X}, \hat{\lambda}_t)_{t \in [0,\infty)})$ induces a Type A cobordism decomposition $\sigma \in \Sigma_A(\hat{X}, \hat{\lambda}_0)$ by taking $\mathcal{H}_- = \mathcal{H}_2(0)$ and $\mathcal{H}_+ = \mathcal{H}_0(0)$. We say that σ is *induced at zero* by σ^{02} .

Similarly, it follows from property (2) of Definition 6.15 that a Type B cobordism decomposition $\sigma^{02} \in \Sigma_B((\hat{X}, \hat{\lambda}_t)_{t \in [0,\infty)})$ induces a pair of Type A decompositions $\sigma^{01} \in \Sigma_A(\hat{X}^{01}, \hat{\lambda}^{01})$ and $\sigma^{12} \in \Sigma_A(\hat{X}^{02}, \hat{\lambda}^{02})$. We say that the pair $(\sigma^{01}, \sigma^{12})$ is *induced at infinity* by σ^{02} .

Definition 6.17 With the notation as above, we define the energy of a Type B cobordism decomposition $\sigma \in \Sigma_B(\hat{X}, \hat{\lambda}_t)$ to be $\mathcal{E}(\sigma) := \sup_t (C_2(t) + C_0(t) - C_1(t) - \tilde{C}_1(t))$. We let

(6-37)
$$\mathcal{E}(\hat{X}, \hat{\lambda}_t) := \inf_{\sigma \in \Sigma_B(\hat{X}, \hat{\lambda}_t)} \mathcal{E}(\sigma) \in \mathbb{R} \cup \{-\infty\}.$$

Given $C \in \mathbb{R}$, let $\Sigma(\hat{X}, \hat{\lambda}_t)_{\leq C} \subset \Sigma(\hat{X}, \hat{\lambda}_t)$ (resp. $\leq C$) denote the subset of Type B cobordism decompositions of energy strictly less than C (resp. at most C).

The following lemma asserts that our notions of energy for Type A and Type B decompositions are compatible with the map which associates to a Type B decomposition the Type A decomposition induced at zero or infinity. It will be used implicitly in the sequel.

Lemma 6.18 Let σ^{02} be a Type *B* cobordism decomposition. Suppose that σ is induced at zero by σ^{02} and that $(\sigma^{01}, \sigma^{12})$ is induced at infinity. Then $\mathcal{E}(\sigma) \leq \mathcal{E}(\sigma^{02})$ and $\mathcal{E}(\sigma^{01}) + \mathcal{E}(\sigma^{12}) \leq \mathcal{E}(\sigma^{02})$.

Proof The first claim follows from (1) in Definition 6.15 and the definition of energy for Type A and Type B cobordism decomposition. The second claim follows similarly from (2) in Definition 6.15. \Box

Corollary 6.19 We have $\mathcal{E}(\hat{X}, \hat{\lambda}_t) = -\infty$ if and only if $Y^2 = \emptyset$.

Proof One direction follows from Lemmas 6.11 and 6.18. The other one can be checked by inspection, using the backwards Liouville flow as in the proof of the corresponding statement in Lemma 6.11. \Box

Definition 6.20 Suppose that $(\hat{X}, \hat{\lambda})$ and $(\hat{X}', \hat{\lambda}')$ are exact symplectic cobordisms. For $C \in \mathbb{R}$, let $\Sigma_A((\hat{X}, \hat{\lambda}), (\hat{X}', \hat{\lambda}'))_{\leq C} \subset \Sigma_A(\hat{X}, \hat{\lambda}) \times \Sigma_A(\hat{X}, \hat{\lambda})$ (resp. $(-)_{\leq C}$) be the subspace of pairs (σ, σ') of Type A cobordism decompositions such that $\mathcal{E}(\sigma) + \mathcal{E}(\sigma') < C$ (resp. $\leq C$).

Lemma 6.21 Given $C \in \mathbb{R}$ such that $\Sigma(X^{02,t}, \hat{\lambda}^{02,t})_{\leq C}$ is nonempty, the map which associates to a decomposition $\sigma^{02} \in \Sigma(X^{02,t}, \hat{\lambda}^{02,t})_{\leq C}$ the pair $(\sigma^{01}, \sigma^{12}) \in \Sigma_A((\hat{X}^{01}, \hat{\lambda}^{01}), (\hat{X}^{12}, \hat{\lambda}^{12}))_{\leq C}$ induced by σ^{02} at infinity is surjective. If $\lambda^0 = \lambda^1 = \lambda^2$ and $(\hat{X}^{01}, \hat{\lambda}^{01}), (\hat{X}^{12}, \hat{\lambda}^{12})$ are symplectizations, the same statement holds with \leq in place of <.

Proof Choose a Type B decomposition $\tilde{\sigma}^{02}$. Let $(\tilde{\sigma}^{01}, \tilde{\sigma}^{12})$ be the Type A decompositions induced by $\tilde{\sigma}^{02}$ at infinity. According to Definition 6.15, this means that there exists a T > 0 so that for $t \ge T$, we have that $\mathcal{H}_0(t)$ and $\tilde{\mathcal{H}}_1(t)$ are independent of t after pulling back via the canonical embedding ι_{01} (and similarly $\mathcal{H}_1(t)$ and $\mathcal{H}_2(t)$ are independent of t after pulling back by ι_{12}). By a routine modification of the arguments of Lemma 6.12(b), one can now construct a Type B decomposition σ^{02} so that $\sigma_t^{02} = \tilde{\sigma}_t^{02}$ for $t \in [0, T]$, $\mathcal{E}(\sigma^{02}) \le \mathcal{E}(\tilde{\sigma}^{02})$ and σ^{02} induces the pair $(\sigma^{01}, \sigma^{12})$.

Lemma 6.22 Suppose that $(\hat{X}^{02,t}, \hat{\lambda}^{02,t})$ is the (t + T)-gluing of two exact symplectic cobordisms $(\hat{X}^{01}, \hat{\lambda}^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12})$, for $T \ge 0$ an arbitrary fixed constant and $t \in [0, \infty)$ a parameter; see Definition 2.7 for the definition of this gluing and Lemma 2.13 for the parametric version. Suppose that either $(\hat{X}^{01}, \hat{\lambda}^{01})$ or $(\hat{X}^{12}, \hat{\lambda}^{12})$ is a symplectization; see Example 2.6. Then $\mathcal{E}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}) = \mathcal{E}(\hat{X}^{01}, \hat{\lambda}^{01}) + \mathcal{E}(\hat{X}^{12}, \hat{\lambda}^{12})$.

Proof By Lemma 6.11 and Corollary 6.19, we may assume that \hat{X}^{12} has a nonempty negative end.

We only treat the case where $(\hat{X}^{01}, \hat{\lambda}^{01})$ is a symplectization and T = 0, since the other cases are analogous.

Choose σ^{01} so that $\mathcal{E}(\sigma^{01}) \leq \mathcal{E}(\hat{X}^{01}, \hat{\lambda}^{01}) + \epsilon$, and choose σ^{12} so that $\mathcal{E}(\sigma^{12}) \leq \mathcal{E}(\hat{X}^{12}, \hat{\lambda}^{12}) + \epsilon$. Let $\tilde{X}^{01} \subset \hat{X}^{01}$ and $\tilde{X}^{12} \subset \hat{X}^{12}$ be the Liouville subdomains which determine the Type A decompositions σ^{01} and σ^{12} , respectively.

Note that $\hat{X}^{02,t}$ comes equipped with tautological embeddings $\iota_{0,t}: \hat{X}^{01} \to \hat{X}^{02,t}$ and $\iota_{2,t}: \hat{X}^{12} \to \hat{X}^{02,t}$; see Definition 2.7. For T' large enough and $t \ge T'$, note that $\iota_{0,t}(\mathcal{H}^0_-)$ is in the image of $\iota_{2,t}(\mathcal{H}^2_+)$ under the Liouville flow. These hypersurfaces therefore bound Liouville domains $([-C_1(t), \tilde{C}_1(t)] \times Y^1, e^s \lambda_1)$.

Let $f: [0, \infty) \to \mathbb{R}$ be a function which equals $-(C_1(T') + \tilde{C}_1(T'))$ on [0, T'], is nondecreasing on [T', T' + 1] and is zero on $[T' + 1, \infty)$. Let $\tau_f: \hat{X}^{01} \times [0, \infty) \to \hat{X}^{01}$ be defined by $\tau_f(x, t) = \phi_{f(t)+T'+1-t}^{01}(x)$, where ϕ_t^{01} is the time-t Liouville flow on \hat{X}^{01} . Now define a map $\overline{\iota}_{0,t}: \hat{X}^{01} \to \hat{X}^{02,t}$ by letting $\overline{\iota}_{0,t}(x) = \iota_{0,t} \circ \tau_f(x, t)$.

We now define the data of a Type B cobordism decomposition by letting

(6-38)
$$\mathcal{H}_2(t) = \iota_{2,t}(\mathcal{H}_-^2), \quad \mathcal{H}_1(t) = \iota_{2,t}(\mathcal{H}_+^2), \quad \widetilde{\mathcal{H}}_1 = \overline{\iota}_{0,t}(\mathcal{H}_-^0(t)), \quad \mathcal{H}_0 = \overline{\iota}_{0,t}(\mathcal{H}_+^0(t)).$$

One can check that this data indeed defines a Type B cobordism decomposition, which has energy precisely equal to $\mathcal{E}(\sigma^{01}) + \mathcal{E}(\sigma^{12}) \leq \mathcal{E}(\hat{X}^{01}, \hat{\lambda}^{01}) + \mathcal{E}(\hat{X}^{12}, \hat{\lambda}^{12}) + 2\epsilon$. Since ϵ was arbitrary, we conclude that $\mathcal{E}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}) \leq \mathcal{E}(\hat{X}^{01}, \hat{\lambda}^{01}) + \mathcal{E}(\hat{X}^{12}, \hat{\lambda}^{12})$.

We now discuss almost complex structures for Setups II-IV.

Setup II Fix a datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, \hat{\lambda}, \hat{J})$ for Setup II, with $\mathcal{D}^{\pm} = (Y^{\pm}, \lambda^{\pm}, J^{\pm})$. Let $(V^{\pm}, \tau^{\pm}) \subset (Y^{\pm}, \xi^{\pm})$ be framed codimension 2 contact submanifolds; let $\alpha^{\pm} := \lambda^{\pm}|_{V^{\pm}}$ and assume that V^{\pm} is a strong contact submanifold with respect to λ^{\pm} . Let J^{\pm} be $d\lambda^{\pm}$ -compatible almost complex structures on $\xi^{\pm} \subset TY^{\pm}$ which preserve $\xi^{\pm} \cap TV^{\pm}$.

Let $H \subset \hat{X}$ be a codimension 2 symplectic submanifold such that $(\hat{X}, \hat{\lambda}, H)$ is an exact relative symplectic cobordism from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) . We will also consider (see Notation 2.4) the strict symplectic cobordisms $(\hat{X}, \hat{\lambda})_{\lambda^-}^{\lambda^+}$ and $(H, \hat{\lambda}|_H)_{\alpha^-}^{\alpha^+}$.

Definition 6.23 Fix a Type A cobordism decomposition $\sigma \in \Sigma(H, \hat{\lambda}|_H)$, which is specified by a pair of hypersurfaces $\mathcal{H}_- = \{-C_-\} \times V^-$ and $\mathcal{H}_+ = \{C_+\} \times V^+$. We say that an almost complex structure \hat{J} on \hat{X} is *adapted* to σ if the following properties hold:

- \hat{J} is compatible with $d\hat{\lambda}$,
- *Ĵ* coincides with *Ĵ*[±] near the ends (where *Ĵ*[±] is the canonical cylindrical almost complex structure induced on (*Ŷ*, *λ*[±]) by *J*[±]),
- $H \subset \hat{X}$ is a \hat{J} -complex hypersurface,
- \hat{J} preserves ker $\alpha^+ \subset TV^+$ on $[C_2, \infty) \times V^+$ (resp. preserves ker $\alpha^- \subset TV^-$ on $(-\infty, -C_1] \times V^-$) and the induced almost complex structure is $d\alpha^+$ -compatible (resp. $d\alpha^-$ -compatible).

In the case that $V^- = \emptyset$, the conditions involving V^- are considered to be vacuously satisfied.

Definition 6.24 Given an almost complex structure \hat{J} on $(\hat{X}, \hat{\lambda})$ we define its energy

(6-39)
$$\mathcal{E}(\widehat{J}) := \inf\{\mathcal{E}(\sigma) \mid \sigma \in \Sigma(H, \widehat{\lambda}|_H), \ \widehat{J} \text{ is adapted to } \sigma\} \in \mathbb{R} \cup \{\pm \infty\}.$$

We define $\mathcal{E}(\hat{J}) = \infty$ if \hat{J} is not adapted to any cobordism decomposition.

Let $\mathcal{J}(\hat{X}, \hat{\lambda}, H)_{\leq C}$ (resp. $\leq C$) be the set of almost complex structures of energy less than *C* (resp. at most *C*). Let $\mathcal{J}(\hat{X}, \hat{\lambda}, H) := \mathcal{J}(\hat{X}, \hat{\lambda}, H)_{\leq \infty}$ be the set of almost complex structures adapted to some decomposition $\sigma \in \Sigma(\hat{H}, \hat{\lambda}|_H)$.

Lemma 6.25 The set $\mathcal{J}(\hat{X}, \hat{\lambda}, H)_{\leq C}$ is

- (a) nonempty for $C > \mathcal{E}(H, \hat{\lambda}|_H)$, and
- (b) path-connected for all $C \in \mathbb{R}$. (Note that the empty set is path-connected.)

If moreover $(H, \hat{\lambda}|_H) = (SV, \lambda_V)$ is a symplectization and $\alpha^+ = \alpha^-$, then $\mathcal{J}(\hat{X}, \hat{\lambda}, H)_{\leq 0}$ is nonempty and path-connected.

Proof To prove (a), it is enough to show that given any cobordism decomposition σ , there exists an almost complex structure adapted to it, ie meeting the conditions of Definition 6.23. To prove (b), it follows from Lemma 6.12 that it is enough to prove a similar statement in families: namely, if $\{\sigma_t\}_{t\in[a,b]}$ is a family of cobordism decompositions and J_a and J_b are almost complex structures adapted to σ_a and σ_b , respectively, then there is a family $\{J_t\}_{t\in[a,b]}$ adapted to σ_t . All of these statements can be proved by standard arguments, using the fact that the space of almost complex structures compatible with a given symplectic structure can be viewed as the space of sections of a bundle with contractible fibers; see eg [52, Proposition 2.6.4].

If $\alpha^+ = \alpha^-$ and $(H, \hat{\lambda}|_H)$ is a symplectization, then Lemma 6.12 implies that $\Sigma(H, \hat{\lambda}|_H)_{\leq 0}$ is nonempty and path-connected. Hence the same arguments involving extensions of almost complex structures imply that $\mathcal{J}(\hat{X}, \hat{\lambda}, H)_{\leq 0}$ is nonempty and path-connected. **Setup IV** Fix a datum \mathcal{D} for Setup IV. We write $\mathcal{D} = (\mathcal{D}^{01}, \mathcal{D}^{12}, (\hat{X}^{02,t}, \hat{\lambda}^{02,t}, \hat{J}^{02,t})_{t \in [0,\infty)})$, where

$$\mathcal{D}^{01} = (\mathcal{D}^0, \mathcal{D}^1, \hat{X}^{01}, \hat{\lambda}^{01}, \hat{J}^{01}),$$

$$\mathcal{D}^{12} = (\mathcal{D}^1, \mathcal{D}^2, \hat{X}^{12}, \hat{\lambda}^{12}, \hat{J}^{12}),$$

$$\mathcal{D}^i = (Y^i, \lambda^i, J^i) \text{ for } i = 0, 1, 2$$

Let $(V^i, \tau^i) \subset (Y^i, \xi)$ be framed codimension 2 contact submanifolds; set $\alpha_{V^i} = \lambda^i|_{V^i}$ and assume that V^i are strong contact submanifolds with respect to λ^i . Let

$$H^{01} \subset \hat{X}^{01}, \quad H^{12} \subset \hat{X}^{12} \text{ and } (H^{02,t} \subset \hat{X}^{02,t})_{t \in [0,\infty)}$$

be cylindrical symplectic submanifolds such that $(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})_{t \in [0,\infty)}$ is a family of relative symplectic cobordisms that agrees for *t* large with the *t*-gluing of the relative symplectic cobordisms $(\hat{X}^{01}, \hat{\lambda}^{01}, H^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12}, H^{12})$. Note that $\{H^{02,t}\}$ forms a family of Liouville manifolds with respect to (the restriction of) $\hat{\lambda}^{02,t}$.

Definition 6.26 Fix a Type B cobordism decomposition $\sigma^{02} \in \Sigma_B(\hat{H}^{02,t}, \hat{\lambda}_{H^{02,t}}^{02,t})$. Recall that σ^{02} consists of the data of hypersurfaces

$$\mathcal{H}_2(t) = \{-C_2(t)\} \times V^2, \quad \mathcal{H}_1(t) = \{-C_1(t)\} \times V^1, \quad \tilde{\mathcal{H}}_1(t) = \{\tilde{C}_1(t)\} \times V^1, \quad \mathcal{H}_0(t) = \{C_0(t)\} \times V^0.$$

We say that an almost complex structure $\hat{J}^{02,t}$ is *adapted* to σ^{02} if the following properties hold:

- $\hat{J}^{02,t}$ is compatible with $d\hat{\lambda}^{02,t}$.
- $\hat{J}^{02,t}$ coincides with \hat{J}^0 (resp. \hat{J}^2) near the positive (resp. negative) end.
- $H^{02,t}$ is a $\hat{J}^{02,t}$ -complex hypersurface. $\hat{J}^{02,t}$ is compatible with the restriction of $d\hat{\lambda}^{02,t}$ to $H^{02,t}$, and
- $\hat{J}^{02,t}$ preserves ker α_0 on $[C_0(t), \infty) \times V^0$, and preserves ker α_1 on $[-C_1(t), \tilde{C}_1(t)] \times V_1$, and preserves ker α_2 on $(-\infty, -C_2(t)] \times V_2$). Moreover, the induced almost complex structure is $d\alpha_0$ -compatible, $d\alpha_1$ -compatible, and $d\alpha_2$ -compatible.

In the case that $V^2 = \emptyset$, all conditions involving V^2 are considered to be vacuously satisfied.

Definition 6.27 Given a family of almost complex structures \hat{J}_t , we define its energy by

(6-40)
$$\mathcal{E}(\hat{J}_t) := \inf\{\mathcal{E}(\sigma) \mid \sigma \in \Sigma(\hat{X}^{02,t}, \hat{\lambda}^{02,t}), \ \hat{J}_t \text{ is adapted to } \sigma\} \in \mathbb{R} \cup \{\pm \infty\}.$$

If \hat{J}_t is not adapted to any cobordism decomposition, we set $\mathcal{E}(\hat{J}_t) = \infty$.

Let $\mathcal{J}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})$ be the set of almost complex structures adapted to some Type B decomposition $\sigma \in \Sigma_B(H^{02,t}, \hat{\lambda}^{02,t}, \hat{\lambda}^{02,t}|_{H^{02,t}})$. For $C \in \mathbb{R}$, let $\mathcal{J}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})_{\leq C}$ (resp. $\leq c$) be the set of all such decompositions having energy less than C (resp. at most C).

Let $\mathcal{J}((\hat{X}^{01}, \hat{\lambda}^{01}), (\hat{X}^{12}, \hat{\lambda}^{12}))_{\leq C} \subset \mathcal{J}(\hat{X}^{01}, \hat{\lambda}^{01}) \times \mathcal{J}(\hat{X}^{12}, \hat{\lambda}^{12})$ (resp. $\leq C$) be the subspace of pairs (J, J') with the property that $\mathcal{E}(J) + \mathcal{E}(J') < C$ (resp. $\leq C$).

The following lemma is an analog of Lemma 6.25 and can be proved by similar arguments.

Lemma 6.28 The set $\mathcal{J}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})_{\leq C}$ is nonempty for $C > \mathcal{E}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})$. If moreover $\alpha_0 = \alpha_1 = \alpha_2$ and $(H^{01}, \hat{\lambda}^{01}|_{H^{01}}), (H^{12}, \hat{\lambda}^{12}|_{H^{12}})$ and $(H^{02,t}, \hat{\lambda}^{02,t}|_{H^{02,t}})$ are symplectizations, then $\mathcal{J}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})_{\leq 0}$ is nonempty.

We will also need the following lemma, which follows from Lemma 6.21 and standard arguments for extending compatible almost complex structures.

Lemma 6.29 Suppose that $\mathcal{J}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})_{< C}$ is nonempty. The map that associates to an almost complex structure $\hat{J}_t \in \mathcal{J}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})_{< C}$ the pair $(\hat{J}^{01}, \hat{J}^{12}) \in \mathcal{J}((\hat{X}^{01}, \hat{\lambda}^{01}), (\hat{X}^{12}, \hat{\lambda}^{12}))_{< C}$ is surjective for all C > 0.

If moreover $\alpha_0 = \alpha_1 = \alpha_2$, and $(H^{01}, \hat{\lambda}^{01}|_{H^{01}}), (H^{12}, \hat{\lambda}^{12}|_{H^{12}})$ and $(H^{02,t}, \hat{\lambda}^{02,t}|_{H^{02,t}})$ are symplectizations, then the same statement holds for C = 0 with \leq in place of <.

7 Enriched setups and twisted moduli counts

7.1 Enriched setups

The construction of invariants of codimension 2 contact submanifolds in this paper follows the same general scheme as Pardon's construction of contact homology. However, we work with a class of "enriched" Setups I*–IV*, which contain more information than the standard Setups I–IV considered by Pardon and reviewed in Section 4.1.

We will show in Section 7.2 that the data associated to our enriched setups give rise to twisting maps. These twisting maps are constructed using Siefring's intersection theory, and will be used to define "twisted" moduli counts, following the construction of Section 6.1.

Given a datum \mathcal{D} for any of Setups I*–IV*, there is a "forgetful functor" which allows one to view \mathcal{D} as a datum of Setups I–IV. However, it is not the case that every datum of Setups I–IV admits an enrichment. Nevertheless, we will show in Section 8.1 that the class of enriched data is large enough for the purpose of defining invariants in the spirit of contact homology.

Setup I* A datum $\mathcal{D} = ((Y, \xi, V), \mathfrak{r}, \lambda, J)$ for Setup I* consists of

- a TN contact pair (Y, ξ, V) ,
- an element $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V),$
- a contact form ker $\lambda = \xi$ which is adapted to \mathfrak{r} , and
- an almost complex structure J which is compatible with $d\lambda$ and preserves ξ_V .

Observe that there is a "forgetful functor" from Setup I* to Setup I which remembers (Y, λ, J) but forgets V and \mathfrak{r} . One has analogous forgetful functors for the other setups.

Setup II* A datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, \hat{\lambda}, H, \hat{J})$ for Setup II* consists of

- data $\mathcal{D}^{\pm} = ((Y^{\pm}, \xi^{\pm}, V^{\pm}), \mathfrak{r}^{\pm}, \lambda^{\pm}, J^{\pm})$ for Setup I*, where we write $\mathfrak{r}^{\pm} = (\alpha_V^{\pm}, \tau^{\pm}, r^{\pm})$,
- an exact relative symplectic cobordism (X̂, λ̂, H) with positive end (Y⁺, λ⁺, V⁺) and negative end (Y⁻, λ⁻, V[±]), and
- an $d\hat{\lambda}$ -tame almost complex structure \hat{J} on \hat{X} which agrees with \hat{J}^{\pm} at infinity.

This datum is moreover subject to the conditions that

- there exists a trivialization of the normal bundle of H which restricts to τ^+ (resp. τ^-) on the positive (resp. negative) end, and
- $\mathcal{E}(\hat{J}) < \infty$ and $r^+ \ge e^{\mathcal{E}(\hat{J})}r^-$.

Setup III* A datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, \hat{\lambda}^t, H^t, \hat{J}^t)_{t \in [0,1]}$ for Setup III* consists of

- data $\mathcal{D}^{\pm} = ((Y^{\pm}, \xi^{\pm}, V^{\pm}), \mathfrak{r}^{\pm}, \lambda^{\pm}, J^{\pm})$ for Setup I*,
- a family of exact relative symplectic cobordisms (X̂, λ̂^t, Ĥ^t) for t ∈ [0, 1], with positive end (Y⁺, λ⁺, V⁺) and negative end (Y⁻, λ⁻, V⁻), and
- a family $d\hat{\lambda}^t$ -tame almost complex structures \hat{J}^t on \hat{X} , which agree with \hat{J}^{\pm} at infinity.

This datum is moreover subject to the conditions that

- there exists a trivialization of the normal bundle of H which restricts to τ^+ (resp. τ^-) on the positive (resp. negative) end, and
- $\mathcal{E}(\hat{J}^t) < \infty$ and $r^+ \ge e^{\mathcal{E}(\hat{J}^t)}r^-$.

Setup IV* A datum $\mathcal{D} = (\mathcal{D}^{01}, \mathcal{D}^{12}, (\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t}, \hat{J}^{02,t})_{t \in [0,\infty)})$ for Setup IV* consists of

- data $\mathcal{D}^i = ((Y^i, \xi^i, V^i); \mathfrak{r}^i, \lambda^i, J^i)$ for Setup I* for i = 0, 1, 2,
- a datum $\mathcal{D}^{01} = (\mathcal{D}^0, \mathcal{D}^1, \hat{X}^{01}, \hat{\lambda}^{01}, H^{01}, \hat{J}^{01})$ for Setup II*,
- a datum $\mathcal{D}^{12} = (\mathcal{D}^1, \mathcal{D}^2, \hat{X}^{12}, \hat{\lambda}^{12}, H^{12}, \hat{J}^{12})$ for Setup II*, and
- a family of cylindrical symplectic submanifolds $H^{02,t} \subset \hat{X}^{02,t}$ for $t \in [0,\infty)$, such that

$$(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})_{t \in [0,\infty)}$$

is a family of exact relative symplectic cobordisms that agrees for t large with the t-gluing of the relative symplectic cobordisms $(\hat{X}^{01}, \hat{\lambda}^{01}, H^{01})$ and $(\hat{X}^{12}, \hat{\lambda}^{12}, H^{12})$.

This datum is moreover subject to the conditions that

- there exists a trivialization of the normal bundle of $H^{02,t}$ which restricts to τ^0 (resp. τ^2) on the positive (resp. negative) end,
- there exists a trivialization of the normal bundle of H^{01} which restricts to τ^0 (resp. τ^1) on the positive (resp. negative) end,

- there exists a trivialization of the normal bundle of H^{12} which restricts to τ^1 (resp. τ^2) on the positive (resp. negative) end,
- $\mathcal{E}(\hat{J}^{02,t}) < \infty, \mathcal{E}(\hat{J}^{01}) < \infty$ and $\mathcal{E}(\hat{J}^{12}) < \infty$, and
- $r^0 \ge e^{\mathcal{E}(\hat{J}^{02,t})} r^2; r^0 \ge e^{\mathcal{E}(\hat{J}^{01})} r^1; \text{ and } r^1 \ge e^{\mathcal{E}(\hat{J}^{12})} r^2.$

We note that the requirement in Setups II*–IV* that the almost complex structures have finite energy is of course vacuous if r^- , r_1 and r_2 are nonzero (or equivalently, V^- , V_1 and V_2 are nonempty).

7.2 Twisting maps associated to enriched setups

In this section, we construct twisting maps on the contact homology algebra. These maps depend on geometric data involving codimension 2 contact submanifolds and relative symplectic cobordisms.

Setup I* Let $\mathcal{D} = ((Y, \xi, V), \mathfrak{r}, \lambda, J)$ be a datum for Setup I*, where $\mathfrak{r} = (\alpha_V, \tau, r)$. There is an obvious functor from $\mathcal{S}_{\mathrm{I}}(\mathcal{D})$ to the category $\hat{\mathcal{S}}(\hat{Y})$ defined in Section 5.3. We therefore have a well-defined intersection number $T * \hat{V}$ for $T \in \mathcal{S}_{\mathrm{I}}(\mathcal{D})$. We now introduce twisting maps associated to the above setup.

Definition 7.1 We define a map $\psi_V(T): \mathcal{S}_{I}^{\neq \emptyset}(\mathcal{D}) \to \mathbb{Q}[U]$ by

(7-1) $\psi_V(T) = U^{T * \hat{V} + \Gamma^-(T,V)},$

where $\Gamma^{-}(T, V)$ denotes the number of output edges *e* of *T* such that the corresponding Reeb orbit γ_e is contained in *V*. Corollary 5.14 ensures that the exponents appearing in these definitions are nonnegative.

Remark 7.2 Corollary 5.14 only applies to trees T such that $\overline{\mathcal{M}}(T) \neq \emptyset$. This is why the definition of twisting maps only requires them to be defined on $S^{\neq\emptyset}$ and not on the whole category S.

Definition 7.3 We define a map $\tilde{\psi}_V : S_{\mathrm{I}}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}$ by

(7-2)
$$\widetilde{\psi}_V(T) = \begin{cases} 1 & \text{if } T * \widehat{V} = 0 \text{ and } |\gamma_e| \cap V = \emptyset \text{ for every } e \in E(T), \\ 0 & \text{otherwise.} \end{cases}$$

We must now check that the maps in Definitions 7.1 and 7.3 satisfy the axioms of Definition 6.1.

Proposition 7.4 The map ψ_V introduced in Definition 7.1 is a twisting map.

Proof It follows from Proposition 5.11 that $\psi_V(T) = \psi_V(T')$ for any morphism $T \to T'$.

Let $\{T_i\}_i$ be a concatenation in $\mathcal{S}_1^{\neq \varnothing}$. We need to show that $\psi_V(\#_i T_i) = \prod_i \psi_V(T_i)$, ie

(7-3)
$$(\#_i T_i) * \hat{V} + \Gamma^-(\#_i T_i, V) = \sum_i T_i * \hat{V} + \Gamma^-(T_i, V).$$

We assume $V \neq \emptyset$ (otherwise there is nothing to say). Since the contact form λ is positive-elliptic near V, we have $p_N(\gamma) = 1$ for every Reeb orbit γ contained in V by Proposition 3.8. Remark 5.5 and Corollary 5.4 therefore imply that $\hat{\gamma} * \hat{V}$ is equal to -1 if γ is contained in V and 0 otherwise. By Proposition 5.12, this means that

(7-4)
$$T * \hat{V} = \sum_{v \in V(T)} \beta_v * \hat{V} + \Gamma^{\text{int}}(T, V)$$

for all $T \in S_{I}$, where $\Gamma^{int}(T, V)$ denotes the number of edges $e \in E^{int}(T)$ such that γ_{e} is contained in *V*. Equation (7-3) is therefore equivalent to

(7-5)
$$\Gamma^{\text{int}}(\#_i T_i, V) + \Gamma^-(\#_i T_i, V) = \sum_i (\Gamma^{\text{int}}(T_i, V) + \Gamma^-(T_i, V)).$$

The result now follows from the observation that there is a (canonical) label-preserving bijection between $E^{\text{int}}(\#_i T_i) \cup E^-(\#_i T_i)$ and $\bigcup_i (E^{\text{int}}(T_i) \cup E^-(T_i))$ —this is an immediate consequence of the definition: every interior edge of T_j corresponds to an interior edge of $\#_i T_i$, and every output edge of T_j corresponds either to an interior or an output edge of $\#_i T_i$ depending on whether it is identified with another edge in the concatenation or not.

It will be convenient to introduce the following definition.

Definition 7.5 Given a tree $T \in S_I$, a vertex $v \in V(T)$ is *mean* if it is an interior vertex and $|\gamma_e| \subset V$ for all $e \in e^+(v) \sqcup E^-(v)$. All other vertices are said to be *nice*. These sets are denoted by $V_m(T) \subset V(T)$ and $V_n(T) \subset V(T)$, respectively.

Remark 7.6 This notion of *nice/mean* vertices is purely auxiliary (and has nothing to do with good/bad Reeb orbits!). Geometrically, mean vertices correspond to holomorphic buildings which have intermediate orbits intersecting \hat{V} . Nice orbits do not affect the intersection number of the building, but mean orbits do affect it and must therefore be treated carefully (hence the adjective).

Proposition 7.7 The map $\tilde{\psi}_V$ introduced in Definition 7.3 is a twisting map.

Proof Fix a tree $T \in S_{I}^{\neq \emptyset}$. We first show that $\tilde{\psi}_{V}(T') = \tilde{\psi}_{V}(T)$ for any tree $T' \in S_{I}^{\neq \emptyset}$ admitting a morphism $T' \to T$. Observe that we may assume without loss of generality that T' is representable by a \hat{J} -holomorphic building; see Definition 5.13. Indeed, since $T, T' \in S_{I}^{\neq \emptyset}$, there exists $T'' \to T \to T$ such that T'' is representable by a \hat{J} -holomorphic building. So we may as well prove that $\tilde{\psi}_{V}(T'') = \tilde{\psi}_{V}(T')$ and $\tilde{\psi}_{V}(T'') = \tilde{\psi}_{V}(T)$.

Let us therefore fix $T' \in S_{I}^{\neq \emptyset}$ such that T' is representable by a \hat{J} -holomorphic building, and a morphism $T' \to T$. It follows from Proposition 5.11 that $T' * \hat{V} = T * \hat{V}$. Note that T' and T have the same exterior edges. If one of these edges is contained in V, then $\tilde{\psi}_{V}(T') = \tilde{\psi}_{V}(T) = 0$. So we can assume that the exterior edges of T' and T are not contained in V.

Suppose now that T' has an interior edge contained in V. For i = 0, 1, 2, let $X_i \ge 0$ be the number of edges $e \in E(T')$ such that $|\gamma_e| \subset V$ and e is adjacent to exactly i mean vertices. By assumption, we have $X_2 + X_1 + X_0 \ge 1$. According to Proposition 5.12, we have

(7-6)
$$T' * \hat{V} = \sum_{v \in V_n(T')} \beta_v * \hat{V} + \sum_{v \in V_m(T')} \beta_v * \hat{V} + X_2 + X_1 + X_0$$

According to Proposition 5.3, we also have that $\sum_{v \in V_n(T')} \beta_v * \hat{V} \ge 0$ —here we use the fact that T' is representable by a \hat{J} -holomorphic building. If there are no mean vertices, then we have that $\sum_{v \in V_m(T')} \beta_v * \hat{V} = 0$, $X_1 = X_2 = 0$ and $X_0 \ge 1$. So $T' * \hat{V} > 0$. If there exists at least one mean vertex, observe that we have $X_2 \le \#V_m(T')-1$. Moreover, given $v \in V_m(T')$, Proposition 5.6 together with the fact that T' is representable by a \hat{J} -holomorphic building imply that $\beta_v * \hat{V} \ge 1 - p_v$, where p_v is the number of edges adjacent to v. It follows that $\sum_{v \in V_m(T')} \beta_v * \hat{V} + X_2 + X_1 \ge (\#V_m(T') - X_1 - 2X_2) + X_2 + X_1 = \#V_m(T') - X_2 \ge 1$. It thus follows again that $T' * \hat{V} > 0$. We conclude that $\tilde{\psi}_H(T') = \tilde{\psi}_H(T) = 0$ if T' has an interior edge contained in V.

We are left with the case where T' and hence T have no edges contained in V. It is then immediate that $\tilde{\psi}_V(T') = \tilde{\psi}_V(T)$.

We now show that any concatenation $\{T_i\}_i$ satisfies $\tilde{\psi}_V(\#_i T_i) = \prod_i \tilde{\psi}_V(T_i)$. If one of the T_i has an edge contained in V, then $\#_i T_i$ also has an edge contained in V and we have $\tilde{\psi}_V(\#_i T_i) = \prod_i \tilde{\psi}_V(T_i) = 0$. If none of the T_i have an edge contained in V, then the same is true for $\#_i T_i$. Hence Proposition 5.12 implies that $\#_i T_i * \hat{V} = \sum_i T_i * \hat{V}$. By positivity of intersection (Proposition 5.3), $\sum_i T_i * \hat{V} = 0$ if and only if $T_i * \hat{V} = 0$ for all i. It then follows that $\tilde{\psi}_V(\#_i T_i) = \prod_i \tilde{\psi}_V(T_i)$.

Setup II* Fix a datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, H, \hat{\lambda}, \hat{J})$ for Setup II*, where we write

$$\mathcal{D}^{\pm} = ((Y^{\pm}, \xi^{\pm}, V^{\pm}), \mathfrak{r}^{\pm}, \lambda^{\pm}, J^{\pm}) \quad \text{and} \quad \mathfrak{r}^{\pm} = (\alpha^{\pm}, \tau^{\pm}, r^{\pm}).$$

We now introduce the following twisting maps.

Definition 7.8 We define a map $\psi_H : \mathcal{S}_{\mathrm{II}}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}[U]$ by (7-7) $\psi_H(T) = U^{T*H + \Gamma^-(T,V^-)}.$

Definition 7.9 We define a map $\widetilde{\psi}_H : \mathcal{S}_{II}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}$ by

(7-8)
$$\widetilde{\psi}_H(T) = \begin{cases} 1 & \text{if } T * H = 0 \text{ and } |\gamma_e| \cap V^{\pm} = \emptyset \text{ for every } e \in E(T), \\ 0 & \text{otherwise.} \end{cases}$$

We need to verify that the above definitions satisfy the axioms of twisting maps. The first step is to prove that the $\psi_H(T)$ are nonnegative powers of U. This is the content of Corollary 7.12, whose proof requires some preparatory lemmas. (In the next two lemmas, $\dot{\Sigma}$ always denotes an arbitrary punctured Riemann surface.)

Homological invariants of codimension 2 contact submanifolds

Lemma 7.10 For $n \ge 1$, suppose that $\beta \in \pi_2(\hat{X}, \gamma^+ \sqcup (\bigcup_{i=1}^n \gamma_i^-))$ is represented by a \hat{J} -holomorphic curve $u: \dot{\Sigma} \to \hat{X}$ which is contained in H. Then $P^+ - e^{-\mathcal{E}(\hat{J})}(\sum_{i=1}^n P_i^-) \ge 0$, where P^+ (resp. P_i^-) is the period of $\gamma^+ \subset (Y^+, \lambda^+)$ (resp. the period of $\gamma_i^- \subset (Y^-, \lambda^-)$).

Proof The claim is trivial if $\mathcal{E}(\hat{J}) = \infty$, so let us assume that $\hat{J} \in \mathcal{J}(\hat{X}, \hat{\lambda}, H)$. We may therefore fix a Type A decomposition σ of (H, λ_H) , which is specified by a pair of hypersurfaces $\mathcal{H}_- = \{-C_1\} \times V^-$ and $\mathcal{H}_+ = \{C_+\} \times V^+$.

It will be convenient to define the regions $R^- := (-\infty, -C_1] \times V^-$, $R^+ := [C_2 \times \infty) \times V^+$ and $\tilde{H} = H - (int(R^-) \cup int(R^+))$. Let us first assume that u is transverse to the boundary of \tilde{H} . Consider now the sum

(7-9)
$$\int_{u^{-1}(R^{-})} e^{-C_1} u^* d\alpha^- + \int_{u^{-1}(\tilde{H})} u^* d\hat{\lambda} + e^{C_2} \int_{u^{-1}(R^+)} u^* d\alpha^+.$$

Each summand is nonnegative due to the fact that u is \hat{J} -holomorphic and that \hat{J} is adapted to σ . By Stokes' theorem, the sum of the integrals is $e^{C_2}P^+ - e^{-C_1}(\sum_{i=1}^n P_i^-) \ge 0$. This implies that $P^+ \ge e^{-\mathcal{E}(\sigma)}(\sum_{i=1}^n P_i^-)$.

If *u* is not transverse to the boundary of \tilde{H} , observe by Sard's theorem that transversality can be achieved for a sequence of domains $\tilde{H}^n := \tilde{H} \cup [-C_1^n, -C_1] \cup [C_2, C_2^n]$, where $\{C_i^n\}_{n=0}^{\infty}$ is monotonically decreasing and $C_i^n \to C_i$. It is easy to verify that \hat{J} is still adapted to the Type A decompositions induced by the boundary of \tilde{H}^n , so the above argument goes through and passing to the limit gives $P^+ \ge e^{-\mathcal{E}(\sigma)} (\sum_{i=1}^n P_i^-).$

The lemma now follows from the definition of $\mathcal{E}(\hat{J})$.

Lemma 7.11 For $n \ge 0$, suppose that $\beta \in \pi_2(\hat{X}, \gamma^+ \sqcup (\bigcup_{i=1}^n \gamma_i^-))$ is represented by a \hat{J} -holomorphic curve $u: \dot{\Sigma} \to \hat{X}$. (Note that unlike in Lemma 7.10, we allow n = 0 in which case the union is interpreted as being empty.) Then $\beta * H \ge -n_u$, where n_u is the total number of negative punctures of u contained in $V^- \subset Y^-$.

Proof According to Proposition 5.3, we only need to consider the case where the image of u is contained in H. By definition of a datum for Setup II*, the trivializations τ^{\pm} extend to a global trivialization τ of the normal bundle of H, which implies that $u^{\tau} \cdot H = 0$. Using the fact that $u^{\tau} \cdot H = 0$, we have (see Definition 5.1 and the proof of Proposition 5.6)

(7-10)
$$u * H = \alpha_N^{\tau;-}(\gamma^+) - \sum_{i=1}^n \alpha_N^{\tau;+}(\gamma_i^-) = \lfloor CZ_N^{\tau}(\gamma^+)/2 \rfloor - \sum_{i=1}^n \lceil CZ_N^{\tau}(\gamma_i^-)/2 \rceil$$
$$= \lfloor r^+P^+ \rfloor - \sum_{i=1}^n \lfloor r^-P_i \rfloor,$$

where the sum is interpreted as zero if u has no negative punctures.

We may assume that $n \ge 1$ and $r^- > 0$ (otherwise the lemma is automatic). Let $p_u = n_u + 1$ be the total number of punctures (positive and negative) of u contained in $V^{\pm} \subset Y^{\pm}$. Using the trivial bounds $x - 1 < \lfloor x \rfloor \le x$, we obtain

(7-11)
$$u * H > (r^+P^+ - 1) - \sum_{i=1}^n (1 + r^-P_i^-) = -p_u + r^+P^+ - r^- \sum_{i=1}^n P_i^-.$$

Using now Lemma 7.10 and the fact that $r^+ \ge e^{\mathcal{E}(\hat{J})}r^-$, we have

(7-12)
$$-p_u + r^+ P^+ - r^- \sum_{i=1}^n P_i^- \ge -p_u + r^+ \epsilon^{-\mathcal{E}(\hat{J})} \sum_{i=1}^n P_i^- - r^- \sum_{i=1}^n P_i^- \ge -p_u.$$

The claim follows.

Corollary 7.12 We have
$$T * H \ge -\Gamma^-(T, V^-)$$
 for any $T \in \mathcal{S}_{II}^{\neq \emptyset}(\mathcal{D})$. Hence $\psi_H(T) \in \mathbb{Q}[U]$.

Proof Since $T \in S_{\text{II}}^{\neq \varnothing}(\mathcal{D})$, there exists $T' \to T$ such that T' is representable by a holomorphic building. Since the Siefring number is invariant under gluing (Proposition 5.11), we may assume that T is representable by a holomorphic building. We now apply Proposition 5.17: it therefore suffices to check that for each $v \in T$, the intersection number $\beta_v * \eta_{*(v)}$ is bounded below by $-\#\{E^-(v)\}$. In the case that *(v) = 01, this follows from Lemma 7.11. In the case that *(v) = 00 or *(v) = 11, this follows either from Lemma 7.11 or (more directly) from Proposition 5.6.

Proposition 7.13 Let $\{T_i\}_i$ be a concatenation in S_{II} . Then we have

$$(\#_{i} T_{i}) * H + \Gamma^{-}(\#_{i} T_{i}, V^{-}) = \sum_{T_{i} \in \mathcal{S}_{I}^{+}} (T_{i} * \hat{V}^{+} + \Gamma^{-}(T_{i}, V^{+})) + \sum_{T_{i} \in \mathcal{S}_{II}} (T_{i} * H + \Gamma^{-}(T_{i}, V^{-})) + \sum_{T_{i} \in \mathcal{S}_{I}^{-}} (T_{i} * \hat{V}^{-} + \Gamma^{-}(T_{i}, V^{-})).$$

Proof As in the proof of Proposition 7.4, our assumptions imply that $\hat{\gamma} * V^{\pm} = -1$ if γ is contained in V^{\pm} and 0 otherwise. By Proposition 5.15, we have

(7-13)
$$T * H = \sum_{\substack{v \in V(T) \\ *(v) = 00}} \beta_v * \hat{V}^+ + \sum_{\substack{v \in V(T) \\ *(v) = 01}} \beta_v * H + \sum_{\substack{v \in V(T) \\ *(v) = 11}} \beta_v * \hat{V}^- + \Gamma^{\text{int}}(T, V^+) + \Gamma^{\text{int}}(T, V^-)$$

for all $T \in S_{II}$. By applying this formula to $T = \#_i T_i$ (and also using (7-4)), we see that it suffices to prove that

$$(7-14) \quad \Gamma^{\text{int}}(\#_{i}T_{i}, V^{+}) + \Gamma^{\text{int}}(\#_{i}T_{i}, V^{-}) + \Gamma^{-}(\#_{i}T_{i}, V^{-}) \\ = \sum_{T_{i} \in \mathcal{S}_{I}^{+}} \Gamma^{\text{int}}(T_{i}, V^{+}) + \Gamma^{-}(T_{i}, V^{+}) + \sum_{T_{i} \in \mathcal{S}_{II}} \Gamma^{\text{int}}(T_{i}, V^{+}) + \Gamma^{\text{int}}(T_{i}, V^{-}) + \Gamma^{-}(T_{i}, V^{-}) \\ + \sum_{T_{i} \in \mathcal{S}_{I}^{-}} \Gamma^{\text{int}}(T_{i}, V^{-}) + \Gamma^{-}(T_{i}, V^{-}).$$

Geometry & Topology, Volume 28 (2024)

As in the proof of Proposition 7.4, this is just a matter of understanding how the edges of $\#_i T_i$ are obtained from the edges of the T_i , following the discussion in [61, Section 2.2]. More precisely, let us check that every edge counted on the right-hand side of (7-14) is also counted on the left-hand side. Note that under concatenation, interior edges remain interior edges. Output edges either remain output edges, or they become interior edges. The output edges corresponding to $\Gamma^-(T_i, V^+)$ for $T_i \in S_I^+$ must all become interior edges of $\#_i T_i$: indeed, any such output edge has label *(e) = 0, but the output edges of $\#_i T_i$ have label *(e) = 1. These output edges are thus counted in $\Gamma^{int}(\#_i T_i, V^+)$.

The output edges corresponding to $\Gamma^{-}(T_i, V^{-})$ for $T_i \in S_{\Pi}$ may either become interior edges of $\#_i T_i$ (in which case they are counted in $\Gamma^{\text{int}}(\#_i T_i, V^{-})$), or remain output edges (in which case they are counted in $\Gamma^{-}(\#_i T_i, V^{-})$). Similarly, the output edges corresponding to $\Gamma^{-}(T_i, V^{-})$ for $T_i \in S_{\Pi}^{-}$ may either become interior edges of $\#_i T_i$ (counted in $\Gamma^{\text{int}}(\#_i T_i, V^{-})$) or remain output edges (counted in $\Gamma^{-}(\#_i T_i, V^{-})$). \Box

Corollary 7.14 Under the assumptions of Proposition 7.13, $\psi_H \in \Psi_{II}(\mathcal{D}; \psi_{V^+}, \psi_{V^-})$.

Proof Proposition 5.11 implies that $\psi_H(T) = \psi_H(T')$ for any morphism $T \to T'$. Proposition 7.13 implies that ψ_H acts correctly on concatenations.

We now want to show that $\tilde{\psi}_H$ is a twisting map. We will need the following definition.

Definition 7.15 Given a tree $T \in S_{II}$, a vertex $v \in V(T)$ is *mean* if it is an interior vertex and $|\gamma_e| \subset V^{\pm}$ for all $e \in e^+(v) \sqcup E^-(v)$. All other vertices are said to be *nice*. These sets are denoted by $V_m(T) \subset V(T)$ and $V_n(T) \subset V(T)$, respectively.

Proposition 7.16 Under the assumptions of Definition 7.9, $\tilde{\psi}_H \in \Psi_{II}(\mathcal{D}; \tilde{\psi}_{V^+}, \tilde{\psi}_{V^-})$.

Proof Consider a tree $T' \in S_{II}^{\neq \emptyset}$ with a morphism $T' \to T$. We wish to show that $\tilde{\psi}_H(T') = \tilde{\psi}_H(T)$. As in the proof of Proposition 7.7, we may assume that T' is representable by a building; see Definition 5.16.

It follows from Proposition 5.11 that T' * H = T * H. Note that T' and T have the same exterior edges. If one of these edges is contained in V^{\pm} , then $\tilde{\psi}_H(T') = \tilde{\psi}_H(T) = 0$. So we can assume that the exterior edges of T' and T are not contained in V^{\pm} .

Suppose now that T' has an interior edge contained in V^{\pm} . Arguing as in the proof of Proposition 7.7, let $X_i \ge 0$ for i = 0, 1, 2 denote the number of edges $e \in E(T')$ such that $|\gamma_e| \subset V^{\pm}$ and e is adjacent to exactly i mean vertices. By assumption $X_0 + X_1 + X_2 \ge 1$. By Proposition 5.15, we have

(7-15)
$$T' * H = \sum_{v \in V_n(T')} \beta_v * H_v + \sum_{v \in V_m(T')} \beta_v * H_v + X_2 + X_1 + X_0,$$

where we write $H_v = \hat{V}^+$ if *(v) = 00, $H_v = H$ if *(v) = 01 and $H_v = \hat{V}^-$ if *(v) = 11. According to Proposition 5.6 and the fact that T' is representable by a building, we have that $\sum_{v \in V_n(T')} \beta_v * H_v \ge 0$.

If there are no mean vertices, then $\sum_{v \in V_m(T')} \beta_v * H_v = X_1 = X_2 = 0$ and $X_0 \ge 1$. Hence $T' * H \ge 1$. If there exists at least one mean vertex, observe that $X_2 \le \#V_m(T') - 1$. According to Lemma 7.11 and the fact that T' is representable by a building, we have that $\sum_{v \in V_m(T)} \beta_v * H_v + X_2 + X_1 \ge \#V_m(T') - X_1 - 2X_2 + X_2 + X_1 = \#V_m(T') - X_2 \ge 1$. It thus follows again that $T' * \hat{V} \ge 1$. We conclude that $\tilde{\psi}_V(T') = \tilde{\psi}_V(T) = 0$ if T' has an interior edge contained in V^{\pm} .

We are left with the case where T' and hence T have no edges contained in V^{\pm} . It's then immediate that $\tilde{\psi}_H(T') = \tilde{\psi}_H(T)$.

If $\{T_i\}_i$ is a concatenation, then the argument is the same as in the proof of Proposition 7.7 (using Proposition 5.15 instead of Proposition 5.12).

Setup III* Fix a datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, \hat{\lambda}^t, H^t, \hat{J}^t)_{t \in [0,1]}$ for Setup III*, where $\mathcal{D}^{\pm} = ((Y^{\pm}, \xi^{\pm}, V^{\pm}), \mathfrak{r}^{\pm}, \lambda^{\pm}, J^{\pm}).$

We now introduce the following twisting maps.

Definition 7.17 We define a map
$$\psi_{H^t} : S_{\text{III}}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}[U]$$
 by
(7-16) $\psi_{H^t}(T) = U^{T * H^t + \Gamma^-(T, V^-)}.$

Definition 7.18 We define a map $\tilde{\psi}_{H^t} : S_{\text{III}}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}$ by

(7-17)
$$\widetilde{\psi}_{H^t}(T) = \begin{cases} 1 & \text{if } T * H^t = 0 \text{ and } |\gamma_e| \cap V^{\pm} = \emptyset \text{ for every } e \in E(T), \\ 0 & \text{otherwise.} \end{cases}$$

There is no difference between S_{III} and S_{II} from the point of view of the intersection theory defined in Section 5.3. It can therefore be shown by essentially the same arguments as in the previous section that the above definitions do indeed satisfy the axioms for twisting maps.

Corollary 7.19 We have $\psi_{H^{t}} \in \Psi_{\mathrm{III}}(\mathcal{D}; \psi_{H^{0}}, \psi_{H^{1}})$ and $\tilde{\psi}_{H^{t}} \in \Psi_{\mathrm{III}}(\mathcal{D}; \tilde{\psi}_{H^{0}}, \tilde{\psi}_{H^{1}})$.

Setup IV* Fix datum $\mathcal{D} = (\mathcal{D}^{01}, \mathcal{D}^{12}, (\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t}, \hat{J}^{02,t})_{t \in [0,\infty)})$ for Setup IV*. Here,

• $\mathcal{D}^{01} = (\mathcal{D}^0, \mathcal{D}^1, \hat{X}^{01}, \hat{\lambda}^{01}, H^{01}, \hat{J}^{01})$ is a datum for Setup II*,

• $\mathcal{D}^{12} = (\mathcal{D}^1, \mathcal{D}^2, \hat{X}^{12}, \hat{\lambda}^{12}, H^{12}, \hat{J}^{12})$ is a datum for Setup II*, and

• $\mathcal{D}^i = ((Y^i, \xi^i, V^i); \mathfrak{r}^i, \lambda^i, J^i)$ is a datum for Setup I* for i = 0, 1, 2.

We introduce the following twisting maps.

Definition 7.20 We define $\psi_{H^{02,l}} : S_{\mathrm{IV}}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}[U]$ by (7-18) $\psi_{H^{02,l}}(T) = U^{T*\eta + \Gamma^{-}(T,V^{2})}.$

Homological invariants of codimension 2 contact submanifolds

Definition 7.21 We define $\widetilde{\psi}_{H^{02,t}} : S_{\mathrm{IV}}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}$ by

(7-19)
$$\widetilde{\psi}_{H^{02,t}}(T) = \begin{cases} 1 & \text{if } T * \eta = 0 \text{ and } |\gamma_e| \cap V^i = \emptyset \text{ for all } e \in E(T) \text{ and } i \in \{0, 1, 2\}, \\ 0 & \text{otherwise.} \end{cases}$$

We need to show that the powers of U appearing in Definition 7.20 are nonnegative. This will be the content of Corollary 7.24, which requires some preparatory lemmas.

Lemma 7.22 For $n \ge 1$, suppose that $u: \dot{\Sigma} \to \hat{X}^{02,t}$ is $\hat{J}^{02,t}$ -holomorphic with positive orbit γ^+ and negative orbits $\bigcup_{i=1}^{n} \gamma_i^-$. Then we have $P^+ - e^{-\mathcal{E}(\hat{J}^{02,t})} \sum_i P_i^- \ge 0$, where P^+ (resp. P_i^-) is the period of $\gamma^+ \subset V^0$ (resp. $\gamma_i^- \subset V^2$).

Proof The proof is analogous to that of Lemma 7.10. If $\mathcal{E}(\hat{J}^{02,t}) = \infty$, the result is trivial. Hence we may assume that $\hat{J}^{02,t} \in \mathcal{J}(\hat{X}^{02,t}, \hat{\lambda}^{02,t}, H^{02,t})$ and fix a Type B cobordism decomposition $\sigma^{02,t}$ of $(H^{02,t}, \hat{\lambda}^{02,t}|_{H^{02,t}})$ to which $\hat{J}^{02,t}$ is adapted. The decomposition $\sigma^{02,t}$ is specified by a family of hypersurfaces

$$\mathcal{H}_2(t) = \{-C_2(t)\} \times V^2, \quad \mathcal{H}_1(t) = \{-C_1(t)\} \times V^1, \quad \tilde{\mathcal{H}}_1(t) = \{\tilde{C}_1(t)\} \times V^1, \quad \mathcal{H}_0(t) = \{C_0(t)\} \times V^0.$$

It will be convenient to define the regions $R_2(t) = (-\infty, -C_2(t)] \times V^2$, $R_0(t) = [C_0(t), \infty) \times V^0$ and $R_1(t) = [-C_1(t), \tilde{C}_1(t)] \times V^1$.

Suppose first that $R_1(t)$ is empty. Then $\tilde{C}_1(t) + C_1(t) = 0$, and hence $\mathcal{E}(\sigma^{02,t}) = C_0(t) + C_2(t)$. Hence $\mathcal{E}(\sigma^{02,t})$ coincides with the energy of $\sigma^{02,t}$ if it is viewed as a Type A cobordism decomposition (Definition 6.8) by forgetting C_1 and \tilde{C}_1 . Hence, when $R_1(t)$ is empty, the claim reduces to Lemma 7.10.

We now assume that $R_1(t)$ is nonempty. We suppose that $\tilde{H}_{12}^{02,t} \sqcup \tilde{H}_{01}^{02,t}$ are the connected components of $X^{02,t} - int(R_2(t) \cup R_1(t) \cup R_0(t))$. Let us first assume that the image of u intersects the boundaries of $\tilde{H}_{01}^{02,t}$ and $\tilde{H}_{12}^{02,t}$ transversally. We then have the following computations:

$$\begin{split} & \cdot \int_{u^{-1}(R_{2}(t))} u^{*} \alpha_{2} = \int_{u^{-1}(\mathcal{H}_{2}(t))} u^{*} \alpha_{2} - \sum_{i=1}^{n} P_{i}^{-} \ge 0, \\ & \cdot \int_{u^{-1}(\mathcal{H}_{21}^{02,t})} u^{*} d(e^{s} \alpha_{2}) = e^{-C_{1}(t)} \int_{u^{-1}(\mathcal{H}_{1}(t))} u^{*} \alpha_{1} - e^{-C_{2}(t)} \int_{u^{-1}(\mathcal{H}_{2}(t))} u^{*} \alpha_{2} \ge 0 \\ & \cdot \int_{u^{-1}(R_{1}(t))} u^{*} \alpha_{1} = \int_{u^{-1}(\tilde{\mathcal{H}}_{1}(t))} u^{*} \alpha_{1} - \int_{u^{-1}(\mathcal{H}_{1}(t))} u^{*} \alpha_{1} \ge 0, \\ & \cdot \int_{u^{-1}(\tilde{\mathcal{H}}_{01}^{02,t})} u^{*} d(e^{s} \alpha_{1}) = e^{C_{0}(t)} \int_{u^{-1}(\mathcal{H}_{0}(t))} u^{*} \alpha_{0} - e^{\tilde{C}_{1}(t)} \int_{u^{-1}(\tilde{\mathcal{H}}_{1}(t))} u^{*} \alpha_{1} \ge 0, \\ & \cdot \int_{u^{-1}(R_{0}(t))} u^{*} \alpha_{0} = P_{i}^{+} - \int_{u^{-1}(\mathcal{H}_{0}(t))} u^{*} \alpha_{0} \ge 0. \end{split}$$

After appropriate rescalings, these terms form a telescoping sum. We find that

$$P^{+} - e^{C_{2}(t) + C_{0}(t) - C_{1}(t) - \tilde{C}_{1}(t)} \sum_{i} P_{i}^{-} = P^{+} - e^{-\mathcal{E}(\sigma^{02,t})} \sum_{i} P_{i}^{-} \ge 0.$$

Suppose now that the image of u does not intersect the boundaries of $\tilde{H}_{01}^{02,t}$ and $\tilde{H}_{12}^{02,t}$ transversally. For $\epsilon_n \downarrow 0$, set

$$R_2^{(n)} := (-\infty, -C_2(t) - \epsilon_n] \times V^2, \quad R_0^{(n)} = [C_0(t) + \epsilon_n, \infty) \times V^0, \quad R_1^{(n)} = [-C_1(t) + \epsilon_n, \tilde{C}_1(t) - \epsilon_n] \times V^1.$$

By Sard's theorem, we may assume by choosing ϵ_n appropriately that u intersects the boundary of the $R_i^{(n)}$ transversally. Now repeat the above argument with $R_i^{(n)}$ in place of $R_i(t)$. This yields the inequality $P^+ - e^{C_2(t) + C_0(t) - C_1(t) - \tilde{C}_1(t) + 4\epsilon_n} \sum_i P_i^- \ge 0$. The claim now follows by passing to the limit. \Box

Lemma 7.23 For $n \ge 0$, suppose that $u: \dot{\Sigma} \to \hat{X}^{02,t}$ is a $\hat{J}^{02,t}$ -holomorphic curve in the homotopy class $\beta \in \pi_2(\hat{X}^{02,t}, \gamma^+ \sqcup (\bigcup_{i=1}^n \gamma_i^-))$ for $t < \infty$. Then $\beta * [H^{02,t}] \ge -n_u$, where n_u is the total number of negative punctures. (Note that unlike in Lemma 7.22, we allow n = 0 here, in which case the union is interpreted as being empty.)

Proof We argue as in the proof of Lemma 7.11. It is enough to consider the case where the image of u is contained in $H^{02,t}$. The trivialization τ extends to a global trivialization along $H^{02,t}$, implying that $\beta \cdot H^{02,t} = 0$.

We thus have

$$u * H^{02,t} = \alpha_N^{\tau;-}(\gamma^+) - \sum_{i=1}^n \alpha_N^{\tau;+}(\gamma_i^-) = \lfloor CZ_N^{\tau}(\gamma^+)/2 \rfloor - \sum_{i=1}^n \lceil CZ_N^{\tau}(\gamma_i^-)/2 \rceil$$
$$= \lfloor r_0 P^+ \rfloor - \sum_{i=1}^n + \lfloor rP_i \rfloor,$$

where the sum is interpreted as zero if u has no negative punctures. Thus the lemma is verified if n = 0 or $r_2 = 0$. It remains only to consider the case where $n \ge 1$ and $r_2 > 0$.

Using the trivial bounds $x - 1 < \lfloor x \rfloor \le x$, we obtain

$$u * \hat{V} > \sum_{z \in \mathbf{p}_{u}^{+}} (r_{0}P_{z} - 1) - \sum_{z \in \mathbf{p}_{u}^{-}} (1 + r_{2}P_{z}) \ge -p_{u} + e^{\mathcal{E}(\sigma^{02})} r_{2} \left(\sum_{z \in \mathbf{p}_{u}^{+}} P_{z}\right) - r_{2} \sum_{z \in \mathbf{p}_{u}^{-}} P_{z}.$$

It follows from Lemma 7.22 that $e^{\hat{J}^{02,t}} r_2 \left(\sum_{z \in p_u^+} P_z \right) - r_2 \sum_{z \in p_u^-} P_z \ge 0$. The claim follows. \Box

Corollary 7.24 We have that $T * \eta + \Gamma^{-}(T, V^{2}) \ge 0$.

Proof We need to consider two cases. If $\mathfrak{s}(T) \in [0, \infty)$, then the claim follows by combining Proposition 5.17 and Lemma 7.23. If $\mathfrak{s}(T) = \{\infty\}$, then the argument is the same as in the proof of Corollary 7.12.

$$\begin{aligned} (\#_{i} T_{i}) * \eta + \Gamma^{-}(T, V^{2}) \\ &= \sum_{T_{i} \in \mathcal{S}_{1}^{0}} (T_{i} * \hat{V}^{0} + \Gamma^{-}(T_{i}, V^{0})) + \sum_{T_{i} \in \mathcal{S}_{1}^{01}} (T_{i} * H^{01} + \Gamma^{-}(T_{i}, V^{1})) + \sum_{T_{i} \in \mathcal{S}_{1}^{1}} (T_{i} * \hat{V}^{1} + \Gamma^{-}(T_{i}, V^{1})) \\ &+ \sum_{T_{i} \in \mathcal{S}_{1}^{12}} (T_{i} * H^{12} + \Gamma^{-}(T_{i}, V^{2})) + \sum_{T_{i} \in \mathcal{S}_{1}^{2}} (T_{i} * \hat{V}^{2} + \Gamma^{-}(T_{i}, V^{2})). \end{aligned}$$

Proof As in the proof of Proposition 7.4, our assumptions imply that $\hat{\gamma} * V^j = -1$ if γ is contained in V^j and 0 otherwise. Proposition 5.15 implies that

$$\begin{aligned} &(\#_{i}T_{i})*\eta \\ &= \sum_{\substack{v \in V(\#_{i}T_{i})*(v)=00}} \beta_{v}*\hat{V}^{0} + \sum_{\substack{v \in V(\#_{i}T_{i})*(v)=01}} \beta_{v}*H^{01} + \sum_{\substack{v \in V(\#_{i}T_{i})*(v)=11}} \beta_{v}*\hat{V}^{1} + \sum_{\substack{v \in V(\#_{i}T_{i})*(v)=12}} \beta_{v}*H^{12} + \sum_{\substack{v \in V(\#_{i}T_{i})*(v)=22}} \beta_{v}*\hat{V}^{2} \\ &+ \Gamma^{\text{int}}(\#_{i}T_{i},V^{0}) + \Gamma^{\text{int}}(\#_{i}T_{i},V^{1}) + \Gamma^{\text{int}}(\#_{i}T_{i},V^{2}). \end{aligned}$$

As in the proof of Proposition 7.13, it follows that the result is equivalent to

$$\begin{split} \Gamma^{\text{int}}(\#_{i}T_{i}, V^{0}) + \Gamma^{\text{int}}(\#_{i}T_{i}, V^{1}) + \Gamma^{\text{int}}(\#_{i}T_{i}, V^{2}) + \Gamma^{-}(T, V^{2}) \\ &= \sum_{T_{i} \in \mathcal{S}_{I}^{0}} \Gamma^{\text{int}}(T_{i}, V^{0}) + \Gamma^{-}(T_{i}, V^{0}) + \sum_{T_{i} \in \mathcal{S}_{II}^{01}} \Gamma^{\text{int}}(T_{i}, V^{0}) + \Gamma^{\text{int}}(T_{i}, V^{1}) + \Gamma^{-}(T_{i}, V^{1}) \\ &+ \sum_{T_{i} \in \mathcal{S}_{I}^{1}} \Gamma^{\text{int}}(T_{i}, V^{1}) + \Gamma^{-}(T_{i}, V^{1}) + \sum_{T_{i} \in \mathcal{S}_{II}^{12}} \Gamma^{\text{int}}(T_{i}, V^{1}) + \Gamma^{\text{int}}(T_{i}, V^{2}) + \Gamma^{-}(T_{i}, V^{2}) \\ &+ \sum_{T_{i} \in \mathcal{S}_{I}^{2}} \Gamma^{\text{int}}(T_{i}, V^{2}) + \Gamma^{-}(T_{i}, V^{2}), \end{split}$$

which is a consequence of the way the edges of $\#_i T_i$ are obtained from the edges of the T_i .

Corollary 7.26 We have $\psi_{H^{02,t}} \in \Psi_{IV}(\mathcal{D}; \psi_{H^{01}}, \psi_{H^{12}}, \psi_{H^{02,0}})$ and we have that

$$\widetilde{\psi}_{H^{02,t}} \in \Psi_{\mathrm{IV}}(\mathcal{D}; \widetilde{\psi}_{H^{01}}, \widetilde{\psi}_{H^{12}}, \widetilde{\psi}_{H^{02,0}}).$$

Proof Proposition 7.25 shows that $\psi_{H^{02,t}}$ acts correctly on concatenations of type (2); the proof that $\psi_{H^{02,t}}$ behaves well with respect to the other two types of concatenation is virtually identical. Proposition 5.11 implies that $\psi_{H^{02,t}}(T) = \psi_{H^{02,t}}(T')$ for any morphism $T \to T'$. The argument that $\tilde{\psi}_{H^{02,t}}(T) = \tilde{\psi}_{H^{02,t}}(T')$ is essentially the same as the proof of Proposition 7.16, except that we appeal to Lemma 7.23 instead of Lemma 7.11.

The results from the previous sections can be conveniently packaged into the following theorem.

Geometry & Topology, Volume 28 (2024)

Theorem 7.27 (cf [61, Theorem 1.1]) Let \mathcal{D} be a datum for any one of Setups I*–IV*. Then there exists a set of perturbation data $\Theta(\mathcal{D})$ and twisted moduli counts

$$\#_{\psi}\overline{\mathcal{M}}(T)_{\theta}^{\operatorname{vir}} \in \mathbb{Q}[U] \quad and \quad \#_{\widetilde{\psi}}\overline{\mathcal{M}}(T)_{\theta}^{\operatorname{vir}} \in \mathbb{Q}$$

for $\theta \in \Theta(D)$ and $T \in S_*(D)$ for * = I, II, III, IV, satisfying the obvious analogs of (i)–(v) in Theorem 1.1 in [61].

Proof There is a forgetful functor taking a datum for the enriched setups I^*-IV^* (Section 7.1) to a datum for the standard Setups I–IV (Section 4.1). So the set of perturbation data is furnished by [61, Theorem 1.1]. We showed in Section 7.2 a datum for Setups I^*-IV^* gives rise to twisting maps, from which we may define our twisted moduli counts as in Section 6.1. The properties (i)–(iv) are tautological and (v) is a consequence of the axioms of twisting maps, as explained in Section 6.1.

8 Construction of the main invariants

In this section, we construct the invariants which are the central objects of this paper. To the data of a TN contact pair (Y, ξ, V) and an element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$, we associate a unital, $\mathbb{Z}/2$ -graded $\mathbb{Q}[U]$ -algebra

(8-1)
$$CH_{\bullet}(Y,\xi,V;\mathfrak{r}).$$

There is a natural map to ordinary contact homology $CH_{\bullet}(Y, \xi, V; \mathfrak{r}) \to CH_{\bullet}(Y, \xi)$ given by setting U = 1.

A contactomorphism $f: (Y, \xi, V) \to (Y', \xi', V')$ induces an identification

$$CH_{\bullet}(Y,\xi,V;\mathfrak{r}) = CH_{\bullet}(Y',\xi',V';f_{*}\mathfrak{r}).$$

An exact relative symplectic cobordism $(\hat{X}, \hat{\lambda}, H)$ from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) satisfying an energy condition induces a map $CH_{\bullet}(Y^+, \xi^+, V^+; \mathfrak{r}^+) \rightarrow CH_{\bullet}(Y^-, \xi^-, V^-; \mathfrak{r}^-)$. Unfortunately, our notions of energy are not well behaved under compositions of arbitrary relative symplectic cobordisms, so the composition of maps is not always defined.

We also define a reduced version of (8-1), which only counts Reeb orbits in the complement of a codimension 2 submanifold, and certain "asymptotic invariants" which have good functoriality properties.

8.1 Construction and basic properties of the invariants

The following subsection is entirely parallel to [61, Section 1.7]. More precisely, Pardon constructs (ordinary) contact homology by applying [61, Theorem 1.1.] to data from Setups I–IV. We construct our new invariants by applying Theorem 7.27 to data from Setups I*–IV*.

Setup I* Fix a TN contact pair (Y, ξ, V) and an element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. According to Proposition 3.9, we may choose a contact form $\xi = \ker \lambda$ which is adapted to \mathfrak{r} . Let $J : \xi \to \xi$ be a $d\lambda$ -compatible almost complex structure which preserves ξ_V . We therefore obtain a datum \mathcal{D} for Setup I*. Theorem 7.27 applied to \mathcal{D} furnishes a $\mathbb{Z}/2$ -graded, unital $\mathbb{Q}[U]$ -algebra

(8-2)
$$CH_{\bullet}(Y,\xi,V;\mathfrak{r})_{\lambda,J,\theta}$$

for any choice of perturbation datum $\theta \in \Theta_{I}(\mathcal{D})$; cf (6-7) and (6-8).

Setup II* Fix pairs $\mathcal{D}^{\pm} = ((Y^{\pm}, \xi^{\pm}, V^{\pm}), \mathfrak{r}^{\pm}, \lambda^{\pm}, J^{\pm})$ of data for Setup I*, where we write $\mathfrak{r}^{\pm} = (\alpha^{\pm}, \tau^{\pm}, r^{\pm})$. Let $(\hat{X}, \hat{\lambda}, H)$ be an exact relative symplectic cobordism with positive end (Y^+, λ^+, V^+) and negative end (Y^-, λ^-, V^-) , and suppose that there exists a trivialization of the normal bundle of H which restricts to τ^{\pm} on the positive/negative end.

Proposition 8.1 Suppose that $r^+ > e^{\mathcal{E}(H,\hat{\lambda}|_H)}r^-$. Then there is an induced map on homology

(8-3)
$$\Phi(\hat{X},\hat{\lambda},H)_{\hat{J},\theta}: CH_{\bullet}(Y^+,\xi^+,V^+;\mathfrak{r}^+)_{\lambda^+,J^+,\theta^+} \to CH_{\bullet}(Y^-,\xi^-,V^-;\mathfrak{r}^-)_{\lambda^-,J^-,\theta^-}.$$

If $\alpha^+ = \alpha^-$ and $(H, \hat{\lambda}|_H)$ is a symplectization, then the same conclusion holds provided that $r^+ \ge r^-$.

Proof According to Lemma 6.25, we can choose an almost complex structure \hat{J} on \hat{X} which is $d\hat{\lambda}$ compatible and agrees with \hat{J}^{\pm} at infinity, and is such that $r^{+} \geq e^{\mathcal{E}(\hat{J})}r^{-}$. We thus obtain a datum $\mathcal{D} = (\mathcal{D}^{+}, \mathcal{D}^{-}, \hat{X}, H, \hat{\lambda}, \hat{J})$ for Setup II*.

Given $(\theta^+, \theta^-) \in \Theta_{\mathrm{I}}(\mathcal{D}^+) \times \Theta_{\mathrm{I}}(\mathcal{D}^-)$, Theorem 7.27 thus provides a perturbation datum $\theta \in \Theta_{\mathrm{II}}(\mathcal{D})$ with $\theta \mapsto (\theta^+, \theta^-)$, and twisted moduli counts which give rise to the map (8-3); cf (6-12).

Setup III* We have the following proposition.

Proposition 8.2 Under the assumptions of Proposition 8.1, the map (8-3) is independent of the pair (\hat{J}, θ) .

Proof Let (\hat{J}_0, θ_0) and (\hat{J}_1, θ_1) be two possible choices of such pairs. Let us first treat the case where $(H, \hat{\lambda}|_H)$ is not a symplectization. For any $\epsilon > 0$, Lemma 6.25 provides an interpolating family of almost complex structures $\{\hat{J}_t\}_{t \in [0,1]}$ such that $\mathcal{E}(\hat{J}_t) \leq \max(\mathcal{E}(\hat{J}_0), \mathcal{E}(\hat{J}_1)) + \epsilon$. Choosing ϵ small enough so that $r^+ > e^{\mathcal{E}(\hat{J}_t)}r^-$, we thus get a datum \mathcal{D} for Setup III*.

Theorem 7.27 now provides perturbation data $\theta \in \Theta_{\text{III}}(\mathcal{D})$ mapping to (θ_0, θ_1) , and a chain homotopy between the maps $\Phi(\hat{X}, \hat{\lambda}, H)_{\hat{J}_0, \theta_0}$ and $\Phi(\hat{X}, \hat{\lambda}, H)_{\hat{J}_1, \theta_1}$; cf (6-18) and (6-20).

If $\alpha^+ = \alpha^-$ and $(H, \hat{\lambda}|_H)$ is a symplectization, then Lemma 6.25 implies that we may repeat the above argument for a family of almost complex structures \hat{J}_t which have vanishing energy. Tracing through the proof, it is straightforward to check that the desired conclusion goes through provided that $r^+ \ge r^-$. \Box

Setup IV* Let us consider data $\widetilde{\mathcal{D}} = (\widetilde{\mathcal{D}}^{01}, \widetilde{\mathcal{D}}^{12}, (\widehat{X}^{02,t}, \widehat{\lambda}^{02,t})_{t \in [0,\infty)})$, where

$$\begin{split} \widetilde{\mathcal{D}}^{01} &= (\mathcal{D}^0, \mathcal{D}^1, \widehat{X}^{01}, \widehat{\lambda}^{01}, H^{01}), \\ \widetilde{\mathcal{D}}^{12} &= (\mathcal{D}^1, \mathcal{D}^2, \widehat{X}^{12}, \widehat{\lambda}^{12}, H^{12}), \\ \mathcal{D}^i &= ((Y^i, \xi^i, V^i), \mathfrak{r}^i, \lambda^i, J^i) \quad \text{for } i = 0, 1, 2 \end{split}$$

Here $\tilde{\mathcal{D}}^{01}$, $\tilde{\mathcal{D}}^{12}$ and $\tilde{\mathcal{D}}$ are "partial data" for Setups II* and IV*, since they do not contain any information about almost complex structures. These "partial data" are assumed to obey all the axioms stated in Section 7.1 which do not involve complex structures.

The \mathcal{D}^i are (ordinary) data for Setup I*.

Proposition 8.3 Suppose that the following conditions hold:

- $r_0 > e^{\mathcal{E}(H^{02,t}, \hat{\lambda}^{02,t}|_{H^{02,t}})} r_2$,
- $r_0 > e^{\mathcal{E}(H^{01}, \hat{\lambda}^{01}|_{H^{01}})} r_1.$
- $r_1 > e^{\mathcal{E}(H^{12}, \hat{\lambda}^{12}|_{H^{12}})} r_2.$

Then the following diagram commutes:

$$CH_{\bullet}(Y^{1},\xi^{1},V^{1};\mathfrak{r}^{1})_{\lambda^{1},J^{1},\theta^{1}} \xrightarrow{\Phi(\hat{X}^{12},\hat{\lambda}^{12},H^{12})} \\ CH_{\bullet}(Y^{0},\xi^{0},V^{0};\mathfrak{r}^{0})_{\lambda^{0},J^{0},\theta^{0}} \xrightarrow{\Phi(\hat{X}^{02},\hat{\lambda}^{02},H^{02})} \xrightarrow{CH_{\bullet}(Y^{2},\xi^{2},V^{2};\mathfrak{r}^{2})_{\lambda^{2},J^{2},\theta^{2}}}$$

If $\alpha_0 = \alpha_1 = \alpha_2$ and if $(H^{01}, \hat{\lambda}^{01}|_{H^{01}}), (H^{12}, \hat{\lambda}^{12}|_{H^{12}})$ and $(H^{02,t}, \hat{\lambda}^{02,t}|_{H^{02,t}})$ are symplectizations, then the conclusion still holds if we only assume that $r_i \ge r_j$ for $i \ge j$.

Proof According to Lemma 6.28, one can choose a family of almost complex structures $\hat{J}^{02,t}$ so that $r_0 > e^{\mathcal{E}(\hat{J}^{02,t})}r_2$. Moreover, by Lemma 6.29, one may also assume that $r_0 > e^{\mathcal{E}(\hat{J}^{01})}r_1$ and $r_1 > e^{\mathcal{E}(\hat{J}^{12})}r_2$, where J^{01} and J^{12} are the almost complex structures induced at infinity by $J^{02,t}$. We therefore obtain a datum for Setup IV* by considering $\mathcal{D}^{01} = (\tilde{\mathcal{D}}^{01}, \hat{J}^{01}), \mathcal{D}^{12} = (\tilde{\mathcal{D}}^{12}, \hat{J}^{12})$ and $\mathcal{D} = (\tilde{\mathcal{D}}, \hat{J}^{02,t})$. Theorem 7.27 applied to \mathcal{D} now implies the commutativity of the above diagram; cf (6-27).

Under the additional hypotheses that $\alpha_0 = \alpha_1 = \alpha_2$ and that the relevant cobordisms are symplectizations, Lemmas 6.28 and 6.29 allow us to work with (families of) almost complex structures with vanishing energy. Retracing through the above argument, we find that the desired conclusion follows if $r_i \ge r_j$ for $i \ge j$.

Proposition 8.4 Let $(\hat{X}, \hat{\lambda}, H)$ be a relative symplectic cobordism from (Y^+, ξ^+, V^+) to (Y^-, ξ^-, V^-) . Let $(\hat{V}^{\pm}, \hat{\lambda}_{\hat{V}}^{\pm})$ be the Liouville structure induced on \hat{V}^{\pm} from the canonical Liouville structure of the symplectization $(\hat{Y}^{\pm}, \lambda_{Y^{\pm}})$. For $i \in \{1, 2\}$, consider elements $\mathfrak{r}_i^{\pm} \in \mathfrak{R}(Y^{\pm}, \xi^{\pm}, V^{\pm})$ and let λ_i^{\pm} be a contact form on Y^{\pm} which is adapted to \mathfrak{r}_i^{\pm} . Suppose finally that we have:

Homological invariants of codimension 2 contact submanifolds

(1) $r_i^+ > e^{\mathcal{E}(H,\hat{\lambda}|_H)}r_i^-.$ (2) $r_1^+ > e^{\mathcal{E}(\hat{V}^+,\hat{\lambda}_{\hat{V}}^+)}r_2^+ \text{ and } r_1^- > e^{\mathcal{E}(\hat{V}^-,\hat{\lambda}_{\hat{V}}^-)}r_2^-.$

Then the following diagram commutes:

As usual, if $\alpha_1^+ = \alpha_1^- = \alpha_2^+ = \alpha_2^-$ and $(H, \hat{\lambda}|_H)$ is a symplectization, then it is enough to assume that $r_i^+ \ge r_i^-, r_1^+ \ge r_2^+$ and $r_1^- \ge r_2^-$.

Proof Observe first that the conditions (1) and (2) along with Proposition 8.1 ensure that the maps appearing in the commutative diagram are well-defined. Let us now consider the strict exact symplectic cobordisms $(\hat{X}, \hat{\lambda}, H)_{\lambda_1^-}^{\lambda_1^+}$ and $(\hat{Y}^-, \lambda_2^-, \hat{V}^-)_{\lambda_2^-}^{\lambda_1^-}$. For $t \in [0, \infty)$ and $T_0 > 0$ large enough, we can consider their $(t + T_0)$ -gluing $(\hat{X}^t, \hat{\lambda}^t, H^t)$; cf Definition 2.23. According to Lemma 6.22, we have that (cf Notation 2.19)

(8-4)
$$\mathcal{E}((H^t, \hat{\lambda}^t|_{H^t})^{\alpha_1^+}_{\alpha_2^-}) = \mathcal{E}((H, \hat{\lambda}|_H)^{\alpha_1^+}_{\alpha_1^-}) + \mathcal{E}((\hat{V}^-, \hat{\lambda}_{\hat{V}}^-)^{\alpha_1^-}_{\alpha_2^-}).$$

It then follows from (1) and (2) that $r_1^+ > e^{\mathcal{E}((H^t, \hat{\lambda}^t|_{H^t})_{\alpha_2}^{\alpha_1^+})} r_2^-$.

We can now appeal to Proposition 8.3, which implies that the composition $\Phi(\hat{Y}, \hat{V}^-) \circ \Phi(\hat{X}, \hat{\lambda}, H)$ agrees with the map induced by $(\hat{X}^0, \hat{\lambda}^0, H^0) = (\hat{X}, \hat{\lambda}, H)_{\lambda_2^-}^{\lambda_1^+}$; see Example 2.11. The same argument shows that composition along the upper right-hand side of the diagram agrees with the map induced by $(\hat{X}, \hat{\lambda}, H)_{\lambda_2^-}^{\lambda_1^+}$. This proves the claim.

We obtain the following corollary by putting together the results of the previous section.

Corollary 8.5 Consider a TN contact pair (Y, ξ, V) and fix an element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let $\mathcal{D}^{\pm} = (Y, \xi, V), \mathfrak{r}, \lambda^{\pm}, J^{\pm})$ be a pair of data for Setup I*, and fix $\theta^{\pm} \in \Theta_{\mathrm{I}}(Y^{\pm}, \lambda^{\pm}, J^{\pm})$.

The map

(8-5)
$$\Phi(\hat{Y}, \hat{\lambda}, \hat{V}) \colon CH_{\bullet}(Y, \xi, V; \mathfrak{r})_{\lambda^{+}, J^{+}, \theta^{+}} \to CH_{\bullet}(Y, \xi, V; \mathfrak{r})_{\lambda^{-}, J^{-}, \theta^{-}}$$

defined in Proposition 8.1 is an isomorphism.

Proof In light of Proposition 8.4 and Lemma 6.9, it's enough to consider the case $\lambda^+ = \lambda^- = \lambda$ and $J^+ = J^- = J$. Let $\theta \in \Theta_{\text{II}}(\hat{Y}, \hat{\lambda}, \hat{J})$ be a lift of (θ^+, θ^-) under the forgetful map $\Theta_{\text{II}}(\hat{Y}, \hat{\lambda}, \hat{J}) \rightarrow 0$

 $\Theta_{I}(Y, \lambda, J) \times \Theta_{I}(Y, \lambda, J)$. The proof of [61, Lemma 1.2] can be adapted to show that the map

(8-6)
$$\Phi(\hat{Y},\hat{\lambda},\hat{V})_{\hat{J},\theta}: CC_{\bullet}(Y,\xi,V;\mathfrak{r})_{\lambda,J,\theta^{+}} \to CC_{\bullet}(Y,\xi,V;\mathfrak{r})_{\lambda,J,\theta^{-}}$$

is an isomorphism of chain complexes: one simply needs to observe that the twisted counts of trivial cylinders coincide with the usual counts. \Box

We now arrive at the definition of our main invariants.

Definition 8.6 (full invariant) Consider a TN contact pair (Y, ξ, V) and choose an element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let

be the limit (or equivalently the colimit) of $\{CH_{\bullet}(Y, \xi, V; \mathfrak{r})_{\lambda,J,\theta}\}_{\lambda,J,\theta}$ along the maps (8-5). Proposition 8.4 and Corollary 8.5 imply that $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$ is canonically isomorphic to $CH_{\bullet}(Y, \xi, V; \mathfrak{r})_{\lambda,J,\theta}$ for any admissible choice of (λ, J, θ) .

Given $s \in \mathbb{Q}$, define

(8-8)
$$CH^{U=s}_{\bullet}(Y,\xi,V;\mathfrak{r}) := CH_{\bullet}(Y,\xi,V;\mathfrak{r}) \otimes_{\mathbb{Q}[U]} \mathbb{Q},$$

where the map $\mathbb{Q}[U] \to \mathbb{Q}$ sends $U \mapsto s$. There is a natural evaluation morphism of $\mathbb{Q}[U]$ -algebras

(8-9)
$$\operatorname{ev}_{U=s}: CH_{\bullet}(Y,\xi,V;\mathfrak{r}) \to CH_{\bullet}^{U=s}(Y,\xi,V;\mathfrak{r}).$$

It follows tautologically from the construction that $CH^{U=1}_{\bullet}(Y,\xi,V;\mathfrak{r}) = CH_{\bullet}(Y,\xi)$. The invariant $CH_{\bullet}(Y,\xi,V;\mathfrak{r})$ therefore admits a $\mathbb{Q}[U]$ algebra morphism to ordinary contact homology (which is viewed as a $\mathbb{Q}[U]$ -algebra by letting U act by the identity).

We can also define a "reduced" variant of the invariants (8-7), which are based on the twisting map $\tilde{\psi}$. These invariants are naturally \mathbb{Q} -algebras (as opposed to $\mathbb{Q}[U]$ -algebras) and only take into account Reeb orbits in the complement of the codimension 2 submanifold.

More precisely, given a datum $((Y, \xi, V), \mathfrak{r}, \lambda, J)$ for Setup I*, we may proceed as in Setup I* in Section 8.1 and let

(8-10)
$$(\widetilde{CC}_{\bullet}(Y,\xi,V;\mathfrak{r})_{\lambda}, d_{\widetilde{\Psi}_{\bullet},I,\theta})$$

be the complex generated by the (good) Reeb orbits *not contained in* $V \subset Y$, for some perturbation datum $\theta \in \Theta_{I}(\mathcal{D})$. By repeating the above arguments with the twisting maps $\tilde{\psi}_{-}$ in place of the twisting maps ψ_{-} , one can establish the obvious analogs of Proposition 8.4 and Corollary 8.5. In particular, given choices of data $(\lambda^+, J^+, \theta^+)$ and $(\lambda^-, J^-, \theta^-)$ as in Corollary 8.5, there is an isomorphism

(8-11)
$$\Phi(\widehat{Y},\widehat{\lambda},\widehat{V}):\widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r})_{\lambda^{+},J^{+},\theta^{+}}\to\widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r})_{\lambda^{+},J^{+},\theta^{+}}.$$
Definition 8.7 (reduced invariant) Consider a TN contact pair (Y, ξ, V) and fix an element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let

(8-12)
$$\widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r})$$

be the limit (or equivalently the colimit) of the algebras $\{\widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r})_{\lambda,J,\theta}\}_{\lambda,J,\theta}$ along the maps (8-11).

For future reference, we record the following corollary of the above discussion.

Corollary 8.8 Let $(Y^{\pm}, \xi^{\pm}, V^{\pm})$ be TN contact pairs, and choose elements $\mathfrak{r}^{\pm} = (\alpha^{\pm}, \tau^{\pm}, r^{\pm}) \in \mathfrak{R}(Y^{\pm}, \xi^{\pm}, V^{\pm})$. Let $(\hat{X}, \hat{\lambda}, H)$ be an exact relative symplectic cobordism with positive end (Y^+, ξ^+, V^+) and negative end (Y^-, ξ^-, V^-) , and suppose that τ^+ and τ^- extend to a global trivialization of the normal bundle of H. If $r^+ \ge e^{\mathcal{E}((H, \hat{\lambda}|_H)_{\alpha}^{\alpha^+})}r^-$, then there is an induced map

(8-13)
$$\Phi(\hat{X}, \hat{\lambda}, H) \colon CH_{\bullet}(Y^+, \xi^+, V^+; \mathfrak{r}^+) \to CH_{\bullet}(Y^-, \xi^-, V^-; \mathfrak{r}^-).$$

Similarly, suppose that $(\hat{X}, \hat{\lambda}^t, H^t)_{t \in [0,1]}$ is a family of exact relative symplectic cobordisms with ends $(V^{\pm}, \xi^{\pm}, V^{\pm})$ and such that τ^{\pm} extends to a global trivialization of the normal bundle of H^t . If $r^+ \ge e^{\mathcal{E}(H^t, (\hat{\lambda}^t)|_{H^t})}r^-$, then

(8-14)
$$\Phi(\hat{X}, \hat{\lambda}^0, H^0) = \Phi(\hat{X}, \hat{\lambda}^1, H^1)$$

The analogous statement holds for the reduced invariants $\widetilde{CH}_{\bullet}(-)$.

8.2 (Bi)gradings

The $\mathbb{Z}/2$ -grading by parity on the deformed invariants $CH_{\bullet}(-)$, $\widetilde{CH}_{\bullet}(-)$ shall be referred to as the *homological grading*. As in the case of (ordinary) contact homology, the homological grading can be lifted to a \mathbb{Z} -grading under certain topological assumptions. We will also to refer to this \mathbb{Z} -grading as the homological grading when it exists.

Definition 8.9 [61, Section 1.8] Let (Y^{2n-1}, ξ, V) be a TN contact pair and choose $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Suppose that $H_1(Y; \mathbb{Z}) = 0$ and $c_1(\xi) = 0$. Then the homological $\mathbb{Z}/2$ -grading lifts to a canonical \mathbb{Z} -grading defined on generators by

$$|\gamma| = CZ^{\tau}(\gamma) + n - 3,$$

where τ is any trivialization of the contact distribution along γ — this is independent of τ due to our assumption that $c_1(\xi) = 0$.

Remark 8.10 In Definition 8.9, our assumption that $c_1(\xi) = 0$ is equivalent to the statement that the canonical bundle $\Lambda_{\mathbb{C}}^{n-1}\xi$ is trivial. The grading in general depends on a trivialization of the canonical bundle; however, our assumption that $H_1(Y; \mathbb{Z}) = 0$ along with the universal coefficients theorem implies that $H^1(Y; \mathbb{Z}) = 0$. Hence the canonical bundle admits a unique trivialization.

Lemma 8.11 (see [61, equation (2.50)]) With the notation of Corollary 8.8, suppose that $H_1(Y^{\pm}; \mathbb{Z}) = 0$ and that $c_1(\xi^{\pm}) = c_1(TX) = 0$. Then the cobordism maps described in Corollary 8.8 preserve the homological \mathbb{Z} -grading.

Under certain topological assumptions, the reduced invariant $\widetilde{CH}_{\bullet}(-; -)$ admits an additional \mathbb{Z} -grading, which we will refer to as the *linking number grading*.

Definition 8.12 Let (Y, ξ, V) be a TN contact pair and choose $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Suppose that $H_1(Y; \mathbb{Z}) = H_2(Y; \mathbb{Z}) = 0$. Then the *linking number grading* $|\cdot|_{\text{link}}$ on $\widetilde{CH}_{\bullet}(Y, \xi, V; \mathfrak{r})$ is given on generators by

(8-16)
$$|\gamma|_{\text{link}} = \text{link}_V(\gamma).$$

(See Definition 5.19.) The linking number grading of a word of generators is then defined to be the sum of the linking number grading of each letter. One can verify using Lemma 5.21 that this grading is well-defined.

We let

(8-17)
$$\widetilde{CH}_{\bullet,\bullet}(Y,\xi,V;\mathfrak{r})$$

be the (super)commutative bigraded \mathbb{Q} -algebra, where

- the first bullet refers to the homological Z-grading (which exists in view of our topological assumption and the universal coefficients theorem, see Definition 8.9);
- the second bullet refers to the linking number \mathbb{Z} -grading.

We sometimes drop the second grading in our notation, so the reader should keep in mind that the notation $\widetilde{CH}_{\bullet}(-; -)$ always refers to the homological grading.

We have the following lemma as a consequence of Lemmas 5.21 and 8.11.

Lemma 8.13 With the notation of Corollary 8.8, suppose that $H_1(Y^{\pm}; \mathbb{Z}) = H_2(X, Y^+; \mathbb{Z}) = 0$. Then the cobordism maps described in Corollary 8.8 preserve the linking number \mathbb{Z} -grading. If we also have that $c_1(\xi^{\pm}) = c_1(TX) = 0$, then the cobordism maps preserve the $(\mathbb{Z} \times \mathbb{Z})$ -bigrading (8-17).

8.3 Asymptotic invariants

Given a TN contact pair (Y, ξ, V) and a trivialization τ of the normal bundle $N_{Y/V}$, let

(8-18)
$$\mathfrak{R}^{\tau}(Y,\xi,V) = \{\mathfrak{r} = (\alpha,\tau',r') \in \mathfrak{R}(Y,\xi,V) \mid \tau' = \tau\} \subset \mathfrak{R}(Y,\xi,V)$$

We equip $\mathfrak{R}^{\tau}(Y,\xi,V)$ with a preorder⁶ \leq defined by setting $(\alpha_V^-,\tau,r^-) \leq (\alpha_V^+,\tau,r^+)$ if $r^+ \geq e^{-\min f} r^-$, where $\alpha_V^+ = e^f \alpha_V^-$. We let \leq^{op} denote the opposite preorder.

 $^{^{6}}$ A preorder on a set is a binary relation which is reflexive and transitive. Equivalently, a preordered set is a category with at most one morphism from any object x to any other object y.

We now define a functor $\mathcal{F}(Y, \xi, V)$ from the preordered set $(\mathfrak{R}^{\tau}(Y, \xi, V), \leq^{\mathrm{op}})$ to the category of $\mathbb{Q}[U]$ -algebras. On objects, the functor takes \mathfrak{r} to $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$. It remains to define the functor on morphisms.

Given elements $\mathfrak{r}^{\pm} = (\alpha_V^{\pm}, \tau, r^{\pm}) \in \mathfrak{R}^{\mathfrak{r}}(Y, \xi, V)$, let λ^{\pm} be a contact form on *Y* which is adapted to \mathfrak{r}^{\pm} . Consider the symplectization $(\hat{Y}, \hat{\lambda}, \hat{V})_{\lambda^{-}}^{\lambda^{+}}$. If $\mathfrak{r}^{-} \leq \mathfrak{r}^{+}$, then Lemma 6.9 and Propositions 8.1 and 8.4 imply that there is a map

(8-19)
$$\Phi(\hat{Y},\hat{\lambda},\hat{V}): CH_{\bullet}(Y,\xi,V;\mathfrak{r}^{+}) \to CH_{\bullet}(Y,\xi,V;\mathfrak{r}^{-}).$$

This defines $\mathcal{F}(Y, \xi, V)$ on morphisms. One can check using Proposition 8.4 that $\mathcal{F}(Y, \xi, V)$ is indeed a functor.

We can similarly define a functor $\mathcal{F}_+(Y,\xi,V)$ from $(\mathfrak{R}^{\tau}(Y,\xi,V), \leq^{\mathrm{op}})$ to the category of \mathbb{Q} -algebras using $\widetilde{CH}_{\bullet}(-)$.

Definition 8.14 (asymptotic invariants) Noting that the category of $\mathbb{Q}[U]$ -algebras is complete and cocomplete, we denote by

(8-20)
$$CH_{\bullet}(Y,\xi,V;\tau)$$
 and $CH_{\bullet}(Y,\xi,V;\tau)$

the limit (resp. colimit) of the $\mathbb{Q}[U]$ -algebras { $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$ } over the preordered set $(\mathfrak{R}^{\tau}(Y, \xi, V), \preceq^{\mathrm{op}})$. We let $\widetilde{CH}_{\bullet}(Y, \xi, V; \tau)$ and $\widetilde{CH}_{\bullet}(Y, \xi, V; \tau)$ be defined similarly over the category of \mathbb{Q} -algebras.

It's easy to check that $(\mathfrak{R}^{\mathfrak{r}}(Y, \xi, V), \leq^{\mathrm{op}})$ is a filtered preordered set. In particular, (co)limits can be computed by restricting to (co)final subsets. In contrast to the invariants defined in Section 8.1, the asymptotic invariants are fully functorial under compositions of arbitrary relative symplectic cobordisms which respect normal trivializations. (A verification of this is tedious and essentially consists of repeating the arguments of Section 8.1—the key is that the energy conditions can always be satisfied by sending the rotation parameter to zero or infinity.)

8.4 Mixed morphisms

Consider a TN contact pair (Y, ξ, V) and elements $\mathfrak{r}^{\pm} = (\alpha^{\pm}, \tau^{\pm}, r^{\pm}) \in \mathfrak{R}(Y, \xi, V)$. In this section, we exhibit a \mathbb{Q} -algebra map

(8-21)
$$CH^{U=0}_{\bullet}(Y,\xi,V;\mathfrak{r}^+) \to \widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r}^-)$$

under certain assumptions on r^+ and r^- . Precomposing with (8-9) gives a \mathbb{Q} -algebra map

(8-22)
$$CH_{\bullet}(Y,\xi,V;\mathfrak{r}^+) \to \widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r}^-).$$

Let us begin by considering a datum $\mathcal{D} = (\mathcal{D}^+, \mathcal{D}^-, \hat{X}, H, \hat{\lambda}, \hat{J})$ for Setup II*, where we let $\mathcal{D}^{\pm} = ((Y, \xi, V), \mathfrak{r}^{\pm}, \lambda^{\pm}, J^{\pm}).$

Definition 8.15 We define a map $\psi_{mix} \colon \mathcal{S}_{II}^{\neq \varnothing}(\mathcal{D}) \to \mathbb{Q}$ by

(8-23)
$$\psi_{\min}(T) = \begin{cases} 1 & \text{if } T * \hat{V} = 0 \text{ and } |\gamma_e| \cap V^- = \emptyset \text{ for every } e \in E(T) \\ 0 & \text{otherwise.} \end{cases}$$

Proposition 8.16 The map $\psi_{mix}(-)$ is a twisting map provided that the following properties hold:

(i) r⁺ ≥ 2e^{ε(Ĵ)}r⁻, and
(ii) r⁺ > 2/R_α^{min}, where R_α^{min} denotes the smallest action of all Reeb orbits of α.

We need the following lemma for proving Proposition 8.16.

Lemma 8.17 Suppose that $\beta \in \pi_2(\hat{Y}, \gamma^+ \sqcup (\bigcup_{i=1}^n \gamma_i^-))$ is represented by a \hat{J} -holomorphic curve $u: \dot{\Sigma} \to \hat{X}$ contained in \hat{V} . Suppose also that r^+ and r^- satisfy assumptions (i) and (ii) in Definition 8.15. Then $\beta * \hat{V} \ge 2 - p_u$, where p_u is the total number of punctures (positive and negative) of u contained in V^{\pm} .

Proof By Lemma 7.10, we have $r^+P^+ - r^- \sum_{i=1}^n P_i^- \ge r^+P^+ - r^-P^+e^{\mathcal{E}(\hat{J})} \ge P^+(r^+ - r^-e^{\mathcal{E}(\hat{J})}) \ge R_{\alpha}^{\min}((r^+/2 - r^-e^{\mathcal{E}(\hat{J})}) + r^+/2) \ge R_{\alpha}^{\min}r^+/2 \ge 1$. The lemma now follows from Proposition 5.7.

Definition 8.18 Given a tree $T \in S_{\Pi}^{\neq \emptyset}$, we say that a vertex $v \in V(T)$ is *mean* if all adjacent edges are contained in V^{\pm} . Otherwise, we say that $v \in V(T)$ is *nice*. We denote by $V_b(T)$ (resp. $V_n(T)$) the set of mean (resp. nice) vertices of T.

Proof of Proposition 8.16 Choose a tree $T' \in S_{II}^{\neq \emptyset}$. Let $T' \to T$ be a morphism. It follows from Proposition 5.11 that $T' * \hat{V} = T * \hat{V}$. If T' has no edges contained in V^- , then neither does T' and we see that $\psi_{mix}(T') = \psi_{mix}(T)$.

Let us now suppose that T' has an edge contained in V^- . Note that T' and T have the same exterior edges. If one of these edges is contained in V^- , then $\psi_{\text{mix}}(T') = \psi_{\text{mix}}(T) = 0$. Let us therefore assume that the exterior edges of T' and T are not contained in V^- .

We are left with the case where T' has at least one interior edge contained in V^- . If T' had no mean vertices, then it would follow from Proposition 5.15 that there are no interior edges contained in V^{\pm} , which is a contradiction. It follows that T' has at least one mean vertex. Let $E_b^{int}(T') \subset E^{int}(T')$ be the set of interior edges which occur as an outgoing edge of some mean vertex. According to Proposition 5.15, we have

$$T' * \hat{V} = \sum_{v \in V_n(T')} \beta_v * \hat{V} + \sum_{v \in V_m(T')} \beta_v * \hat{V} + |E^{\text{int}}(T') - E^{\text{int}}_b(T')| + |E^{\text{int}}_b(T')|$$

$$\geq \sum_{v \in V_m(T')} \beta_v * \hat{V} + |E^{\text{int}}_b(T')|$$

$$= \sum_{v \in V_m(T')} (\beta_v * \hat{V} + p_v^-),$$

where p_v^- denotes the number of outgoing edges of v. (Here, we have used the fact that the outgoing edges of a mean vertex are all interior edges, which follows from our assumption that the exterior edges of T' and T are not contained in V^- .) It now follows from Lemma 8.17 and the fact that T' has at least one mean vertex that $\sum_{v \in V_m(T')} (\beta_v * \hat{V} + p_v) \ge \sum_{v \in V_m(T')} (2 - p_v + p_v^-) \ge \sum_{v \in V_m(T')} 1 \ge 1$. Hence $\psi_{\min}(T') = \psi_{\min}(T) = 0$. This completes the proof that $\psi_{\min}(T') = \psi_{\min}(T)$.

If $\{T_i\}_i$ is a concatenation, then the argument is the same as in the proof of Proposition 7.13 since every edge in $\#_i T_i$ appears in at least one of the T_i .

Proposition 8.19 Consider a TN contact pair (Y, ξ, V) and elements $\mathfrak{r}^{\pm} = (\alpha^{\pm}, \tau^{\pm}, r^{\pm}) \in \mathfrak{R}(Y, \xi, V)$. If $r' > 2e^{\mathcal{E}(\hat{V}, \hat{\lambda}|_{\hat{V}})}r^{-}$ and $r^{+} > 2/R_{\alpha}^{\min}$, then there is a map of \mathbb{Q} -algebras

(8-24)
$$CH^{U=0}_{\bullet}(Y,\xi,V;\mathfrak{r}^+) \to \widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r}^-).$$

Proof The argument is essentially the same as the proof of Proposition 8.1. Choose data of Type I* $\mathcal{D}^{\pm} = ((Y, \xi, V), \mathfrak{r}^{\pm}, \lambda^{\pm}, J^{\pm})$. Now consider the symplectization $(\hat{Y}, \hat{\lambda}, \hat{V})$. Lemma 6.25 furnishes an almost complex structure \hat{J} on \hat{Y} which is $d\hat{\lambda}$ -compatible and agrees with \hat{J}^{\pm} at infinity, and such that $r^+ \ge e^{\mathcal{E}(\hat{J})}r^-$. It now follows as in Setup II of Section 6.1 that we have a Q-algebra chain map

(8-25)
$$\Phi(\widehat{Y},\widehat{\lambda},\psi_{\mathrm{mix}})_{\widehat{J},\Theta}: CC^{U=0}_{\bullet}(Y,\xi,V;\mathfrak{r}')_{J^+,\theta^+} \to \widetilde{CC}_{\bullet}(Y,\xi,V;\mathfrak{r})_{J^-,\theta^-}$$

for perturbation data $\Theta \mapsto (\theta^+, \theta^-)$.

Remark 8.20 The proof of Proposition 8.19 does not show that (8-24) is independent of auxiliary choices (ie $\hat{J}, J^{\pm}, \Theta, \theta^{\pm}$). To show this, one needs to extend the definition of the twisting map ψ_{mix} to Setups III* and IV*. One can then prove analogs of Theorem 7.27 and Corollary 8.8 for ψ_{mix} . All of the ingredients for this are already in place, but we omit the details since Proposition 8.19 is sufficient for our applications.

Corollary 8.21 Suppose that $\widetilde{CH}_{\bullet}(Y, \xi, V; \mathfrak{r}) \neq 0$ for some $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V)$. Letting $\mathfrak{r}' = (\alpha'_V, \tau', r')$, we have $CH_{\bullet}(Y, \xi, V; \mathfrak{r}') \neq 0$ if r' is large enough. In particular, $CH_{\bullet}(Y, \xi, V; \tau) \neq 0$. \Box

9 Augmentations and linearized invariants

9.1 Differential graded algebras

Let *R* be a commutative ring containing \mathbb{Q} . The following two categories occur naturally in contact topology: the category **cdga** of unital \mathbb{Z} -graded (super)commutative dg *R*-algebras; the category **dga** of unital \mathbb{Z} -graded associative dg *R*-algebras. When we speak of a *dg algebra*, we mean an object of either one of these categories.

Let us say that a dg algebra is *action-filtered* if the underlying graded algebra is the free algebra on a free graded module U having the following property: U admits a basis $\{x_{\alpha} \mid \alpha \in A\}$ for some well-ordered set A such that dx_{α} is a sum of words in the letters x_{β} for $\beta < \alpha$. Note that the dg algebras which arise in Symplectic Field Theory are naturally action-filtered by Reeb length.

The categories **cdga** and **dga** each carry a model structure described by Hinich [34] (see also Section 1.11 of [66]) with the following properties:

- (i) the weak equivalences are the quasi-isomorphisms, and
- (ii) all objects are fibrant and the set of cofibrant objects includes all objects which are action-filtered.⁷

Note that in any model category, there is a notion of two maps being left (resp. right) homotopic, defined in terms of cylinder (resp. path) objects [35, Section 7.3.1]. These notions coincide on objects which are fibrant and cofibrant.

We let **hcdga** and **hdga** be the associated homotopy categories: the objects are those objects of **cdga** (resp. **dga**) which are fibrant and cofibrant, and the morphisms are the homotopy classes of morphisms in **cdga** (resp. **dga**). *In particular*, **hcdga** *and* **hdga** *contain all action-filtered dg algebras*.

An *augmentation* of a dg *R*-algebra *A* is a morphism of dg algebras $\epsilon: A \to R$, where *R* is viewed as a dg algebra concentrated in degree 0. A dg algebra equipped with an augmentation is said to be augmented. We let $\mathbf{cdga}_{/R}$ and $\mathbf{dga}_{/R}$ be the overcategory of augmented objects, which naturally inherit model structures. We let $\mathbf{hcdga}_{/R}$ (resp. $\mathbf{hdga}_{/R}$) be the associated homotopy categories.

Definition 9.1 Given an augmented dg algebra $\epsilon : (A, d) \to R$, we consider the graded *R*-module $A_{\epsilon} := \ker \epsilon / (\ker \epsilon)^2$. The differential *d* descends to a differential d_{ϵ} on A_{ϵ} . The resulting differential graded module $(A_{\epsilon}, d_{\epsilon})$ is called the *linearization*⁸ of (A, d) at the augmentation ϵ .

It follows from the definition that linearization defines a functor from $\mathbf{cdga}_{/R}$ (resp. $\mathbf{dga}_{/R}$) to the category of chain complexes of *R*-modules. It is an important fact that this functor is left Quillen [51, Proposition 12.1.7], and therefore induces a functor of homotopy categories. We state this as a corollary:

Corollary 9.2 Linearization defines a functor from $hcdga_{/R}$ (resp. $hdga_{/R}$) to the homotopy category of chain complexes of *R*-modules.

⁷Both [34] and [51] work in the setting of dg algebras over operads. The categories **cdga** and **dga** are, respectively, the category of dg algebras over the operads uComm and uAss (*u* stands for *unital*), so they can be treated on equal footing. A good reference for the material in this section, which mostly avoids operadic language (but only treats the case of **cdga**), is [47].

⁸The linearization is sometimes also called the "indecomposable quotient" in the rational homotopy theory literature. In the contact topology literature, one also often encounters an equivalent construction of the linearization in which one twists the differential by the augmentation; see [57, Remark 2.8].

In particular, Corollary 9.2 implies that homotopic augmentations of action-filtered dg algebras induce isomorphic linearizations. Special cases of this statement have already appeared in the Legendrian contact homology literature; see eg [58, Section 5.3.2].

Remark 9.3 Let (A, d) be a dg algebra, where A is the free R-algebra generated by the set $\{x_{\alpha} \mid \alpha \in A\}$. Let $\epsilon : A \to R$ be the unique R-algebra map sending the generators x_{α} to zero. Then ϵ is an augmentation if and only if dx_{α} is contained in the ideal $(x_{\beta} \mid \beta \in A)$ for all $\alpha \in A$ (or equivalently, if and only if the differential has no constant term). If ϵ is an augmentation, it is called the *zero augmentation*.

Suppose now that (A, d) is the (possibly deformed) contact algebra of some contact manifold, ie (A, d) is the commutative *R*-algebra generated by good Reeb orbits (for $R = \mathbb{Q}$ or $\mathbb{Q}[U]$) and the differential is defined as in Section 6.1. Suppose that $\epsilon: A \to R$ is the zero augmentation. Then ker ϵ is the free *R*-module generated by (good) Reeb orbits, and d_{ϵ} counts curves with one input and one output (ie d_{ϵ} is defined as in (6-7), where the sum is restricted to curves with $|\Gamma^-| = 1$). It follows that the homology of the complex $(A_{\epsilon}, d_{\epsilon})$ can be interpreted as the (possibly deformed) *cylindrical* contact homology. This latter invariant admits a rigorous definition for contact structures under certain assumptions, such as the existence of a contact form with no contractible Reeb orbits; see [4; 42] in dimension 3 and [61, Section 1.8] in general.

9.2 Cyclic homology

Definition 9.4 (cf [17]) Let *S* be a countable, well-ordered set equipped with a map $|\cdot|: S \to \mathbb{Z}$. Let $A = R\langle S \rangle$ be the free \mathbb{Z} -graded *R*-algebra generated by *S*, where the \mathbb{Z} -grading is induced by extending $|\cdot|$ multiplicatively.

Let $d: A \to A$ be a differential of degree -1. Let $\overline{A} := A/R$, and consider the cyclic permutation map $\tau: \overline{A} \to \overline{A}$ which is defined on monomials by $\tau(\gamma_1 \cdots \gamma_l) = (-1)^{|\gamma_1|(|\gamma_2|+\cdots+|\gamma_l|)}\gamma_2 \cdots \gamma_l \gamma_1$ and extended *R*-linearly. Let $\overline{A}^{\tau} := \overline{A}/(1-\tau)$ be the \mathbb{Z} -graded *R*-module of coinvariants. Observe that *d* passes to the quotient. We denote the induced differential by d^{τ} .

We now define

(9-1)
$$\overline{HC}_{\bullet}(A) := H_{\bullet}(\overline{A}^{\tau}, d^{\tau}),$$

and refer to this invariant as the *reduced cyclic homology* of the dg algebra (A, d).

Remark 9.5 Definition 9.4 agrees with other definitions of reduced cyclic homology of dg algebras (such as [50, Section 5.3]) which may be more familiar to the reader, when both are defined. We adopt the present definition for consistency with [7].

In the special case where *A* is the Chekanov–Eliashberg dg algebra of a Legendrian knot in a contact manifold satisfying the assumptions of [7, Section 4.1], the algebraic invariants considered in [7, Section 4] can be translated as follows: $L\mathbb{H}^{cyc}(A) = \overline{HC}_{\bullet}(A)$, $L\mathbb{H}^{Ho+}(A) = \overline{HH}_{\bullet}(A)$ and $L\mathbb{H}^{Ho}(A) = HH_{\bullet}(A)$. Here $HH_{\bullet}(-)$ and $\overline{HH}_{\bullet}(-)$ denote, respectively, Hochschild homology and reduced Hochschild homology.

We record the following computation which will be useful to us later on.

Lemma 9.6 Under the assumptions of Definition 9.4, if (A, d) is acyclic, then

(9-2)
$$\overline{HC}_k(A) = \begin{cases} R & \text{if } k \text{ is odd and positive} \\ 0 & \text{otherwise.} \end{cases}$$

Proof Let us first prove the lemma under the assumption that S is a finite set. Note first that the Hochschild homology of an acyclic finitely generated dg algebra vanishes identically. Moreover, we have an exact triangle

(9-3)
$$R[0] \to HH_{\bullet}(A) \to \overline{HH}_{\bullet}(A) \xrightarrow{[-1]},$$

which implies that $\overline{HH}_{\bullet}(A)$ is just a copy of *R* concentrated in degree 1.

We now consider the following Gysin-type exact triangle (see [7, Proposition 4.9]):

(9-4)
$$\overline{HC}_{\bullet}(A) \xrightarrow{[-2]} \overline{HC}_{\bullet}(A) \xrightarrow{[+1]} \overline{HH}_{\bullet}(A) \xrightarrow{[0]} .$$

The desired result now follows immediately by induction, using (9-4) and the fact that $\overline{HC}_{\bullet}(A)$ vanishes in sufficiently large positive and negative degrees due to our finiteness hypotheses. We remark that (9-4) is constructed from a spectral sequence, whose convergence can only be verified under finiteness assumptions.

Let us now drop our assumption that S is a finite set. We instead consider an exhaustion of S by finite subsets $S^{(1)} \subset S^{(2)} \subset \cdots \subset S$. Let $(A^{(k)}, d) \subset (A, d)$ be the dg subalgebra generated by $S^{(k)}$. One can readily verify that

(9-5)
$$\lim_{k \to \infty} \overline{HC}_{\bullet}(A^{(k)}) = \overline{HC}_{\bullet}(\lim_{k \to \infty} A^{(k)}) = \overline{HC}_{\bullet}(A).$$

Observe that $(A^{(k)}, d)$ is acyclic for k large enough and satisfies the assumption of Definition 9.4. Since we have already proved the lemma under the assumption that S is finite, it is enough to prove that the natural maps $\overline{HC}_{\bullet}(A^{(k)}) \rightarrow \overline{HC}_{\bullet}(A^{(k+1)})$ are isomorphisms for k large enough.

To this end, note that the exact triangles (9-3) and (9-4) can be shown to be functorial under morphisms of bounded dg algebras. Since quasi-isomorphisms induce isomorphisms on Hochschild homology, it follows from (9-3) that the natural map $\overline{HH}_{\bullet}(A^{(k)}) \rightarrow \overline{HH}_{\bullet}(A^{(k+1)})$ is an isomorphism. Since $\overline{HH}_{\bullet}(A^{(k)}) = \overline{HH}_{\bullet}(A^{(k+1)})$ is concentrated in degree 1, and since $\overline{HC}_i(A^{(k)})$ and $\overline{HC}_i(A^{(k+1)})$ vanish for |i| sufficiently large, the desired claim can be checked by inductively applying the five lemma; cf [50, Section 2.2.3].

9.3 Augmentations from relative fillings

According to the philosophy of [24], contact homology is supposed to be well-defined as a differential graded \mathbb{Q} -algebra up to strict isomorphism. However, [61] only proves that contact homology is well-defined as a graded \mathbb{Q} -algebra, ie after passing to homology. Similarly, the invariants introduced in

Section 8 are merely graded *R*-algebras for $R = \mathbb{Q}$ or $\mathbb{Q}[U]$. For the purpose of linearizing the invariants of Section 8, we saw in Section 9.1 that the following intermediate assumption — weaker than what is conjectured in [24] but stronger than what is proved in [61] — is sufficient:

Assumption 9.7 The constructions of Section 8 can be lifted to the category **hcdga**. (That is, cobordisms induce a unique morphism of commutative dg algebras up to homotopy, and composition of cobordism is functorial up to homotopy.)

The rest of this section depends on the unproved Assumption 9.7; all statements which depend on this assumption are therefore starred, in accordance with the convention stated in the introduction.

Definition* 9.8 Fix a TN contact pair (Y, ξ, V) and an element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let

(9-6)
$$\mathcal{A}(Y,\xi,V;\mathfrak{r}) \in \mathbb{Q}[U]$$
-hcdga

be the limit (or equivalently the colimit) of the dg algebras $\{(CC_{\bullet}(Y, \xi, V; \mathfrak{r})_{\lambda}, d_{\psi,J,\theta})\}_{\lambda,J,\theta}$ under the lifts of the maps (8-5) which are furnished by Assumption 9.7. We also analogously define

(9-7)
$$\widetilde{\mathcal{A}}(Y,\xi,V;\mathfrak{r}) \in \mathbb{Q}$$
-hcdga.

Remark 9.9 (bigradings) With the notation of Definition* 9.8, suppose that $H_1(Y; \mathbb{Z}) = H_2(Y; \mathbb{Z}) = 0$. Combining Assumption 9.7 with the discussion of Section 8.2, it then follows that $\tilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r})$ is a $(\mathbb{Z} \times \mathbb{Z})$ -bigraded differential algebra, where the differential has bidegree (-1, 0).

Definition* 9.10 Given an augmentation $\epsilon : \mathcal{A}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}[U]$, we let $\mathcal{A}^{\epsilon}(Y, \xi, V; \mathfrak{r})$ be the linearized chain complex (in the sense of Definition 9.1) with respect to ϵ and let $CH^{\epsilon}_{\bullet}(Y, \xi, V; \mathfrak{r})$ be the resulting homology.

We have analogous invariants in the reduced case, which are denoted by $\widetilde{\mathcal{A}}^{\widetilde{\epsilon}}(Y, \xi, V; \mathfrak{r})$ and $\widetilde{CH}^{\widetilde{\epsilon}}(Y, \xi, V; \mathfrak{r})$ for an augmentation $\widetilde{\epsilon}: \widetilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}$.

Definition 9.11 Given a contact manifold (Y, ξ) and a codimension 2 contact submanifold *V*, a *relative filling* $(\hat{X}, \hat{\lambda}, H)$ is a relative symplectic cobordism from (Y, ξ, V) to the empty set.

Let $(\hat{X}, \hat{\lambda}, H)$ be a relative filling of (Y, ξ, V) and fix $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Suppose τ extends to a normal trivialization of H. Then Lemma 8.11 and Assumption 9.7 furnish an augmentation $\epsilon(\hat{X}, \hat{\lambda}, H) : \mathcal{A}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}[U]$. Similarly, we have an augmentation $\tilde{\epsilon}(\hat{X}, \hat{\lambda}, H) : \tilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}$.

If we suppose that $H_1(Y; \mathbb{Z}) = H_2(Y; \mathbb{Z}) = H_2(X, Y; \mathbb{Z}) = 0$ and $c_1(TX) = 0$, then Lemma 8.13 and Assumption 9.7 imply that $\tilde{\epsilon}(\hat{X}, \hat{\lambda}, H)$ preserves the $(\mathbb{Z} \times \mathbb{Z})$ -bigrading defined in Remark 9.9. It follows that the linearized complex

(9-8)
$$\widetilde{\mathcal{A}}^{\widetilde{\epsilon}}(Y,\xi,V;\mathfrak{r})$$

inherits a $(\mathbb{Z} \times \mathbb{Z})$ -bigrading with differential of bidegree (-1, 0). Hence

(9-9) $\widetilde{CH}^{\widetilde{\epsilon}}_{\bullet}(Y,\xi,V;\mathfrak{r})$

is a $(\mathbb{Z} \times \mathbb{Z})$ -bigraded \mathbb{Q} -vector space.

We end this section by collecting some lemmas which will be useful later. The reader is referred to Section 3.3 for a review of open book decompositions.

Lemma* 9.12 Suppose that $(\hat{W}, \hat{\lambda}, H)$ is a relative filling of (Y, ξ, V) . Suppose that *V* is the binding of an open book decomposition (Y, V, π) which supports ξ . Fix an element $\mathfrak{r} = (\alpha_V, \tau, r) \in \mathfrak{R}(Y, \xi, V)$, where τ is the canonical trivialization induced by the open book.

Suppose that *H* admits a normal trivialization which restricts to τ . Suppose also that $H_1(Y; \mathbb{Z}) = H_2(Y; \mathbb{Z}) = H_2(W, Y; \mathbb{Z}) = 0$ and that $c_1(TW) = 0$. Then the augmentation

(9-10) $\widetilde{\epsilon}: \widetilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}$

is the zero augmentation. In particular, it depends only on $(\hat{W}, \hat{\lambda})$ and not on H.

Proof It is shown in Corollary 3.15 that there exists a nondegenerate contact form α for (Y, ξ) which is adapted to \mathfrak{r} and has the property that all Reeb orbits are transverse to the pages of the open book decomposition. Given auxiliary choices of almost complex structures and perturbation data, the augmentation $\tilde{\epsilon}$ counts (possibly broken) holomorphic planes u which are asymptotic to a Reeb orbit γ disjoint from H, and such that [u] * H = 0. However, our topological assumptions and Lemma 5.21 implies that [u] * H is precisely the linking number of γ with the binding V, which is strictly positive by assumption; see Remark 5.20.

Lemma* 9.13 Let $(\hat{X}, \hat{\lambda}, H)$ and $(\hat{X}', \hat{\lambda}', H')$ be relative symplectic fillings of (Y, ξ, V) and (Y', ξ', V') . Let $f: (Y, \xi, V) \to (Y', \xi', V')$ be a contactomorphism. Suppose that there exists a symplectomorphism $\phi: (\hat{X}, \hat{\lambda}) \to (\hat{X}', \hat{\lambda}')$ which coincides near infinity with the induced map $\tilde{f}: SY \to SY'$.

Given any $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$, the following diagram commutes:

$$(9-11) \qquad \begin{array}{c} CH_{\bullet}(Y,\xi,V;\mathfrak{r}) & \xrightarrow{\Phi(\hat{X},\hat{\lambda},H)} & \mathbb{Q}[U] \\ \downarrow = & \downarrow \\ CH_{\bullet}(Y',\xi',V';f_{*}\mathfrak{r}) & \xrightarrow{\Phi(\hat{X}',\phi_{*}\hat{\lambda},\phi(H))} & \mathbb{Q}[U] \\ \downarrow = & \downarrow \\ CH_{\bullet}(Y',\xi',V';f_{*}\mathfrak{r}) & \xrightarrow{\Phi(\hat{X}',\hat{\lambda}',\phi(H))} & \mathbb{Q}[U] \end{array}$$

The analogous statement holds for $\widetilde{CH}_{\bullet}(-)$, with \mathbb{Q} in place of $\mathbb{Q}[U]$. In addition, if $H_1(Y;\mathbb{Z}) = H_2(Y;\mathbb{Z}) = 0$ and $c_1(TX) = 0$, then all arrows can be assumed to preserve the bigrading in Definition 8.12.

Proof The commutativity of the top square is essentially tautological; more precisely, it follows from the functoriality of the moduli counts in Theorem 7.27. The commutativity of the bottom square follows from the observation that ϕ preserves the Liouville form outside a compact set. Hence $(\hat{X}', \phi_* \hat{\lambda})$ and $(\hat{X}', \hat{\lambda}')$ are deformation equivalent. It follows by Corollary 8.8 that they induce the same morphism on homology. The fact that the maps preserve the bigradings on $\widetilde{CH}(-; -)$ (under the above topological assumptions) is a consequence on Lemma 8.13.

Corollary* 9.14 Let $(\hat{X}, \hat{\lambda}, H)$ (resp. $(\hat{X}, \hat{\lambda}, H')$) be relative fillings for (Y, ξ, V) (resp. (Y, ξ, V')). Suppose that *V* is the binding of an open book decomposition of (Y, ξ) and fix $\mathfrak{r} = (\alpha, \tau, r) \in \mathfrak{R}(Y, \xi, V)$, where τ is induced by the open book. Let $f : (Y, \xi, V) \to (Y, \xi, V')$ be a contactomorphism.

Suppose that $H_1(Y; \mathbb{Z}) = H_2(Y; \mathbb{Z}) = H_2(X, Y; \mathbb{Z}) = 0$ and that $c_1(TX) = 0$. Then

(9-12)
$$\widetilde{CH}^{\widetilde{\epsilon}}_{\bullet,\bullet}(Y,\xi,V;\mathfrak{r}) = \widetilde{CH}^{\widetilde{\epsilon}'}_{\bullet,\bullet}(Y,\xi,V';f_*\mathfrak{r}),$$

where both augmentations are induced by the relative fillings and the $(\mathbb{Z} \times \mathbb{Z})$ -bigrading is defined in Definition 8.12.

Proof Indeed, since the lift of a contactomorphism to the symplectization is a Hamiltonian symplectomorphism, it is easy to construct a symplectic automorphism of $(X, d\hat{\lambda})$ satisfying the conditions of Lemma* 9.13; see eg [14, Section 3.2]. The claim now follows from Lemma* 9.12.

10 Invariants of Legendrian submanifolds

10.1 Invariants of contact pushoffs

Definition 10.1 (see Definition 3.1 in [10]) Let (Y, ξ) be a contact manifold and let $\Lambda \hookrightarrow Y$ be a Legendrian embedding. By the Weinstein neighborhood theorem, the map extends to an embedding $Op(\Lambda) \subset (J^1\Lambda, \xi_{std}) \to (Y, \xi)$, where $Op(\Lambda) \subset (J^1\Lambda, \xi_{std})$ denotes an open neighborhood of the zero section.

Let $\tau(\Lambda)$ be the induced codimension 2 contact embedding

(10-1) $\tau(\Lambda): \partial(D^*_{\epsilon,g}\Lambda) = \partial(D^*_{\epsilon,g}\Lambda) \times 0 \subset T^*\Lambda \times \mathbb{R} = J^1\Lambda \hookrightarrow (Y,\xi).$

Here $D_{\epsilon,g}^* \Lambda$ is the sphere bundle of covectors of length ϵ with respect to some metric g, which is a contact manifold with respect to the restriction of the canonical 1-form on $T^*\Lambda$. We refer to $\tau(\Lambda)$ as the *contact pushoff* of $\Lambda \hookrightarrow Y$.

Standard arguments establish that the contact pushoff is canonical up to isotopy through codimension 2 contact embeddings. By abuse of notation, we will routinely identify $\tau(\Lambda)$ with its image. Observe that it follows that $CH_{\bullet}(Y, \xi, \tau(\Lambda); \mathfrak{r})$ and $\widetilde{CH}_{\bullet}(Y, \xi, \tau(\Lambda); \mathfrak{r})$ can be viewed as invariants of Λ .

10.2 Deformations of the Chekanov–Eliashberg dg algebra

In the spirit of the previous sections, we now consider deformations of the Chekanov–Eliashberg dg algebra of a Legendrian induced by a codimension 2 contact submanifold. We begin with some preliminary definitions.

Definition 10.2 Let $\Lambda \subset (Y, \xi)$ be a Legendrian submanifold. Given a contact form ker $\alpha = \xi$, consider a Reeb chord $c : [0, R] \to Y$. The linearized Reeb flow defines a path of symplectomorphisms $\mathcal{P}_r : \xi|_{c(0)} \to \xi|_{c(r)}$. We say that the Reeb chord c is nondegenerate if $\mathcal{P}_R(T_{c(0)}\Lambda) \cap T_{c(R)}\Lambda = \{0\}$.

Definition 10.3 (cf Section 2.1 in [7]) With the notation of Definition 10.2, let $\bigwedge_{\mathbb{C}}^{n-1}(\xi, d\alpha)$ be the canonical bundle of ξ and suppose that it admits a trivialization σ . Let $\Lambda_1, \ldots, \Lambda_k$ be an enumeration of the components of Λ . Suppose that each Λ_i has vanishing Maslov class.

Suppose first of all that k = 1 (ie Λ is connected). Given a nondegenerate Reeb chord c, pick a path c_{-} in Λ connecting c(R) to c(0). Observe that $\bigwedge^{n-1} T_{c_{-}} \Lambda_i \subset \bigwedge_{\mathbb{C}}^{n-1}(\xi, d\alpha)$ is a path of Lagrangian subspaces along c_{-} . We call this path $L_{c_{-}}$. The map \mathcal{P}_r also defines a path of Lagrangian subspaces $\bigwedge^{n-1} \mathcal{P}_r(T_{c(0)}\Lambda_i) \subset \bigwedge_{\mathbb{C}}^{n-1}(\xi, d\alpha)$ along c. We call this path L_c .

Let $\tilde{c} = c_- * c$ be obtained by concatenating c_- and c (the concatenation is from left to right). Now consider the path of Lagrangian subspaces $L_{\tilde{c}} = L_{c^-} * L_c * P^+$, where P^+ is a positive rotation from $\mathcal{P}_R(T_{c(0)}\Lambda)$ to $T_{c(R)}\Lambda$ (this is well-defined by our assumption that c is nondegenerate).

The Conley–Zehnder index for chords of c with respect to σ is denoted by $CZ^{+,\sigma}(c)$ and defined by

(10-2)
$$CZ^{+,\sigma}(c) = \mu^{\sigma}(L_{\tilde{c}}),$$

where $\mu^{\sigma}(-)$ is the Maslov index with respect to σ ; see [52, Theorem 2.3.7]. This definition is independent of the choice of c_{-} due to our assumption that Λ has vanishing Maslov class. Note also that the resulting index depends on σ , but its parity does not.

In the case that k > 1, the definition of the Conley–Zehnder index for chords is more complicated, and depends on additional choices. We refer the reader to [7, Section 2.1] — we warn the reader that there is a typo in the formula stated there: the correct formula for the Conley–Zehnder index for chords should read $CZ^{+,\sigma}(c) = |c| - 1 = (\phi_{-} - \phi_{\Lambda}(x_1))/\pi + (n-1)/2$.

Remark 10.4 It may happen that a Reeb orbit can also be viewed as a Reeb chord with same starting and end point. In this case, we have in general that $CZ^+(c) \neq CZ(c)$.

Let us now consider a TN contact pair (Y, ξ, V) and a Legendrian submanifold $\Lambda \subset (Y - V, \xi)$. We let $\Lambda_1, \ldots, \Lambda_k$ be an enumeration of the connected components of Λ . As in Definition 10.3, we assume that the Λ_i have vanishing Maslov class.

Definition/Assumption* 10.5 (cf Proposition 2.8.2 in [24]) Fix $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let us also choose the following additional data:

- A contact form λ ∈ Ω¹(Y) which is adapted to r and has the property that all Reeb orbits and Λ-Reeb chords are nondegenerate.
- A $d\lambda$ -compatible almost complex structure J on ξ which preserves TV.

Given a class $\beta \in \pi_2(\hat{Y}; c^+, \Gamma_{\Lambda}^-, \Gamma^-)$, we let

(10-3)
$$\mathcal{M}(c^+, \Gamma_{\Lambda}^-, \Gamma^-; \beta)_J$$

be the moduli space of connected \hat{J} -holomorphic curves, modulo \mathbb{R} -translation representing the class β . (Here we follow the notation of Section 2.2, where c^+ is a Reeb chord of $\Lambda \subset (Y, \lambda)$, $\Gamma_{\Lambda}^- = \{c_1^-, \ldots, c_{\sigma}^-\}$ is an ordered collections of (not necessarily distinct) Reeb chords, and Γ^- is a collection of Reeb orbits.) Since $V \subset (Y - \Lambda, \lambda)$ is a strong contact submanifold, a straightforward extension of Siefring's intersection theory defines an intersection number $\hat{V} * \beta \in \mathbb{Z}$.

Let us now consider the semisimple ring

(10-4)
$$\mathcal{R} = \bigoplus_{i=1}^{k} \mathbb{Q}[U]$$

and let e_1, \ldots, e_k be the idempotents corresponding to the unit in each summand.

Let $CL_{\bullet}(Y, \xi, V, \Lambda; \mathfrak{r})_{\lambda}$ be the free \mathcal{R} algebra generated by (good) Reeb orbits of (Y, α) and Reeb chords of Λ , subject to the following relations:

- $\gamma_1 \gamma_2 = (-1)^{|\gamma_1||\gamma_2|} \gamma_2 \gamma_1$ for Reeb orbits *a* and *b*.
- If c_{ij} is a Reeb chord from Λ_i to Λ_j , then $e_k c_{ij} e_l = \delta_{jk} c_{ij} \delta_{il}$.

We assume that there exists a suitable virtual perturbation framework compatible with [61], so that we can define a differential d_J (squaring to zero) on generators as follows:

• For a Reeb chord c^+ , we let

(10-5)
$$d_J(c^+) = \sum \frac{1}{|\operatorname{Aut}|} \# \overline{\mathcal{M}}(c^+, \Gamma_{\Lambda}^-, \Gamma^-; \beta)_J U^{\widehat{V} * \beta} c_1^- \cdots c_{\sigma}^- \gamma_1 \cdots \gamma_s,$$

where the sum is over choices of $\beta \in \pi_2(\hat{Y}; c^+, \Gamma_{\Lambda}^-, \Gamma^-)$ for all possible choices of Γ_{Λ}^- and Γ^- .

• For a Reeb orbit γ , we let $d_J(\gamma)$ be the usual deformed contact homology differential, as described in Section 8.1.

We assume that $(CL_{\bullet}(Y, \xi, V, \Lambda; \mathfrak{r})_{\lambda}, d_J)$ is independent of λ and J up to canonical isomorphism in $\mathbb{Q}[U]$ -hdga. We denote the resulting object by

(10-6)
$$\mathcal{L}(Y,\xi,V,\Lambda;\mathfrak{r})$$

and we let $CH_{\bullet}(Y, \xi, V, \Lambda; \mathfrak{r})$ be its homology. We assume that $\mathcal{L}(Y, \xi, V, \Lambda; \mathfrak{r})$ satisfies the limited functoriality described in Proposition* 10.7.

Definition/Assumption* 10.6 (cf Proposition 2.8.2 in [24]) Carrying over the hypotheses and notation from Definition/Assumption* 10.5, let us consider the semisimple ring

(10-7)
$$\tilde{\mathcal{R}} = \bigoplus_{i=1}^{k} \mathbb{Q}.$$

where we again let e_1, \ldots, e_k be the idempotents corresponding to the unit in each summand.

We let $\widetilde{CL}_{\bullet}(Y, \xi, V, \Lambda; \mathfrak{r})_{\lambda}$ be the free $\widetilde{\mathcal{R}}$ algebra generated by (good) Reeb orbits of (Y, α) which are not contained in V and Λ Reeb chords, subject to the following relations:

- $\gamma_1 \gamma_2 = (-1)^{|\gamma_1||\gamma_2|} \gamma_2 \gamma_1$ for Reeb orbits *a* and *b*.
- If c_{ij} is a Reeb chord from Λ_i to Λ_j , then $e_k c_{ij} e_l = \delta_{ik} c_{ij} \delta_{il}$.

This algebra is again $\mathbb{Z}/2$ -graded in general, and \mathbb{Z} -graded when the canonical bundle is trivialized. We assume again that there exists a suitable virtual perturbation framework so that one can define a differential \tilde{d}_L (squaring to zero) on generators as follows:

• For a Reeb chord c^+ , we let

(10-8)
$$d_J(c^+) = \sum \frac{1}{|\operatorname{Aut}|} # \overline{\mathcal{M}}(c^+, \Gamma_{\Lambda}^-, \Gamma^-; \beta)_J \delta(\widehat{V} * \beta) c_1^- \cdots c_{\sigma}^- \gamma_1 \cdots \gamma_s,$$

where $\delta \colon \mathbb{R} \to \{0, 1\}$ satisfies $\delta(0) = 1$ and $\delta(s) = 0$ for $s \neq 0$ and the sum is over all possible choices of homotopy classes as in Definition/Assumption* 10.5.

• For a Reeb orbit γ , we let $\tilde{d}_J(\gamma)$ be the reduced contact homology differential associated to the twisted moduli counts $\#_{\tilde{u}}\mathcal{M}$, which only counts curves disjoint from \hat{V} .

We assume that $(\widetilde{CL}_{\bullet}(Y, \xi, V, \Lambda; \mathfrak{r})_{\lambda}, \widetilde{d}_J)$ is independent of λ, J up to canonical isomorphism in \mathbb{Q} -hdga. We denote the resulting object by

(10-9)
$$\widetilde{\mathcal{L}}(Y,\xi,V,\Lambda;\mathfrak{r})$$

and we let $CH_{\bullet}(Y, \xi, V, \Lambda; \mathfrak{r})$ be its homology. We also assume that $\widetilde{\mathcal{L}}(Y, \xi, V, \Lambda; \mathfrak{r})$ satisfies the limited functoriality described in Proposition* 10.7.

Proposition* 10.7 (cf Corollary 8.8) Let $(Y^{\pm}, \xi^{\pm}, V^{\pm})$ be TN contact pairs and choose elements $\mathfrak{r}^{\pm} = (\alpha^{\pm}, \tau^{\pm}, r^{\pm}) \in \mathfrak{R}(Y^{\pm}, \xi^{\pm}, V^{\pm})$. Consider an exact relative symplectic cobordism $(\hat{X}, \hat{\lambda}, H)$ with positive end (Y^+, ξ^+, V^+) and negative end (Y^-, ξ^-, V^-) , and suppose that τ^+ and τ^- extend to a global trivialization of the normal bundle of H.

Suppose that $L \subset (\hat{X}, \hat{\lambda}, H)$ is a cylindrical Lagrangian submanifold which is disjoint from H, with ends $\Lambda^{\pm} \subset (Y^{\pm} - V^{\pm}, \xi^{\pm}).$

Homological invariants of codimension 2 contact submanifolds

If $r^+ \ge e^{\mathcal{E}((H,\hat{\lambda}_H)_{\alpha}^{\alpha^+})}r^-$, then there is an induced map

(10-10)
$$\Phi(\hat{X}, \hat{\lambda}, H, L) \colon \mathcal{L}(Y^+, \xi^+, V^+; \mathfrak{r}^+) \to \mathcal{L}(Y^-, \xi^-, V^-; \mathfrak{r}^-).$$

The analogous statement holds for the reduced invariants $\widetilde{\mathcal{L}}(-)$.

Definition* 10.8 Let $\epsilon : \mathcal{A}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}[U]$ be an augmentation. Then we let

(10-11)
$$\mathcal{L}^{\epsilon}(Y,\xi,V,\Lambda;\mathfrak{r}) := \mathcal{L}(Y,\xi,V,\Lambda;\mathfrak{r}) \otimes_{\mathcal{A}(Y,\xi,V;\mathfrak{r})} \mathbb{Q}[U],$$

with differential $d_L \otimes 1$. The structure maps implicit in forming the tensor product (10-11) are, respectively, furnished by the inclusion $\mathcal{A}(Y, \xi, V; \mathfrak{r}) \subset \mathcal{L}(Y, \xi, V, \Lambda; \mathfrak{r})$ and the augmentation $\epsilon : \mathcal{A}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}[U]$. The resulting tensor product is naturally also a differential graded $\mathbb{Q}[U]$ algebra.

We similarly define

(10-12)
$$\widetilde{\mathcal{L}}^{\widetilde{\epsilon}}(Y,\xi,V,\Lambda;\mathfrak{r}) := \widetilde{\mathcal{L}}(Y,\xi,V,\Lambda;\mathfrak{r}) \otimes_{\widetilde{\mathcal{A}}(Y,\xi,V;\mathfrak{r})} \mathbb{Q},$$

using the maps $\widetilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \subset \widetilde{\mathcal{L}}(Y, \xi, V, \Lambda; \mathfrak{r})$ and $\widetilde{\epsilon} : \widetilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}$, which is naturally a differential graded \mathbb{Q} -algebra.

Remark 10.9 The algebra $\mathcal{L}^{\epsilon}(Y, \xi, V, \Lambda; \mathfrak{r})$ is the twisted analog of the Legendrian homology dg algebra (or Chekanov–Eliashberg dg algebra) described in [7, Section 4.1].

We now discuss gradings on the above Legendrian invariants.

Definition* 10.10 Let (Y, ξ, V) be a TN contact pair and choose $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let $\Lambda \subset (Y - V, \xi)$ be a Legendrian submanifold. Suppose that $H_1(Y; \mathbb{Z}) = 0$ and that $c_1(\xi) = 0$. Then the Legendrian homological $\mathbb{Z}/2$ -grading of $\mathcal{L}(Y, \xi, V, \Lambda; \mathfrak{r})$ (resp. $\tilde{\mathcal{L}}(Y, \xi, V, \Lambda; \mathfrak{r})$) lifts to a canonical \mathbb{Z} -grading given on orbits by (8-15) and given on chords by

(10-13)
$$|c| = CZ^{+,\tau}(c) - 1,$$

which is well-defined due to our topological assumptions.

The invariants $\mathcal{L}^{\epsilon}(Y, \xi, V, \Lambda; \mathfrak{r})$, $\overline{HC}(\mathcal{L}^{\epsilon}(Y, \xi, V, \Lambda; \mathfrak{r}))$, $\widetilde{\mathcal{L}}^{\epsilon}(Y, \xi, V, \Lambda; \mathfrak{r})$ and $\overline{HC}(\widetilde{\mathcal{L}}^{\epsilon}(Y, \xi, V, \Lambda; \mathfrak{r}))$ inherit a \mathbb{Z} -grading, which we also refer to as the homological grading.

Lemma* 10.11 With the notation of Proposition* 10.7, suppose that $H_1(Y^{\pm}; \mathbb{Z}) = 0$ and that $w_2(L) = c_1(\xi^{\pm}) = c_1(TX) = 0$. Then the cobordism maps described in Proposition* 10.7 preserve the Legendrian homological \mathbb{Z} -grading.

As in Section 8.2, there is also a linking number \mathbb{Z} -grading on the reduced Legendrian invariants under certain topological assumptions.

Definition* 10.12 Let (Y, ξ, V) be a TN contact pair and choose $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let $\Lambda \subset (Y, \xi)$ be a Legendrian submanifold. Suppose that $H_1(Y;\mathbb{Z}) = H_2(Y;\mathbb{Z}) = \pi_0(\Lambda) = \pi_1(\Lambda) = 0$. Then the *linking* number grading $|\cdot|_{\text{link}}$ on $\widetilde{\mathcal{L}}(Y,\xi,V,\Lambda;\mathfrak{r})$ is given on Reeb chords by

(10-14)
$$|c|_{\text{link}} = \text{link}_V(c; \Lambda).$$

It is given on Reeb orbits by (8-16). The grading is extended to arbitrary words by defining the grading of a word to be the sum of the gradings of its letters. One can verify using Lemma 5.22 that this grading is well-defined.

(10-15)
$$\widetilde{\mathcal{L}}_{\bullet,\bullet}(Y,\xi,V,\Lambda;\mathfrak{r})$$

be the bigraded differential \mathbb{Q} -algebra of bidegree (-1, 0), where

- the first bullet refers to the (Legendrian) homological \mathbb{Z} -grading (which is well-defined in view of our topological assumptions and the universal coefficients theorem, see Definition* 10.10), and
- the second bullet refers to the (Legendrian) linking number grading.

We also have the following lemma, which follows from Lemma* 10.11 and Lemma 5.22.

Lemma* 10.13 With the notation of Proposition* 10.7, suppose that $H_1(Y^{\pm};\mathbb{Z}) = H_2(Y^{\pm};\mathbb{Z}) =$ $H_2(X, Y^+; \mathbb{Z}) = 0$ and $\pi_0(\Lambda^{\pm}) = \pi_1(\Lambda^{\pm}) = 0$. Then the cobordism maps described in Proposition* 10.7 preserve the linking number \mathbb{Z} -grading. In the case that we also have $w_2(\Lambda^{\pm}) = c_1(\xi^{\pm}) = c_1(TX) = 0$, then the cobordism maps preserve the $(\mathbb{Z} \times \mathbb{Z})$ -bigrading (10-15).

Corollary* 10.14 Consider a TN contact pair (Y, ξ, V) and an element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$. Let $\Lambda \subset (Y, \xi)$ be a Legendrian submanifold. Let (W, λ, H) be a relative filling for (Y, ξ, V) and let $\tilde{\epsilon} : \tilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \to \mathbb{O}$ be the induced augmentation. Suppose that $H_1(Y;\mathbb{Z}) = H_2(Y;\mathbb{Z}) = H_2(W,Y;\mathbb{Z}) = 0$, that $\pi_0(\Lambda) =$ $\pi_1(\Lambda) = 0$ and that $w_2(\Lambda) = c_1(\xi) = c_2(TW) = 0$.

Then

(10-16)
$$\widetilde{\mathcal{L}}_{\bullet,\bullet}^{\widetilde{\epsilon}}(Y,\xi,V;\mathfrak{r})$$

inherits the structure of a $(\mathbb{Z} \times \mathbb{Z})$ -bigraded \mathbb{Q} -algebra with differential of bidegree (-1, 0). Moreover,

(10-17)
$$\overline{HC}_{\bullet,\bullet}(\widetilde{\mathcal{L}}^{\widetilde{\epsilon}}(Y,\xi,V;\mathfrak{r}))$$

inherits the structures of a $(\mathbb{Z} \times \mathbb{Z})$ -bigraded \mathbb{Q} -vector space.

Proof According to Lemma 5.21 and our topological hypotheses, the augmentation $\tilde{\epsilon}$ preserves the linking number. The first claim follows. For the second claim, note that both the homological grading and linking number grading are preserved by the cyclic permutation operator τ , and hence pass to reduced cyclic homology; see Definition 9.4.

10.3 The effect of Legendrian surgery

The familiar procedure of attaching a handle in differential topology can be performed in the symplectic category. There are various essentially equivalent approaches to doing this in the literature. For concreteness, we exclusively follow in this paper the construction described in [46, Section 3.1], which we now summarize.

Construction 10.15 (attaching a handle) Let (X_0^{2n}, λ_0) be a Liouville cobordism with positive boundary $(Y_0^{2n-1}, \xi_0 = \ker(\lambda_0|Y_0))$. Let $\Lambda \subset (Y_0 - V, \xi_0)$ be an isotropic sphere with trivialized conformal symplectic normal bundle (the latter condition is vacuous if Λ is a Legendrian). Choose an arbitrary open neighborhood \mathcal{U} of Λ , which we refer to as the *attaching region*.

We may now glue a model handle H along Y_0 inside U, following the detailed construction given in [46, Section 3.1]. The gluing is carried out by identifying the Liouville flow near Λ with the flow on H. We note that this gluing procedure involves some auxiliary choices, which we do not state here.

The outcome of the procedure (for any of the above auxiliary choices) is a Liouville cobordism (X, λ) with positive boundary $(Y, \xi = \ker(\lambda|_Y))$. We say that this domain is obtained from (X_0, λ_0) by *attaching a handle along* Λ , or *Legendrian surgery on* Λ . As is well-known from differential topology, *Y* differs from Y_0 by surgery along Λ .

In [7], Bourgeois, Ekholm and Eliashberg study the effect of handle attachment on various flavors of symplectic and contact homology. In particular, they describe conjectural exact sequences which govern the change in these invariants and describe the moduli spaces of holomorphic curves which should underlie the existence of these exact sequences. In Theorem/Assumption* 10.16 below (see also Remark 10.17), we state an analog of the surgery exact sequence for linearized contact homology in [7, Theorem 5.1]. The proofs sketched in [7, Section 6] also apply *mutatis mutandis* in the present setting, so will not be repeated. As discussed in Section 1.3, we expect that if [7, Theorem 5.1] can be made rigorous in the setting of Pardon's VFC package, it should pose no substantial additional difficulties to also establish Theorem/Assumption* 10.16.

To set the stage for Theorem/Assumption* 10.16, let (Y_0^{2n-1}, ξ_0) be a contact manifold and let $V \subset (Y_0, \xi_0)$ be a codimension 2 contact submanifold with trivial normal bundle. Let (X_0, λ_0) be a Liouville domain with positive boundary $(Y_0, \xi = \ker \lambda_0)$ and let $H_0 \subset (X, \lambda_0)$ be a symplectic submanifold which is preserved setwise by the Liouville flow near $\partial X_0 = Y_0$ and such that $\partial H = V$.

Let $\Lambda \subset (Y_0 - V, \xi_0)$ be an isotropic sphere with trivialized conformal symplectic normal bundle (the latter condition is vacuous if Λ is a Legendrian). Let (X, λ) be the Liouville domain obtained by attaching a Weinstein handle along Λ according to Construction 10.15 and let $(Y, \xi = \ker \lambda)$ be the positive boundary. We may assume that the attaching region is disjoint from $V \subset Y_0$. By abuse of notation, we therefore view V as a codimension 2 contact submanifold of (Y_0, ξ_0) and (Y, ξ) and view H_0 as a submanifold of X_0 and X. We also identify $\Re(Y_0, \xi_0, V) = \Re(Y, \xi, V)$.

We let $(\hat{X}_0, \hat{\lambda}_0, H)$ be the completion of (X_0, λ_0, H_0) , and let $(\hat{X}, \hat{\lambda}, H)$ be the completion of (X, λ, H_0) . There are natural (strict) markings $e_0 : \mathbb{R} \times Y_0 \to \hat{X}_0$ taking $(t, y_0) \mapsto \psi_t^0(y_0)$ and $e : \mathbb{R} \times Y \to \hat{X}$ taking $(t, y) \mapsto \psi_t(y_0)$, where ψ^0 (resp. ψ) is the Liouville flow in \hat{X}_0 (resp. in \hat{X}).

Finally, in order to have well-defined homological \mathbb{Z} -gradings, we assume that $H_1(Y_0; \mathbb{Z}) = H_1(Y; \mathbb{Z}) = 0$ and that $c_1(TX_0) = c_1(TX) = 0$.

Theorem/Assumption* 10.16 (cf Theorem 5.1 in [7]) With the above setup and $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$, consider the augmentations $\tilde{\epsilon}((\hat{X}_0, \hat{\lambda}_0, H), e_0) : \tilde{\mathcal{A}}(Y, \xi, V) \to \mathbb{Q}$ and $\tilde{\epsilon}((\hat{X}, \hat{\lambda}, H), e) : \tilde{\mathcal{A}}(Y, \xi, V) \to \mathbb{Q}$. If Λ is a Legendrian sphere, we have the following exact triangle, where the top horizontal arrow is the natural map induced by an exact relative symplectic cobordism:



If Λ is an isotropic sphere of dimension $k \leq n-2$, then we have that

(10-19)
$$H_*\left(\operatorname{Cone}\left(\widetilde{CH}_{\bullet-(n-3)}^{\widetilde{\epsilon}}(Y,\xi,V;\mathfrak{r})\to\widetilde{CH}_{\bullet-(n-3)}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V;\mathfrak{r})\right)\right) = \begin{cases} \mathbb{Q} & \text{if } *=n-k+2\mathbb{N}, \\ 0 & \text{otherwise.} \end{cases}$$

Remark 10.17 There is a natural analog of Theorem/Assumption* 10.16 involving the invariants $CH_{\bullet-(n-3)}^{\epsilon_0}(Y_0,\xi_0,V;\mathfrak{r}), CH_{\bullet-(n-3)}^{\epsilon}(Y,\xi,V;\mathfrak{r})$ and $\overline{HC}_{\bullet}(\mathcal{L}^{\epsilon_0}(Y_0,\xi_0,V,\Lambda;\mathfrak{r}))$ which we also expect to hold. We do not state it here since we do not use it in this paper.

Remark 10.18 With the setup of Theorem/Assumption* 10.16, let us in addition assume that $H_2(Y_0; \mathbb{Z}) = H_2(X_0, Y_0) = H_2(Y; \mathbb{Z}) = H_2(X, Y; \mathbb{Z}) = 0$. Then Lemma 8.13 and Corollary* 10.14 provide an additional linking number \mathbb{Z} -grading on the invariants appearing in the surgery exact sequences.

The resulting $(\mathbb{Z}\times\mathbb{Z})$ -bigrading is preserved by the maps in the surgery exact sequence. Indeed, Lemma 8.13 ensures the top horizontal map preserves the linking number \mathbb{Z} -grading. The bottom right map counts holomorphic disks with one positive interior puncture, *k* negative boundary punctures, and with boundary mapping to $S\Lambda$ (the relevant moduli space is described in [7, Section 2.6]). Hence one can readily verify (cf Lemma 5.22) that this map also preserves the linking number grading. Finally, the bottom left map is defined algebraically as the connecting map in the long exact sequence. Since the internal differentials of the relevant chain complexes preserve the linking number grading, this connecting map does too.

11 Some computations

11.1 Vanishing results

Recall that a contact manifold (Y^{2n-1}, ξ) is said to be *overtwisted* if it contains an overtwisted disk; see [5, Section 1]. In general, if (Y, ξ) is overtwisted and $C \subset Y$ is a closed subset, then $(Y - C, \xi)$ may not be overtwisted.

Theorem 11.1 Suppose that (Y, ξ, V) is a TN contact pair such that $(Y - V, \xi)$ is overtwisted. Given any element $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$, we have

(11-1) $CH_{\bullet}(Y,\xi,V;\mathfrak{r}) = \widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r}) = 0.$

We collect some definitions which will be useful in proving Theorem 11.1. Let $\mathfrak{alm}_{U(n-1)}(S^{2n-1})$ be the set of almost-contact structures on S^{2n-1} ; see Definition 12.1. It follows by the main theorem of [5] that $\mathfrak{alm}_{U(n-1)}(S^{2n-1})$ is in canonical correspondence with the set of overtwisted contact structures on the sphere, a fact which will be used implicitly in the proof of Theorem 11.1.

A folklore result in contact topology (see eg [12, Section 6]) states that for any fixed element $\beta \in \mathfrak{alm}_{U(n-1)}(S^{2n-1})$, the operation of connected sum endows $\mathfrak{alm}_{U(n-1)}(S^{2n-1})$ with a group structure with identity element β . The isomorphism class of the resulting group is moreover independent of β . Since these facts are not to our knowledge available in the literature, we have provided careful proofs in the appendix.

For the remainder of this section, we fix $\beta \in \mathfrak{alm}_{U(n-1)}(S^{2n-1})$ to be the almost-contact structure induced by the standard contact structure on the sphere. Given a pair of contact manifolds (M_1, α_1) and (M_2, α_2) , one can also consider their connected sum $(M_1 \# M_2, \alpha_1 \# \alpha_2)$, which is obtained by gluing-in a neck along Darboux balls in M_1, M_2 . This operation is discussed in Remark A.10. As noted there, the two a priori different notions of a connected sum of (almost-)contact manifolds commute with the forgetful map from contact manifolds to almost-contact manifolds.

Proof of Theorem 11.1 It is enough to prove that the invariants vanish for a particular choice of nondegenerate contact form $\tilde{\alpha}$ on Y which is adapted to r. To construct such a form, we follow arguments of Bourgeois and Van Koert in [8, Section 6.2].

Using Construction 3.6, we define an auxiliary contact form α in a small neighborhood \mathcal{N} of V with the property that V is a strong contact submanifold and that α is adapted to \mathfrak{r} . After possibly shrinking \mathcal{N} , we can assume that $(Y - \mathcal{N}, \xi)$ is overtwisted. We now extend α arbitrarily to a globally defined, nondegenerate contact form on (Y, ξ) . (Since α_V is nondegenerate, Construction 3.6 produces a nondegenerate contact form on \mathcal{N} , so it extends unproblematically to a global nondegenerate contact form.)

Choose a Darboux ball $B \subset Y$ whose closure is disjoint from \mathcal{N} . Let $B' \subset B$ be a smaller Darboux ball and let $A = B - \overline{B'}$. Let β_0 denote the almost-contact structure on B obtained by restricting ξ . Let $\mathfrak{alm}_{U(n-1)}(B, A; \beta_0)$ be the set of almost-contact structures on B agreeing with β_0 near A. The group $\mathfrak{alm}_{U(n-1)}(S^{2n-1})$ acts on $\mathfrak{alm}_{U(n-1)}(B, A; \beta_0)$ by connect-summing with an almost-contact sphere along a disk whose closure is disjoint from A.

We now appeal to work of Bourgeois and van Koert: in [8, Section 2.2], they construct a special contact form α_L on the sphere S^{2n-1} (this form turns out to be overtwisted by [12], although we won't need this). They prove [8, Sections 2–5] that α_L admits a Reeb orbit γ which bounds a single, transversally cut-out *J*-holomorphic plane, for some suitable *J* on the symplectization.

We now form the connected sum of (S^{2n-1}, α_L) with (Y, α) , where we assume that the gluing happens entirely inside of B'. Bourgeois and van Koert (following earlier work of Ustilovsky [65]) explain how to perform this connected sum so that the orbit γ survives, and still has the property that it bounds a single transversally cut-out plane (basically, one can suitably adjust the neck to ensure that the plane cannot cross the neck; see [8, Section 6.2]).

Finally, we connect sum with another overtwisted contact sphere (S^{2n-1}, α'_L) so that $(B, \alpha \# \alpha_L \# \alpha'_L |_B)$ is formally contact isotopic to $(B, \alpha |_B)$, through a contact isotopy fixed near A. Note that we may freely assume that the two connected sums happen in disjoint regions of B', so they do not interfere with each other. Unwinding the definitions, this means that there exists a diffeomorphism $\psi : B \to B \# S^{2n-1} \# S^{2n-1}$ fixed near the boundary and a formal contact isotopy from ker $\psi^*(\alpha \# \alpha_L \# \alpha'_L)|_B$ to $\xi_B = \ker \alpha|_B$, which is fixed near A.

If we extend ψ to a diffeomorphism $Y \to Y \# S^{2n-1} \# S^{2n-1}$ by letting it be the identity outside of B, we observe that ker $\psi^*(\alpha \# \alpha_L \# \alpha'_L)$ is formally isotopic to $\xi = \ker \alpha$. Moreover, these contact structures agree on $Y - B \supset \mathcal{N}$. Since $(Y - \mathcal{N}, \xi)$ is overtwisted, it follows from the relative h-principle for overtwisted contact structures (see [5, Theorem 1.2]) that there is a smooth isotopy ϕ_t fixed on \mathcal{N} so that $\tilde{\alpha} := \phi_1^* \psi^*(\alpha \# \alpha_L \# \alpha'_L)$ is a contact form for (Y, ξ) . By construction, $\tilde{\alpha} = \alpha$ on \mathcal{N} , so $\tilde{\alpha}$ is adapted to \mathfrak{r} . Finally, it follows from the above discussion that $CH_{\bullet}(Y, \xi, V; \mathfrak{r})$ vanishes when we compute it using the form $\tilde{\alpha}$ (since γ bounds a rigid plane). An analogous argument shows that $\widetilde{CH}_{\bullet}(Y, \xi, V; \mathfrak{r})$ vanishes as well.

We also state a vanishing result for the deformed Chekanov–Eliashberg dg algebra of certain loose Legendrians. To set the notation, let us now assume that (Y, ξ, V) is an arbitrary TN contact pair and fix $\mathfrak{r} \in \mathfrak{R}(Y, \xi, V)$.

Proposition* 11.2 Suppose that $\Lambda \subset (Y - V, \xi)$ is a loose Legendrian submanifold. Then $\mathcal{L}(Y, \xi, V, \Lambda; \mathfrak{r})$ and $\widetilde{\mathcal{L}}(Y, \xi, V, \Lambda; \mathfrak{r})$ are acyclic. Given augmentations $\epsilon : \mathcal{A}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}[U]$ and $\widetilde{\epsilon} : \widetilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}$, the invariants $\mathcal{L}^{\epsilon}(Y, \xi, V, \Lambda; \mathfrak{r})$ and $\widetilde{\mathcal{L}}^{\epsilon}(Y, \xi, V, \Lambda; \mathfrak{r})$ are also acyclic.

Proof The argument is the same as that which shows that the (undeformed) Chekanov–Eliashberg dg algebra of a loose Legendrian is acyclic (see eg [55, Section 5]): up to Legendrian isotopy in Y - V, we can find a chord c of arbitrarily small action which bounds a single half-disk. This disk can be assumed to stay in a small ball disjoint from V for action reasons. Hence we have d(c) = 1.

11.2 Nonvanishing results: bindings of open books

The following theorem is the main result of this section.

Theorem 11.3 Consider a TN contact pair (Y, ξ, V) . Suppose that Y admits an open book decomposition (Y, B, π) which supports the contact structure ξ and realizes V = B as its binding. Viewing (B, τ) as a framed contact submanifold, where τ denote the trivialization of $B \subset Y$ induced by the open book decomposition, we have

(11-2) $\widetilde{CH}_{\bullet}(Y,\xi,B;\mathfrak{r}) \neq 0$

for any $\mathfrak{r} = (\alpha_B, \tau, r) \in \mathfrak{R}(Y, \xi, B)$.

By combining Theorem 11.3 with Corollary 8.21, we obtain the following result.

Corollary 11.4 Under the hypotheses of Theorem 11.3, if r' is large enough and we write $r' = (\alpha_B, \tau, r')$, then

(11-3)
$$CH_{\bullet}(Y,\xi,B;\mathfrak{r}')\neq 0.$$

Proof of Theorem 11.3 According to Corollary 3.15, the open book decomposition (Y, B, π) supports a nondegenerate Giroux form α which is adapted to \mathfrak{r} for any $\mathfrak{r} = (\alpha_B, \tau, r) \in \mathfrak{R}(Y, \xi, B)$.

Consider the algebra $\widetilde{CC}_{\bullet}(Y, \xi, B; \mathfrak{r})$ generated by (good) Reeb orbits of α not contained in *B*. After fixing an almost complex structure $J: \xi \to \xi$ which is compatible with $d\alpha$ and preserves $\xi|_B$, and a choice of perturbation data $\theta \in \Theta_{\mathrm{I}}((Y, \xi, B), \alpha, J)$, we get a differential $d_J = d(\tilde{\psi}_B, J, \theta)$ and the homology of the resulting chain complex is (canonically isomorphic to) $\widetilde{CH}_{\bullet}(Y, \xi, B; \mathfrak{r})$.

Let us suppose for contradiction that $\widetilde{CH}_{\bullet}(Y, \xi, B; \mathfrak{r}) = 0$. This means that 1 is in the image of the differential. By the Leibnitz rule, this implies that there exists some good Reeb orbit $\gamma: S^1 \to Y$ and a relative homotopy class $\beta \in \pi_2(Y, \gamma)$ such that the twisted moduli count of planes positively asymptotic to γ in the homotopy class β is nonzero. To state this more formally in the language of Section 5.3, let $T \in S_{\mathrm{I}}((Y, \xi, B), \alpha, J)$ be the tree with a single input edge *e* and a single vertex *v*, where *e* is decorated with the Reeb orbit γ and *v* is decorated with the $\beta \in \pi_2(Y, \gamma)$. Then we have that $\widetilde{\psi}_B(T) \neq 0$.

In particular, this implies that $\overline{\mathcal{M}}(T) \neq \emptyset$. Hence there exists $T' \to T$ such that T' is representable by a J-holomorphic building. The proof of Proposition 7.7 shows that we may assume that T' does not have any edges contained in B (since otherwise we would have $\tilde{\psi}_B(T') = \tilde{\psi}_B(T) = 0$).

It follows by Proposition 5.12 that $T' * \hat{B} = \sum_{v \in V(T')} \beta_v * \hat{B}$, and Corollary 5.14 implies that all the terms on the right-hand side are nonnegative. Since $\tilde{\psi}_B(T') = \tilde{\psi}_B(T) \neq 0$, it follows by definition of the reduced twisting maps that $T' * \hat{V} = 0$. Hence $\beta_v * \hat{B} = 0$ for all $v \in V(T')$.

For topological reasons, there exists $\tilde{v} \in V(T')$ such that \tilde{v} has a single incoming edge and no outgoing edges. Hence \tilde{v} is represented by a *J*-holomorphic plane *u* which is asymptotic to some Reeb orbit $\tilde{\gamma}$. By positivity of intersection (see Proposition 5.3) and the fact that $\beta_{\tilde{v}} * \hat{B} = 0$, the image of *u* is contained in $\mathbb{R} \times (Y \setminus B)$. Thus $\tilde{\gamma}$ is contractible in $Y \setminus B$, which implies that the composition $\pi \circ \tilde{\gamma} : S^1 \to Y \setminus B \to S^1$ has degree 0. This is a contradiction: since α is a Giroux form, $\pi \circ \tilde{\gamma}$ must be an immersion by Remark 3.12, and hence have nonzero degree.

We also state a vanishing result for the (reduced) Chekanov–Eliashberg dg algebra introduced in Section 10.2.

Theorem* 11.5 Let (Y, ξ, V) be a TN contact pair and let $\Lambda \subset (Y - V, \xi)$ be a Legendrian submanifold. Suppose that ξ supports an open book decomposition π with binding B = V such that Λ is contained in a single page. Let τ be the trivialization of $N_{Y/V}$ induced by the open book. Then we have

(11-4)
$$\widetilde{\mathcal{L}}(Y,\xi,V,\Lambda;\mathfrak{r}) \neq 0.$$

Proof The proof is identical to that of Theorem 11.3; namely, one argues that the image of any Reeb orbit or chord under the differential cannot contain a term of degree zero, which immediately implies the claim. \Box

We note that is was proved by Honda and Huang [39, Corollary 1.3.3] that any Legendrian Λ in a contact manifold (Y, ξ) is contained in the page of some compatible open book decomposition. Hence it follows from Theorem* 11.5 that every Legendrian is tight in the complement of some codimension 2 contact submanifold.

11.3 Explicit computations in open books

We now perform certain explicit computations in open book decompositions which will be used in applications in the next sections. We assume throughout this section that $n \ge 4$. This assumption is needed for the purpose of obtaining a $(\mathbb{Z} \times \mathbb{Z})$ -bigrading on $\widetilde{CH}(-)$; see Lemma 11.6 and Corollary 12.17.

Let us endow S^{n-1} with a Riemannian metric *h* having the property that all geodesics are nondegenerate. Such metrics, which are typically referred to as "bumpy" in the literature, are generic in the space of Riemannian metrics; see [1] or [44, 3.3.9]. It can be shown [1] that any manifold endowed with a bumpy metric admits a closed geodesic of minimal length. We let $\rho > 0$ be the length of the shortest geodesic of (S^{n-1}, h) . To set the stage for this section, it will be useful to recall some general facts about coordinate systems. Given a system of local coordinates (q_1, \ldots, q_m) on some manifold M, the *dual coordinates* (p_1, \ldots, p_m) in the fibers of the T^*M are characterized by the property that

(11-5)
$$T^*M \ni (q, p) = (q_1, \dots, q_m, p_1, \dots, p_m) = \sum_{i=1}^m p_i dq_i.$$

Unless otherwise indicated, a pair (q, p) refers to a system of local coordinates in the cotangent bundle of a manifold, where p is dual to q.

It will sometimes also be useful to work with Riemannian normal coordinates. Recall that on a Riemannian manifold (M, g), a system of normal coordinates (x_1, \ldots, x_m) has the property that for any vector $a \in T_x M$, the path $\theta \mapsto at$ is a geodesic. If (q_1, \ldots, q_n) is a system of Riemannian normal coordinates, then the path $t \mapsto (\gamma(t), \dot{\gamma}^{\flat})$ can be written in coordinates (q, p) as

$$(11-6) t \mapsto (at, a) \in T^*M.$$

We now introduce a Liouville manifold which will be studied throughout the remainder of this section. For a > 0, define

(11-7)
$$(\widehat{W}_0, \widehat{\lambda}^a) = \left(D^2 \times T^* S^{n-1}, \widehat{\lambda}^a := \frac{1}{a} s^2 \, d\theta + \lambda_{\text{std}}\right),$$

where we have chosen local coordinates (s, θ, q, p) . We emphasize that the Liouville structure depends on the parameter a > 0.

Let $\phi: \widehat{W}_0 \to \mathbb{R}$ be the function

(11-8)
$$\phi(s,\theta,\boldsymbol{q},\boldsymbol{p}) = s^2 + \|\boldsymbol{p}\|^2$$

We consider the Liouville domain

(11-9)
$$(W_0, \lambda^a) = (\{\phi \le 1\}, \lambda^a := \hat{\lambda}^a | W_0)$$

and its contact-type boundary

(11-10)
$$(Y_0, \xi_0) = (\{\phi = 1\}, \xi = \ker \lambda_0),$$

where $\lambda_0 = (\hat{\lambda}^a)|_{Y_0}$ is the induced contact form. Consider also the codimension 2 contact submanifold

(11-11)
$$V = \{\phi = 1, s = 0\} \subset (Y_0, \xi_0)$$

and the Legendrian

(11-12)
$$\Lambda := \{ \phi = 1, \theta = \text{constant}, s = 1, \|p\| = 0 \}.$$

We define $\alpha := (\lambda_0)|_V$ and let τ be the trivialization of $N_{Y_0/V}$, which is unique by Lemma 11.6. We set

(11-13)
$$\mathfrak{r} = (\alpha, \tau, a) \in \mathfrak{R}(Y_0, \xi_0, V).$$

Finally, we let $H = \{0\} \times T^* S^{n-1} \subset \widehat{W}_0$.

Observe that \mathfrak{r} depends on our choice of a > 0. More generally, the contact form $\lambda_0 = (\hat{\lambda}^a)|_{Y_0}$ on (Y_0, ξ_0) obviously depends on a > 0. The plan is now to study the Reeb dynamics on (Y_0, ξ_0) with respect to this contact form. By taking $a \gg 0$ large enough, we will be able to obtain a sufficiently good understanding of the Reeb dynamics to compute the invariant $\widetilde{CH}(Y_0, \xi_0, V; \mathfrak{r})$ in low degrees; see Proposition* 11.17.

Lemma 11.6 The manifolds W_0 and Y_0 have vanishing first and second homology and cohomology with \mathbb{Z} -coefficients. In addition, we have $H^1(V; \mathbb{Z}) = 0$ and $w_2(\Lambda) = 0$.

Proof The first claim is proved in Corollary 12.17. To compute $H^1(V; \mathbb{Z})$, note that V is the sphere bundle associated to T^*S^{n-1} . Hence, we have a fibration $S^{n-2} \hookrightarrow V \to S^{n-1}$ giving rise to a Gysin sequence

(11-14)
$$\cdots \to H^k(S^{n-1};\mathbb{Z}) \to H^k(V;\mathbb{Z}) \to H^{k-(n-2)}(S^{n-1};\mathbb{Z}) \to \cdots$$

Taking k = 1 immediately gives the desired result since $n \ge 4$. Finally, note that $\Lambda = S^{n-1}$, which has vanishing homology (with any coefficients) in degrees $1 \le i \le n-2$. Hence $w_2(\Lambda) = 0$ for $n \ge 4$. \Box

Observe that there is a natural marking

(11-15)
$$e_0: \mathbb{R} \times Y_0 \to (\widehat{W}_0, \widehat{\lambda}^a, H), \quad (t, y) \mapsto \psi_t(y),$$

where $\psi_{(-)}$ is the Liouville flow associated to $\hat{\lambda}^a$.

This endows $(\hat{W}_0, \hat{\lambda}^a, H)$ with the structure of a (strict) relative exact symplectic cobordism. We thus obtain an augmentation

(11-16)
$$\widetilde{\epsilon}_0 : \widetilde{\mathcal{A}}(Y_0, \xi_0, V; \mathfrak{r}) \to \mathbb{Q}$$

It follows from Lemma 11.6 and the discussion following Definition* 9.10 that $\widetilde{\mathcal{A}}(Y_0, \xi_0, V; \mathfrak{r})$ and $\widetilde{\mathcal{A}}^{\widetilde{\epsilon}_0}(Y_0, \xi_0, V; \mathfrak{r})$ admit a $(\mathbb{Z} \times \mathbb{Z})$ -bigrading.

We now analyze the structure of (Y_0, λ_0) in more detail. First, observe that $(Y_0 - V, \lambda_0|_{Y_0-V})$ is strictly contactomorphic to

(11-17)
$$\left(S^1 \times D^* S^{n-1}, \alpha_V := \frac{1}{a} (1 - \|\boldsymbol{p}\|^2) \, d\theta + \lambda_{\text{std}}\right)$$

via the map

(11-18)
$$S^1 \times D^* S^{n-1} \to Y_0 - V, \quad (\theta, \boldsymbol{q}, \boldsymbol{p}) \mapsto \left(\sqrt{1 - \|\boldsymbol{p}\|^2}, \theta, \boldsymbol{q}, \boldsymbol{p}\right),$$

where $D^*S^{n-1} = \{(q, p) \in T^*S^{n-1} | ||p|| < 1\}$. We let $\mathcal{N} \subset Y_0$ denote the image of $S^1 \times S^{n-1}$ under this map; equivalently, $\mathcal{N} = \{||p|| = 0\}$. The complement $(Y_0 - \mathcal{N}, \lambda_0|_{Y_0 - \mathcal{N}})$ is strictly contactomorphic to

$$\Big(B \times U, \alpha_{\mathcal{N}} := \frac{1}{a} (x \, dy - y \, dx) + \sqrt{1 - x^2 - y^2} \alpha_U \Big),$$

where $B \subset \mathbb{R}^2$ denotes the open unit disk and (U, α_U) denotes the unit cotangent bundle of (S^{n-1}, h) , equipped with the contact form $\alpha_U := \lambda_{\text{std}}$ induced by the canonical Liouville form on T^*S^{n-1} . A contactomorphism is given by

(11-19)
$$B \times U \to Y_0 - \mathcal{N}, \quad (x, y, q, p) \mapsto \left(x, y, q, \sqrt{1 - x^2 - y^2}p\right).$$

Observe that the size of the tubular neighborhood $B \times U$ depends on our choice of a > 0.

Our first task is to study the Reeb orbits of λ_0 which are in the complement of \mathcal{N} . In particular, we wish to show that they are nondegenerate for a generic choice of *a*, and moreover that their Conley–Zehnder indices depend linearly on *a*. This is the content of Proposition 11.7 and Corollary 11.8 below.

Proposition 11.7 Let $\gamma_U : \mathbb{R}/\mathbb{Z} \to U$ be a Reeb orbit of α_U of period P_U . Then:

(A) The map

(11-20)
$$\gamma_1 \colon \mathbb{R}/\mathbb{Z} \to B \times U, \quad t \mapsto (0, 0, \gamma_U(t)),$$

is a Reeb orbit of α_N of period $P_1 := P_U$.

(B) Given any $r_0 \in (0, 1)$ and integers m, n > 0 such that

(11-21)
$$\frac{a P_U}{4\pi \sqrt{1-r_0^2}} = \frac{m}{n}$$

the map

(11-22)
$$\gamma_2: \mathbb{R}/\mathbb{Z} \to B \times U, \quad t \mapsto (r_0 \cos(2\pi m t), r_0 \sin(2\pi m t), \gamma_U(nt)),$$

is a Reeb orbit of α_N of period

$$P_2 := (2 - r_0^2) \frac{2\pi m}{a} = (2 - r_0^2) \frac{n P_U}{2\sqrt{1 - r_0^2}}.$$

Every Reeb orbit of α_N is of the form (A) or (B) for some choice of γ_U , r_0 , m and n.

If α_U is nondegenerate and *a* satisfies

(11-23)
$$a^{-1} \notin \bigcup_{q \in \mathbb{Q}_{>0}} \frac{1}{4\pi \sqrt{q}} \mathcal{S}(\alpha_U),$$

where $S(\alpha_U) \subset \mathbb{R}$ is the action spectrum of α_U , then α_N is nondegenerate. Moreover, given any trivialization τ_0 of $\gamma_U^* \ker(\alpha_U)$, there exist trivializations τ_i of $\gamma_i^* \ker(\alpha_N)$ for i = 1, 2 such that

(11-25)
$$CZ^{\tau_2}(\gamma_2) = 1 + 2 \left\lfloor \frac{P_2 a}{\pi (2 - r_0^2)^2} \right\rfloor + CZ^{\tau_0}(\gamma_U^n).$$

If τ_0 extends to a disk spanning γ_U , then τ_i extends to a disk spanning γ_i .

Proof The Reeb vector field of α_N is given by

(11-26)
$$R_{\alpha_{\mathcal{N}}} = \frac{1}{2 - x^2 - y^2} (a(x\partial_y - y\partial_x) + 2\sqrt{1 - x^2 - y^2} R_U),$$

where R_U denotes the Reeb vector field of α_U (recall that a > 0 is a constant fixed above). A simple computation shows that γ_1 and γ_2 are Reeb orbits with periods as claimed, and that there are no other orbits.

Note that the contact structure $\xi = \ker(\alpha_N)$ splits as

(11-27)
$$\xi = \langle e_1, e_2 \rangle \oplus \ker(\alpha_U),$$

where e_1 and e_2 are the vector fields on $B \times U$ defined by

(11-28)
$$e_1 = \partial_x + \frac{y}{a\sqrt{1 - x^2 - y^2}} R_U,$$

(11-29)
$$e_2 = \partial_y - \frac{x}{a\sqrt{1 - x^2 - y^2}} R_U.$$

In particular, given a trivialization τ_0 of $\gamma_U^* \ker(\alpha_U)$, we get trivializations $\tau_1 = \langle \gamma_1^* e_1, \gamma_1^* e_2 \rangle \oplus \tau_0$ and $\tau_2 = \langle \gamma_2^* e_1, \gamma_2^* e_2 \rangle \oplus \tau_0^n$ of $\gamma_1^* \xi$ and $\gamma_2^* \xi$, where τ_0^n denotes the trivialization of $(\gamma_U^n)^* \ker(\alpha_U)$ induced by τ_0 .

We have

(11-30)
$$\mathcal{L}_{e_1} R_{\alpha_N} = -\partial_x \left(\frac{ay}{2 - x^2 - y^2} \right) e_1 + \partial_x \left(\frac{ax}{2 - x^2 - y^2} \right) e_2$$
$$= \frac{a}{(2 - x^2 - y^2)^2} \left(-2xye_1 + (2 + x^2 - y^2)e_2 \right),$$
(11-31)
$$\mathcal{L}_{e_2} R_{\alpha_N} = \partial_y \left(\frac{ay}{2 - x^2 - y^2} \right) e_1 - \partial_y \left(\frac{ax}{2 - x^2 - y^2} \right) e_2$$

$$= \frac{a}{(2-x^2-y^2)^2} \left(-(2-x^2+y^2)e_1 + 2xye_2 \right).$$

Moreover, for any vector field X on U such that $X \in ker(\alpha_U)$, we have

(11-32)
$$\mathcal{L}_X R_{\alpha_N} = \frac{2\sqrt{1-x^2-y^2}}{2-x^2-y^2} \mathcal{L}_X R_U.$$

Hence, if $\Psi_i(t): \xi_{\gamma_i(0)} \to \xi_{\gamma_i(t)}$ denotes the linearized Reeb flow along γ_i (viewed as a matrix via the trivialization τ_i) for i = 1, 2, then $\Psi'_i(t) = S_i(t)\Psi_i(t)$ with

(11-33)
$$S_1(t) = \frac{aP_U}{2} \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \oplus P_U S_U(t),$$

(11-34)
$$S_2(t) = \frac{2\pi m}{2 - r_0^2} \begin{pmatrix} -r_0^2 \sin(4\pi mt) & -2 + r_0^2 \cos(4\pi mt) \\ 2 + r_0^2 \cos(4\pi mt) & r_0^2 \sin(4\pi mt) \end{pmatrix} \oplus n P_U S_U(nt),$$

where $S_U(t)$ is the matrix such that the linearized Reeb flow $\Psi_U: (\xi_U)_{\gamma_U(0)} \to (\xi_U)_{\gamma_U(t)}$ of R_U along γ_U satisfies $\Psi'_U(t) = P_U S_U(t) \Psi_U(t)$.

It follows that $CZ^{\tau_1}(\gamma_1) = CZ(\psi_1) + CZ^{\tau_0}(\gamma_U)$ and $CZ^{\tau_2}(\gamma_2) = CZ(\psi_2) + CZ^{\tau_0}(\gamma_U^n)$, where ψ_1 and ψ_2 are paths of 2×2 matrices given by $\psi_i(t) = \exp(P_i(t))$ with

(11-35)
$$P_1(t) = t \frac{aP_U}{2} \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix},$$

(11-36)
$$P_{2}(t) = \int_{0}^{t} \frac{2\pi m}{2 - r_{0}^{2}} \begin{pmatrix} -r_{0}^{2} \sin(4\pi ms) & -2 + r_{0}^{2} \cos(4\pi ms) \\ 2 + r_{0}^{2} \cos(4\pi ms) & r_{0}^{2} \sin(4\pi ms) \end{pmatrix} ds,$$
$$= \frac{2\pi m}{2 - r_{0}^{2}} \begin{pmatrix} \frac{r_{0}^{2}}{4\pi m} (\cos(4\pi mt) - 1) & -2t + \frac{r_{0}^{2}}{4\pi m} \sin(4\pi mt) \\ 2t + \frac{r_{0}^{2}}{4\pi m} \sin(4\pi mt) & -\frac{r_{0}^{2}}{4\pi m} (\cos(4\pi mt) - 1) \end{pmatrix}$$

Note that $P_1(t)$ and $P_2(t)$ are diagonalizable with eigenvalues $\pm 2\pi i \lambda_1(t)$ and $\pm 2\pi i \lambda_2(t)$, respectively, where

(11-37)
$$\lambda_1(t) = t \frac{a P_U}{4\pi},$$

(11-38)
$$\lambda_2(t) = \frac{1}{2 - r_0^2} \sqrt{4m^2 t^2 - \frac{r_0^4}{8\pi^2} (1 - \cos(4\pi mt))}.$$

It follows that ker($\psi_i(t)$ -Id) is either \mathbb{R}^2 or 0, depending on whether $\lambda_i(t)$ is an integer or not. Assumption (11-23) implies that $\lambda_i(1)$ is not an integer and hence that $\psi_i(1)$ doesn't have 1 as an eigenvalue, ie ψ_i is nondegenerate. This is clear for $\lambda_1(1)$; to check this for $\lambda_2(1)$, note that it follows from (11-21) that

(11-39)
$$\lambda_2(1) = \frac{2m}{2 - r_0^2} = \frac{2m}{1 + \frac{n^2 a^2 P_U^2}{(4\pi)^2 m^2}} = \frac{2m^3 (4\pi)^2}{(4\pi)^2 m^2 + n^2 a^2 P_U^2}.$$

If this expression were integral, then the reciprocal would be rational; hence $n^2 a^2 P_U^2 / (2m^3(4\pi)^2)$ would be rational, contradicting Assumption (11-23).

Since $-J_0 P'_i(t)$ is positive-definite for all t, it follows from [33, Proposition 52] that

(11-40)
$$CZ(\psi_i) = 1 + 2 \# \{ t \in (0, 1) \mid \lambda_i(t) \in \mathbb{Z} \}.$$

Since λ_i is strictly increasing with $\lambda_i(0) = 0$ and $\lambda_i(1) \notin \mathbb{Z}$, the right-hand side is equal to $1 + 2\lfloor \lambda_i(1) \rfloor$. Thus

as desired.

Geometry & Topology, Volume 28 (2024)

Corollary 11.8 Suppose that γ is a closed Reeb orbits of $(Y_0, \xi = \ker \lambda_0)$ which is contained in the complement of $\mathcal{N} \subset Y_0$. Then

(11-43)
$$\operatorname{CZ}^{\tau}(\gamma) > \left\lfloor \frac{a\rho}{\pi} \right\rfloor,$$

where τ is a trivialization which extends to a spanning disk and $\rho > 0$ is as on page 92.

Proof It is well-known that the Reeb orbits on U correspond bijectively to geodesics on (S^{n-1}, h) ; with our normalization, the action of a closed Reeb orbit equals twice the length of the corresponding unit-speed geodesic; see eg [29, Section 1.5]. Moreover, according to [24, Proposition 1.7.3], given a Reeb orbit $\tilde{\gamma}$ which corresponds to a geodesic γ , we have

(11-44)
$$\mu_M(\gamma) = CZ^{\tau}(\widetilde{\gamma}),$$

where μ_M is the Morse index of the geodesic and τ extends to a spanning disk; see Remark 11.9. Since the Morse index of a geodesic is nonnegative by definition, the corollary follows from Proposition 11.7. \Box

Remark 11.9 The trivialization considered in [24, Proposition 1.7.3] is in fact constructed as follows. Choose a spanning disk $\tilde{v}: D^2 \to U \subset T^*S^{n-1}$ for $\tilde{\gamma}$ and let $v := \pi \circ \tilde{v}$, where $\pi: T^*S^{n-1} \to S^{n-1}$ is the projection. Let $\{\sigma^1, \ldots, \sigma^{n-1}\}$ be a trivialization of v^*TS^{n-1} . For points $\pi: \tilde{x} \mapsto x$, let $Q_{x;\tilde{x}}: \pi^{-1}(x) \to T_{\tilde{x}}(\pi^{-1}(x))$ be the canonical identification. Now define $\tilde{\sigma}_p^i = Q_{v(p);\tilde{v}(p)}\sigma_p$ for $p \in D^2$. Then $\{\tilde{\sigma}^1, \ldots, \tilde{\sigma}^{n-1}\}$ defines a Lagrangian subbundle of the symplectic vector bundle $(\tilde{v}^*(\xi), d\lambda_0)$. Hence it induces a unique trivialization of $\tilde{v}^*\xi$, which restricts on the boundary to a trivialization of $\tilde{\gamma}^*\xi$.

We now turn to the Reeb dynamics near \mathcal{N} . Recall from page 94 that \mathcal{N} is contained in $(Y_0 - V, \lambda_0)$, which is strictly contactomorphic to $(S^1 \times D^* S^{n-1}, \alpha_V)$, where $\alpha_V = (1/a)(1 - \|\boldsymbol{p}\|^2) d\theta + \lambda_{\text{std}}$.

Lemma 11.10 Let $q = (q_1, \ldots, q_{n-1})$ be Riemannian normal coordinates in some open set $U \subset (S^{n-1}, h)$ and let $p = (p_1, \ldots, p_n)$ be the dual coordinates. The Reeb vector field of α_V is given by

(11-45)
$$R_{\alpha_{V}} = \frac{1}{1 + \|p\|^{2}} \left(a \partial_{\theta} + 2 \sum_{i,j} h^{ij} p_{i} \partial_{q_{j}} - \sum_{i,j,k} p_{i} p_{j} \partial_{k} h^{ij} \partial_{p_{k}} \right)$$

on $S^1 \times D^*\mathcal{U}$. (Here we follow the convention of using superscripts $(h^{ij}) = (h_{ij})^{-1}$ to denote the coefficients of the metric induced by h on T^*S^{n-1} .)

Proof A direct computation using the formulas

(11-46)
$$\alpha_V = \frac{1}{a} \left(1 - \sum_{i,j} p_i p_j h^{ij} \right) d\theta + \sum_i p_i dq_i,$$

(11-47)
$$d\alpha_V = -\frac{2}{a} \sum_{i,j} h^{ij} p_i \, dp_j \wedge d\theta - \frac{1}{a} \sum_{i,j,k} p_i p_j \partial_k h^{ij} \, dq_k \wedge d\theta + \sum_i dp_i \wedge dq_i$$

shows that $\alpha_V(R_{\alpha_V}) = 1$ and $d\alpha_V(R_{\alpha_V}, -) = 0$.

Geometry & Topology, Volume 28 (2024)

Lemma 11.11 Consider the map $\pi: Y_0 - V \to S^1$ given by $\pi(s, \theta, q, p) = \theta$. Then the pair (V, π) defines an open book decomposition of *Y*. Moreover, λ_0 is a Giroux form for the contact structure $\xi_0 = \ker \lambda_0$.

Proof It is clear that (V, π) defines an open book decomposition of *Y*. To verify that λ_0 is a Giroux form, observe by Lemma 11.10 that the Reeb vector field is transverse to the pages of π . The claim then follows by combining Definition 3.11 and Remark 3.12.

By Lemma 11.10, the map $\gamma_0: \mathbb{R}/\mathbb{Z} \to S^1 \times D^*\mathcal{U}$ given by the formula $\gamma_0(t) = (2\pi t, 0, 0)$ defines a simple Reeb orbit in Y_0 . Let γ_0^k denote its *k*-fold cover. There is an obvious trivialization τ_0 of $\xi|_{\gamma_0^k}$ given by

(11-48)
$$\tau_0 = \{\partial_{p_1}, \dots, \partial_{p_{n-1}}, \partial_{q_1}, \dots, \partial_{q_{n-1}}\}.$$

Let τ be the trivialization of $\xi|_{\gamma_0^k}$ defined as

(11-49) $\tau = \{\sin(2\pi kt)\partial_{q_1} + \cos(2\pi kt)\partial_{p_1}, \partial_{p_2}, \dots, \partial_{p_n}, \cos(2\pi kt)\partial_{q_1} - \sin(2\pi kt)\partial_{p_1}, \partial_{q_2}, \dots, \partial_{q_n}\}.$

Observe that τ extends to a disk spanning γ_0 in Y_0 .

Lemma 11.12 With respect to the trivialization τ_0 , the linearized Reeb flow along γ_0^k is given by the matrix

$$(11-50) \qquad \qquad \begin{pmatrix} 1 & 0 \\ 2t & 1 \end{pmatrix}$$

where each entry of this matrix should be viewed as an $(n-1) \times (n-1)$ diagonal matrix.

Proof Note that τ_0 can be extended to a trivialization $\tilde{\tau}_0$ of ker (α_V) over $S^1 \times D^* \mathcal{U}$, where

(11-51)
$$\widetilde{\tau}_0 = \left\{ \partial_{p_1}, \dots, \partial_{p_{n-1}}, \partial_{q_1} - \frac{ap_1}{1 - \|\boldsymbol{p}\|^2} \partial_{\theta}, \dots, \partial_{q_{n-1}} - \frac{ap_{n-1}}{1 - \|\boldsymbol{p}\|^2} \partial_{\theta} \right\}.$$

Using the formula for R_{α_V} given in Lemma 11.10, one can easily compute

(11-52)
$$\mathcal{L}_{\partial_{p_i}} R_{\alpha_V} |_{\boldsymbol{p}=\boldsymbol{0}, \boldsymbol{q}=\boldsymbol{0}} = 2\partial_{q_i},$$

(11-53)
$$\mathcal{L}_{\partial_{q_i} - (ap_i/(1 - \|\boldsymbol{p}\|^2)) \partial_{\theta}} R_{\alpha_V}|_{\boldsymbol{p} = 0, \boldsymbol{q} = 0} = 0.$$

Hence, the matrix A(t) representing the linearized Reeb flow $\xi_{\gamma_0^k(0)} \to \xi_{\gamma_0^k(t)}$ with respect to the trivialization τ_0 is given by

(11-54)
$$A(t) = \exp\left(t\begin{pmatrix} 0 & 0\\ 2 & 0 \end{pmatrix}\right) = \begin{pmatrix} 1 & 0\\ 2t & 1 \end{pmatrix}$$

where each entry should be interpreted as a multiple of the $(n-1) \times (n-1)$ identity matrix.

Geometry & Topology, Volume 28 (2024)

Corollary 11.13 The Robbin–Salamon index satisfies

(11-55)
$$\mu_{\rm RS}^{\tau_0}(\gamma_0^k) = \frac{1}{2}(n-1).$$

Hence,

(11-56)
$$\mu_{\rm RS}^{\tau}(\gamma_0^k) = \frac{1}{2}(n-1) + 2k.$$

Proof The first computation follows from [33, Proposition 54]; there is a sign change due to the fact that the matrix we are considering is the transpose of that considered in [33, Proposition 54], but the proof is entirely analogous. The second computation follows from the fact (see the proof of Lemma 57 in [33]) that the Robbin–Salamon index satisfies the so-called "loop property", ie given a path of symplectic matrices $\phi: [0, 1] \rightarrow \text{Sp}(2n, \mathbb{R})$ with $\phi(0) = \phi(1) = \text{id}$, and given a path $\psi: [0, 1] \rightarrow \text{Sp}(2n, \mathbb{R})$, we have

(11-57)
$$\mu_{\rm RS}(\phi\psi) = \mu_{\rm RS}(\psi) + 2\mu(\phi)$$

where ϕ is the Maslov index of the path.

By Lemma 11.10, $\mathcal{N} = \{ \|p\|=0 \}$ is preserved by the Reeb flow and is foliated by Reeb orbits in a Morse–Bott family.

Given $\epsilon > 0$ which will be fixed later, let $\mathcal{U}_{\epsilon} = \{ \|p\| < \epsilon \} \cap Y_0$. This is a neighborhood of \mathcal{N} , which we identify with $S^1 \times D_{\epsilon}^* S^{n-1}$ via the contactomorphism defined on page 94. Let $f : \mathcal{U}_{\epsilon} \to \mathbb{R}$ be the function corresponding under this identification to

(11-58)
$$S^1 \times D^*_{\epsilon} S^{n-1} \to \mathbb{R}, \quad (\theta, q, p) \mapsto \rho(||p||) g(q),$$

where g is a perfect Morse function on S^{n-1} and $\rho \colon \mathbb{R} \to [0, 1]$ is a smooth bump function with $\rho(x) = 1$ for x near 0 and $\rho(x) = 0$ for $x > \epsilon/2$; cf [6, Section 2.2].

Lemma 11.14 Fix P > 0. If ϵ is small enough, all closed Reeb orbits of (Y_0, λ_0) which are contained in $\mathcal{U}_{\epsilon} - \mathcal{N}$ have action at least P.

We now consider a perturbed contact form $\lambda_{\delta} := (1 + \delta f)\lambda_0$. Since *f* is compactly supported in \mathcal{U}_{ϵ} , the form λ_{δ} can be viewed as a contact form both on \mathcal{U}_{ϵ} and on Y_0 .

Lemma 11.15 Fix P > 0. If ϵ and δ are small enough, then there are exactly two simple Reeb orbits in U_{ϵ} with action $\langle P \rangle$. We label them γ_a and γ_b , and they correspond respectively to the minimum and maximum of f.

Proof Combine Lemma 11.14 with the argument of [6, Lemma 2.3].

Geometry & Topology, Volume 28 (2024)

100

Lemma 11.16 Let P > 0 be as in Lemma 11.15. After possibly further shrinking ϵ and δ , we may assume that any Reeb orbit of (Y_0, λ_{δ}) contained in U_{ϵ} and having Conley–Zehnder index (measured with respect to a trivialization which extends to a spanning disk) less than P/a is a multiple of γ_a or γ_b . In addition, we have

(11-59)
$$CZ^{\tau}(\gamma_a^k) = \mu_{RS}^{\tau}(\gamma_0^k) - \frac{1}{2}(n-1) + \operatorname{ind}_a(\delta f) = 2k,$$

(11-60) $CZ^{\tau}(\gamma_b^k) = \mu_{RS}^{\tau}(\gamma_0^k) - \frac{1}{2}(n-1) + \operatorname{ind}_b(\delta f) = (n-1) + 2k.$

Proof First of all, observe by Lemma 11.10 that the boundary of U_{ϵ} is preserved by the Reeb flow of λ_0 . It follows that the Reeb flow of λ_0 has "bounded return time", in the terminology of [6, Definition 2.5].

Next, it follows from (11-56) that the Robbin–Salamon index of any Reeb orbit γ contained in the Morse–Bott submanifold $\mathcal{N} = \{ \|p\| = 0 \} \subset Y_0$ satisfies

(11-61)
$$\mu_{\rm RS}(\gamma) = \frac{1}{2}(n-1) + 2\operatorname{wind}(\gamma) = \frac{1}{2}(n-1) + 2T_{\gamma}a,$$

where P_{γ} is the length of γ . It follows that these orbits satisfy "index positivity" (with constant 2/a), in the terminology of [6, Definition 2.6].

The first claim now follows from [6, Lemma 2.7]. The index computations follow by combining (11-56) with [6, Lemma 2.4]. \Box

We now put together the above results. For any integer N > 0, let us define

(11-62) $\Sigma_N^1 = \{k \in \mathbb{Z} \mid 0 < k < N, k \text{ even}\},\$

and let

(11-63)
$$\Sigma_N^2 = \{k \in \mathbb{Z} \mid k < N, \, k = n - 1 + 2j \text{ for } j \ge 1\}.$$

Proposition* 11.17 Given any N > 0, there exists A > 0 such that

(11-64)
$$CH_{k-(n-3)}^{U=0}(Y_0,\xi_0,V;\mathfrak{r}) = \widetilde{CH}_{k-(n-3)}(Y_0,\xi_0,V;\mathfrak{r})$$
$$= \begin{cases} \mathbb{Q} \oplus \mathbb{Q} & \text{if } k \in \Sigma_N^1 \cap \Sigma_N^2, \\ \mathbb{Q} & \text{if } k \in \Sigma_N^1 \cup \Sigma_N^2 - (\Sigma_N^1 \cap \Sigma_N^2), \\ 0 & \text{if } k \notin \Sigma_N^1 \cup \Sigma_N^2 \text{ and } k < N, \end{cases}$$

whenever a > A. (Recall from (11-13) that \mathfrak{r} depends on a > 0 and hence on N > 0.)

Proof According to Corollary 11.8, we may fix A > 0 large enough so that all Reeb orbits for (Y_0, λ_0) in the complement of $\mathcal{N} \subset Y_0$ have index at least N. We now choose ϵ and δ small enough so that the conclusions of Lemma 11.16 hold with P = N. Since f_{δ} is compactly supported in \mathcal{U}_{ϵ} , we find that the only Reeb orbits of (Y_0, λ_{δ}) having index less than N are multiples of γ_a and γ_b .

According to (11-59) and (11-60), it is now enough to check that the differential vanishes on the set

$$\Omega_N = \{\gamma_a^{k_a}, \gamma_b^{k_b} \mid CZ^{\tau}(\gamma_a^{k_a}) < N, CZ^{\tau}(\gamma_b^{k_b}) < N\}.$$

To see this, observe that for $\gamma \in \Omega_N$ we have

(11-65)
$$CZ^{\tau}(\gamma) = 2 \operatorname{wind}(\gamma) \mod (n-1)$$

Suppose that there exists a homotopy class β of curves of index 1 with $\hat{V} * \beta = 0$. Then the linking number of the positive puncture equals the sum of the linking numbers of the negative punctures. Hence, by (11-65), the index of the positive puncture equals the sum of the indices of the negative punctures mod (n-1). Since β has index 1, this means that $1 = 0 \mod (n-1)$. This is a contradiction since n > 2. \Box

Corollary* 11.18 Let N > 0 be as in Proposition* 11.17. Then for all integers k < N we have

(11-66)
$$\widetilde{CH}_{k-(n-3)}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V;\mathfrak{r}) = \widetilde{CH}_{k-(n-3)}(Y_0,\xi_0,V;\mathfrak{r}),$$

where the right-hand side was computed in Proposition* 11.17.

We now turn out attention to computing certain Legendrian invariants. Let $\Lambda \subset (Y_0, \xi_0)$ be defined as above; see (11-12). Recall that the relative symplectic filling $(\hat{W}_0, \hat{\lambda}_0, H)$ gives an augmentation $\tilde{\epsilon}_0: \tilde{\mathcal{A}}(Y_0, \xi_0, V; \mathfrak{r}) \to \mathbb{Q}$.

It follows from Corollary* 10.14 and Lemma 11.6 that $\widetilde{\mathcal{L}}^{\widetilde{\epsilon}_0}(Y_0, \xi_0, V, \Lambda; \mathfrak{r})$ is a $(\mathbb{Z} \times \mathbb{Z})$ -bigraded algebra with differential of bidegree (-1, 0), and that $\overline{HC}_{\bullet, \bullet}(\widetilde{\mathcal{L}}^{\widetilde{\epsilon}}(Y, \xi, V; \mathfrak{r}))$ is a $(\mathbb{Z} \times \mathbb{Z})$ -bigraded \mathbb{Q} -vector space.

We now have the following computation.

Proposition* 11.19 Given a positive integer $N \gg 1$, let

(11-67)
$$\bigoplus_{j \le N} \widetilde{\mathcal{L}}_{\bullet,j}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V,\Lambda;\mathfrak{r}) \subset \widetilde{\mathcal{L}}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V,\Lambda;\mathfrak{r})$$

be the bigraded submodule of elements of winding number at most *N*. Then this submodule can be generated by products of total winding number $\leq N$ of Reeb chords $\{a_k\}_{k \in \mathbb{N}_+}$ and $\{b_k\}_{k \in \mathbb{N}_+}$, where $|a_k| = 2k - 1$ and $|b_k| = n - 2 + 2k$. (Note that we do not say anything about the differential.)

Proof Since $(Y_0 - V, \lambda_0)$ is strictly contactomorphic to $(S^1 \times D^* S^{n-1}, \alpha_V)$, we have that $(Y_0 - V, \lambda_\delta)$ is strictly contactomorphic to $(S^1 \times D^* S^{n-1}, \alpha_\delta := (1 + \delta f) \alpha_V)$.

Recall that f depends on a positive real parameter $\epsilon > 0$, which can be taken to be arbitrarily small. Moreover, the restriction of f to the Legendrian $\Lambda = \{0\} \times S^{n-1}$ is equal to g, a Morse function with exactly two critical points: one minimum a and one maximum b.

Let *c* denote either *a* or *b*. As in Lemma 11.16, we let γ_c denote the simple Reeb chord (which is also a Reeb orbit) passing through *c*, and let γ_c^k denote its *k*-fold cover. Observe first of all that all Reeb chords for Λ are contained in \mathcal{U}_{ϵ} – this follows from the fact that *f* is compactly supported in \mathcal{U}_{ϵ} ; see page 100. Hence, as in Lemma 11.14, if we assume that $\epsilon > 0$ is small enough, then there exists P > 0large enough so that all Reeb chords of action greater than *P* have winding number greater than *N*. By a routine adaptation of Lemma 11.15 (or rather the proof of [6, Lemma 2.3]), one concludes that the only Reeb chords of winding number less than or equal to *N* are the γ_a^k and γ_b^k .

We can assume without loss of generality that there are normal coordinates $q = (q_1, \ldots, q_{n-1})$ defined in a neighborhood $U_c \subset \Lambda$ of c in which g is given by

(11-68)
$$g = g(c) + \epsilon \sum_{i=1}^{n-1} q_i^2,$$

where $\epsilon = 1$ if c = a, and $\epsilon = -1$ if c = b. The Reeb vector field of α_{δ} is given by

(11-69)
$$R_{\alpha_{\delta}} = \frac{1}{1+\delta f} R_{\alpha} + \frac{2\epsilon\delta}{(1+\delta f)^2} \sum_{i} \left(q_i - \frac{2p_i}{1+\|p\|^2} \sum_{j,k} h^{jk} p_k q_j \right) \partial_{p_i}$$

on $S^1 \times D^* U_c$ for $||\mathbf{p}||$ sufficiently small, is satisfying $\rho(||\mathbf{p}||) = 1$. We will now show that for every $k \ge 1$, the indices of γ_a^k and γ_b^k as Reeb chords are given by

(11-70)
$$CZ^+(\gamma_a^k) = 2k \text{ and } CZ^+(\gamma_b^k) = 2k + n - 1.$$

Hence, setting $a_k = \gamma_a^k$ and $b_k = \gamma_b^k$, we have

$$|a_k| = CZ^+(a_k) - 1 = 2k - 1,$$

 $|b_k| = CZ^+(b_k) - 1 = 2k + n - 2,$

as desired.

To compute $CZ^+(\gamma_c^k)$, we start by computing the linearized Reeb flow along γ_c^k with respect to the trivialization τ_0 ; see (11-48). We proceed as in Lemma 11.12: we have

(11-71)
$$\mathcal{L}_{\partial_{p_i}} R_{\alpha_\delta} \Big|_{\boldsymbol{p}=0, \boldsymbol{q}=0} = \frac{2}{1+\delta g(c)} \partial_{q_i},$$

(11-72)
$$\mathcal{L}_{\partial_{q_i} - (ap_i/(1 - \|\boldsymbol{p}\|^2)) \partial_{\theta}} R_{\alpha_{\delta}} \Big|_{\boldsymbol{p} = 0, \boldsymbol{q} = 0} = \epsilon \frac{2\delta}{(1 + \delta g(c))^2} \partial_{p_i}$$

Hence, the matrix A(t) representing the linearized Reeb flow $\xi_{\gamma_c^k(0)} \rightarrow \xi_{\gamma_c^k(t)}$ with respect to the trivialization τ_0 satisfies A'(t) = SA(t) with

(11-73)
$$S = \begin{pmatrix} 0 & \epsilon \frac{2\delta}{(1+\delta g(c))^2} \\ \frac{2}{1+\delta g(c)} & 0 \end{pmatrix},$$

where each entry should be interpreted as a multiple of the $(n-1) \times (n-1)$ identity matrix.

Setting $\mu = 2\delta/(1 + \delta g(c))^2$ and $\nu = 2/(1 + \delta g(c))$ for ease of notation, it follows that

(11-74)
$$A(t) = \exp(tS) = \begin{cases} \begin{pmatrix} \cosh(t\sqrt{\mu\nu}) & \sqrt{\mu/\nu}\sinh(t\sqrt{\mu\nu}) \\ \sqrt{\nu/\mu}\sinh(t\sqrt{\mu\nu}) & \cosh(t\sqrt{\mu\nu}) \end{pmatrix} & \text{if } \epsilon = 1, \\ \begin{pmatrix} \cos(t\sqrt{\mu\nu}) & -\mu/\nu\sin(t\sqrt{\mu\nu}) \\ \nu/\mu\sin(t\sqrt{\mu\nu}) & \cos(t\sqrt{\mu\nu}) \end{pmatrix} & \text{if } \epsilon = -1. \end{cases}$$

(Note that $\mu, \nu > 0$ if δ is sufficiently small.)

Let $L(t) \subset \xi_{\gamma_c^k(t)}$ be the path of Lagrangian subspaces obtained by applying the linearized Reeb flow to the tangent space $T_c \Lambda \subset \xi_c$ and let $\tilde{L}(t)$ be the loop obtained by closing up L(t) by a positive rotation. Since $T_c \Lambda$ is represented by

$$\begin{pmatrix} 0\\ I_{n-1} \end{pmatrix}$$

in the trivialization τ_0 , L(t) is represented by

$$A(t) \begin{pmatrix} 0 \\ I_{n-1} \end{pmatrix}.$$

In the two-dimensional case (ie n - 1 = 1), one can easily deduce (eg using the standard properties of the Maslov index stated in [52, Theorem 2.3.7]) that

(11-75)
$$\mu^{\tau_0}(\widetilde{L}(t)) = \begin{cases} 0 & \text{if } \epsilon = 1, \\ 1 & \text{if } \epsilon = -1. \end{cases}$$

In general, L(t) splits as a direct sum of n - 1 copies of the two-dimensional case, so the additivity property of the Maslov index [52, Theorem 2.3.7] implies that

(11-76)
$$\mu^{\tau_0}(\widetilde{L}(t)) = \begin{cases} 0 & \text{if } \epsilon = 1, \\ n-1 & \text{if } \epsilon = -1 \end{cases}$$

According to Definition 10.3 and the definition of the Maslov index [52, Theorem 2.3.7], we have $CZ^+(\gamma_c^k) = \mu^{\tau}(\Lambda_{\mathbb{C}}^{n-1}\tilde{\Lambda}) = \mu^{\tau}(\tilde{L}(t))$, where τ is a trivialization of the contact structure along γ_c^k which extends to a spanning disk. For example, we can take τ to be the trivialization defined in equation (11-49). The difference $\mu^{\tau}(\tilde{L}(t)) - \mu^{\tau_0}(\tilde{L}(t))$ is equal to twice the Maslov index of the loop of symplectic matrices relating τ and τ_0 , ie

(11-77)
$$\mu^{\tau}(\widetilde{L}(t)) - \mu^{\tau_0}(\widetilde{L}(t)) = 2\mu \begin{pmatrix} \cos(2\pi kt) & -\sin(2\pi kt) \\ \sin(2\pi kt) & \cos(2\pi kt) \end{pmatrix} = 2k.$$

It follows that

(11-78)
$$CZ^+(\gamma_a^k) = 2k \text{ and } CZ^+(\gamma_b^k) = 2k + n - 1$$

as desired.

It will be useful to record the following consequence of the above computation.

Geometry & Topology, Volume 28 (2024)

Corollary* 11.20 Suppose that $n \ge 4$ is even. Then we have

(11-79)
$$\operatorname{rk} \overline{HC}_{2n,2}(\widetilde{\mathcal{L}}^{\epsilon_0}(Y_0,\xi_0,V,\Lambda;\mathfrak{r})) = 1.$$

Proof Indeed, note that the generators described in Proposition* 11.19 satisfy $link(a_k) = link(b_k) = k$. It thus follows that

(11-80)
$$\overline{CC}_{2n-1,2}(\widetilde{\mathcal{L}}^{\epsilon}(Y_0,\xi_0,V;\mathfrak{r})) = \overline{CC}_{2n+1,2}(\widetilde{\mathcal{L}}^{\epsilon}(Y_0,\xi_0,V;\mathfrak{r})) = 0.$$

On the other hand, $\overline{CC}_{2n,2}(\widetilde{\mathcal{L}}^{\epsilon}(Y_0,\xi_0,V;\mathfrak{r}))$ is generated by the word b_1b_1 .

12 Applications to contact topology

12.1 Contact and Legendrian embeddings

We begin by introducing some standard definitions in the theory of contact and Legendrian embeddings.

Definition 12.1 Given a smooth manifold Y^{2n-1} , a *formal contact structure* (or *almost-contact structure*) is the data of a pair (η, ω) , where $\eta \subset TY$ is a codimension 1 distribution and $\omega \in \Omega^2(Y)$ is a 2-form whose restriction to η is nondegenerate. A formal contact structure is said to be *genuine* if it is induced by a contact structure.

If Y^{2n-1} is orientable, then a formal contact structure is the same thing as a lift of the classifying map $Y \to BSO(2n+1)$ to a map $Y \to B(U(n) \times id) = BU(n)$.

Definition 12.2 (see Definition 2.2 in [10]) Let $(Y^{2n-1}, \xi = \ker \alpha)$ be a contact manifold. Given a formal contact manifold (V^{2m-1}, η, ω) where $1 \le m \le n-1$, a formal (iso)contact embedding is a pair (f, F_s) where

- F_s is a fiberwise injective bundle map $TV \rightarrow TY$ defined for $s \in [0, 1]$,
- $f: V \to Y$ is a smooth map and $df = F_0$, and
- F_1 defines a fiberwise conformally symplectic map $(\eta, \omega) \rightarrow (\xi, d\alpha)$.

Observe that the above properties are independent of the choice of contact form α .

Two formal contact embeddings $i_0, i_1 : (V, \zeta, \omega) \to (Y, \xi)$ are said to be formally isotopic if they can be connected by a family $\{i_t\}_{t \in [0,1]}$ of formal contact embeddings.

A (genuine) contact embedding $(V, \zeta) \to (Y, \xi)$ is simply a smooth embedding $\phi: V \to Y$ such that $\phi_*(\zeta) = \xi_{\phi(V)}$. In particular, every contact embedding induces a formal contact embedding by taking $F_s = F_0 = df$.

Definition 12.3 (see Definition 2.1 in [10]) Let (Y^{2n-1}, ξ) be a contact manifold. Given a smooth *n*-dimensional manifold Λ , a formal Legendrian embedding is a pair (f, F_s) where

- F_s is a fiberwise injective bundle map $TV \rightarrow TY$ defined for $s \in [0, 1]$,
- $df = F_0$, and
- $\operatorname{im}(F_1) \subset \xi$.

Two formal Legendrian embeddings are said to be formally isotopic if they can be connected by a family of Legendrian embeddings. A (genuine) Legendrian embedding $\Lambda \rightarrow (Y, \xi)$ is a smooth embedding $\phi: \Lambda \rightarrow Y$ such that $d\phi(T\Lambda) \subset \xi \subset TY$. In particular, a Legendrian embedding canonically induces a formal Legendrian embedding.

We now review some foundational facts about loose Legendrians. Recall that a Legendrian A in a (possibly noncompact) contact manifold (Y, ξ) of dimension at least five is defined to be loose if it admits a loose chart. For concreteness, we adopt as our definition of a loose chart the one given in [16, Section 7.7].

Loose Legendrians satisfy the following h-principle due to Murphy [55, Theorem 1.2]: given a pair of loose Legendrian embeddings $f_0, f_1 \colon \Lambda \to (Y, \xi)$ which are formally isotopic, then f_0 and f_1 are genuinely isotopic, is isotopic through Legendrian embeddings.

Given an arbitrary Legendrian submanifold Λ_0 in a contact manifold (Y_0, ξ_0) of dimension at least five, one can perform a local modification, called *stabilization*, which makes Λ_0 loose without changing the formal isotopy class of the tautological embedding $\Lambda_0 \xrightarrow{id} \Lambda_0$. This modification can be realized in multiple essentially equivalent ways. In this paper, we will take as our definition of stabilization any construction which satisfies the properties stated in the following lemma.

Lemma 12.4 Given a Legendrian submanifold $\Lambda \subset (Y_0, \xi_0)$ and an open set $U \subset Y_0$ such that $U \cap \Lambda_0$ is nonempty, there exists a Lagrangian embedding $f_1 \colon \Lambda_0 \to Y$ which is formally isotopic to the tautological embedding $\Lambda_0 \xrightarrow{\text{id}} \Lambda_0$ via a family of formal Legendrian embeddings $\{(f_t, F_s^t)\}_{t \in [0,1]}$ which are independent of t on $\Lambda_0 \cap (Y - U)$. We put $\Lambda := f_1(\Lambda_0)$ and say that Λ is the **stabilization** of Λ_0 inside U.

Proof To construct Λ , we follow the procedure described in [16, Section 7.4]. As the reader may verify, this construction can be assumed to happen entirely inside a suitably chosen Darboux chart $\mathcal{U} \subset U$. In addition, the construction depends on the choice of a function f; using that Y has dimension at least five, we may (and do) assume that $\chi(\{f \ge 1\}) = 0$. To construct the formal isotopy, we simply follow the proof of [16, Proposition 7.23] (using the assumption that $\chi(\{f \ge 1\}) = 0$). The argument there is entirely local, so that the isotopy can be assumed to be fixed outside of \mathcal{U} (and in particular, outside of U).
12.2 Embeddings into overtwisted contact manifolds

We begin with the following proposition.

Proposition 12.5 Suppose that (Y, ξ) is an overtwisted contact manifold and let

(12-1) $i: (V, \zeta) \to (Y, \xi)$

be a formal contact embedding. Then there exists an open subset $\Omega \subset Y$ such that $Y - \overline{\Omega}$ is overtwisted, and a **genuine** contact embedding

(12-2) $j: (V, \xi) \to \Omega \subset (Y, \xi)$

such that *i* and *j* are formally contact isotopic in (Y, ξ) .

Proof We will assume for simplicity that *V* is connected, but the proof can easily be generalized. Let $D_{ot} \subset (Y, \xi)$ be an overtwisted disk. Let f_t be a family of formal contact embeddings such that f_0 is the underlying smooth map induced by *i*, and $Im(f_1) \cap D_{ot} = \emptyset$. Let $\Omega \subset Y$ be a connected open subset such that $Im(f_1) \subset \Omega \subset \overline{\Omega} \subset Y - D_{ot}$. According to [5, Proposition 3.8], we can assume by choosing Ω large enough that (Ω, ξ) is overtwisted.

For purely algebrotopological reasons, there exists a family ξ_t of formal contact structures on Y with the following properties:

- $\xi_0 = \xi$,
- ξ_t is constant in the complement of $\overline{\Omega}$, and
- ξ₁ is a genuine contact structure in a neighborhood V ⊂ Ω of Im(f₁), and f₁ is a genuine contact embedding with respect to ξ₁.

Since ξ_1 is genuine on $\mathcal{V} \cup (Y - \overline{\Omega})$, it follows from the relative h-principle for overtwisted contact structures [5, Theorem 1.2] that ξ_1 is homotopic to a genuine overtwisted contact structure through a homotopy fixed on $\mathcal{V} \cup (Y - \overline{\Omega})$. Thus we may as well assume in the third property above that ξ_1 is genuine everywhere.

Since (Ω, ξ) is overtwisted, it follows from [5, Theorem 1.2] that there exists a homotopy $\tilde{\xi}_t$ of genuine contact structures such that $\tilde{\xi}_0 = \xi_0 = \xi$, $\tilde{\xi}_1 = \xi_1$ and $\tilde{\xi}_t$ is independent of t on $Y - \overline{\Omega}$.

By Gray's theorem, there is an ambient isotopy $\psi_t: Y \to Y$ which is fixed on $Y - \overline{\Omega}$ and has the property that $\psi_t^* \tilde{\xi}_t = \tilde{\xi}_0 = \xi_0$. The composition $\psi_1^{-1} \circ f_1$ is in the same class of formal contact embeddings as f_1 , and gives the desired genuine embedding.

We now describe a procedure for constructing pairs of codimension 2 contact embeddings in overtwisted manifolds which are formally isotopic but fail to be isotopic as contact embeddings.

Construction 12.6 Let (Y_0^{2n-1}, ξ_0) be a closed, overtwisted contact manifold and let (Y_0, B, π) be an open book decomposition which supports ξ_0 . Let $i_0: (B, (\xi_0)_B) \to (Y_0, \xi_0)$ be the tautological embedding of the binding and let $j_0: (B, (\xi_0)_B) \to (Y_0, \xi_0)$ be a contact embedding with overtwisted complement which is formally isotopic to i_0 (the existence of such an embedding follows from Proposition 12.5). Let $D_{\text{ot}} \subset Y_0$ be an overtwisted disk which is disjoint from $j_0(B)$.

Choose an open subset $\mathcal{U} \subset Y_0$ whose closure is disjoint from $i_0(B) \cup j_0(B) \cup D_{\text{ot}}$, and such that i_0 and j_0 are formally isotopic in the complement of $\overline{\mathcal{U}}$. Now let (Y, ξ) be obtained by attaching handles of arbitrary index along isotropic submanifolds contained inside \mathcal{U} ; see Construction 10.15. We let $(\hat{X}, \hat{\lambda})$ denote the resulting Weinstein cobordism with positive end (Y, ξ) and negative end (Y_0, ξ) .

Observe that (Y, ξ) is still overtwisted and that i_0 and j_0 can also be viewed as codimension 2 contact embeddings into (Y, ξ) . We denote these latter embeddings by $i, j : (B, (\xi_0)_B) \to (Y, \xi)$. By construction, the embeddings i and j are formally isotopic.

Theorem 12.7 The embeddings *i* and *j* which arise from Construction 12.6 are not genuinely isotopic. In fact, *i* is not genuinely isotopic to any reparametrization of *j* in the source, meaning that the codimension 2 submanifolds $(i(B), \xi_{i(B)}), (j(B), \xi_{j(B)})$ are not contact isotopic.

Proof According to Corollary 8.8, the cobordism $(\hat{X}, \hat{\lambda})$ induces a map of unital \mathbb{Q} -algebras

$$\widetilde{CH}_{\bullet}(Y,\xi,B;\mathfrak{r}) \to \widetilde{CH}_{\bullet}(Y_0,\xi_0,B;\mathfrak{r})$$

for any element $\mathfrak{r} \in \mathfrak{R}(Y_0, \xi_0, B) \equiv \mathfrak{R}(Y, \xi, B)$. Moreover, Theorem 11.3 implies that $\widetilde{CH}_{\bullet}(Y_0, \xi_0, B; \mathfrak{r}) \neq 0$ for appropriate $\mathfrak{r} \in \mathfrak{R}(Y, \xi, B)$. It follows that $\widetilde{CH}_{\bullet}(Y, \xi, B; \mathfrak{r}) \neq 0$.

If we assume that $(i(B), \xi_{i(B)}), (j(B), \xi_{j(B)})$ are isotopic as codimension 2 contact submanifolds, then $\widetilde{CH}_{\bullet}(Y, \xi, B; \mathfrak{r}) = \widetilde{CH}_{\bullet}(Y, \xi, j(B); \mathfrak{r}')$ for some datum $\mathfrak{r}' \in \mathfrak{R}(Y, \xi, j(B))$. On the other hand, observe that $(Y - j(B), \xi)$ is overtwisted by construction. Hence Theorem 11.1 implies $\widetilde{CH}_{\bullet}(Y, \xi, j(B); \mathfrak{r}') = 0$. \Box

Example 12.8 By a well-known theorem of Giroux and Mohsen [29, Theorem 7.3.5], any contact manifold (Y, ξ) admits an open book decomposition (Y, B, π) which supports ξ . Hence Construction 12.6 and Theorem 12.7 can be applied to any overtwisted contact manifold.

We also consider the following modification of Construction 12.6.

Construction 12.9 Let (Y_0^{2n-1}, ξ_0) be a closed, overtwisted contact manifold and let (Y_0, B, π) be an open book decomposition which supports ξ_0 . Suppose that there exists a Legendrian submanifold $\Lambda \subset Y_0$ such that $B = \tau(\Lambda)$ is a contact pushoff of Λ . Let $D_{ot} \subset Y_0$ be an overtwisted disk.

Let $\mathcal{U}_1 \subset Y_0 - B - D_{\text{ot}}$ be an open ball which intersects Λ . Let $\Lambda' \subset (Y_0, \xi_0)$ be obtained by stabilizing Λ inside \mathcal{U}_1 ; see Lemma 12.4. Let \mathcal{U}_2 be the union of \mathcal{U}_1 with a tubular neighborhood of Λ . Let $V' = \tau(\Lambda') \subset \mathcal{U}_2$ be a choice of contact pushoff for Λ' .

Let $i_0: (B, (\xi_0)_B) \to (Y_0, \xi_0)$ be the tautological embedding. By [10, Lemma 3.4], i_0 is formally isotopic to some codimension 2 contact embedding $j_0: (B, (\xi_0)_B) \to (Y_0, \xi_0)$, where $j_0(B) = B'$. Choose such a formal isotopy and let $\mathcal{T} \subset Y_0$ be its trace.

Let $\mathcal{U} \subset Y_0$ be an open set whose closure is disjoint from $\mathcal{T} \cup \mathcal{U}_2 \cup D_{\text{ot}}$. As in Construction 12.6, let (Y, ξ) be obtained by attaching handles of arbitrary index along some collection of isotropics inside \mathcal{U} . Let $(\hat{X}, \hat{\lambda})$ denote the resulting Weinstein cobordism with positive end (Y, ξ) and negative end (Y_0, ξ) .

It follows from our choice of \mathcal{U} that (Y, ξ) is overtwisted and that Λ , Λ' , V and V' can be viewed as submanifolds of (Y, ξ) . It also follows that Λ' is the stabilization of Λ as submanifolds of (Y, ξ) , and that V (resp. V') is the contact pushoff of Λ (resp. Λ').

Corollary 12.10 The submanifolds (V, ξ_V) and $(V', \xi_{V'})$ are not isotopic through codimension 2 contact submanifolds. Hence the Legendrian submanifolds $\Lambda, \Lambda' \subset (Y, \xi)$ are not isotopic through Legendrian submanifolds.

Proof The proof of the first statement is identical to that of Theorem 12.7. The second statement follows from the fact that *V* and *V'* are, respectively, the contact pushoff of Λ and Λ' .

Example 12.11 Let $(Y_0, \xi_0) = \text{obd}(T^*S^{n-1}, \tau^{-1})$, where τ^{-1} is a left-handed Dehn twist. Note by [12, Theorem 1.1] that (Y_0, ξ_0) is overtwisted — in fact, (Y, ξ) is contactomorphic to (S^{2n-1}, ξ_{ot}) . Let $P = T^*S^{n-1} \subset Y_0$ be a page of the open book and let $\Lambda \subset (Y_0, \xi_0)$ be the Legendrian which corresponds to the zero section of P. Then the binding of the open book decomposition is also a contact pushoff of Λ . We may therefore apply Construction 12.9 to this data.

Remark 12.12 Consider the special case of Constructions 12.6 and 12.9 where \mathcal{U} is empty, ie one does not attach any handles. In this case, Theorem 12.7 and Corollary 12.10 are essentially equivalent to the statement that the binding of an open book decomposition is tight, ie must intersect any overtwisted disk. This statement was proved in dimension 3 by Etnyre and Vela-Vick [28, Theorem 1.2]; the higher-dimensional case follows from work of Klukas [45, Corollary 3], who proved (following an outline of Wendl [67, Remark 4.1]) the stronger fact that a local filling obstruction (such as an overtwisted disk) in a closed contact manifold must intersect the binding of any supporting open book.

12.3 Contact embeddings into the standard contact sphere

In this section, we exhibit examples of pairs of codimension 2 contact embeddings into tight contact manifolds which are formally isotopic but are not isotopic through genuine contact embeddings. We begin with the following construction.

Construction 12.13 Let (Y_0^{2n-1}, ξ_0) be a contact manifold for $n \ge 3$. Let $V \subset (Y_0, \xi_0)$ be a codimension 2 contact submanifold and let $\Lambda \subset (Y_0, \xi_0)$ be a loose Legendrian such that $\Lambda \cap V = \emptyset$.

Choose an open ball $\mathcal{U} \subset Y_0$ such that $(\mathcal{U}, \mathcal{U} \cap \Lambda)$ is a loose chart for Λ . Next, choose an open ball $\mathcal{O} \subset Y_0 - \mathcal{V} - \mathcal{U}$. (By definition of a loose chart, $\mathcal{U} \cap \Lambda$ is a proper subset of Λ , so it is clear that such choices exist.)

Let Λ' be obtained by stabilizing Λ inside \mathcal{O} . It follows from Lemma 12.4 that Λ and Λ' are formally isotopic via a formal isotopy fixed outside of \mathcal{O} .

According to Lemma 12.14 below, we can (and do) fix a contactomorphism $f: (Y_0, \xi_0) \to (Y_0, \xi_0)$ with the following properties:

- (1) f is isotopic to the identity,
- (2) $f(\Lambda) = \Lambda'$, and
- (3) the tautological contact embedding $i'_0: (V, (\xi_0)_V) \to (Y_0 \Lambda', \xi_0)$ is formally isotopic to the embedding $i'_1 := f \circ i'_0: (V, (\xi_0)_V) \to (Y_0 \Lambda', \xi_0)$. (We emphasize here that the formal isotopy is contained in the open contact manifold $(Y_0 \Lambda', \xi_0)$.)

Finally, let (Y, ξ) be obtained by attaching a Weinstein *n*-handle along $\Lambda' \subset (Y_0, \xi_0)$ as described in Construction 10.15. We assume without loss of generality that the attaching region $\Lambda' \subset \mathcal{V}$ disjoint from *V* and f(V), and that i'_0 and i'_1 are formally isotopic in $Y_0 - \mathcal{V}$. We let $\iota: Y_0 - \mathcal{V} \hookrightarrow Y$ be the canonical inclusion.

Let

(12-3)
$$i_0 = \iota \circ i'_0 \colon (V, \xi_V) \to (Y, \xi)$$

be the tautological contact embedding and define

(12-4)
$$i_1 := \iota \circ i'_1 : (V, \xi_V) \to (Y, \xi).$$

It is an immediate consequence of (3) and our choice of V that i_0 and i_1 are formally isotopic.

Lemma 12.14 With the notation of Construction 12.13, there exists a contactomorphism $f: (Y_0, \xi_0) \rightarrow (Y_0, \xi_0)$ satisfying the properties (1)–(3) stated in Construction 12.13.

Proof Recall that \mathcal{U} is disjoint from \mathcal{O} . Recall also that $(\mathcal{U}, \mathcal{U} \cap \Lambda)$ is a loose chart for Λ , which means in particular that \mathcal{U} deformation retracts onto $\mathcal{U} \cap \Lambda$. Using these two facts, it is not hard to verify that there exists a family of *formal* contact embeddings $j_t : (V, (\xi_0)_V) \to (Y_0, \xi_0)$ for $t \in [0, 1]$, with the following properties:

- $j_0 = i'_0$,
- $j_t(V)$ is disjoint from $\mathcal{O} \cup \Lambda$ for all $t \in [0, 1]$, and
- $j_1(V)$ is disjoint from $\mathcal{U} \cup \mathcal{O} \cup \Lambda$.

Geometry & Topology, Volume 28 (2024)

By the *h*-principle for loose Legendrian embeddings [55, Theorem 1.2], there exists a global contact isotopy ϕ_t for $t \in [0, 1]$ such that $\phi_0 = \text{Id}$ and $\phi_1(\Lambda) = \Lambda'$. By the Legendrian isotopy extension theorem [29, Theorem 2.6.2], this isotopy can be assumed to be compactly supported and constant in a neighborhood W of $j_1(V)$, where W is disjoint from $U \cup O \cup \Lambda$.

Let $f := \phi_1$ and observe that f satisfies (1)–(2). Observe that $f \circ j_t$ defines a formal contact isotopy from $f \circ i'_0 = i'_1$ to j_1 in the complement of $\Lambda' = f(\Lambda)$. Since i'_0 is formally isotopic to j_1 in the complement of $\Lambda' \subset \Lambda \cup \mathcal{O}$, we find that f satisfies (3).

It will be useful to record the following basic observation, which is a consequence of the fact stated in Definition 2.1 that an isotopy of contactomorphisms induces a Hamiltonian isotopy of symplectizations.

Lemma 12.15 (cf Definition 2.1) Let $(\hat{X}, \hat{\lambda})$ be a relative cobordism from (Y^+, λ^+) to (Y^-, λ^-) . Given contactomorphisms $f^{\pm}: (Y^{\pm}, \lambda^{\pm}) \to (Y^{\pm}, \lambda^{\pm})$ which are contact isotopic to the identity, there is a symplectomorphism $F: (\hat{X}, \hat{\lambda}) \to (\hat{X}, \hat{\lambda})$ which agrees near infinity with the lifts $\tilde{f}^{\pm}: (SY^{\pm}, \lambda_{Y^{\pm}}) \to (SY^{\pm}, \lambda_{Y^{\pm}})$.

Let us now return to the geometric setup considered in Section 11.3. In particular, we let

(12-5)
$$(\widehat{W}_0, \widehat{\lambda}^a) := \left(D^2 \times T^* S^{n-1}, \frac{1}{a} r^2 d\theta + \lambda_{\text{std}}\right),$$

where a > 0 is a constant which will be fixed later; see (11-7).

We let (W_0, λ^a) , $(Y_0, \xi_0 = \ker \lambda_0)$, $V \subset Y_0$, $\Lambda \subset Y_0$ and $H = \{0\} \times T^* S^{n-1}$ be defined as in Section 11.3. Note that Λ is a loose Legendrian according to [11, Proposition 2.9].

Construction 12.13 applied to the above data produces a contactomorphism $f: (Y_0, \xi_0) \to (Y_0, \xi_0)$, a Liouville domain (X, λ) with positive contact boundary $(Y, \xi = \ker \lambda)$, and a pair of formally isotopic contact embeddings $i_0, i_1: (V, \xi_V) \to (Y, \xi)$.

Let $\mathfrak{r} = (\alpha, \tau, a) \in \mathfrak{R}(Y_0, \xi_0, V)$, where $\alpha := (\lambda_0)|_V$ and the trivialization τ is unique since $H^1(V; \mathbb{Z}) = 0$; see Corollary 12.17. We let $\mathfrak{r}' = ((i'_1)_*\alpha, \tau, a) \in \mathfrak{R}(Y_0, \xi_0, V)$, where i'_1 is defined as in Construction 12.13 and τ is again unique. Since the surgery resulting from Construction 12.13 is away from V and $i'_1(V)$, we may identify $\mathfrak{R}(Y, \xi, V) = \mathfrak{R}(Y_0, \xi_0, V)$ and $\mathfrak{R}(Y, \xi, i_1(V)) = \mathfrak{R}(Y_0, \xi_0, i'_1(V))$.

As in Section 11.3, let $e_0: \mathbb{R} \times Y_0 \to (\hat{W}_0, \hat{\lambda}_0, H)$ be the canonical marking furnished by the Liouville flow and let $\tilde{\epsilon}_0: \tilde{\mathcal{A}}(Y_0, \xi_0, V; \mathfrak{r}) \to \mathbb{Q}$ be the associated augmentation.

By Lemma 12.15, there is a symplectomorphism $\psi: (\hat{W}_0, \hat{\lambda}_a) \to (\hat{W}_0, \hat{\lambda}_a)$ which coincides near infinity with the lift $\tilde{f}: SY_0 \to SY_0$. Let $H' \subset \hat{W}_0$ be a symplectic submanifold which is cylindrical at infinity and coincides with the symplectization of $f(V) = i'_1(V)$ on $[0, \infty) \times Y_0$. Such a surface can be constructed by taking the backwards Liouville flow of $\psi(H)$.

Let $\tilde{\epsilon}'_0: \tilde{\mathcal{A}}(Y_0, \xi_0, i'_1(V); \mathfrak{r}') \to \mathbb{Q}$ be the augmentation induced by the relative symplectic cobordism $((\hat{W}_0, \hat{\lambda}_a, H'), \tilde{\epsilon}_0).$

Observe that $(Y_0, \xi_0, V) \in \mathcal{G}$ and hence also $(Y_0, f_*\xi_0, f(V)) = (Y_0, \xi_0, i'_1(V)) \in \mathcal{G}$; see Definition 3.13. The following lemma shows that we also have $(Y, \xi, i_1(V)) \in \mathcal{G}$.

Lemma 12.16 Up to contactomorphism, $(Y, \xi) = ob(T^*S^{n-1}, \tau_S) = (S^{2n-1}, \xi_{std})$, where τ_S denotes a right-handed Dehn twist. Moreover, the first contactomorphism can be assumed to take $i_1(V)$ to the binding of the open book decomposition $ob(T^*S^{n-1}, \tau_S)$.

Proof By construction, there is an open book decomposition of (Y_0, ξ_0) agreeing (up to contactomorphism) with $ob(T^*S^{n-1}, id)$, such that $i_1(V)$ is the binding and Λ' is the zero section of a page. Note now that attaching a handle to the zero section of a page of $(Y_0, \xi_0) = ob(T^*S^{n-1}, id)$ simply changes the monodromy of the open book by a positive Dehn twist [46, Theorem 4.6]. Hence, $i_1(V)$ is the binding of $ob(T^*S^{n-1}, \tau_S) = (S^{2n-1}, \xi_{std})$.

Corollary 12.17 The manifolds Y_0 , W_0 , Y and W have vanishing first and second homology and cohomology with \mathbb{Z} -coefficients. Hence the same is also true for the pairs (W_0, Y_0) and (W, Y). Finally, we have $H^1(V; \mathbb{Z}) = 0$.

Proof By construction, W is obtained by attaching a handle of index n to W_0 . The union of the core and co-core of this handle has codimension n. Hence, for $i \le n-2$, we have $H_i(W_0; \mathbb{Z}) = H_i(W; \mathbb{Z})$ and $H_i(Y_0; \mathbb{Z}) = H_i(Y; \mathbb{Z})$. Now, W_0 is homotopy equivalent to S^{n-1} by definition, while Y is homeomorphic to S^{2n-1} by Lemma 12.16. Since $n \ge 4$, it follows that Y_0 , W_0 , Y and W have vanishing first and second homology. The vanishing of cohomology in the same degrees follows by the universal coefficients theorem for cohomology. The vanishing of $H^1(V; \mathbb{Z})$ was proved in Lemma 11.6.

As a result of Corollary 12.17, Definition 8.12, Lemma 8.13 and Definition* 10.12, the invariants considered in the proof of Theorem* 12.18 below, as well as the maps between these invariants, are all canonically $(\mathbb{Z} \times \mathbb{Z})$ -bigraded.

Theorem* 12.18 For $n \ge 4$ even and $a \gg 0$ large enough, the contact embeddings $i_0, i_1: (V, \xi_V) \rightarrow (Y, \xi) = (S^{2n-1}, \xi_{std})$ are not isotopic through contact embeddings.

Proof We suppose for contradiction that i_0 and i_1 are isotopic through contact embeddings. This means that there exists a contactomorphism $g: (Y, \xi, V) \rightarrow (Y, \xi, i_1(V))$. It follows by Lemma 12.16 that $(Y, \xi, V) \in \mathcal{G}$.

According to Corollary* 11.18 (and the description of the generators in Proposition* 11.17), we may (and do) fix $a \gg 0$ large enough that

(12-6)
$$\widetilde{CH}_{k-(n-3),2}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V;\mathfrak{r}) = \begin{cases} \mathbb{Z} & \text{if } k = 4,4+(n-1) \\ 0 & \text{otherwise.} \end{cases}$$

In particular, since $n \ge 4$, we have

(12-7)
$$\widetilde{CH}_{2n-(n-3),2}^{\tilde{\epsilon}_0}(Y_0,\xi_0,V;\mathfrak{r}) = \widetilde{CH}_{2n+1-(n-3),2}^{\tilde{\epsilon}_0}(Y_0,\xi_0,V;\mathfrak{r}) = 0.$$

Homological invariants of codimension 2 contact submanifolds

Since $(Y_0, \xi_0, V) \in \mathcal{G}$, it follows by Corollary* 9.14 that

(12-8)
$$\widetilde{CH}_{\bullet,\bullet}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V;\mathfrak{r}) = \widetilde{CH}_{\bullet,\bullet}^{\widetilde{\epsilon}_0'}(Y_0,\xi_0,i_1(V);\mathfrak{r}').$$

Similarly, it follows by Definition/Assumption* 10.6 that

(12-9)
$$\widetilde{\mathcal{L}}_{\bullet,\bullet}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V,\Lambda;\mathfrak{r}) = \widetilde{\mathcal{L}}_{\bullet,\bullet}^{\widetilde{\epsilon}_0'}(Y_0,\xi_0,i_1(V),\Lambda';\mathfrak{r}').$$

Let $e : \mathbb{R} \times Y \to \hat{W}$ be the canonical marking and consider the resulting relative filling $((\hat{W}, \hat{\lambda}, H), e)$. Let $\tilde{\epsilon} : \tilde{\mathcal{A}}(Y, \xi, V; \mathfrak{r}) \to \mathbb{Q}$ be the induced augmentation. Let $\phi : (\hat{W}, \hat{\lambda}, H) \to (\hat{W}, \hat{\lambda}, H)$ be a symplectomorphism which agrees with the lift of g near infinity. Let $\tilde{\epsilon}' : \tilde{\mathcal{A}}(Y, \xi, i_1(V); \mathfrak{r}') \to \mathbb{Q}$ be the augmentation induced by $((\hat{W}, \hat{\lambda}, H'), e)$. Then according to Lemma* 9.13 and Corollary* 9.14, we have

(12-10)
$$\widetilde{CH}_{\bullet,\bullet}^{\widetilde{\epsilon}}(Y,\xi,V;\mathfrak{r}) = \widetilde{CH}_{\bullet,\bullet}^{\widetilde{\epsilon}'}(Y,\xi,i_1(V);\mathfrak{r}').$$

It then follows by Lemma* 12.19 that $\widetilde{CH}_{2n-(n-3),2}^{\tilde{\epsilon}}(Y,\xi,V;\mathfrak{r}) \neq 0$. Hence Lemma* 12.20 implies that

(12-11)
$$\widetilde{CH}_{2n-(n-3),2}^{\tilde{\varepsilon}_0}(Y_0,\xi_0,V;\mathfrak{r}) \neq 0$$

This contradicts (12-7).

Lemma* 12.19 We have

(12-12)
$$\widetilde{CH}_{2n-(n-3),2}^{\widetilde{\epsilon}'}(Y,\xi,i_1(V);\mathfrak{r}')\neq 0.$$

Proof On the one hand, Corollary* 11.20 and (12-9) imply that

(12-13)
$$\operatorname{rk} \overline{HC}_{2n,2}(\widetilde{\mathcal{L}}^{\widetilde{\epsilon}'_0}(Y_0,\xi_0,i_1(V),\Lambda';\mathfrak{r}')) = 1.$$

On the other hand, by (12-7) and (12-8), we have that

(12-14)
$$\widetilde{CH}_{2n-(n-3),2}^{\widetilde{\epsilon}'_0}(Y_0,\xi_0,i_1(V);\mathfrak{r}') = \widetilde{CH}_{2n+1-(n-3),2}^{\widetilde{\epsilon}'_0}(Y_0,\xi_0,i_1(V);\mathfrak{r}') = 0.$$

It then follows by Theorem/Assumption* 10.16 and Remark 10.18 that

(12-15)
$$\widetilde{CH}_{2n-(n-3),2}^{\widetilde{\epsilon}'}(Y,\xi,i_1(V);\mathfrak{r}') \simeq \overline{HC}_{2n,2}(\widetilde{\mathcal{L}}^{\widetilde{\epsilon}'_0}(Y_0,\xi_0,i_1(V),\Lambda';\mathfrak{r}')).$$

This proves the claim.

Lemma* 12.20 The natural map

(12-16)
$$\widetilde{CH}_{2n-(n-3),2}^{\widetilde{\epsilon}}(Y,\xi,V;\mathfrak{r}) \to \widetilde{CH}_{2n-(n-3),2}^{\widetilde{\epsilon}_0}(Y_0,\xi_0,V;\mathfrak{r})$$

is injective.

Proof Since Λ' is loose in $Y_0 - V$, it follows by Proposition* 11.2 and Lemma 9.6 that

(12-17)
$$\overline{HC}_{2k}(\mathcal{L}^{\epsilon_0}(Y_0,\xi_0,V,\Lambda';\mathfrak{r})) = 0$$

for all $k \in \mathbb{Z}$. The lemma thus follows from Theorem/Assumption* 10.16 and Remark 10.18.

Geometry & Topology, Volume 28 (2024)

Remark 12.21 One can slightly tweak Construction 12.13 so that Λ and Λ' are disjoint and $\Lambda \cup \Lambda'$ is a loose Legendrian link. One can then upgrade Lemma 12.14 to require that $f(\Lambda) = \Lambda'$ and $f(\Lambda') = \Lambda$ in (2) of Construction 12.13. In particular, this means that Λ' is a stabilization of Λ , and Λ is a stabilization of Λ' .

Let us apply this tweaked construction to the setup considered in Construction 12.13, where $(Y_0, \xi_0) = ob(T^*S^{n-1}, id)$ for $n \ge 4$, $V \subset (Y_0, \xi_0)$ is the binding and $\Lambda \subset (Y_0, \xi_0)$ is the zero section of a page. It is well known that the zero section of a page in $ob(T^*S^{n-1}, id)$ is the standard Legendrian unknot. Hence Lemma 12.16 implies that $i_1(V)$ is the pushoff of the standard unknot. By construction, it now also follows that $i_0(V)$ is the contact pushoff of a stabilization of the unknot. Theorem* 12.18 thus provides an alternative way to distinguish (for $n \ge 4$ even) the basic example considered by Casals and Etnyre in [10, Section 5].

12.4 Relative symplectic and Lagrangian cobordisms

In this final section, we exhibit some constraints on relative symplectic and Lagrangian cobordisms. In particular, we prove the results which were advertised in Section 1.5 of the introduction.

Proof of Theorem 1.9 Suppose for contradiction that such a relative symplectic cobordism exists. According to Theorem 11.3, we have $\widetilde{CH}_{\bullet}(Y, \xi, V; \mathfrak{r}) \neq 0$ for some $\mathfrak{r} = (\alpha, \tau, r) \in \mathfrak{R}(Y, \xi, V)$, which we now view as fixed. According to Theorem 11.1, we also have $\widetilde{CH}_{\bullet}(Y, \xi, V'; \mathfrak{r}') = 0$ for all $\mathfrak{r}' = (\alpha', \tau', r') \in \mathfrak{R}(Y, \xi, V')$. Choose \mathfrak{r}' depending on our previous choice of \mathfrak{r} so that $r' \geq e^{\mathcal{E}((H,\lambda_H)_{\alpha}^{\alpha'})}r$. Then Corollary 8.8 along with our topological assumptions on H furnishes a unital Q-algebra map

(12-18)
$$\widetilde{CH}_{\bullet}(Y,\xi,V';\mathfrak{r}') \to \widetilde{CH}_{\bullet}(Y,\xi,V;\mathfrak{r})$$

This gives the desired contradiction.

In contrast to Theorem 1.9, we expect that one could prove that V is concordant to V' by adapting work of Eliashberg and Murphy [26], but we do not pursue this here. Note that we could also have proved Theorem 1.9 using the full invariant $CH_{\bullet}(-; -)$ instead of its reduced counterpart.

For Lagrangian cobordisms, we have the following result.

Proposition 12.22 Let (Y, ξ) be a contact manifold. Let Λ and Λ' be Legendrian knots such that $H^1(\tau(\Lambda'); \mathbb{Z}) = H^2(\tau(\Lambda'); \mathbb{Z}) = 0$. Suppose that (SY, λ_Y, L) is a Lagrangian concordance from Λ' to Λ . Given $\mathfrak{r} = (\alpha, \tau, r) \in \mathfrak{R}(Y, \xi, \tau(\Lambda))$, there is a map of \mathbb{Q} -algebras

(12-19)
$$\widetilde{CH}_{\bullet}(Y,\xi,\tau(\Lambda');\mathfrak{r}') \to \widetilde{CH}_{\bullet}(Y,\xi,\tau(\Lambda);\mathfrak{r})$$

for some $\mathfrak{r}' = (\alpha', \tau', r') \in \mathfrak{R}(Y, \xi, \tau(\Lambda'))$. (A similar statement holds for the nonreduced invariants $CH_{\bullet}(-)$.)

Geometry & Topology, Volume 28 (2024)

114

Proof Observe that the trivial Lagrangian cobordism $L = \mathbb{R} \times \Lambda \subset \mathbb{R} \times Y$ admits a "symplectic push-off" $\tau(L) := \mathbb{R} \times \tau(\Lambda) \subset \mathbb{R} \times Y$. It follows by the Lagrangian neighborhood theorem that any Lagrangian concordance (SY, λ_Y, L) also admits a symplectic push-off (SY, λ_Y, H) , which is a relative symplectic cobordism from $(Y, \xi, \tau(\Lambda'))$ to $(Y, \xi, \tau(\Lambda))$. Fix α' arbitrarily and choose r' so that $r' \ge e^{\mathcal{E}((H, \lambda_H)_{\alpha}^{\alpha'})}r$ (note that τ' is unique since $H^1(\tau(\Lambda'); \mathbb{Z}) = 0$). The claim now follows from Corollary 8.8.

Proof of Theorem 1.10 Suppose for contradiction that Λ' is concordant to Λ . As in the proof of Corollary 12.10, we have

(12-20)
$$\widetilde{CH}_{\bullet}(Y,\xi,\tau(\Lambda);\mathfrak{r})\neq 0$$

for a suitable choice $\mathfrak{r} \in \mathfrak{R}(Y, \xi, \tau(\Lambda))$. On the other hand, we have $\widetilde{CH}_{\bullet}(Y, \xi, \tau(\Lambda'); \mathfrak{r}') = 0$ for all $\mathfrak{r}' \in \mathfrak{R}(Y, \xi, \tau(\Lambda'))$. This gives a contradiction in view of Proposition 12.22.

We end by exhibiting examples of Lagrangian cobordisms which cannot be displaced from a codimension 2 symplectic submanifold.

Construction 12.23 Let $(Y_0, \xi_0) = \text{obd}(T^*S^{n-1}, \text{id})$ for $n \ge 3$. Let $V \subset (Y_0, \xi_0)$ be the binding and let $\Lambda \subset (Y_0, \xi_0)$ be the zero section of a page, which is a loose Legendrian by [11, Proposition 2.9]. Let $\mathcal{U}_1 \subset Y_0 - V$ be a small ball which intersects Λ in an (n-1)-ball and let Λ' be obtained by stabilizing Λ inside \mathcal{U}_1 .

Let $\mathcal{U}_2 \subset Y_0 - (V \cup \Lambda \cup \mathcal{U}_1)$ be an open subdomain. Let (Y, ξ) be obtained by attaching a sequence of handles along isotropics contained in \mathcal{U}_2 . Observe that V, Λ and Λ' can be viewed as submanifolds of both Y_0 and Y; we will not distinguish these embeddings in our notation. We let $(\hat{X}, \hat{\lambda}, \hat{V})$ be the associated relative symplectic cobordism from (Y, ξ, V) to (Y_0, ξ_0, V) .

Proof of Theorem* 1.12 We can identify $\Re(Y_0, \xi_0, V) = \Re(Y, \xi, V)$. According to Theorem* 11.5, $\mathcal{L}(Y_0, \xi_0, V, \Lambda; \mathfrak{r}) \neq 0$ for suitable $\mathfrak{r} \in \Re(Y_0, \xi_0, V)$. In contrast, $\mathcal{L}(Y, \xi, V, \Lambda'; \mathfrak{r}) = 0$ by Proposition* 11.2, since Λ' is loose in Y - V by construction. By Proposition* 10.7, the existence of a concordance from Λ' to Λ which doesn't intersect \widehat{V} would imply that there is a unital map of $\mathbb{Q}[U]$ -algebras

(12-21)
$$\mathcal{L}(Y,\xi,V,\Lambda';\mathfrak{r}) \to \mathcal{L}(Y_0,\xi_0,V,\Lambda;\mathfrak{r}).$$

This gives a contradiction.

We remark that Construction 12.23 could be generalized in various directions without affecting the validity of Theorem* 1.12, but we do not pursue this here.

Geometry & Topology, Volume 28 (2024)

Appendix Connected sums of almost-contact manifolds

Let G be a connected⁹ subgroup of SO(n). An *almost* G-*structure* on a smooth oriented manifold M is a homotopy class of maps $M \to BG$ lifting the classifying map of the tangent bundle of M:



An almost G-manifold is a manifold equipped with an almost G-structure.

Example A.1 Taking $G = U(n) \subset SO(2n)$ yields the usual notion of an almost complex manifold. Almost-contact manifolds correspond to $G = U(n) \subset SO(2n + 1)$.

If the *n*-dimensional sphere S^n admits an almost *G*-structure, a result of Kahn [43, Theorem 2] implies that for any two *n*-dimensional almost *G*-manifolds *M* and *N*, there exists an almost *G*-structure on M # N which is compatible with the given ones on *M* and *N* in the complement of the disks used to form the connected sum. In general, this structure is not unique, so the connected sum M # N is not well-defined as an almost *G*-manifold. However, we will show in Section A.1 that a choice of almost *G*-structure β on S^n induces a canonical almost *G*-structure on the connected sum of any two almost *G*-manifolds. Hence, any such β gives rise to a connected sum operation $(M, N) \mapsto M \#_{\beta} N$ for almost *G*-manifolds. Moreover, the set of almost *G*-structures on S^n forms a group under this operation (with identity β). In Section A.2, we will show that this group acts on the set of almost *G*-structures of any *n*-dimensional almost *G*-manifold.

A.1 Connected sums of almost *G*-manifolds

Let S^n be the unit sphere in \mathbb{R}^{n+1} , equipped with its standard orientation as the boundary of the unit disk D^{n+1} . We will write its points as pairs $(x, z) \in \mathbb{R}^n \times \mathbb{R}$. Define

$$D_{-} = \{(x, z) \in S^{n} \mid z < \frac{1}{2}\}, \quad D_{+} = \{(x, z) \in S^{n} \mid z > -\frac{1}{2}\}, \quad A = D_{-} \cap D_{+}, \quad C_{\pm} = D_{\pm} \setminus A.$$

Note that D_- and D_+ are open disks, C_- and C_+ are closed disks, A is an open annulus, and $S^n = D_- \cup D_+ = C_- \sqcup A \sqcup C_+$.

Let *M* and *N* be smooth connected oriented *n*-dimensional manifolds. Choose orientation-preserving embeddings $i_+: \overline{D}_+ \to M$ and $i_-: \overline{D}_- \to N$. We define the connected sum $M \# N = M \#_{i_+,i_-} N$ by

$$M # N = (M \setminus i_+(C_+) \sqcup N \setminus i_-(C_-))/\sim,$$

where $i_+(x) \sim i_-(x)$ for every $x \in A$.

⁹This assumption is used in the proof of Proposition A.6.

We will now explain how to construct a classifying map for the tangent bundle of M # N. The following elementary fact from topology will be useful.

Proposition A.2 Let $i: A \to X$ be a cofibration. Assume A is contractible. Then for any connected space Y and continuous maps $F: X \to Y$ and $f: A \to Y$, there exists a map $F': X \to Y$ homotopic to F such that $F' \circ i = f$.

Let $\tau_S: S^n \to BSO(n)$ be a classifying map for TS^n . Let τ_M and τ_N be classifying maps for TM and TN such that $\tau_M \circ i_+ = \tau_S|_{\overline{D}_+}$ and $\tau_N \circ i_- = \tau_S|_{\overline{D}_-}$ (such maps always exist by Proposition A.2). Define $\tau_{M\#N}$ to be the unique map $M \# N \to BSO(n)$ which coincides with τ_M on $M \setminus i_+(C_+)$ and with τ_N on $N \setminus i_-(C_-)$.

Proposition A.3 The map $\tau_{M\#N}$ is a classifying map for T(M # N).

We start with an easy topological lemma.

Lemma A.4 Let *E* be an oriented vector bundle over a manifold M^n and let $i: D^n \to M^n$ be an embedding. Then any automorphism of i^*E can be extended to an automorphism of *E*.

Proof Let ϕ be an automorphism of i^*E . Since D^n is contractible, we can trivialize i^*E and think of ϕ as a map $D^n \to \operatorname{GL}^+(n)$. Clearly $\phi|_{\partial D^n}$ is nullhomotopic, and since $\operatorname{GL}^+(n)$ is connected, we can extend ϕ to a map $\tilde{\phi}: D_2^n \to \operatorname{GL}^+(n)$ which is constant with value $\operatorname{Id} \in \operatorname{GL}^+(n)$ near ∂D_2^n . Using a tubular neighborhood of $i(\partial D^n) \subset M$, we can also extend i to an embedding $\tilde{i}: D_2^n \to M^n$. Then $\tilde{\phi}$ gives us an automorphism of \tilde{i}^*E which is equal to the identity over a neighborhood of $\partial D_2^n \subset D_2^n$ and hence extends trivially to an automorphism of E.

Proof of Proposition A.3 Let $\tilde{\gamma}_n \to BSO(n)$ be the universal bundle over BSO(n). We want to show that the tangent bundle T(M # N) of the connected sum is isomorphic to $\tau^*_{M \# N} \tilde{\gamma}_n$.

T(M # N) is obtained by gluing $T(M \setminus i_+(C_+))$ and $T(N \setminus i_-(C_-))$ along the maps $di_+: TA \to T(M \setminus i_+(C_+))$ and $di_-: TA \to T(N \setminus i_-(C_-))$. Because of our assumption that $\tau_M \circ i_+ = \tau_S|_{\overline{D}_+}$ and $\tau_N \circ i_- = \tau_S|_{\overline{D}_-}$, we have that $\tau^*_{M\#N}\widetilde{\gamma}_n$ is obtained by gluing $(\tau_M|_{M \setminus i_+(C_+)})^*\widetilde{\gamma}_n$ and $(\tau_N|_{N \setminus i_-(C_-)})^*\widetilde{\gamma}_n$ along bundle maps $(\tau_S|_A)^*\widetilde{\gamma}_n \to (\tau_M|_{M \setminus i_+(C_+)})^*\widetilde{\gamma}_n$ and $(\tau_S|_A)^*\widetilde{\gamma}_n \to (\tau_M|_{M \setminus i_+(C_+)})^*\widetilde{\gamma}_n$ covering $i_+: A \to M \setminus i_+(C_+)$ and $i_-: A \to N \setminus i_-(C_-)$, respectively. Hence, in order to show that T(M # N) is isomorphic to $\tau^*_{M\#N}\widetilde{\gamma}_n$, it suffices to construct a commutative diagram



where the horizontal arrows are bundle isomorphisms.

Start by fixing an isomorphism $\phi: TS^n \to \tau_S^* \tilde{\gamma}_n$, and let the middle arrow of the diagram be the restriction of ϕ to *TA*. To get the top and bottom arrows, it suffices to find bundle isomorphisms completing the following commutative squares:

$$\begin{array}{cccc} TM & ---- & \tau_{M}^{*}\widetilde{\gamma}_{n} & T\overline{D}_{-} & \stackrel{\phi}{\longrightarrow} (\tau_{S}|_{\overline{D}_{-}})^{*}\widetilde{\gamma}_{n} \\ di_{+} \uparrow & \uparrow & di_{-} \downarrow & \downarrow \\ T\overline{D}_{+} & \stackrel{\phi}{\longrightarrow} (\tau_{S}|_{\overline{D}_{+}})^{*}\widetilde{\gamma}_{n} & TN & ---- & \tau_{N}^{*}\widetilde{\gamma}_{n} \end{array}$$

This is possible by Lemma A.4.

We are now ready to define the connected sum of two almost G-manifolds.

Definition A.5 Suppose that S^n admits an almost *G*-structure, and fix a choice β of one such structure. Let β_M and β_N be almost *G*-structures on *M* and *N* respectively. We define an almost *G*-structure $\beta_M \#_\beta \beta_N$ on M # N as follows.

Pick maps $\tilde{\tau}_S : S^n \to BG$, $\tilde{\tau}_M : M \to BG$ and $\tilde{\tau}_N : N \to BG$ representing β , β_M and β_N , respectively. By Proposition A.2, we can assume that $\tilde{\tau}_M \circ i_+ = \tilde{\tau}_S|_{\overline{D}_+}$ and $\tilde{\tau}_N \circ i_- = \tilde{\tau}_S|_{\overline{D}_-}$. Hence, there is a unique map

$$\widetilde{\tau}_{M\#N} = \widetilde{\tau}_M \#_{\widetilde{\tau}_S} \widetilde{\tau}_N \colon M \# N \to BG$$

which coincides with $\tilde{\tau}_M$ on $M \setminus i_+(C_+)$ and with $\tilde{\tau}_N$ on $N \setminus i_-(C_-)$. By Proposition A.3, the composition

$$M \# N \xrightarrow{\widetilde{\tau}_{M \# N}} BG \longrightarrow BSO(n)$$

is a classifying map for T(M # N). Hence, we can (and do) define $\beta_M \#_\beta \beta_N$ to be the homotopy class of $\tilde{\tau}_{M \# N}$.

Proposition A.6 The almost *G*-structure $\beta_M \#_\beta \beta_N$ is well-defined, is independent of the choice of $\tilde{\tau}_S$, $\tilde{\tau}_M$ and $\tilde{\tau}_N$.

Proof Let $\tilde{\tau}_{S}^{j}$, $\tilde{\tau}_{M}^{j}$ and $\tilde{\tau}_{N}^{j}$ represent β , β_{M} and β_{N} , respectively, where $j \in \{0, 1\}$. As in Definition A.5, we assume that $\tilde{\tau}_{M}^{j} \circ i_{+} = \tilde{\tau}_{S}^{j}|_{\overline{D}_{+}}$ and $\tilde{\tau}_{N}^{j} \circ i_{-} = \tilde{\tau}_{S}^{j}|_{\overline{D}_{-}}$.

Fix a homotopy $\tilde{\tau}_S^t$ between $\tilde{\tau}_S^0$ and $\tilde{\tau}_S^1$. We will show that there exist homotopies $\tilde{\tau}_M^t$ and $\tilde{\tau}_N^t$ such that $\tilde{\tau}_M^t \circ i_+ = \tilde{\tau}_S^t |_{\overline{D}_+}$ and $\tilde{\tau}_N^t \circ i_- = \tilde{\tau}_S^t |_{\overline{D}_-}$. This implies that $\tilde{\tau}_{M\#N}^0$ is homotopic to $\tilde{\tau}_{M\#N}^1$ and hence that $\beta_M \#_\beta \beta_N$ is well-defined.

Pick an arbitrary homotopy $h: M \times I \to BG$ between $\tilde{\tau}^0_M$ and $\tilde{\tau}^1_M$ and define a map

$$g: \overline{D}_+ \times \partial I^2 \to BG$$

Geometry & Topology, Volume 28 (2024)

by $g(x,t,0) = h(i_+(x),t)$, $g(x,0,s) = \tilde{\tau}_S^0(x)$, $g(x,1,s) = \tilde{\tau}_S^1(x)$ and $g(x,t,1) = \tilde{\tau}_S^t(x)$. We can extend g to a map $\hat{g} : \overline{D}_+ \times I^2 \to BG$ since the obstruction to doing so lies in

$$H^{2}(\overline{D}_{+} \times I^{2}, \overline{D}_{+} \times \partial I^{2}; \pi_{1}(BG)) \cong \pi_{1}(BG) \cong \pi_{0}(G),$$

which is trivial by our assumption that G is connected.

Let

$$f: \left(M \times (I \times \{0\} \cup \{0\} \times I \cup \{1\} \times I)\right) \cup \left(i_+(\overline{D}_+) \times I^2\right) \to BG$$

be defined by

- $f(x,t,0) = h(x,t), f(x,0,s) = \tilde{\tau}_{M}^{0}(x) \text{ and } f(x,1,s) = \tilde{\tau}_{M}^{1}(x) \text{ for } x \in M,$
- $f(x,t,s) = \hat{g}(i_+^{-1}(x),t,s)$ for $x \in i_+(\overline{D}_+)$.

Since $i_+: \overline{D}_+ \to M$ is a cofibration, the domain of f is a retract of $M \times I^2$. We can therefore extend f to a map $\widehat{f}: M \times I^2 \to BG$. Restricting \widehat{f} to $M \times I \times \{1\}$ then provides us with a homotopy $\widetilde{\tau}_M^t$ such that $\widetilde{\tau}_M^t \circ i_+ = \widetilde{\tau}_S^t|_{\overline{D}_+}$.

The same argument gives us a homotopy $\tilde{\tau}_N^t$ such that $\tilde{\tau}_N^t \circ i_- = \tilde{\tau}_S^t |_{\overline{D}_-}$, so this completes the proof. \Box

Definition A.7 If $M = (M, \beta_M)$ and $N = (N, \beta_N)$ are almost *G*-manifolds, their connected sum (with respect to β) is the almost *G*-manifold $M \#_{\beta} N := (M \# N, \beta_M \#_{\beta} \beta_N)$.

As usual, there is an ambiguity in the notation $M \#_{\beta} N$ since the construction of the connected sum involves a choice of embeddings $i_+: \overline{D}_+ \to M$, $i_-: \overline{D}_- \to N$. However, the result is independent of these choices up to the appropriate notion of equivalence, as one would expect.

Definition A.8 A diffeomorphism of almost *G*-manifolds $f: (M, \beta_M) \to (N, \beta_N)$ consists of a smooth diffeomorphism $f: M \to N$ such that $f^*\beta_N = \beta_M$.

Proposition A.9 The connected sum $M \#_{\beta} N$ is well-defined up to diffeomorphism of almost Gmanifolds. More precisely, given any orientation-preserving embeddings $i_+, j_+: \overline{D}_+ \to M$ and $i_-, j_-: \overline{D}_- \to N$, there exists an orientation-preserving diffeomorphism $\phi: M \#_{i_+,i_-} N \to M \#_{j_+,j_-} N$ such that

$$\phi^*(\beta_M \#_{j_+,j_-,\beta} \beta_N) = \beta_M \#_{i_+,i_-,\beta} \beta_N$$

for any almost G-structures β_M and β_N on M and N.

Proof This follows from the isotopy extension theorem as in the smooth case.

Remark A.10 (connected sums of contact manifolds) Suppose that (M_1, α_1) and (M_2, α_2) are contact manifolds. Then one can form the connected sum $(M_1 \# M_2, \alpha_1 \# \alpha_2)$, which is also a contact manifold. The connected sum is obtained by choosing Darboux balls in M_1, M_2 and connecting them by a "neck". This operation can also be understood as a contact surgery along a 0-sphere. We refer to [8, Section 6.2] and [46, Section 3] for more details.

Let $\beta \in \mathfrak{alm}_{U(n)}(S^{2n-1})$ be the almost-contact structure induced by the standard contact structure on the sphere. Then the operation of connected sum (with respect to β) of almost U(n-1)-manifolds defined in Definition A.7, and the operation of connected sum of contact manifolds described above, commute with the forgetful map from contact manifolds to almost-contact manifolds. This can be shown as in the proof of Proposition A.3, replacing BSO(n) with BU(n-1).

The main properties of the connected sum in the smooth case have analogs for almost G-manifolds:

Proposition A.11 Let M, N and P be connected almost G-manifolds of dimension n, and let β and β' be almost G-structures on S^n . Then:

- (1) $M #_{\beta} (S^n, \beta) \cong M.$
- (2) $(M \#_{\beta} N) \#_{\beta'} P \cong M \#_{\beta} (N \#_{\beta'} P).$

Proof If one takes $i_-: \overline{D}_- \to S^n$ to be the inclusion map, then the connected sum $M \# S^n$ is canonically identified with M as a smooth manifold. If one further takes $\tilde{\tau}_N = \tilde{\tau}_S$ in Definition A.5, then this identification is compatible with the almost *G*-structures on $M \# S^n$ and M. This proves that $M \#_\beta(S^n, \beta) \cong M$.

To prove that $(M \#_{\beta} N) \#_{\beta} P \cong M \#_{\beta} (N \#_{\beta} P)$, choose embeddings $i_{+} : \overline{D}_{+} \to M$, $i_{-} : \overline{D}_{-} \to N$, $j_{+} : \overline{D}_{+} \to N$ and $j_{-} : \overline{D}_{-} \to P$. If we assume that i_{-} and j_{+} have disjoint images, then i_{-} induces an embedding $\overline{D}_{-} \to N \#_{j_{+},j_{-}} P$, j_{+} induces an embedding $\overline{D}_{+} \to M \#_{i_{+},i_{-}} N$, and there is a canonical identification of smooth manifolds

$$(M \#_{i_{+},i_{-}} N) \#_{i_{+},i_{-}} P \cong M \#_{i_{+},i_{-}} (N \#_{i_{+},i_{-}} P).$$

Moreover, this identification is compatible with the almost G-structures in the sense that for any choice of maps $\tilde{\tau}_S$, $\tilde{\tau}'_S$, $\tilde{\tau}'_S$, $\tilde{\tau}_N$, $\tilde{\tau}_N$ and $\tilde{\tau}_P$, the following diagram commutes:

$$(M \#_{i_{+},i_{-}} N) \#_{j_{+},j_{-}} P \xrightarrow{(\tilde{\tau}_{M} \#_{\tilde{\tau}_{S}} \tilde{\tau}_{N}) \#_{\tilde{\tau}'_{S}} \tilde{\tau}_{P}} \downarrow \cong \xrightarrow{\tilde{\tau}_{M} \#_{\tilde{\tau}_{S}} (\tilde{\tau}_{N} \#_{\tilde{\tau}'_{S}} \tilde{\tau}_{P})} BG$$
$$M \#_{i_{+},i_{-}} (N \#_{j_{+},j_{-}} P) \Box$$

A.2 The group of almost *G*-structures on the sphere

We will denote the set of almost G-structures on a manifold M by $\mathfrak{alm}_G(M)$. More generally, if $A \subset M$ is a closed subset and β_0 is an almost G-structure on some open neighborhood of A, then $\mathfrak{alm}_G(M, A; \beta_0)$ will denote the set of almost G-structures on M which agree with β_0 near A.

In this section, we will show that $\#_{\beta}$ is a group operation on $\mathfrak{alm}_G(S^n)$, with β as identity element. The resulting group will be denoted by $\mathfrak{alm}_G^{\beta}(S^n)$. We will then show that $\mathfrak{alm}_G^{\beta}(S^n)$ acts on $\mathfrak{alm}_G(M)$, and more generally on $\mathfrak{alm}_G(M, A; \beta_0)$ if $M \setminus A$ is connected.

Proposition A.12 Given any $\beta_1 \in \mathfrak{alm}_G(S^n)$, there exists a $\beta_2 \in \mathfrak{alm}_G(S^n)$ such that $\beta_1 \#_\beta \beta_2 = \beta$.

Proof Recall the decomposition $S^n = C_- \cup A \cup C_+$ introduced at the beginning of Section A.1. We will use the notation $\langle \tau^-, \tau^A, \tau^+ \rangle$ to denote the unique (assuming it exists) map $S^n \to BG$ which coincides with the given maps $\tau^-: C_- \to BG$, $\tau^A: \overline{A} \to BG$ and $\tau^+: C_+ \to BG$ on C_- , \overline{A} and C_+ , respectively.

Let $\tilde{\tau}_S = \langle \tau_S^-, \tau_S^A, \tau_S^+ \rangle$ be a representative for β . Given β_1 and β_2 in $\mathfrak{alm}_G(S^n)$, we can choose representatives of the form $\langle \tau_1^-, \tau_S^A, \tau_S^+ \rangle$ and $\langle \tau_S^-, \tau_S^A, \tau_2^+ \rangle$ by Proposition A.2. Then $\beta_1 \#_\beta \beta_2$ is represented by $\langle \tau_1^-, \tau_S^A, \tau_2^+ \rangle$. Hence, all we need to show is that for any $\tau_1^-: C_- \to BG$, there exists $\tau_2^+: C_+ \to BG$ such that $\langle \tau_1^-, \tau_S^A, \tau_2^+ \rangle$ is homotopic to $\tilde{\tau}_S$. This again follows from Proposition A.2.

Corollary A.13 We have that $(\mathfrak{alm}_G(S^n), \#_\beta)$ is a group with identity β .

Proof This follows from Propositions A.11 and A.12.

(

Remark A.14 The group $(\mathfrak{alm}_G(S^n), \#_\beta)$ is independent of β up to isomorphism. Indeed, given any $x, y, \beta, \beta' \in \mathfrak{alm}_G(S^n)$, it follows from Proposition A.11 that

$$(x \#_{\beta} \beta') \#_{\beta'} (y \#_{\beta} \beta') = (x \#_{\beta} (\beta' \#_{\beta'} y)) \#_{\beta} \beta' = (x \#_{\beta} y) \#_{\beta} \beta',$$

which implies that the map

$$(\mathfrak{alm}_G(S^n), \#_\beta) \to (\mathfrak{alm}_G(S^n), \#_{\beta'}), \quad x \mapsto x \#_\beta \beta',$$

is a group isomorphism.

Given orientation-preserving embeddings $i_+: \overline{D}_+ \to M$ and $i_-: \overline{D}_- \to S^n$, the results of Section A.1 give us a well-defined map

$$\mathfrak{alm}_G(M) \times \mathfrak{alm}_G^\beta(S^n) \to \mathfrak{alm}_G(M \#_{i_+,i_-} S^n).$$

For the remainder of this section, we will take i_- to be the inclusion map $\overline{D}_- \hookrightarrow S^n$. Then $M \#_{i_+,i_-} S^n$ is canonically identified with M (regardless of what i_+ is) and we get a map

(A-1)
$$\mathfrak{alm}_G(M) \times \mathfrak{alm}_G^\beta(S^n) \to \mathfrak{alm}_G(M).$$

By Proposition A.11, this is a group action. Note that the diffeomorphism $\phi: M \to M$ appearing in the statement of Proposition A.9 (applied to $N = S^n$) can be chosen to be isotopic to the identity, which implies that the map (A-1) is independent of i_+ .

If we assume that the image of the embedding $i_+: \overline{D}_+ \to M$ is disjoint from A, then it follows directly from Definition A.5 that the subset $\mathfrak{alm}_G(M, A; \beta_0) \subset \mathfrak{alm}_G(M)$ is invariant under the map (A-1). If $M \setminus A$ is connected, then the resulting action on $\mathfrak{alm}_G(M, A; \beta_0)$ doesn't depend on the choice of i_+ .

References

- R Abraham, *Bumpy metrics*, from "Global analysis" (S-s Chern, S Smale, editors), Proc. Sympos. Pure Math. XIV, Amer. Math. Soc., Providence, RI (1970) 1–3 MR Zbl
- [2] P Albers, B Bramham, C Wendl, On nonseparating contact hypersurfaces in symplectic 4-manifolds, Algebr. Geom. Topol. 10 (2010) 697–737 MR Zbl
- [3] **M R R Alves**, **A Pirnapasov**, *Reeb orbits that force topological entropy*, Ergodic Theory Dynam. Systems 42 (2022) 3025–3068 MR Zbl
- [4] E Bao, K Honda, Definition of cylindrical contact homology in dimension three, J. Topol. 11 (2018) 1002–1053 MR Zbl
- [5] MS Borman, Y Eliashberg, E Murphy, Existence and classification of overtwisted contact structures in all dimensions, Acta Math. 215 (2015) 281–361 MR Zbl
- [6] F Bourgeois, A Morse-Bott approach to contact homology, PhD thesis, Stanford University (2002) Zbl Available at https://www.proquest.com/docview/305591502
- F Bourgeois, T Ekholm, Y Eliashberg, Effect of Legendrian surgery, Geom. Topol. 16 (2012) 301–389 MR Zbl
- [8] F Bourgeois, O van Koert, Contact homology of left-handed stabilizations and plumbing of open books, Commun. Contemp. Math. 12 (2010) 223–263 MR Zbl
- [9] J Bowden, Two-dimensional foliations on four-manifolds, PhD thesis, Ludwig Maximilian University of Munich (2010) Available at https://edoc.ub.uni-muenchen.de/12551/1/Bowden_Jonathan.pdf
- [10] R Casals, J B Etnyre, Non-simplicity of isocontact embeddings in all higher dimensions, Geom. Funct. Anal. 30 (2020) 1–33 MR Zbl
- [11] R Casals, E Murphy, Legendrian fronts for affine varieties, Duke Math. J. 168 (2019) 225–323 MR Zbl
- [12] R Casals, E Murphy, F Presas, Geometric criteria for overtwistedness, J. Amer. Math. Soc. 32 (2019) 563–604 MR Zbl
- [13] R Casals, D M Pancholi, F Presas, The Legendrian Whitney trick, Geom. Topol. 25 (2021) 3229–3256 MR Zbl
- [14] B Chantraine, Lagrangian concordance of Legendrian knots, Algebr. Geom. Topol. 10 (2010) 63–85 MR Zbl
- [15] B Chantraine, G Dimitroglou Rizell, P Ghiggini, R Golovko, Floer theory for Lagrangian cobordisms, J. Differential Geom. 114 (2020) 393–465 MR Zbl

- [16] K Cieliebak, Y Eliashberg, From Stein to Weinstein and back: symplectic geometry of affine complex manifolds, American Mathematical Society Colloquium Publications 59, Amer. Math. Soc., Providence, RI (2012) MR Zbl
- [17] A Connes, Noncommutative differential geometry, Inst. Hautes Études Sci. Publ. Math. 62 (1985) 257–360 MR Zbl
- [18] M Dörner, H Geiges, K Zehmisch, Open books and the Weinstein conjecture, Q. J. Math. 65 (2014) 869–885 MR Zbl
- [19] T Ekholm, Rational SFT, linearized Legendrian contact homology, and Lagrangian Floer cohomology, from "Perspectives in analysis, geometry, and topology" (I Itenberg, B Jöricke, M Passare, editors), Progr. Math. 296, Springer (2012) 109–145 MR Zbl
- [20] **T Ekholm**, Holomorphic curves for Legendrian surgery, preprint (2019) arXiv 1906.07228
- [21] T Ekholm, J Etnyre, L Ng, M Sullivan, Filtrations on the knot contact homology of transverse knots, Math. Ann. 355 (2013) 1561–1591 MR Zbl
- [22] T Ekholm, J B Etnyre, L Ng, M G Sullivan, Knot contact homology, Geom. Topol. 17 (2013) 975–1112 MR Zbl
- [23] T Ekholm, K Honda, T Kálmán, Legendrian knots and exact Lagrangian cobordisms, J. Eur. Math. Soc. 18 (2016) 2627–2689 MR Zbl
- [24] Y Eliashberg, A Givental, H Hofer, Introduction to symplectic field theory, from "Visions in mathematics: GAFA 2000 special volume, II" (N Alon, J Bourgain, A Connes, M Gromov, V Milman, editors), Birkhäuser, Boston, MA (2000) 560–673 MR Zbl
- [25] Y Eliashberg, N Mishachev, Introduction to the h-principle, Graduate Studies in Mathematics 48, Amer. Math. Soc., Providence, RI (2002) MR Zbl
- [26] Y Eliashberg, E Murphy, Lagrangian caps, Geom. Funct. Anal. 23 (2013) 1483–1514 MR Zbl
- [27] JB Etnyre, LL Ng, Legendrian contact homology in R³, from "Surveys in differential geometry 2020: surveys in 3–manifold topology and geometry" (I Agol, D Gabai, editors), Surv. Differ. Geom. 25, International, Boston, MA ([2022] ©2022) 103–161 MR Zbl
- [28] JB Etnyre, DS Vela-Vick, Torsion and open book decompositions, Int. Math. Res. Not. 2010 (2010) 4385–4398 MR Zbl
- [29] H Geiges, An introduction to contact topology, Cambridge Studies in Advanced Mathematics 109, Cambridge Univ. Press (2008) MR Zbl
- [30] M Gerstenhaber, C W Wilkerson, On the deformation of rings and algebras, V: Deformation of differential graded algebras, from "Higher homotopy structures in topology and mathematical physics" (J McCleary, editor), Contemp. Math. 227, Amer. Math. Soc., Providence, RI (1999) 89–101 MR Zbl
- [31] E Giroux, Géométrie de contact: de la dimension trois vers les dimensions supérieures, from "Proceedings of the International Congress of Mathematicians, II" (T Li, editor), Higher Ed., Beijing (2002) 405–414 MR Zbl
- [32] P Griffiths, J Morgan, Rational homotopy theory and differential forms, 2nd edition, Progr. Math. 16, Springer (2013) MR Zbl
- [33] J Gutt, The Conley–Zehnder index for a path of symplectic matrices, preprint (2012) arXiv 1201.3728v2
- [34] V Hinich, Homological algebra of homotopy algebras, Comm. Algebra 25 (1997) 3291–3323 MR Zbl

- [35] **PS Hirschhorn**, Model categories and their localizations, Mathematical Surveys and Monographs 99, Amer. Math. Soc., Providence, RI (2003) MR Zbl
- [36] **H Hofer, K Wysocki, E Zehnder**, A general Fredholm theory, I: A splicing-based differential geometry, J. Eur. Math. Soc. 9 (2007) 841-876 MR Zbl
- [37] **H Hofer**, **K Wysocki**, **E Zehnder**, *A general Fredholm theory, II: Implicit function theorems*, Geom. Funct. Anal. 19 (2009) 206-293 MR Zbl
- [38] **H Hofer**, **K Wysocki**, **E Zehnder**, *A general Fredholm theory, III: Fredholm functors and polyfolds*, Geom. Topol. 13 (2009) 2279-2387 MR Zbl
- [39] K Honda, Y Huang, Convex hypersurface theory in contact topology, preprint (2019) arXiv 1907.06025
- [40] U Hryniewicz, A Momin, PAS Salomão, A Poincaré–Birkhoff theorem for tight Reeb flows on S³, Invent. Math. 199 (2015) 333-422 MR Zbl
- [41] M Hutchings, Mean action and the Calabi invariant, J. Mod. Dyn. 10 (2016) 511–539 MR Zbl
- [42] M Hutchings, J Nelson, Cylindrical contact homology for dynamically convex contact forms in three dimensions, J. Symplectic Geom. 14 (2016) 983-1012 MR Zbl
- [43] PJ Kahn, Obstructions to extending almost X-structures, Illinois J. Math. 13 (1969) 336–357 MR Zbl
- [44] W Klingenberg, Lectures on closed geodesics, Grundl. Math. Wissen. 230, Springer (1978) MR Zbl
- [45] M Klukas, Open books and exact symplectic cobordisms, Internat. J. Math. 29 (2018) 1850026, 19 MR Zbl
- [46] **O van Koert**, Lecture notes on stabilization of contact open books, Münster J. Math. 10 (2017) 425–455 MR Zbl
- [47] A Lazarev, M Markl, Disconnected rational homotopy theory, Adv. Math. 283 (2015) 303–361 MR Zbl
- [48] O Lazarev, Maximal contact and symplectic structures, J. Topol. 13 (2020) 1058–1083 MR Zbl
- [49] W Li, Lagrangian cobordism functor in microlocal sheaf theory, I, J. Topol. 16 (2023) 1113–1166 MR Zbl
- [50] J-L Loday, Cyclic homology, 2nd edition, Grundl. Math. Wissen. 301, Springer (1998) MR Zbl
- [51] J-L Loday, B Vallette, Algebraic operads, Grundl. Math. Wissen. 346, Springer (2012) MR Zbl
- [52] D McDuff, D Salamon, Introduction to symplectic topology, 3rd edition, Oxford Univ. Press (2017) MR Zbl
- [53] A Momin, Contact homology of orbit complements and implied existence, J. Mod. Dyn. 5 (2011) 409–472 Zbl
- [54] A Moreno, R Siefring, Holomorphic curves in the presence of holomorphic hypersurface foliations, preprint (2019) arXiv 1902.02700v1
- [55] E Murphy, Loose Legendrian embeddings in high dimensional contact manifolds, preprint (2019) arXiv 1201.2245v5
- [56] L Ng, Framed knot contact homology, Duke Math. J. 141 (2008) 365–406 MR Zbl
- [57] L Ng, A topological introduction to knot contact homology, from "Contact and symplectic topology" (F Bourgeois, V Colin, A Stipsicz, editors), Bolyai Soc. Math. Stud. 26, János Bolyai Math. Soc., Budapest (2014) 485-530 MR Zbl
- [58] L Ng, D Rutherford, V Shende, S Sivek, E Zaslow, Augmentations are sheaves, Geom. Topol. 24 (2020) 2149-2286 MR Zbl

- [59] D M Pancholi, S Pandit, Iso-contact embeddings of manifolds in codimension 2, J. Symplectic Geom. 20 (2022) 471–497 MR Zbl
- [60] J Pardon, An algebraic approach to virtual fundamental cycles on moduli spaces of pseudo-holomorphic curves, Geom. Topol. 20 (2016) 779–1034 MR Zbl
- [61] J Pardon, Contact homology and virtual fundamental cycles, J. Amer. Math. Soc. 32 (2019) 825–919 MR Zbl
- [62] A Ranicki, High-dimensional knot theory: algebraic surgery in codimension 2, Springer (1998) MR Zbl
- [63] JM Sabloff, L Traynor, Obstructions to Lagrangian cobordisms between Legendrians via generating families, Algebr. Geom. Topol. 13 (2013) 2733–2797 MR Zbl
- [64] R Siefring, Intersection theory of punctured pseudoholomorphic curves, Geom. Topol. 15 (2011) 2351–2457 MR Zbl
- [65] I Ustilovsky, Contact homology and contact structures on S^{4m+1}, PhD thesis, Stanford University (1999) Available at https://www.proquest.com/docview/304526632
- [66] B Vallette, Homotopy theory of homotopy algebras, Ann. Inst. Fourier (Grenoble) 70 (2020) 683–738 MR Zbl
- [67] C Wendl, A hierarchy of local symplectic filling obstructions for contact 3–manifolds, Duke Math. J. 162 (2013) 2197–2283 MR Zbl
- [68] C Wendl, Lectures on symplectic field theory, preprint (2016) arXiv 1612.01009
- [69] C Wendl, Lectures on contact 3-manifolds, holomorphic curves and intersection theory, Cambridge Tracts in Mathematics 220, Cambridge Univ. Press (2020) MR Zbl
- [70] **Z Zhou**, Infinite not contact isotopic embeddings in (S^{2n-1}, ξ_{std}) for $n \ge 4$, Bull. Lond. Math. Soc. 55 (2023) 149–155 MR Zbl

Department of Mathematics, Harvard University Cambridge, MA, United States Department of Mathematics, Stanford University Stanford, CA, United States

lcote@math.harvard.edu, fsimon@stanford.edu

Proposed: Leonid Polterovich Seconded: Ciprian Manolescu, Paul Seidel Received: 19 December 2020 Revised: 23 January 2022



GEOMETRY & TOPOLOGY

msp.org/gt

MANAGING EDITOR

András I Stipsicz

cz Alfréd Rényi Institute of Mathematics stipsicz@renyi.hu

BOARD OF EDITORS

Mohammed Abouzaid	Stanford University abouzaid@stanford.edu	Frances Kirwan	University of Oxford frances.kirwan@balliol.oxford.ac.uk
Dan Abramovich	Brown University dan_abramovich@brown.edu	Bruce Kleiner	NYU, Courant Institute bkleiner@cims.nyu.edu
Ian Agol	University of California, Berkeley ianagol@math.berkeley.edu	Urs Lang	ETH Zürich urs.lang@math.ethz.ch
Mark Behrens	University of Notre Dame mbehren1@nd.edu	Marc Levine	Universität Duisburg-Essen marc.levine@uni-due.de
Mladen Bestvina	University of Utah bestvina@math.utah.edu	John Lott	University of California, Berkeley lott@math.berkeley.edu
Martin R Bridson	University of Oxford bridson@maths.ox.ac.uk	Ciprian Manolescu	University of California, Los Angeles cm@math.ucla.edu
Jim Bryan	University of British Columbia jbryan@math.ubc.ca	Haynes Miller	Massachusetts Institute of Technology hrm@math.mit.edu
Dmitri Burago	Pennsylvania State University burago@math.psu.edu	Tomasz Mrowka	Massachusetts Institute of Technology mrowka@math.mit.edu
Tobias H Colding	Massachusetts Institute of Technology colding@math.mit.edu	Aaron Naber	Northwestern University anaber@math.northwestern.edu
Simon Donaldson	Imperial College, London s.donaldson@ic.ac.uk	Walter Neumann	Columbia University neumann@math.columbia.edu
Yasha Eliashberg	Stanford University eliash-gt@math.stanford.edu	Jean-Pierre Otal	Université Paul Sabatier, Toulouse otal@math.univ-toulouse.fr
Benson Farb	University of Chicago farb@math.uchicago.edu	Peter Ozsváth	Princeton University petero@math.princeton.edu
Steve Ferry	Rutgers University sferry@math.rutgers.edu	Leonid Polterovich	Tel Aviv University polterov@post.tau.ac.il
David M Fisher	Rice University davidfisher@rice.edu	Colin Rourke	University of Warwick gt@maths.warwick.ac.uk
Mike Freedman	Microsoft Research michaelf@microsoft.com	Roman Sauer	Karlsruhe Institute of Technology roman.sauer@kit.edu
David Gabai	Princeton University gabai@princeton.edu	Stefan Schwede	Universität Bonn schwede@math.uni-bonn.de
Stavros Garoufalidis	Southern U. of Sci. and Tech., China stavros@mpim-bonn.mpg.de	Natasa Sesum	Rutgers University natasas@math.rutgers.edu
Cameron Gordon	University of Texas gordon@math.utexas.edu	Peter Teichner	Max Planck Institut für Mathematik teichner@mac.com
Lothar Göttsche	Abdus Salam Int. Centre for Th. Physics gottsche@ictp.trieste.it	Richard P Thomas	Imperial College, London richard.thomas@imperial.ac.uk
Jesper Grodal	University of Copenhagen jg@math.ku.dk	Gang Tian	Massachusetts Institute of Technology tian@math.mit.edu
Misha Gromov	IHÉS and NYU, Courant Institute gromov@ihes.fr	Ulrike Tillmann	Oxford University tillmann@maths.ox.ac.uk
Mark Gross	University of Cambridge mgross@dpmms.cam.ac.uk	Nathalie Wahl	University of Copenhagen wahl@math.ku.dk
Rob Kirby	University of California, Berkeley kirby@math.berkeley.edu	Anna Wienhard	Universität Heidelberg wienhard@mathi.uni-heidelberg.de

See inside back cover or msp.org/gt for submission instructions.

The subscription price for 2024 is US \$805/year for the electronic version, and \$1135/year (+\$70, if shipping outside the US) for print and electronic. Subscriptions, requests for back issues and changes of subscriber address should be sent to MSP. Geometry & Topology is indexed by Mathematical Reviews, Zentralblatt MATH, Current Mathematical Publications and the Science Citation Index.

Geometry & Topology (ISSN 1465-3060 printed, 1364-0380 electronic) is published 9 times per year and continuously online, by Mathematical Sciences Publishers, c/o Department of Mathematics, University of California, 798 Evans Hall #3840, Berkeley, CA 94720-3840. Periodical rate postage paid at Oakland, CA 94615-9651, and additional mailing offices. POSTMASTER: send address changes to Mathematical Sciences Publishers, c/o Department of Mathematics, University of California, 798 Evans Hall #3840, Berkeley, CA 94720-3840.

GT peer review and production are managed by EditFLOW® from MSP.

PUBLISHED BY

mathematical sciences publishers

nonprofit scientific publishing

http://msp.org/

© 2024 Mathematical Sciences Publishers

GEOMETRY & TOPOLOGY

Volume 28 Issue 1 (pages 1–496) 2024			
Homological invariants of codimension 2 contact submanifolds	1		
LAURENT CÔTÉ and FRANÇOIS-SIMON FAUTEUX-CHAPLEAU			
The desingularization of the theta divisor of a cubic threefold as a moduli space			
Arend Bayer, Sjoerd Viktor Beentjes, Soheyla Feyzbakhsh, Georg Hein, Diletta Martinelli, Fatemeh Rezaee and Benjamin Schmidt			
Coarse-median preserving automorphisms			
Elia Fioravanti			
Classification results for expanding and shrinking gradient Kähler–Ricci solitons	267		
RONAN J CONLON, ALIX DERUELLE and SONG SUN			
Embedding calculus and smooth structures			
BEN KNUDSEN and ALEXANDER KUPERS			
Stable maps to Looijenga pairs	393		

PIERRICK BOUSSEAU, ANDREA BRINI and MICHEL VAN GARREL